AMREP CORP. Form 10-K July 16, 2013

UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-K ANNUAL REPORT PURSUANT TO SECTIONS 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

(Mark One)

[x] Annual Report pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 For the fiscal year ended April 30, 2013

OR

[] Transition Report pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 For the transition period from to

Commission File Number 1-4702

AMREP CORPORATION

(Exact name of Registrant as specified in its charter)

Oklahoma 59-0936128 (State or other jurisdiction of incorporation or organization) Identification No.)

300 Alexander Park, Suite 204, Princeton, New

08540

Jersey

(Address of principal executive offices)

(Zip Code)

Registrant's telephone number, including area code: (609) 716-8200

Securities registered pursuant to Section 12(b) of the Act:

Title of Each Class Name of Each Exchange on Which Registered

Common Stock \$.10 par value New York Stock Exchange

Securities registered pursuant to Section 12(g) of the Act: None

Indicate by check mark whether the Registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act of 1933.

Yes NoX

Indicate by check mark if the Registrant is not required to file reports pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 (the "Exchange Act").

Yes NoX

Indicate by check mark whether the Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Exchange Act during the preceding 12 months (or for such shorter period that the Registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

YesXNo

Indicate by check mark whether the Registrant has submitted electronically and posted on its corporate website, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the Registrant was required to submit and post such files).

YesX No

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229.405 of this chapter) is not contained herein, and will not be contained, to the best of the Registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. [X]

Indicate by check mark whether the Registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See definitions of "large accelerated filer", "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated Accelerated filer

filer

Non-accelerated Smaller reporting X

filer company (Do not check if a smaller reporting company)

Indicate by check mark whether the Registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act).

Yes NoX

As of October 31, 2012, which was the last business day of the Registrant's most recently completed second fiscal quarter, the aggregate market value of the Common Stock held by non-affiliates of the Registrant was \$18,482,705. Such aggregate market value was computed by reference to the closing sale price of the Registrant's Common Stock as quoted on the New York Stock Exchange on such date. For purposes of making this calculation only, the Registrant has defined affiliates as including all directors and executive officers and certain persons related to them. In making such calculation, the Registrant is not making a determination of the affiliate or non-affiliate status of any holders of shares of Common Stock.

As of July 5, 2013, there were 7,195,454 shares of the Registrant's Common Stock outstanding.

DOCUMENTS INCORPORATED BY REFERENCE

As stated in Part III of this annual report on Form 10-K, portions of the Registrant's definitive proxy statement to be filed within 120 days after the end of the fiscal year covered by this annual report on Form 10-K are incorporated herein by reference.

PART I

Item 1. Business

GENERAL

The Company* was organized in 1961 as an Oklahoma corporation and, through its subsidiaries, is primarily engaged in four business segments: the Subscription Fulfillment Services business operated by Palm Coast Data LLC ("Palm Coast"), the Newsstand Distribution Services business and the Product Packaging and Fulfillment Services and Other businesses operated by Kable Media Services, Inc. and its subsidiaries ("Kable") (the businesses operated by Palm Coast and Kable are collectively referred to as "Media Services"), and the real estate business operated by AMREP Southwest Inc. and its subsidiaries (collectively, "AMREP Southwest"). On December 31, 2012, a newly-formed wholly-owned subsidiary of Palm Coast, FulCircle Media, LLC ("FulCircle"), acquired certain assets from a third party. The results of this new subsidiary are included in the Subscription Fulfillment Services business since December 31, 2012. Information concerning industry segments is set forth in Note 19 of the notes to the consolidated financial statements included in this annual report on Form 10-K.

As of July 1, 2013, the Company employed approximately 1,200 full time employees, including approximately 370 temporary employees of the Company's staffing business. The Company's foreign sales and activities are not significant. Many of the amounts and percentages presented in this Part I have been rounded for convenience of presentation. All references in this Part I to 2013, 2012 and 2011 mean the Company's fiscal years ended April 30, 2013, 2012 and 2011, unless otherwise qualified.

SUBSCRIPTION FULFILLMENT SERVICES, NEWSSTAND DISTRIBUTION SERVICES AND PRODUCT PACKAGING AND FULFILLMENT SERVICES AND OTHER BUSINESSES

The Company (i) through its Palm Coast and FulCircle subsidiaries conducts its Subscription Fulfillment Services business in which it performs subscription fulfillment and related services for publishers and other customers, (ii) through its Kable Newsstand Distribution Services subsidiaries distributes periodicals nationally and in Canada and, to a small degree, in other foreign countries, and (iii) through its Kable Product Packaging and Fulfillment Services and Staffing Resources subsidiaries, provides internet order processing and shipment for e-commerce retailers, packaging design, procurement and product fulfillment services and temporary staffing services. Total Media Services revenues were \$82,105,000 for 2013.

Subscription Fulfillment Services

The Subscription Fulfillment Services business performs fulfillment and fulfillment-related activities, principally magazine subscription fulfillment services and ancillary services, and it accounted for approximately 70% of Media Services revenues in 2013. In the magazine subscription fulfillment services operation, Palm Coast maintains subscriber lists and databases, processes new orders, receives and accounts for payments, prepares and transmits to each publisher's printer the labels or tapes containing the names and addresses of subscribers for mailing each issue, handles subscriber telephone inquiries and correspondence, prepares renewal and statement notifications for mailing, generates marketing and statistical reports, processes internet orders and prints forms and promotional materials. List services clients are primarily publishers for whom Palm Coast maintains client customer lists, selects names for clients who rent their lists, merges rented lists with a client's lists to eliminate duplication for the client's promotional mailings, and sorts and sequences mailing labels to provide optimum postal discounts. These services are performed for many clients, but some clients may only utilize certain of them. Although by far the largest number of magazine titles for which subscription fulfillment services are performed are consumer publications, Palm Coast also performs services for membership organizations, trade (business) publications and government agencies that utilize the broad

capabilities of Palm Coast's extensive database systems.

Palm Coast performs subscription fulfillment services for approximately 422 different magazine titles for approximately 96 clients and maintains databases of approximately 39 million active subscribers for its client publishers and membership organizations. In a typical month, Palm Coast produces or provides data

^{*}As used herein, "Company" includes the Registrant and its subsidiaries unless the context requires or indicates otherwise.

for approximately 39 million mailing labels for its clients and also processes over 13 million pieces of outgoing mail for these clients.

There are a number of companies that perform subscription fulfillment services for publishers and with which Palm Coast competes, including one that is larger than Palm Coast. Since publishers often utilize only a single fulfillment company for a particular publication, there is intense competition to obtain subscription fulfillment contracts with publishers. Competition for non-publisher clients is also intense. Palm Coast has a sales staff whose primary task is to solicit subscription fulfillment business.

FulCircle is engaged in the marketing strategy, call center service, product distribution and operational solutions businesses. The acquisition of these assets is expected to allow the Company's Media Services business to move further into non-publishing relationships.

Newsstand Distribution Services

In its Newsstand Distribution Services business, Kable distributes over 450 publications for approximately 200 publishers. Among the titles are many special interest magazines, including various hobbyist, celebrity, puzzle, automotive, comics, women's service and sports magazines. In a typical month, Kable distributes approximately 43 million copies of various titles to wholesalers. Kable coordinates the movement of the publications from its publisher clients to approximately 100 independent wholesalers in North America and to wholesalers in over 80 countries worldwide. The wholesalers in turn sell the publications to retail chains and independent retail outlets. All parties generally have full return rights for unsold copies. The Newsstand Distribution Services business accounted for approximately 11% of Media Services revenues in 2013.

While Kable may not handle all publications of an individual publisher client, it usually is the exclusive distributor into the consumer marketplace for the publications it distributes. Kable has a distribution sales and marketing force that works with wholesalers and retailers to promote magazine sales and assist in determining the appropriate number of copies of an individual magazine to be delivered to each wholesaler and ultimately each retailer serviced by that wholesaler. Kable generally does not physically handle any product. Kable generates and delivers to each publisher's printer shipping instructions with the addresses of the wholesalers and the number of copies of product to be shipped to each. All magazines have a defined "off sale" date following which the retailers return unsold copies to the wholesalers, who destroy them after accounting for returned merchandise in a manner satisfactory to and auditable by Kable.

Kable generally makes substantial cash advances to publishers against future sales that publishers may use to help pay for printing, paper and production costs prior to the product going on sale. Kable is usually not paid by wholesalers for product until some time after the product has gone on sale, and is therefore exposed to credit risks with both publishers and wholesalers. Kable's ability to limit its credit risk is dependent in part on its skill in estimating the number of copies of an issue that should be distributed and which will be sold, and on limiting its advances to the publisher accordingly.

Kable competes primarily with three other national distributors, each of which is larger than Kable. One of these competitors is affiliated with a magazine publishing company, and one is affiliated with one of the largest magazine wholesalers in North America. The competition for the distribution rights in this business is intense. In addition, over the past five years, there has been a major consolidation and reduction in the number of wholesalers to whom Kable distributes magazines arising from changes within the magazine distribution industry. During 2013, business with three wholesalers accounted for a major portion of the gross billings of the Newsstand Distribution Services business, which is common for the industry. Of Kable's Newsstand Distribution Services aggregate accounts receivable at April 30, 2013, approximately 68% were due from these three wholesalers.

Product Packaging and Fulfillment Services and Other

Together, Kable's Product Packaging and Fulfillment Services business offers an integrated approach to all forms of electronic and traditional commerce for consumer products manufacturers. Specifically, the business unit provides:

- Design, procurement and assembly/packaging of retail packs and point-of-purchase displays;
- Front-end processing, including customer data base management for one-time and subscription order programs;

- Out-reach to consumers via e-commerce, direct mail and outbound phone services;
- Order capturing via electronic mediums as well as traditional call center operations, mail capture and associated data entry;
- Warehousing and pick/pack/ship functions for both business to business and business to consumer shipments; and
 - Billing as well as collection of payments tendered by credit cards, checks and cash.

As an adjunct to the Subscription Fulfillment Services business, the Product Packaging and Fulfillment Services business offers fulfillment services to their publishers, including shipment of premiums (free gifts) provided to subscribers, shipment of replacement copies of newly issued magazines or purchased copies of older editions and shipment and payment processing for product advertised for sale in their publishers' magazines or associated web sites. The Product Packaging and Fulfillment Services business operates from a 191,000 square foot facility owned by the Company in Fairfield, Ohio, with an office in Mt. Morris, Illinois providing customer service, phone center, systems and accounting support.

Kable Staffing Resources ("Staffing Resources") operates as a separate business which provides temporary employees to local companies in the Fairfield, Ohio area, including Kable's Product Packaging and Fulfillment Services business. Many of its clients experience fluctuations in their businesses and therefore choose to use temporary employees from an agency rather than trying to manage the labor themselves. In this business, Staffing Resources hires and pays the employees it provides to its clients, and charges its clients a rate that includes its margin for providing this service. Both Product Packaging and Fulfillment Services and Staffing Resources operate in very competitive environments. Together, they accounted for approximately 19% of Media Services revenues in 2013.

REAL ESTATE OPERATIONS

The Company conducts its real estate business through AMREP Southwest, with these activities occurring primarily in the City of Rio Rancho and certain adjoining areas of Sandoval County, New Mexico. References below to Rio Rancho include the City and such adjoining areas.

Properties – Rio Rancho

Rio Rancho consists of 91,049 acres in Sandoval County near Albuquerque, of which approximately 74,100 acres have been platted into approximately 114,680 residential and commercial lots, approximately 16,500 acres are dedicated to community facilities, roads and drainage and the remainder is unplatted land. At April 30, 2013, approximately 91,000 of these residential and commercial lots had been sold by AMREP Southwest net of lots repurchased and those returned to AMREP Southwest by deeds in lieu of foreclosure. AMREP Southwest currently owns approximately 17,345 acres in Rio Rancho, of which approximately 4,410 acres are in several areas of contiguous properties which are being developed or are suitable for development, and approximately 2,000 acres are in areas with a high concentration of ownership, where AMREP Southwest owns more than 50% of the lots in the area. These high concentration areas are suitable for special assessment districts or city redevelopment areas that may allow for future development under the auspices of local government. The balance of the acreage owned is in scattered lots, where AMREP Southwest owns less than 50% of the lots in the area, that may require the purchase of a sufficient number of adjoining lots to create tracts suitable for development or that AMREP Southwest may offer for sale individually or in small groups.

Activities conducted or arranged by AMREP Southwest include the obtaining of necessary governmental approvals ("entitlements"), installation of utilities and necessary storm drains, and building or improving of roads necessary for land development. At Rio Rancho, AMREP Southwest develops both residential lots and sites for commercial and

industrial use as demand warrants, and also secures entitlements for large development tracts for sale to homebuilders. The engineering work at Rio Rancho is performed by both AMREP Southwest employees and outside firms, but all development work is performed by outside contractors. AMREP Southwest personnel market land at Rio Rancho, both directly and through brokers. AMREP Southwest competes with other owners of land in the Rio Rancho and Albuquerque area that offer for sale developed and undeveloped residential lots and sites for commercial and industrial use.

The City of Rio Rancho is the third largest city in New Mexico with a population of approximately 93,000 whose median age is 36 years. The city's population grew by approximately 80% from calendar year 2000 through 2012. The city has completed significant construction projects within the last five years, including a City Center central business district with a 6,500 seat events center and a city hall, the University of New Mexico West campus at City Center, a Hewlett-Packard technical and customer support center, Central New Mexico Community College and two new hospitals, among others. These new projects and business relocations are anticipated to generate approximately 3,000 jobs in the Rio Rancho area over the next five years. Currently, major non-government employers include Intel Corporation, Hewlett-Packard, U.S. Cotton and customer care call centers of Bank of America, Alliance Data and Sprint PCS. As of March 2013, the City of Rio Rancho's unemployment rate was 6.8%.

In Rio Rancho, AMREP Southwest sells both developed and undeveloped lots to national, regional and local homebuilders, commercial and industrial property developers and others. In the last three fiscal years, its land sales in Rio Rancho have been as follows:

	Acres Sold	Revenues		Revenues Per Acre (a)	
2013:					
Developed					
Residential (b)	-	\$	120,000	\$	476,000
Commercial	3		525,000		175,000
Total Developed	3		645,000		215,000
Undeveloped	9		188,000		21,000
Total	12	\$	833,000	\$	69,000
2012:					
Developed					
Residential	-	\$	-	\$	-
Commercial	4		748,000		178,000
Total Developed	4		748,000		178,000
Undeveloped	100		1,141,000		11,000
Total	104	\$	1,889,000	\$	18,000
2011:					
Developed					
Residential	3	\$	1,031,000	\$	344,000
Commercial (c)	-		35,000		_
Total Developed	3		1,066,000		344,000
Undeveloped	19		714,000		38,000
Total	22	\$	1,780,000	\$	81,000

- (a) Revenues per acre may not calculate precisely due to rounding of acres sold amounts and rounding to nearest thousand for revenues.
- (b) Less than one half an acre of residential developed land was sold in 2013.
 - (c) Revenues recognized under the cost recovery method of sales for real estate. Acres sold were recognized in a prior period.

In connection with certain individual home site sales made prior to 1977 at Rio Rancho, New Mexico, if water, electric and telephone utilities have not reached the lot site when a purchaser is ready to build a home, AMREP Southwest is obligated to exchange a lot in an area then serviced by such utilities for the lot of the purchaser, without cost to the purchaser. AMREP Southwest has not incurred significant costs related to the exchange of lots.

Other Properties

AMREP Southwest also owns two tracts of land in Colorado, consisting of one property of approximately 160 acres planned for approximately 400 homes that AMREP Southwest intends to offer for sale upon obtaining all necessary entitlements, and one property of approximately 10 acres zoned for commercial use, which is being offered for sale.

Available Information

The Company maintains a website at www.amrepcorp.com. The Company's annual reports on Form 10-K, quarterly reports on Form 10-Q, current reports on Form 8-K and amendments to those reports filed or furnished pursuant to Sections 13(a) and 15(d) of the Securities Exchange Act of 1934, as amended, are available free of charge through the Company's website as soon as reasonably practicable after such material is electronically filed with, or furnished to, the Securities and Exchange Commission. The information found on the Company's website is not part of this or any other report that the Company files with, or furnishes to, the Securities and Exchange Commission.

Item 1A. Risk Factors

The risks described below are among those that could materially and adversely affect the Company's business, financial condition or results of operations. These risks could cause actual results to differ materially from historical experience and from the Company's plans, projections or other forward-looking statements included in "Item 7. Management's Discussion and Analysis of Financial Condition and Operations" below and elsewhere in this annual report on Form 10-K. These risks are not the only risks the Company faces, and other risks include those not presently known as well as those that are currently considered to be less significant.

Major Business Risks

The Company's subsidiaries have substantial indebtedness and other financial obligations which could adversely affect the Company's business, financial condition or results of operations.

The Company's primary sources of funding for working capital requirements are cash flow from operations, the Company's Media Services banking facility and working capital made available to the Company by customers. The Company's liquidity is affected by many factors, including some that are based on normal operations and some that are related to the industries in which the Company operates and the economy generally. The Company's Media Services businesses finance operations in part through a revolving credit facility that matures May 12, 2015 (the "Media Services Credit Facility"). AMREP Southwest finances its business from cash flow from operations, which has been minimal in recent years due to the poor conditions in its real estate markets, and from advances made to it by its parent. AMREP Southwest also has a loan agreement that matures December 1, 2017, which does not allow for additional borrowings. If the Company's Media Services businesses are unable to extend or renew on similar terms the Media Services Credit Facility that expires in May 2015, the Media Services businesses would be obligated to repay the outstanding balance of the Media Services Credit Facility; similarly, if AMREP Southwest is unable to extend or renew on similar terms the AMREP Southwest loan that is due in December 2017, AMREP Southwest would be obligated to repay the outstanding balance of the AMREP Southwest loan. If this occurs, the Company's Media Services businesses or AMREP Southwest, as the case may be, may not be able to fund in full or refinance these obligations at such time absent the Company obtaining additional debt or equity funding or raising capital through the sale of assets. Such additional funding may not be available on acceptable terms or at all at such time. Any failure in this regard could have a material adverse effect on the Company's business, financial condition and results of operations.

Similarly, the Company's Media Services businesses have operated with negative working capital (\$25,752,000 at April 30, 2013) primarily through liquidity provided by one significant customer contract that expires June 2014. The negative working capital of the Company's Media Services businesses represents the net payment obligation due to this customer and certain other third parties. On May 26, 2013, the Company's Newsstand Distribution Services business, which is part of the Company's Media Services businesses, received notice that this customer contract would not be renewed upon its scheduled expiration in June 2014. The Company's Newsstand Distribution Services business is evaluating its ability to pay the net payment obligation represented by the negative working capital (\$19,086,000 as of April 30, 2013) upon expiration of the contract. The Company's Newsstand Distribution Services business currently

does not have sufficient capital to fund in full or refinance this obligation, and it may be unable to pay such amount and obtain other sources of working capital absent the Company obtaining additional debt or equity funding or raising capital through the sale of assets. Such additional funding or capital may not be available on acceptable terms or at all. Any failure to obtain capital to pay such obligation or to obtain other sources of working capital could have a material adverse effect on the Company's business, financial condition and results of operations; these circumstances could also trigger a default under the Company's Media Services Credit Facility. In addition, this customer contract represented approximately 8.6% of revenues for the Company's Newsstand Distribution Services business for the year ended April 30, 2013 (or less than 1% of the Company's consolidated revenues for that period), and the loss of this revenue following the expiration of the contract in June 2014 could have a material adverse effect on the Company's Newsstand Distribution Services business, financial condition and results of operations.

Due to the Company's holding company structure, it is dependent on its subsidiaries to distribute funds upstream to the parent company.

AMREP Corporation is a holding company that conducts substantially all of its operations through subsidiaries. As a holding company, AMREP Corporation is dependent on distributions of funds from subsidiaries to pay expenses and fund operations. Because of the adverse conditions currently affecting AMREP Southwest, Media Services is presently the sole source of funding for the parent company's operations, and the parent company in turn is supplying a substantial portion of the funding needed by AMREP Southwest. The continued availability of this funding is dependent upon the results of operations of Media Services and its continued compliance with the covenants in its revolving credit facility. The Company's overall results of operations, future growth or both would be adversely affected if for any reason Media Services were unable to distribute sufficient funds to support the operations of the parent company and AMREP Southwest. If the cash available for distribution by Media Services were insufficient to fund the Company's consolidated operations and the Company was not able to provide the funding needed by AMREP Southwest, the Company would be forced to seek either replacement financing or other sources of capital, such as by selling assets or issuing equity, which replacement financing or other sources of capital might not be available on acceptable terms (or at all).

The Media Services Credit Facility requires the borrowers to meet certain covenants, including maintaining a minimum fixed charge coverage ratio, as defined. However, neither meeting a covenant's requirement in the future nor obtaining relief from the lender if it is not met can be assured. At April 30, 2013, the borrowers were in compliance with the covenants of the Media Services Credit Facility. At April 30, 2013, the borrowing availability under the Media Services Credit Facility was \$12,895,000, and there was \$620,000 outstanding against this availability. The highest amount borrowed during the year ended April 30, 2013 was \$6,770,000. Under the terms of the Media Services Credit Facility, in the event of any future non-compliance, the Media Services business would be barred from repaying indebtedness to or otherwise distributing funds to the parent company and the lender would be entitled to terminate the Media Services Credit Facility and seek immediate payment of any outstanding borrowing.

The Company's defined benefit pension plan is currently substantially underfunded and will require additional cash contributions, some of which are likely to be accelerated.

Under generally accepted accounting principles, the Company's defined benefit pension plan was underfunded at April 30, 2013 by \$13,805,000. Although the Company froze the pension plan effective March 1, 2004, so that from that date there would be no new participants in the plan and the existing participants' future compensation would not affect their pension benefits, the plan remains underfunded. A key assumption underlying the actuarial calculations for the pension plan is an assumed annual investment rate of return of 8.0%. If the pension plan assets do not realize this expected rate of return, or if any other assumptions underlying the actuarial calculations are incorrect or are modified, the Company may be required to make contributions to the pension plan beyond current requirements, which could negatively impact the Company's limited financial resources.

In addition, due to the closing of certain facilities in connection with the consolidation of the Company's Subscription Fulfillment Services business and the associated work force reduction, the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and the regulations thereunder, have given the Pension Benefit Guaranty Corporation (the "PBGC") the right to require the Company to accelerate the funding of \$11,688,000 of accrued pension-related obligations to the pension plan. In August 2012, the Company and the PBGC reached an agreement with respect to this funding obligation, and as a result, the Company made a \$3,000,000 cash contribution to the pension plan on August 16, 2012. The agreement also provided that if, before August 15, 2013, the Company is unable to pay the remaining \$8,688,000 liability or adequately secure it with collateral acceptable to the PBGC, the Company would be required to either (i) provide a letter of credit equal to 110% of the remaining liability or establish a cash escrow for 100% of the remaining liability, to be maintained for five years or until the remaining liability is discharged, if sooner or (ii) discharge the remaining liability in quarterly installments over a five year period and secure it with collateral

acceptable to the PBGC. In the event the Company fails to meet the terms of the agreement, the PBGC could seek immediate payment of the amount due or attempt to force a termination of the pension plan. Although the Company is currently having discussions with the PBGC about a comprehensive restructuring of this \$8,688,000 payment obligation, the Company is unable to offer any assurance that it will be able to discharge the pension plan funding obligation by August 15, 2013 or otherwise meet the PBGC's requirements for securing or paying the undischarged amount, and the Company cannot offer any assurance

that upon such inability it will be able to negotiate with the PBGC to obtain further relief. In addition, the Company may become subject to additional acceleration of its remaining accrued pension-related obligations to the pension plan if the Company closes certain additional facilities and further reduces its work force. Any such acceleration could negatively impact the Company's limited financial resources and could have a material adverse impact on the Company's financial condition.

The Company has incurred significant historical losses and cannot assure shareholders that the Company will be profitable in the near term or at all.

The Company has incurred significant historical losses, including net losses of \$2,838,000, \$1,143,000 and \$7,561,000 for the fiscal years ended April 30, 2013, 2012 and 2011, primarily due to highly competitive conditions in the Media Services businesses and the absence of meaningful real estate development and building activity after 2008 in the markets where AMREP Southwest owns real estate. The Company cannot assure shareholders that the economic or operating conditions affecting either of its principal businesses will improve or that the Company will return to profitability in the near term or at all.

Risks Related to the Company's Media Services Operations

The introduction and increased popularity of alternative technologies for the distribution of news, entertainment and other information and the resulting shift in consumer habits and advertising expenditures from print to other media has adversely affected the Company's Media Services operations.

Revenues in the Company's Media Services operations are principally derived from services the Company performs for traditional publishers. Historically, a reduction in the demand for the Company's newsstand distribution services due to lower sales of magazines at newsstands has often been at least partially offset by an increase in demand for the Company's subscription fulfillment services as consumers affected by the reduction in newsstand distribution instead sought publications through subscription. However, technology changes, particularly digital technology used in the entertainment and media industries, continue to evolve rapidly, and advances in that technology have led to alternative methods for the distribution, storage and consumption of content such as that contained in the products distributed by the Company's Media Services operations. These technological changes have driven and reinforced changes in consumer behavior as consumers increasingly seek control over when, where and how they consume content. For example, the distribution of news, entertainment and other information via the internet has become very popular, and consumers increasingly rely on electronic tablets and readers, personal computers, cellular phones and other electronic devices for such information. The resulting reduction in demand for traditional print media and the shift of advertising dollars from traditional print media to online media has adversely affected the publishing industry in general and has had a negative impact on both the Company's Subscription Fulfillment Services and Newsstand Distribution Services businesses. The Company's failure or inability to adapt to emerging technologies and changes in consumer behavior could have a significant adverse effect on the Company's competitive position and its businesses and results of operations.

The Company's operating results depend in part on successful research, development and marketing of new or improved services and data processing capabilities and could suffer if the Company is not able to continue to successfully implement new technologies.

The Company's Media Services businesses operate in highly competitive markets that are subject to rapid change, and must therefore continue to invest in developing technologies and improving various existing systems in order to remain competitive. There are substantial uncertainties associated with the Company's efforts to develop new technologies and services for the subscription fulfillment and newsstand distribution markets the Company serves, including the Company's lack of financial resources. Particularly in the Subscription Fulfillment Services business, the Company would need to make substantial capital investments in order to convert the Company's business to newer

digital and internet-based technologies. Some of the Company's competitors have already adjusted their businesses for the growing digital market and any improvements the Company makes may not be developed until it is too late to compete effectively. Additionally, the cost and expertise needed to develop these new digital and internet-based technologies may be prohibitive for the Company, and even if the Company makes significant investments in new information processing technologies and services in these or other areas, they may not prove to be profitable. The failure or inability to successfully develop or employ new technologies and services could have a material adverse effect on the Company's competitive position and its businesses and results of operations. Even if these developments are profitable, the operating margins resulting from their application would not necessarily equal, or result in an improvement over, the Company's historical margins.

The Company's Media Services operations could face increased costs and business disruption from instability in the newsstand distribution channel.

The Company's Newsstand Distribution Services business operates a national distribution business that relies on wholesalers to distribute magazines to newsstands and other retail outlets. A small number of wholesalers are responsible for a substantial percentage of the wholesale magazine distribution business in the United States, and the Company extends credit to such wholesalers, whose credit worthiness and financial position may be affected by changes in economic or other external conditions. In recent years there has been instability in the wholesaler channel that has led to one major wholesaler abandoning the business and to certain disruptions to magazine distribution. In addition, another magazine wholesaler is in the process of restructuring itself after failing to timely meet its payment obligations to certain national distributors. The Company has reserved substantially all of the net accounts receivable due from this magazine wholesaler as an estimate for doubtful accounts, which resulted in a \$2,000,000 charge to operations. There is the possibility of further consolidation among wholesalers, and the insolvency or non-payment of its obligations by one or more of these wholesalers would have a material adverse impact on the Company's results of operations and financial condition. In addition, due to the significant concentration in the industry, should there be a disruption in the wholesale channel, it could impede the Company's ability to distribute magazines to the retail marketplace.

The Company's publisher customers face business pressures from reduced advertising revenues and increased costs for paper, printing and postal rates. These factors could have a negative effect on their operating income, and this in turn could negatively affect the Company's Media Services operations.

An important source of revenue for the magazine publishing industry, the principal industry the Company serves, is advertising. As a result of the recent economic slowdown, there was a well-publicized reduction in advertising at all levels which caused a higher attrition rate of magazine titles than had been previously experienced. In addition, the Company's publisher customers' principal raw material is paper. Paper and printing prices have fluctuated over the past several years, and significant increases in paper prices could adversely affect a publisher customer's operating income. Postage for magazine distribution and direct solicitation is another significant operating expense of the Company's publisher customers, which primarily use the U.S. Postal Service to distribute their products. Any softness in advertising revenues or significant increases in paper costs, printing costs or postal rates that publishers are not able to offset could have a negative effect on their operating income and number of titles published, and this in turn could negatively affect the Company's Media Services operations.

Almost all of the revenues of the Company's Newsstand Distribution Services business are derived from sales made on a fully returnable basis, and an error in estimating expected returns could cause a misstatement of revenues for the period affected.

As is customary in the magazine distribution industry, almost all of the commission revenues of the Company's Newsstand Distribution Services business are derived from sales made on a fully returnable basis, meaning that customers may return unsold copies of magazines for credit. From May 1, 2010 through April 30, 2013, customers ultimately returned for credit approximately 65% of the magazines the Company initially distributed. The Company recognizes commission revenues from the distribution of magazines at the time of delivery to the wholesalers, less a reserve for estimated returns that is based on historical experience and recent sales data on an issue-by-issue basis. Although the Company has the contractual right to return these magazines for offsetting credits from the publishers from whom the magazines are purchased, an error in estimating the percentage of returns at the end of an accounting period could have the effect of understating or overstating revenues in the period affected, which misstatement would have to be adjusted in a subsequent period when the actual return information becomes known.

Competitive pressures may result in a decrease in the Company's Media Services revenues and profitability.

The subscription fulfillment and newsstand distribution services businesses are highly competitive, and some of the Company's competitors have financial resources that are substantially greater than the Company's. The Company experiences significant price competition in the markets in which it competes. Competition in the Company's Media Services businesses may come not only from other service providers, but also from the Company's customers, who may choose to develop their own internal subscription fulfillment or newsstand distribution operations, thereby reducing demand for the Company's services. Competitive pressures could cause the Company's Media Services businesses to lose market share or result in significant price erosion that could have an adverse effect on the Company's results of operations.

The Company may not be able to successfully introduce new services and data processing capabilities on a timely and cost-effective basis.

The success of new and improved services depends on their initial and continued acceptance by the publishers and other customers with whom the Company conducts business. The Company's Media Services businesses are affected, to varying degrees, by technological changes and shifts in customer demand. These changes result in the transition of services provided and increase the importance of being "first to market" with new services and information processing innovations. The Company may not have the financial and other resources necessary to successfully and timely develop such services or innovations. Difficulties or delays in the development, production or marketing of new services and information processing capabilities may be experienced, and may adversely affect the Company's results of operations. These difficulties and delays could also prevent the Company from realizing a reasonable return on the investment required to bring new services and information processing capabilities to market on a timely and cost-effective basis.

The Company's operations could be disrupted if its information systems fail, causing increased expenses and loss of sales.

The Company's business depends on the efficient and uninterrupted operation of its systems and communications capabilities, including the maintenance of customer databases for billing and label processing, and the Company's magazine distribution order regulation system. If a key system was to fail or experience unscheduled downtime for any reason, even if only for a short period, the Company's operations and financial results could be adversely affected. The Company's systems could be damaged or interrupted by a security breach, fire, flood, power loss, telecommunications failure or similar event. The Company has a disaster recovery plan in place, but this plan may not prevent delays or other complications that could arise from an information systems failure. The Company's business interruption insurance may not adequately compensate the Company for losses that may occur.

The Company depends on the internet to deliver some services, which may expose the Company to various risks.

Many of the Company's operations and services, including order taking on behalf of customers and communications with customers and suppliers, involve the use of the internet. The Company is therefore subject to factors that adversely affect internet usage, including the reliability of internet service providers that from time to time may have operational problems and experience service outages or concentrated attacks from third parties seeking to disrupt commerce. Additionally, as the Company continues to increase the services it provides using the internet, the Company is increasingly subject to risks related to the secure transmission of confidential information over public networks. Failure to prevent security breaches of the Company's networks or those of its customers, or a security breach affecting the internet in general, could adversely affect the Company's results of operations.

The Company is subject to extensive rules and regulations of credit card associations.

The Company processes a large number of credit card transactions on behalf of its Subscription Fulfillment Services customers and is thus subject to the extensive rules and regulations of the leading credit card associations. The card associations modify their rules and regulations from time to time, and the Company's inability to anticipate changes in such rules and regulations or in the interpretation or application thereof may result in substantial disruption to its business. In the event that the card associations or the sponsoring banks determine that the manner in which the Company processes certain credit card transactions is not in compliance with existing rules and regulations, or if the card associations adopt new rules or regulations that prohibit or restrict the manner in which the Company processes credit card transactions, the Company may be subject to substantial penalties and fines and be forced to modify the manner in which it operates, which may increase costs, or to cease processing certain types of transactions altogether, any of which could have a negative impact on the Company's business.

Changes relating to consumer information collection and use could adversely affect the Company's ability to collect and use data, which could harm its business.

Public concern over methods of information gathering has led to the enactment of legislation in most jurisdictions that restricts the collection and use of consumer information. The Company engages in the collection and use of consumer information in connection with its clients' businesses and the Company's growing digital efforts. Further legislation, government regulations, industry regulations, the issuance of judicial interpretations or a change in customs relating to the collection, management, aggregation and use of consumer information could materially

increase the cost of collecting that data, or limit the Company's ability to provide information to its customers or otherwise utilize telemarketing or e-mail marketing or distribute the Company's digital products across multiple platforms, and could adversely affect the Company's results of operations.

The Company faces government regulation and legal uncertainties related to internet communications, commerce and privacy regulation.

The growth and development of the market for internet commerce and communications has prompted both federal and state laws and regulations concerning the collection and use of personally identifiable information (including consumer credit and financial information), consumer protection, the content of online publications, the taxation of online transactions and the transmission of unsolicited commercial email, popularly known as "spam." More laws and regulations are under consideration by various governments, agencies and industry self-regulatory groups. Although the Company's compliance with applicable federal and state laws, regulations and industry guidelines has not to date had a material adverse effect on it, new laws and regulations may be introduced and modifications to existing laws may be enacted that require the Company to make changes to its business practices. Although the Company believes that its practices are in compliance with applicable laws, regulations and policies, if the Company were required to defend its practices against investigations of state or federal agencies or if the Company's practices were deemed to violate applicable laws, regulations or policies, the Company could be penalized and some of its activities could be enjoined. Any of the foregoing could impact the Company's ability to provide services, increase the cost of conducting online activities, decrease the demand for the Company's services or lessen the Company's ability to effectively market its services, or otherwise materially adversely affect its business, financial condition and results of operations.

The Company's intangible assets, including customer contracts and relationships, are periodically subject to asset impairment tests for accounting purposes, which may result in the incurrence of non-cash impairment charges.

The Company periodically evaluates and reviews its intangible assets, including customer contracts and relationships, for impairment when events or changes in circumstances indicate the carrying value of the asset may not be recoverable. Any such impairment charge would be included in the Company's results of operations. The uncertainty of revenue trends in the media services industry, any decline in the Company's estimate of revenues recoverable from its customer contracts and relationships, the general decline in the magazine publishing industry, which represents the Media Services business's primary customer base, the recent recession that impacted the U.S. economy and consumers and the uncertainty about the economy's future may adversely impact the recoverability of the Company's intangible assets, including customer contracts and relationships, which may result in non-cash impairment charges that could adversely affect the Company's results of operations.

Risks Related to the Company's Real Estate Operations

The Company's real estate assets are concentrated in one market, Rio Rancho, New Mexico, so that the Company's results of operations and future growth may be limited or affected by economic changes in that market.

Substantially all of the Company's real estate assets are located in Rio Rancho, which is adjacent to Albuquerque, New Mexico. As a result of this geographic concentration, the Company has been and will be affected by changes in economic conditions that occur in this region from time to time, including regional economic contraction due to, among other things, the failure or downturn of key industries and employers. The Company's results of operations, future growth or both may be further adversely affected if the regional demand for residential and commercial real estate remains at the current historically low levels due to the prolonged severe decline in the real estate market in the greater Albuquerque-metro and Rio Rancho areas. Land sales have declined from 1,051 acres sold by the Company in fiscal 2007 to 12 acres sold in fiscal 2013, as builders have slowed the pace of building on developed lots previously purchased from the Company in Rio Rancho and delayed or cancelled the purchase of additional developed lots.

A downturn in the business of Rio Rancho's largest employer may adversely affect the Company's real estate development business there.

Intel Corporation ("Intel") is the largest employer in Rio Rancho and operates a large semiconductor manufacturing facility there. Although Intel has made substantial investments in the Rio Rancho plant in recent years, the number of employees and on-site contractors at the plant fluctuates with the demand for the products produced at the plant. If

Intel's presence in Rio Rancho were to diminish for any reason, such as in response to a downturn in its semiconductor manufacturing business or as a result of the relocation of its operations conducted there to another location, the Rio Rancho real estate market and the Company's land development business would likely be adversely affected.

As Rio Rancho's population continues to grow, the Company's land development activities in that market may be subject to greater limitations than they have been historically.

When the Company acquired its core real estate inventory in Rio Rancho nearly 50 years ago, the area was not developed and had a small population. As of April 30, 2013, Rio Rancho was the third largest city in New Mexico with a population of approximately 93,000. As Rio Rancho's population continues to grow, the Company may be unable to engage in development activities comparable to those the Company has engaged in historically. Local community or political groups may oppose the Company's development plans or require modification of those plans, which could cause delays or increase the cost of the Company's development projects. In addition, zoning density limitations, "slow growth" provisions or other land use regulations implemented by state, city or local governments could further restrict the Company's development activities or those of its homebuilder customers, or could adversely affect financial returns from a given project, which could adversely affect the Company's results of operations.

Much of the Company's remaining Rio Rancho real estate is not in contiguous properties, which may adversely affect the Company's ability to sell lots at levels comparable with the levels it experienced prior to the 2008 sales downturn.

Of the approximately 17,345 acres in Rio Rancho that the Company owned at April 30, 2013, approximately 4,410 acres were in several areas of contiguous properties that are being developed or are suitable for development, and approximately 2,000 acres were in areas with a high concentration of ownership, where the Company owns more than 50% of the lots in the area, suitable for special assessment districts or city redevelopment areas that may allow for future development under the auspices of local government. The balance is in scattered lots, where the Company owns less than 50% of the lots in the area, which may require the purchase of a number of adjoining lots to create tracts suitable for development or that the Company may offer for sale individually or in small groups. As the Company's land sales continue and the number of the Company's contiguous and highly concentrated lots diminishes, the Company's ability to continue to be in a position to sell lots and generate land sale revenues at satisfactory levels may be adversely affected, which would have an adverse effect on the Company's results of operations.

The Company's real estate assets are diminishing over time, meaning long-term growth in the Company's real estate business will require the acquisition of additional real estate assets, possibly by expanding into new markets.

Substantially all of the Company's real estate revenues are derived from sales of its core inventory in Rio Rancho, New Mexico. This property was acquired nearly 50 years ago, and each time the Company develops and sells real estate to customers in Rio Rancho, these real estate assets diminish. As of April 30, 2013, the Company owned approximately 17,345 acres in Rio Rancho out of an original purchase of approximately 91,000 acres. The continuity and future growth of the Company's real estate business, if the Company pursues such growth, will require that the Company acquire new properties in or near Rio Rancho or expand to other markets to provide sufficient assets to support a meaningful real estate development business. While the Company owns two properties in Colorado, the Company has not for many years made any significant attempt to identify a development opportunity similar to the one the Company has undertaken in Rio Rancho and has no current plans to do so. If the Company does not acquire new real estate assets, its real estate holdings will continue to diminish, which will adversely affect the Company's ability to continue its real estate operations.

The Company may not be able to acquire properties or develop them successfully.

If the Company elects to pursue and is able to identify real estate development opportunities outside of Rio Rancho, the success of the Company's real estate business will depend in large part upon its ability to acquire additional

properties on satisfactory terms and to develop them successfully. If the Company is unable to do so, its results of operations could be adversely affected.

The acquisition, ownership and development of real estate are subject to many risks that may adversely affect the Company's results of operations, including risks that:

the Company may not be able to acquire a desired property because of competition from other real estate developers or investors who may have greater capital or better access to cash than the Company;

the Company may not be able to obtain or renew financing on acceptable terms, or at all;

an adverse change in market conditions during the interval between acquisition and sale of a property may result in a lower than originally anticipated profit;

the Company may underestimate the cost of development required to bring an acquired property up to standards established for the market position intended for that property;

acquired properties may be located in new markets where the Company may face risks associated with a lack of market knowledge or understanding of the local economy, a lack of business relationships in the area or unfamiliarity with local governmental and permitting procedures; and

the Company may be unable to quickly and efficiently integrate new acquisitions, particularly acquisitions of portfolios of properties, into the Company's existing operations.

In addition, the Company currently has approximately 450 developed lots available for sale in Rio Rancho. Development activities performed in connection with real estate sales include obtaining necessary governmental approvals, acquiring access to water supplies, installing utilities and necessary storm drains and building or improving roads. The development of additional lots for sale in Rio Rancho will require additional financing or other sources of funding, which may not be available. If the Company is unable to obtain such financing, its results of operations could be adversely affected.

The Company's real estate development activities have been primarily limited to a single market, and it may face substantially more experienced competition in acquiring and developing real estate in new markets.

Since the Company's real estate acquisition and development activities have been primarily limited to the Rio Rancho market, the Company does not have extensive experience in acquiring real estate in other markets or engaging in development activities in multiple markets simultaneously. Should the Company seek to acquire additional real estate in new markets, competition from other potential purchasers of real estate could adversely affect the Company's operations. Many of these entities may have substantially greater experience than the Company has in identifying, acquiring and developing real estate opportunities in other markets and in managing real estate developments in multiple markets. These entities may also have greater financial resources than the Company has and may be able to pay more than the Company can or accept more risk than the Company is willing to accept to acquire real estate. These entities also may be less sensitive to risks with respect to the costs or the geographic concentration of their investments. This competition may prevent the Company from acquiring the real estate assets the Company seeks, or increase the cost of properties that the Company does acquire. Competition may also reduce the number of suitable investment opportunities available to the Company or may increase the bargaining power of property owners seeking to sell.

The Company will likely compete for real estate investment opportunities with, among others, insurance companies, pension and investment funds, partnerships, real estate and housing developers, investment companies, real estate investment trusts and owner/occupants.

Properties that the Company acquires may have defects that are unknown to the Company.

Although the Company would expect to perform due diligence on prospective properties before they are acquired, and on a periodic basis after acquisition, any of the properties the Company may acquire may have characteristics or deficiencies unknown to the Company that could adversely affect the property's value or revenue potential or, in the case of environmental or other factors, impose liability on the Company, which could be significant.

Competition from other developers and land owners may adversely affect the Company's ability to sell lots and generate land sale revenues at satisfactory levels.

The real estate business is highly competitive and fragmented. The Company competes with many developers and land owners of varying sizes ranging from local to national level businesses to complete land transactions. These developers and land owners may have greater financial, marketing, sales and other resources than those of the Company. The potential parties interested in the Company's land holdings are primarily residential and commercial developers and commercial businesses and competition for these transactions is based on, among other things, price, location, land use, optionality of land use, and other development activities in the surrounding area. Given the geographic concentration of the Company's real estate assets in Rio Rancho, which is adjacent to Albuquerque, New Mexico, the Company is subject to competition from land development in these areas. Three large real estate parcels not owned by the Company are under development in Albuquerque and Rio Rancho that will result in additional competition for the sale of the Company's developed lots. Competition with other developers and land owners in Rio Rancho and Albuquerque may adversely affect the Company's ability to sell lots and generate land sale revenues at satisfactory levels, which would have an adverse effect on the Company's results of operations.

The Company is subject to substantial legal, regulatory and other requirements regarding the development of land, including obtaining government approvals, which may be delayed or denied, and therefore the Company may encounter difficulties in obtaining entitlements on a timely basis, which could limit its ability to sell land.

There are many legal, regulatory and other requirements regarding the development of land, which may delay the start of planned development activities, increase the Company's expenses or limit the Company's customers' development activities. Development activities performed in connection with real estate sales include obtaining necessary governmental approvals, acquiring access to water supplies, installing utilities and necessary storm drains and building or improving roads. Numerous local, state and federal statutes, ordinances and rules and regulations, including those concerning zoning, resource protection and the environment, regulate these tasks. These regulations often provide broad discretion to the governmental authorities that regulate these matters and from whom the Company must obtain necessary approvals. The approval process can be lengthy and delays can increase the Company's costs, as well as the costs for the primary customers of the Company's real estate business (residential and commercial developers). Failure to obtain necessary approvals on a timely basis may significantly adversely affect the Company's real estate development activities and its results of operations.

The Company may be subject to environmental liability.

Various laws and regulations impose liability on real property owners and operators for the costs of investigating, cleaning up and removing contamination caused by hazardous or toxic substances at a property. In the Company's role as a property owner or developer, the Company could be liable for such costs. This liability may be imposed without regard to the legality of the original actions and without regard to whether the Company knew of, or was responsible for, the presence of the hazardous or toxic substances. If the Company fails to disclose environmental issues, it could also be liable to a buyer or lessee of the property. In addition, some environmental laws create a lien on the contaminated site in favor of the government for damages and costs incurred by the government in connection with the contamination. If the Company incurs any such liability that is material, its results of operations and financial position could be materially adversely affected.

Increases in taxes or governmental fees may increase the Company's costs. Also, adverse changes in tax laws could reduce customer demand for land for commercial and residential development.

Increases in real estate taxes and other local governmental fees, such as fees imposed on developers to fund schools, open space and road improvements or to provide low and moderate income housing, would increase the Company's costs and have an adverse effect on the Company's operations. Municipal and state resources have been particularly

strained as a consequence of the economic downturn that began in 2008 and as a result, many governmental entities have adopted significant tax increases. The Company cannot control these tax increases and may not be able to pass such increased costs on to purchasers, particularly as it generally holds properties for many years. In addition, further increases in local real estate taxes or changes in income tax laws that would reduce or eliminate tax deductions or incentives related to real estate would increase the Company's expenses and could adversely affect homebuilders' potential customer demand and could adversely affect the Company's future land sales to those homebuilders.

Unless the City of Rio Rancho supplements its current water supply, development of the Company's remaining Rio Rancho land may be adversely affected.

All of the Company's future Rio Rancho land development will require water service from the City of Rio Rancho or from another source. While the city has not denied any development in the past due to a shortage of water supply, it has expressed concerns that its current water supply cannot support growth indefinitely. Although the city is currently pursuing various methods to supplement its water supply, if it is unsuccessful, development of the Company's remaining Rio Rancho land could be restricted or adversely affected.

Real estate is a cyclical industry, and the Company's results of operations could be adversely affected during cyclical downturns in the industry.

During periods of economic expansion, the real estate industry typically benefits from an increased demand for land. In contrast, during periods of economic contraction, the real estate industry is typically adversely affected by a decline in demand for land. For example, increased rates of residential mortgage defaults that began in early calendar 2007 led to significant losses for the companies holding such mortgages and contributed to a severe and continuing downturn in the residential housing market. Further, real estate development projects typically begin, and financial and other resources are committed, long before such projects come to market, which could be during a time when the real estate market is depressed. There can be no assurance that an increase in demand or an economic expansion will occur or be sustained in the Rio Rancho market, where the Company's core real estate business is based and operates, or in any new market into which the Company may expand its real estate operations. Any of the following (among other factors, including those mentioned elsewhere in these Risk Factors) could cause a general decline in the demand for residential or commercial real estate which, in turn, could contribute to a downturn in the real estate development industry that could have an adverse effect on the Company's results of operations:

- periods of general economic slowdown or recession;
- change or uncertainty in government regulation;
- rising interest rates or a decline in the general availability or affordability of mortgage financing;
- adverse changes in local or regional economic conditions;
- shifts in population away from the markets that the Company serves;

tax law changes, including potential limits on, or elimination of, the deductibility of certain mortgage interest expense, real property taxes and employee relocation expenses, and uncertainty with respect to these matters; or

acts of God, including hurricanes, earthquakes and other natural disasters.

Changing credit conditions may adversely affect companies in the real estate industry, which rely upon credit in order to finance their purchases of land from the Company.

Changes in interest rates and other economic factors can dramatically affect the availability of capital for the Company's developer customers. Residential and commercial developers to whom the Company frequently sells land typically rely upon third party financing to provide the capital necessary for their acquisition of land. Changes in economic and other external market conditions can and have resulted in the inability of developers to obtain suitable financing, which has, and so long as these circumstances continue will, adversely impact the Company's ability to sell land and could force the Company to sell land at lower prices, which would adversely affect its results of operations.

Changes in general economic, real estate development or other business conditions may adversely affect the Company's business and its financial results.

A significant percentage of the Company's real estate revenues have historically been derived from customers in the residential homebuilding business, which is particularly sensitive to changes in economic conditions and factors such as the level of employment, consumer confidence, consumer income, availability of mortgage financing and interest rates. Adverse changes in these conditions have decreased demand for homes generally, and may continue to do so,

adversely affecting the pricing of homes and in turn the price of land sold to developers, which could adversely affect the Company's results of operations.

The real estate that the Company holds is periodically the subject of third party appraisals used for accounting purposes, which may result in the incurrence of non-cash impairment charges.

The Company recently received third party appraisals of real estate inventory that had been sold but was subsequently repossessed ("take-back lots"). When repossessed, take-back lots are initially taken into inventory at fair market value less estimated costs to sell, and subsequently evaluated for impairment. The recently received appraisals resulted in the carrying value of certain take-back lots located in Rio Rancho being adjusted from \$4,779,000 to fair value of \$3,437,000 resulting in a pre-tax impairment charge of \$1,342,000, which has been included in the results of operations for the fourth quarter of 2013. Changes in economic and other external market conditions may further adversely impact the fair market value of the Company's real estate inventory, which could lead to additional impairment charges that could adversely affect the Company's results of operations.

A number of contracts for individual Rio Rancho home site sales made prior to 1977 require the Company to exchange land in an area that is serviced by utilities for land in areas where utilities are not installed.

In connection with certain individual home site sales made prior to 1977 at Rio Rancho, New Mexico, if water, electric and telephone utilities have not reached the lot site when a purchaser is ready to build a home, the Company is obligated to exchange a lot in an area then serviced by such utilities for the lot of the purchaser, without cost to the purchaser. Although this has not been the case in the past, if the Company were to experience a large number of requests for such exchanges in the future, its results of operations could be adversely impacted.

If subcontractors are not available to assist in completing the Company's land development projects, the Company may not be able to complete those projects on a timely basis.

The development of land on a timely basis is critical to the Company's ability to complete development projects in accordance with the Company's contractual obligations. The availability of subcontractors in the markets in which the Company operates can be affected by factors beyond the Company's control, including the general demand for these subcontractors by other developers. If subcontractors are not available when the Company requires their services, the Company may experience delays or be forced to seek alternative suppliers, which may increase costs or adversely affect the Company's ability to sell land on a timely basis.

Land investments are generally illiquid, and the Company may not be able to sell its properties when it is economically or otherwise important to do so.

Land investments generally cannot be sold quickly, and the Company's ability to sell properties has been and may continue to be affected by market conditions. The Company may not be able to diversify concentration risk or vary its portfolio quickly in response to economic or other conditions. The Company's ability to pay down debt, reduce interest costs and acquire properties is dependent upon its ability to sell the properties it has selected for disposition at the prices and within the deadlines the Company has established for each property.

Other Risks

The Company may engage in future acquisitions and may encounter difficulties in integrating the acquired businesses, and, therefore, may not realize the anticipated benefits of the acquisitions in the time frames anticipated, or at all.

From time to time, the Company may seek to grow through strategic acquisitions intended to complement or expand one or more of its businesses or to enable the Company to enter a new business. The success of these transactions will

depend in part on the Company's ability to integrate the systems and personnel acquired in these transactions into its existing business without substantial costs, delays or other operational or financial problems. The Company may encounter difficulties in integrating acquisitions with the Company's operations or in separately managing a new business. Furthermore, the Company may not realize the degree of benefits that the Company anticipates when first entering into a transaction, or the Company may realize benefits more slowly than it anticipates. Any of these problems or delays could adversely affect the Company's results of operations.

The Company does not have a Chief Executive Officer.

On January 22, 2013, Theodore J. Gaasche, the Company's President and Chief Executive Officer, resigned from those positions which he had held since August 2011. No successor has been appointed or identified, and the Company may continue to operate without a Chief Executive Officer for the foreseeable future. From January 1996 until August 2011, the Company also operated without a Chief Executive Officer. Following his resignation as President and Chief Executive Officer, the Company's board of directors elected Mr. Gaasche as a member of the board of directors and Vice Chairman of the Executive Committee of the board of directors. In the absence of a Chief Executive Officer, the Executive Committee is charged with the oversight of the Company's business between meetings of the board of directors. However, the Executive Committee does not function as an executive officer. Without a Chief Executive Officer, the Company does not have the benefit of the operational oversight and strategic leadership that a person in that position would normally provide, which may adversely affect the Company's business and prospects.

The Company's current management and internal systems may not be adequate to handle the Company's growth, if any.

To manage the Company's future growth, if any, the Company's management must continue to improve operational and financial systems and to expand, train, retain and manage the Company's employee base. If the Company grows, it will also likely need to recruit and retain additional qualified management personnel, and its ability to do so will depend upon a number of factors, including the Company's results of operations and prospects and the level of competition then prevailing in the market for qualified personnel. At the same time, the Company will likely be required to manage an increasing number of relationships with various customers and other parties. If the Company's management personnel, systems, procedures and controls are inadequate to support its operations, expansion could be slowed or halted and the opportunity to gain significant additional market share could be impaired or lost. Any inability of the Company's management to manage the Company's growth effectively may adversely affect its results of operations.

The Company's business could be seriously harmed if the Company's accounting controls and procedures are circumvented or otherwise fail to achieve their intended purposes.

Although the Company evaluates its internal controls over financial reporting and its disclosure controls and procedures at the end of each quarter, any system of controls, however well designed and operated, is based in part on certain assumptions and can provide only reasonable, not absolute, assurances that the objectives of the system will be met. Any failure or circumvention of the controls and procedures or failure to comply with regulations related to controls and procedures could have a material adverse effect on the Company's results of operations.

In addition, there can be no assurance that the Company's internal control systems and procedures, or any future acquisitions and their respective internal control systems and procedures, will not result in or lead to a material weakness in the Company's internal controls, or that the Company or its independent registered public accounting firm will not identify a material weakness in the Company's internal controls in the future. If the Company's internal controls over financial reporting are not considered adequate, the Company's financial statements could become incorrect or misleading and the Company may experience a loss of public confidence, which could subject the Company to liability and have an adverse effect on the Company's business and the price of the Company's common stock.

Further, deficiencies or weaknesses that the Company has not yet identified could emerge and the identification and correction of those deficiencies or weaknesses could have an adverse effect on the Company's results of operations.

The Company's quarterly and annual operating results can fluctuate significantly.

The Company has experienced, and is likely to continue to experience, significant fluctuations in its quarterly and annual operating results, which may adversely affect the Company's stock price. Future quarterly and annual operating results may not align with past trends as a result of numerous factors, including many factors that result from the unpredictability of the nature and timing of land sales, the variability in gross profit margins and competitive pressures.

Changes in the Company's income tax estimates could affect profitability.

In preparing the Company's consolidated financial statements, significant management judgment is required to estimate the Company's income taxes. The Company's estimates are based on its interpretation of federal and state tax laws and regulations. The Company estimates actual current tax due and assesses temporary differences resulting from differing treatment of items for tax and accounting purposes. The temporary differences result in deferred tax assets and liabilities, which are included in the Company's consolidated balance sheet. Adjustments may be required by a change in assessment of the Company's deferred tax assets and liabilities, changes due to audit adjustments by federal and state tax authorities and changes in tax laws. To the extent adjustments are required in any given period, the Company will include the adjustments in the tax provision in its financial statements. These adjustments could have an adverse effect on the Company's financial position, cash flows and results of operations.

The price of the Company's common stock in recent years has been volatile and has been subject to above average selling pressure from time to time. This may make it difficult for shareholders to sell the Company's common stock, and the sale of substantial amounts of the Company's common stock could adversely affect the price of the Company's common stock.

The market price for the Company's common stock varied between a high of \$15.77 and a low of \$4.80 per share between May 1, 2010 and April 30, 2013. This volatility may make it difficult for a shareholder to sell the Company's common stock, and the sale of substantial amounts of the Company's common stock could adversely affect the price of the common stock. The Company believes the price of its common stock has been adversely affected from time to time by above average selling pressure created by institutions that have received charitable gifts of the Company's common stock and engaged in selling programs to liquidate such positions. From October 3, 2011 to May 17, 2013, the Company's largest shareholder, Nicholas Karabots, has donated shares he has beneficially owned representing approximately 24% of the Company's outstanding common stock to such institutions. The stock price may continue to be volatile and subject to significant price fluctuations in response to market and other factors (including those mentioned elsewhere in these Risk Factors), and:

- variations in the Company's quarterly and annual operating results, which could be significant;
- material announcements by the Company or the Company's competitors;
- sales of a substantial number of shares of the Company's common stock; and
- adverse changes in general economic or market conditions.

In addition to the factors discussed above, the Company's common stock is often thinly traded. As a result large transactions in the Company's common stock may be difficult to execute in a short time frame and may cause significant fluctuations in the price of the Company's common stock. Among other reasons, the stock is thinly traded due to the fact that four of the Company's shareholders beneficially own a substantial majority of the outstanding common stock. The average trading volume in the Company's common stock on the New York Stock Exchange over the ten-day trading period ending on April 30, 2013 was approximately 10,100 shares per day. This thin trading market, when combined with additional selling pressure, including selling by institutions that receive charitable gifts as described above, can result in overall negative stock price and trading trends, which can last as long as the additional selling pressure continues. Further, there have been, from time to time, significant "short" positions in the Company's common stock, consisting of borrowed shares sold, or shares sold for future delivery, which may not have been borrowed. Any attempt by the short sellers to liquidate their positions over a short period of time could cause significant volatility in the price of the Company's common stock.

In the past, following periods of volatility in the market price of their stock, many companies have been the subject of securities class action litigation. The Company has not been involved in any securities class action litigation; however, if the Company were to become involved in securities class action litigation in the future, it could result in substantial costs and diversion of the Company's management's attention and resources and could harm the Company's stock price, business, prospects, results of operations and financial condition. In addition, the broader stock market has experienced significant price and volume fluctuations in recent years. This volatility has affected the market prices of securities issued by many companies for reasons unrelated to their operating performance and may adversely affect the price of the Company's common stock.

The Company has a significant shareholder whose interests may conflict with those of other investors.

The Company has a significant shareholder, Nicholas G. Karabots, who, together with certain of his affiliates, beneficially owns 34.8% of the Company's outstanding common stock. Because of his significant voting power, this shareholder, who resigned as a member of the Company's board of directors effective January 22, 2013, could influence the Company to make decisions that might run counter to the wishes of the Company's other shareholders. Additionally, Mr. Karabots may be able to influence the Company's board of directors and management as he (including through companies that he controls) (i) is the employer of Theodore J. Gaasche, a director on the Company's board of directors and Vice Chairman of the Executive Committee of the Company's board of directors, (ii) has a commercial relationship with Lonnie A. Coombs (another member of the Company's board of directors) who is also a director of a private company owned by Mr. Karabots, (iii) is the father-in-law of Michael P. Duloc, who is the President and Chief Executive Officer of the Company's Media Services businesses and (iv) is the owner of Kappa Lending Group, LLC, a company that holds \$16,007,000 of debt of AMREP Southwest, and which could exert substantial additional control over AMREP Southwest and the Company in the event of certain future breaches or defaults concerning such debt. In addition, a publishing company owned by Mr. Karabots is the largest customer of the Company's Newsstand Distribution Services business, as well as a customer of the Company's Subscription Fulfillment Services business. For its fiscal years ended April 30, 2012, the Company's revenues from the newsstand distribution and fulfillment services it provided to Mr. Karabots' publishing company amounted to \$1,545,000, which was approximately 1.8% of the Company's consolidated revenues for that period. For its fiscal years ended April 30, 2013, the Company's revenues from the newsstand distribution and fulfillment services it provided to Mr. Karabots' publishing company amounted to \$1,458,000, which was approximately 1.8% of the Company's consolidated revenues for that period. As a result, Mr. Karabots may have business interests with respect to the Company that differ from or conflict with the interests of other holders of the Company's common stock.

Although the Company has paid dividends in the past, no dividends have been paid since 2008; the Company has no regular dividend policy and can give no assurance of any future dividends.

The Company has not paid any cash dividends on its common stock since fiscal year 2008. The board of directors has stated that it may consider special dividends from time-to-time in the future in light of conditions then existing, including earnings, financial condition, cash position, and capital requirements and other needs. No assurance is given that there will be any such future dividends declared.

Certain provisions of Oklahoma law and the Company's organizational documents may impede or discourage a takeover, which may have a limiting effect on the market price of the Company's common stock.

The Company is an Oklahoma corporation and the anti-takeover provisions of its certificate of incorporation and of Oklahoma law generally prohibit the Company from engaging in "business combinations" with an "interested shareholder," as those terms are defined therein, unless the holders of at least two-thirds of the Company's then outstanding common stock approve the transaction. Consequently, the concurrence of the Company's significant shareholder, Mr. Karabots and his affiliates, is needed for any third party (other than Mr. Karabots and his affiliates) to acquire control of the Company, even if a change in control would be beneficial to the Company's other shareholders. In addition to this restriction, some other provisions of the Company's certificate of incorporation and by-laws may discourage certain acts that would involve the Company undergoing a fundamental change. For example, the Company's certificate of incorporation and its by-laws contain certain provisions that:

elassify the Company's board of directors into three classes, each of which serves for a term of three years, with one class being elected each year; and

prohibit shareholders from calling a special meeting of shareholders.

Because the Company's board of directors is classified and the Company's certificate of incorporation and by-laws do not otherwise provide, Section 1027 of the Oklahoma General Corporation Act permits the removal of any member of the board of directors only for cause.

These factors could impede a merger, takeover or other business combination involving the Company or discourage a potential acquirer from making a tender offer for the Company's common stock, which could have a limiting effect on the market price of the Company's common stock.

Item 1B. Unresolved Staff Comments

Not applicable.

Item 2. Properties

The Company's executive offices are located in approximately 2,600 square feet of leased space in an office building in Princeton, New Jersey. The Company's Subscription Fulfillment Services operation is located in six owned or leased facilities in Palm Coast, Florida and two locations in Colorado comprising approximately 258,000 square feet of space. The Company's Newsstand Distribution Services and Product Packaging and Fulfillment Services and Other businesses are primarily located in nine owned or leased facilities comprising approximately 321,000 square feet of space, with an executive office located in New York City and other principal locations being in Mt. Morris, Illinois and Fairfield, Ohio. Real Estate operations are based in approximately 4,900 square feet of leased space in an office building in Rio Rancho, New Mexico. In addition, other real estate inventory and investment properties are described in Item 1. The Company believes its facilities are adequate for its current requirements.

Item 3. Legal Proceedings

In March 2009, a civil action was commenced against the Company's wholly-owned subsidiary in the United States District Court for the Southern District of New York entitled Anderson News, L.L.C., et al. v. American Media, Inc., et al. Anderson News, L.L.C. ("Anderson") was a wholesaler of magazines. Anderson has alleged that magazine publishers and distributors, including the Company's subsidiary, Kable Distribution Services, Inc. ("Kable Distribution"), conspired to boycott Anderson to drive it out of business, and that other wholesalers participated in this effort. Anderson has asserted claims under Section 1 of the Sherman Act (antitrust), for defamation, for tortious interference with its contracts with retailers, and for civil conspiracy. Damages have not been quantified, but would presumably be alleged to be substantial. Anderson has alleged that the distributor and publisher defendants acted in concert to cut off Anderson from its supply of magazines to enable them to gain control of the single-copy magazine distribution channel. The amended complaint in the lawsuit was filed on September 7, 2012. On September 24, 2012, Kable Distribution filed its answer to the amended complaint denying all allegations of wrongdoing against it. Discovery in the case is continuing. Kable Distribution is vigorously defending the lawsuit. The Company is not in a position to predict the outcome of the lawsuit, nor can it estimate a range of possible losses.

On July 11, 2011, Kable Distribution was served with a summons and complaint in a lawsuit entitled Distribution Integrated Services, Inc. v. Kable Distribution Services, Inc.; Island Periodicals Puerto Rico, LLC brought in the Tribunal de Primera Instancia, Sala de San Juan, in Puerto Rico. Kable Distribution's co-defendant, Island Periodicals Puerto Rico, LLC, is a sub-distributor of magazines for Kable Distribution in Puerto Rico, a position formerly held by plaintiff. In the lawsuit, plaintiff has alleged that the termination by Kable Distribution of plaintiff's former sub-distributorship arrangement with Kable Distribution was in breach of a contract between them, and therefore in violation of Puerto Rico Law 75, a statute that provides remedies to a dealer in property for the unjustified termination of its dealership arrangement. Plaintiff is seeking damages from Kable Distribution in the amount of \$2,000,000 and injunctive relief. In September 2012, an intermediate appellate court reversed the previous decision of the lower court denying the plaintiff's request for a preliminary injunction restoring the plaintiff as Kable Distribution's subdistributor while the lawsuit continues. The defendants have appealed the intermediate appellate court's action to the Puerto Rico Supreme Court and issuance of the preliminary injunction has been stayed pending the outcome of the appeal. Kable Distribution's co-defendant has agreed to indemnify it against the claims asserted by plaintiff. Kable Distribution is vigorously defending the lawsuit. The Company is not in a position to predict the outcome of the lawsuit, nor can it estimate a range of possible losses or whether the co-defendant will be able to indemnify the Company.

On March 29, 2013, Kable Distribution, Kable Media Services, Inc. and Kable News Company, Inc. (collectively, the "Kable Patent Defendants") were named in the fourth amended complaint for patent infringement in the United States District Court for the District of Utah, Central Division in a lawsuit entitled Etagz, Inc. v. Berkeley Publications, Inc., et al. On April 1, 2013, the Kable Patent Defendants were notified that they had been named as defendants in the lawsuit and received service of process of the complaint on May 10, 2013. The Kable Patent Defendants are accused of infringing certain patents owned by Etagz, Inc. in connection with the distribution of certain DVDs and magazines published by Berkeley Publications, Inc., one of the co-defendants in the lawsuit. Berkeley Publications, Inc. has agreed to indemnify the Kable Patent Defendants against the claims asserted by plaintiff. The Kable Patent Defendants are vigorously defending the lawsuit. The Company is not in a position to predict the outcome of the lawsuit, nor can it estimate a range of possible losses or whether the co-defendant will be able to indemnify the Company.

The Company and its subsidiaries are involved in various other claims and legal actions arising in the normal course of business. While the ultimate results of these matters cannot be predicted with certainty, management believes that they will not have a material adverse effect on the Company's consolidated financial position, liquidity or results of operations.

Item 4. Mine Safety Disclosures

Not applicable.

Executive Officers of the Registrant

Set forth below is certain information concerning persons who are the current executive officers of the Company.

Peter M. Pizza, age 62, has been Vice President and Chief Financial Officer of the Company since 2001 and was Vice President and Controller of the Company from 1997 to 2001.

Christopher V. Vitale, age 37, has been Vice President, General Counsel and Secretary of the Company since March 2013. From April 2012 to March 2013, he was Vice President, Legal at Franklin Square Holdings, L.P. and from August 2011 to March 2012, he was Assistant Vice President, Legal at Franklin Square Holdings, L.P., a national sponsor and distributor of investment products, where he was responsible for securities matters, corporate governance and general corporate matters. From March 2011 to July 2011, Mr. Vitale was the Chief Administrative Officer at WorldGate Communications, Inc. ("WorldGate"), and from April 2009 to July 2011 he was Senior Vice President, General Counsel and Secretary at WorldGate, a provider of digital voice and video phone services and video phones, where he was responsible for managing legal, governance, compliance, litigation, securities and governmental affairs. In 2012, WorldGate filed a voluntary petition for relief under Chapter 7 of the United States Bankruptcy Code in the United States Bankruptcy Court for the District of Delaware. Prior to joining WorldGate, Mr. Vitale was an attorney with the law firms of Morgan, Lewis & Bockius LLP and Sullivan & Cromwell LLP.

Michael P. Duloc, age 56, was appointed President and Chief Executive Officer of the Company's Media Services businesses effective August 1, 2011. Since 1993 Mr. Duloc has held various senior executive positions with Media Services companies, including President of Kable Media Services, Inc. since 2007, President of Kable Distribution since 1996, and President of Kable Fulfillment Services, Inc., one of the predecessors of Palm Coast Data LLC, from 2000 until January 2007.

The executive officers are elected or appointed by the board of directors of the Company or its appropriate subsidiary to serve until the appointment or election and qualification of their successors or their earlier death, resignation or removal.

PART II

Many of the amounts and percentages presented in this Part II have been rounded for convenience of presentation. All references in this Part II to 2013, 2012 and 2011 mean the Company's fiscal years ended April 30, 2013, 2012 and 2011, unless otherwise qualified.

Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities

The Company's common stock is traded on the New York Stock Exchange under the symbol "AXR". On June 30, 2013, there were approximately 790 holders of record of the common stock. The range of high and low sales prices of the common stock on the New York Stock Exchange for the last two fiscal years by quarter is presented below:

	Pric	Price Range	
	High	Low	
Fiscal 2012			
First Quarter	\$10.19	\$8.60	
Second Quarter	\$8.95	\$6.06	
Third Quarter	\$8.74	\$5.64	
Fourth Quarter	\$8.99	\$6.00	
		Price Range	
	Pric	e Range	
	Pric High	e Range Low	
Fiscal 2013		U	
Fiscal 2013 First Quarter		U	
	High	Low	
First Quarter	High \$8.17	Low \$5.80	
First Quarter Second Quarter	High \$8.17 \$10.40	Low \$5.80 \$4.80	

Dividend Policy

The Company has paid no cash dividends on its common stock since fiscal 2008. The board of directors has stated that it may consider special dividends from time-to-time in the future in light of conditions then existing, including earnings, financial condition, cash position, capital requirements and other needs. No assurance is given that there will be any such future dividends declared.

Equity Compensation Plan Information

See Item 12 of this annual report on Form 10-K that incorporates such information by reference from the Company's Proxy Statement for its 2013 Annual Meeting of Shareholders.

Item 6. Selected Financial Data

Not required.

Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

INTRODUCTION

For a description of the Company's business, refer to Item 1 of Part I of this annual report on Form 10-K.

As indicated in Item 1, the Company, through its subsidiaries, is primarily engaged in four business segments: the Subscription Fulfillment Services business operated by Palm Coast Data LLC ("Palm Coast"), the Newsstand Distribution Services business and the Product Packaging and Fulfillment Services and Other businesses operated by Kable Media Services, Inc. and its subsidiaries ("Kable") (the businesses operated by Palm Coast and Kable are collectively referred to as "Media Services"), and the real estate business operated by AMREP Southwest Inc. and its subsidiaries (collectively, "AMREP Southwest"). On December 31, 2012, a newly-formed wholly-owned subsidiary of Palm Coast, FulCircle Media, LLC ("FulCircle"), acquired certain assets from a third party. The results of this

new subsidiary are included in the Subscription Fulfillment Services business since December 31, 2012. Data concerning industry segments is set forth in Note 19 of the notes to the consolidated financial statements included in this annual report on Form 10-K. The Company's foreign sales and activities are not significant.

The following provides information that management believes is relevant to an assessment and understanding of the Company's consolidated results of operations and financial condition. The discussion should be read in conjunction with the consolidated financial statements and accompanying notes.

CRITICAL ACCOUNTING POLICIES AND ESTIMATES

The Company prepares its financial statements in conformity with accounting principles generally accepted in the United States of America. The Company discloses its significant accounting policies in the notes to its audited consolidated financial statements.

The preparation of such financial statements requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosures of contingent assets and liabilities at the dates of those financial statements as well as the reported amounts of revenues and expenses during the reporting periods. Areas that require significant judgments and estimates to be made include: (i) the determination of revenue recognition for the Newsstand Distribution Services business, which is based on estimates of allowances for magazine returns to the Company from wholesalers and the offsetting returns of magazines by the Company to publishers for credit; (ii) allowances for doubtful accounts; (iii) real estate cost of sales calculations, which are based on land development budgets and estimates of costs to complete; (iv) cash flow and valuation assumptions in performing asset impairment tests of long-lived assets, goodwill and assets held for sale; (v) actuarially determined benefit obligations and other pension plan accounting and disclosures; (vi) risk assessment of uncertain tax positions; (vii) the determination of the recoverability of net deferred tax assets; and (viii) legal contingencies. Actual results could differ from those estimates.

There are numerous critical assumptions that may influence accounting estimates in these and other areas. Management bases its critical assumptions on historical experience, third-party data and various other estimates that it believes to be reasonable under the circumstances. The most critical assumptions made in arriving at these accounting estimates include the following:

- Newsstand Distribution Services revenues represent commissions earned from the distribution of publications for client publishers, which are recorded by the Company at the time the publications go on sale in accordance with the Financial Accounting Standards Board ("FASB") Accounting Standards Codification ("ASC") 605. The publications generally are sold on a fully returnable basis, which is in accordance with prevailing trade practice. Accordingly, the Company provides for estimated returns by charges to income that are determined on an issue-by-issue basis utilizing historical experience and current sales information. The financial impact to the Company of a change in the sales estimate for magazine returns to it from its wholesalers is substantially offset by the simultaneous change in the Company's estimate of its cost of purchases since it passes on the returns to publishers for credit. As a result, the effect of a difference between the actual and estimated return rates on the Company's commission revenues is the amount of the commission attributable to the difference. The effect of an increase or decrease in the Company's estimated rate of returns of 1% during any period would be dependent upon the mix of magazines involved and the related selling prices and commission rates, but would generally result in a change in that quarter's net commission revenues of approximately \$83,000;
- management determines the allowance for doubtful accounts by attempting to identify troubled accounts by analyzing the credit risk of specific customers and by using historical experience applied to the aging of accounts and, where appropriate within the real estate business, by reviewing any collateral which may secure a receivable;

- real estate development costs are incurred throughout the life of a project, and the costs of initial sales from a project frequently must include a portion of costs that have been budgeted based on engineering estimates or other studies, but not yet incurred;
- asset impairment determinations are based upon the intended use of assets, expected future cash flows and estimates of fair value of assets;

- benefit obligations and other pension plan accounting and disclosures are based upon numerous assumptions and estimates, including the expected rate of investment return on retirement plan assets, the discount rate used to determine the present value of liabilities, and certain employee-related factors such as turnover, retirement age and mortality. As of April 30, 2013, the effect of every 0.25% change in the investment rate of return on retirement plan assets would increase or decrease the pension expense by approximately \$65,000 per year, and the effect of every 0.25% change in the discount rate would increase or decrease the subsequent year's pension cost by approximately \$45,000;
- the Company assesses risk for uncertain tax positions and recognizes the financial statement effects of a tax position when it is more likely than not that the position will be sustained upon examination by tax authorities;
- projected Company earnings (including currently unrealized gains on real estate inventory) for the recoverability of net deferred tax assets in the future; and
- the Company is currently involved in legal proceedings which are described in Part I, Item 3 of this annual report on Form 10-K and the Company estimates whether the legal proceedings relate to a probable loss and if so, then an estimate of probable loss within a range of potential probable losses is made for accrual in the financial statements. It is possible that the consolidated financial position or results of operations for any particular quarterly or annual period could be materially affected by an outcome of litigation that is significantly different from the Company's assumptions. The Company does not accrue for expected future legal costs for such proceedings.

RESULTS OF OPERATIONS

Year Ended April 30, 2013 Compared to Year Ended April 30, 2012

For 2013, the Company recorded a net loss of \$2,838,000, or \$0.47 per share, compared to a net loss of \$1,143,000, or \$0.19 per share, in 2012. The results for 2013 included pre-tax, non-cash impairment charges of \$1,511,000 (\$952,000 after tax, or \$0.16 per share) while the results for 2012 included a pre-tax, non-cash impairment charge of \$570,000 (\$359,000 after tax, or \$0.06 per share), with the charges in both years reflecting the write-down of certain real estate assets. The results for 2013 also included a fourth quarter reserve for doubtful accounts receivable from a magazine wholesaler of \$2,000,000 (\$1,260,000 after tax, or \$0.21 per share). Excluding the impairment charges in both years and accounts receivable reserve in 2013, results for 2013 were a net loss of \$626,000, or \$0.10 per share, compared to a net loss of \$784,000, or \$0.13 per share, for 2012. Revenues for 2013 were \$83,001,000 compared to \$85,360,000 in the prior year.

Revenues from the Company's Media Services operations decreased from \$83,447,000 for 2012 to \$82,105,000 for 2013. Magazine publishers are the principal customers of these businesses, and they have continued to be negatively impacted by increased competition from new media sources and weakness in the U.S. economy. The result has been reduced subscription and newsstand magazine sales, which has caused publishers to close some magazine titles and seek more favorable terms from Palm Coast and Kable and their competitors when contracts are up for bid or renewal. As a consequence of these and other factors, including customer losses, revenues from Subscription Fulfillment Services operations decreased from \$62,230,000 for 2012 to \$57,472,000 for 2013, which includes \$1,759,000 of FulCircle revenues from the acquisition date of December 31, 2012. Revenues from Newsstand Distribution Services operations decreased from \$9,127,000 in 2012 to \$8,867,000 in 2013. Revenues from Product Packaging and Fulfillment Services and Other increased from \$12,090,000 in 2012 to \$15,766,000 in 2013, due to an increase in revenues from both a temporary staffing business and the product packaging and fulfillment services business.

Although there are multiple revenue streams in the Subscription Fulfillment Services business, including revenues from the maintenance of customer computer files and the performance of other fulfillment-related activities, including telephone call center support and graphic arts and lettershop services, a customer generally contracts for and utilizes

all available services as a total package, and the Company would not normally provide ancillary services to a customer unless it is also providing the core service of maintaining a database of subscriber names. Thus, variations in Subscription Fulfillment Services revenues are primarily the result of fluctuations in the number and sizes of customers rather than in the demand for a particular service. This is also true in the Newsstand Distribution Services business where there is only one primary service provided, which results in one revenue source, the commissions earned on the distribution of magazines. The Company competes with other companies, including three

larger companies in the Newsstand Distribution Services business and one larger company in the Subscription Fulfillment Services business, and the competition for new customers is intense in both segments, which results in a price sensitivity that makes it difficult for the Company to increase its prices.

Revenues from land sales at AMREP Southwest decreased from \$1,889,000 in 2012 to \$833,000 in 2013. Results for both periods were substantially lower than the Company experienced prior to fiscal 2009 in its principal market of Rio Rancho, New Mexico, due to the severe decline in the real estate market in the greater Albuquerque-metro and Rio Rancho areas that began late in fiscal 2008. Permits for new home construction in Rio Rancho have generally declined from fiscal 2008 through fiscal 2012. Although the number of permits issued for new home construction in Rio Rancho increased from 328 permits in fiscal 2012 to 490 permits in fiscal 2013, the Company still is confronted with builders using their existing inventories of lots previously purchased from the Company in Rio Rancho and delaying or cancelling the purchase of additional developed lots. The steep decline in the Company's sale of undeveloped land to both builders and investors also reflected these factors.

In Rio Rancho, the Company offers for sale both developed and undeveloped lots to national, regional and local homebuilders, commercial and industrial property developers and others. The Company sold 12 acres of land in 2013, at an average selling price of \$69,000 per acre compared to 104 acres of land in 2012, mostly undeveloped, at an average selling price of \$18,000 per acre, reflecting differences in the mix of properties sold in each period. The average gross profit percentage on land sales before indirect costs was 38% for 2013 compared to 86% for 2012. The sale of undeveloped land in 2012 consisted of land sold to a quasi-governmental agency under threat of condemnation for its use as part of a flood plain, and the nature of the land and its average selling price per acre are not believed to be representative of the Company's inventory and investment properties. As a result of these and other factors, including the nature and timing of specific transactions, revenues, average selling prices and related average gross profits from land sales can vary significantly from period to period and prior results are not necessarily a good indication of what may occur in future periods. In addition, as noted above, AMREP Southwest recorded impairment charges of \$1,511,000 in 2013 and \$570,000 in 2012, which were based on appraisals of portions of AMREP Southwest real estate that in each year showed a deterioration in fair market value from the prior year. Should the adverse real estate market conditions continue, AMREP Southwest may experience future impairment charges.

Operating expenses for the Company's Media Services businesses were \$69,880,000 (85.1% of related revenues) for 2013 compared to \$70,076,000 (84.0% of related revenues) for 2012. The decrease of \$196,000 was primarily due to the net effect of (i) a decrease of \$1,392,000 related to facilities and equipment costs, including depreciation, primarily as a result of assets acquired in 2007 having reached full depreciation and, to a lesser extent, the termination of a warehouse lease at the end of the lease term, (ii) a decrease of \$526,000 related to payroll and benefits costs as a result of both reduced and lost business noted earlier, and (iii) an increase in bad debt expense of \$1,796,000.

Other operating expenses decreased \$286,000 (18%) for 2013 compared to the prior year, primarily due to the net effect of AMREP Southwest's (i) lower real estate tax expense of \$504,000, (ii) increased land maintenance costs of \$118,000 and (iii) increased land appraisal costs of \$26,000. Other operating expenses in 2012 also included a one-time credit of \$73,000 related to the sale of property associated with the Company's restructuring of its Subscription Fulfillment Services business.

General and administrative expenses of Media Services operations decreased \$704,000 (8%) in 2013 compared to 2012, primarily due to the net effect of (i) a decrease of \$643,000 related to payroll and benefits, (ii) a credit to expense for \$310,000 related to an accrual adjustment for an earn-out liability recorded at the time of acquisition of certain assets by FulCircle and (iii) an increase of \$283,000 for outside consulting and legal costs. Real estate operations and corporate general and administrative expenses increased \$69,000 (2%) in 2013 compared to 2012, primarily due to increased actuarially determined pension costs.

The Company's effective tax rate was 38.3% in 2013 compared to 43.9% in 2012. The difference between the statutory tax rate and the effective rate of the tax benefit in both years was primarily due to a reduction of liabilities related to unrecognized tax benefits due to the expiration of the statute of limitations on certain prior year tax benefits in both 2013 and 2012.

Year Ended April 30, 2012 Compared to Year Ended April 30, 2011

For 2012, the Company recorded a net loss of \$1,143,000, or \$0.19 per share, compared to a net loss of \$7,561,000, or \$1.26 per share, in 2011. The results for 2012 included a pre-tax, non-cash impairment charge of \$570,000

(\$359,000 after tax, or \$0.06 per share), reflecting the write-down of certain real estate assets. The results for 2011 included pre-tax, non-cash impairment charges of \$10,720,000 (\$8,194,000 after tax, or \$1.37 per share), reflecting the write-down of certain real estate assets (\$6,827,000 before tax and \$4,301,000, or \$0.72 per share, after tax) and of all of the goodwill of the Company's Newsstand Distribution Services business (\$3,893,000 with no tax benefit, or \$0.65 per share). Excluding the impairment charges in both years, results for 2012 were a net loss of \$783,000, or \$0.13 per share, compared to net income of \$632,000, or \$0.11 per share, for 2011. Revenues for 2012 were \$85,360,000 compared to \$96,837,000 in the prior year.

Revenues from the Company's Media Services operations decreased from \$94,963,000 for 2011 to \$83,447,000 for 2012. Magazine publishers, which are the principal customers of these operations, have continued to be negatively impacted by increased competition from new media distribution sources and also by the effects of the recent recession and the continued weak U.S. economy. The result has been a trend of reduced subscription and newsstand magazine sales, which has caused publishers to close some magazine titles and seek more favorable terms from Palm Coast and Kable and their competitors when contracts are up for bid or renewal. As a consequence of these and other factors, including customer losses, revenues from Subscription Fulfillment Services operations decreased from \$73,618,000 for 2011 to \$62,230,000 for 2012, while revenues from Newsstand Distribution Services operations decreased from \$11,030,000 in 2011 to \$9,127,000 in 2012. Revenues from Product Packaging and Fulfillment Services and Other increased from \$10,315,000 in 2011 to \$12,090,000 in 2012, primarily due to an increase in revenues from a temporary staffing business which were partially offset by declines in revenues from the product packaging and fulfillment services business.

Revenues from land sales at AMREP Southwest increased from \$1,780,000 in 2011 to \$1,889,000 in 2012. Results for both periods were substantially lower than the Company has historically experienced in its principal market of Rio Rancho, New Mexico, due to a severe decline in the real estate market in the greater Albuquerque-metro and Rio Rancho areas that began late in fiscal 2008. The trend of declining permits for new home construction in Rio Rancho also continued, with 15% fewer single-family residential building permits issued during fiscal 2012 than in fiscal 2011. Faced with these adverse conditions, many builders have slowed the pace of building on developed lots previously purchased from the Company in Rio Rancho and delayed or cancelled the purchase of additional developed lots. The steep decline in the Company's sale of undeveloped land to both builders and investors also reflected these factors.

In Rio Rancho, the Company offers for sale both developed and undeveloped lots to national, regional and local homebuilders, commercial and industrial property developers and others. The Company sold 104 acres of land in 2012, mostly undeveloped, at an average selling price of \$18,000 per acre compared to 22 acres of land in 2011, both developed and undeveloped, at an average selling price of \$81,000 per acre, reflecting differences in the mix of properties sold in each period. The average gross profit percentage on land sales before indirect costs was 86% for 2012 compared to 51% for 2011. The sale of undeveloped land in 2012 consisted of land sold to a quasi-governmental agency under threat of condemnation for its use as part of a flood plain, and the nature of the land and resulting average selling price are not believed to be representative of the Company's inventory and investment properties. As a result of these and other factors, including the nature and timing of specific transactions, revenues, average selling prices and related average gross profits from land sales can vary significantly from period to period and prior results are not necessarily a good indication of what may occur in future periods. In addition, as noted above, AMREP Southwest recorded impairment charges of \$570,000 in 2012 and \$6,377,000 in 2011, which were based on appraisals of portions of AMREP Southwest real estate that in each year showed a deterioration in fair market value from the prior year.

Operating expenses for the Company's Media Services businesses were \$70,076,000 (84.0% of related revenues) for 2012 compared to 77,972,000 (82.1% of related revenues) for 2011. The decrease of \$7,896,000 was primarily due to (i) a decrease of \$5,251,000 related to payroll and benefits costs as a result of both reduced and lost business noted earlier and efficiencies achieved in the Company's consolidation of its Subscription Fulfillment Services business from

three locations in Colorado, Illinois and Florida into one location at Palm Coast, Florida that was completed during the second quarter of 2011, and (ii) a decrease of \$1,236,000 related to facilities and equipment costs, including depreciation, primarily as a result of the completed consolidation project.

Interest and other revenues decreased \$70,000 (74%) for 2012 compared to 2011, primarily due to a land lease that expired in 2011 with no similar revenue in 2012.

Other operating expenses decreased \$358,000 (18%) for 2012 compared to the prior year, as costs related to restructuring of the Company's Subscription Fulfillment Service business in 2011 were not duplicated in 2012.

General and administrative expenses of Media Services operations decreased \$602,000 (6%) in 2012 compared to 2011, primarily due to the streamlined operations and increased efficiencies resulting from the Subscription Fulfillment Services business consolidation project. Real estate operations and corporate general and administrative expenses decreased \$373,000 (8%) as 2011 costs associated with the evaluation of a "going private" proposal submitted to the Company by its major shareholder and subsequently withdrawn together with the filing of an S-1 Registration Statement did not reoccur in 2012.

The Company's effective tax rate was 43.9% in 2012 compared to 29.8% in 2011. The difference between the statutory tax rate and the effective rate of the tax benefit was primarily due to (i) a reduction of liabilities related to unrecognized tax benefits due to the expiration of the statute of limitations on certain prior year tax benefits in both 2012 and 2011 and (ii) permanent items, the most significant being the charge against book income associated with non-tax deductible goodwill in 2011.

LIQUIDITY AND CAPITAL RESOURCES

The Company's primary sources of funding for working capital requirements are cash flow from operations, the Company's Media Services banking facility and working capital made available to the Company by customers. The Company's liquidity is affected by many factors, including some that are based on normal operations and some that are related to the industries in which the Company operates and the economy generally. The Company's Media Services businesses finance operations in part through the Media Services Credit Facility that matures May 12, 2015. The Company's Media Services businesses also rely on cash flow from operations, and they are able to operate with negative working capital (\$25,752,000 at April 30, 2013) primarily through liquidity provided by one significant customer contract that expires in June 2014. The negative working capital of the Company's Media Services businesses represents the net payment obligation due to this customer and certain other third parties. On May 26, 2013, the Company's Newsstand Distribution Services business, which is part of the Company's Media Services businesses, received notice that this customer contract would not be renewed upon its scheduled expiration in June 2014. The Company's Newsstand Distribution Services business is evaluating its ability to pay the net payment obligation represented by the negative working capital (approximately \$19,086,000 as of April 30, 2013) upon expiration of the contract. The Company's Newsstand Distribution Services business currently does not have sufficient capital to fund in full or refinance this obligation, and it may be unable to pay such amount and obtain other sources of working capital absent the Company obtaining additional debt or equity funding or raising capital through the sale of assets. Such additional funding or capital may not be available on acceptable terms or at all. Any failure to obtain capital to pay such obligation or to obtain other sources of working capital could have a material adverse effect on the Company's business, financial condition and results of operations. In addition, this customer contract represented approximately 8.6% of revenues for the Company's Newsstand Distribution Services business for the year ended April 30, 2013 (or less than 1% of the Company's consolidated revenues for that period), and the loss of this revenue following the expiration of the contract in June 2014 could have a material adverse effect on the Company's Newsstand Distribution Services business, financial condition and results of operations.

AMREP Southwest finances its business from cash flow from operations, which has been minimal in recent years due to the poor conditions in its real estate markets, and from advances made to it by its parent. AMREP Southwest also has a loan agreement that matures December 1, 2017, which does not allow for additional borrowings.

The Company has a defined benefit retirement plan for which accumulated benefits were frozen and future service credits were curtailed as of March 1, 2004. Due to the closing of certain facilities in connection with the consolidation of the Company's Subscription Fulfillment Services business and the associated work force reduction, the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), and the regulations thereunder, have given the Pension Benefit Guaranty Corporation (the "PBGC") the right to require the Company to accelerate the funding of \$11,688,000 of accrued pension-related obligations to the Company's defined benefit pension plan. In August 2012, the Company and the PBGC reached an agreement with respect to this funding obligation, and as a result, the

Company made a \$3,000,000 cash contribution to the pension plan on August 16, 2012. The agreement also provided that if, before August 15, 2013, the Company is unable to pay the remaining \$8,688,000 liability or adequately secure it with collateral acceptable to the PBGC, the Company would be required to either (i) provide a letter of credit equal to 110% of the remaining liability or establish a cash escrow for 100% of the remaining liability, to be maintained for five years or until the remaining liability is discharged, if sooner or (ii) discharge the remaining liability in quarterly installments over a five year period and secure it with collateral acceptable to the PBGC. In the event the Company fails to meet the terms of the agreement, the PBGC could seek immediate payment of the amount due or attempt to force a termination of the pension plan. Although the Company is currently having

discussions with the PBGC about a comprehensive restructuring of this \$8,688,000 payment obligation, the Company is unable to offer any assurance that it will be able to discharge the pension plan funding obligation by August 15, 2013 or otherwise meet the PBGC's requirements for securing or paying the undischarged amount, and the Company cannot offer any assurance that upon such inability it will be able to negotiate with the PBGC to obtain further relief. In addition, the Company may become subject to additional acceleration of its remaining accrued pension-related obligations to the pension plan if the Company closes certain additional facilities and further reduces its work force. Any such acceleration could negatively impact the Company's limited financial resources and could have a material adverse impact on the Company's financial condition. Refer to Note 11 to the consolidated financial statements included in this annual report on Form 10-K for additional pension plan information.

In June 2009, Palm Coast received \$3,000,000 pursuant to an agreement with the State of Florida (the "Award Agreement") as part of the incentives made available in connection with the Company's project, completed in the second quarter of fiscal 2011, to consolidate its Subscription Fulfillment Services operations at its Palm Coast, Florida location. The Award Agreement includes certain performance requirements in terms of job retention, job creation and capital investment which, if not met by Palm Coast, entitles the State of Florida to obtain the return of a portion, or all, of the \$3,000,000. Accordingly, the \$3,000,000 has been recorded as a liability in the accompanying balance sheet. The award monies, if any, to which Palm Coast becomes irrevocably entitled will be amortized into income over the life of the assets acquired with those funds. As of April 30, 2013, Palm Coast had not met certain of the performance requirements, in large part due to the adverse economic conditions experienced by the magazine publishing industry since the Award Agreement was executed. During 2013, the State of Florida made a demand of Palm Coast to repay approximately \$1,305,000 of the \$3,000,000 incentive award by July 31, 2013. Palm Coast is continuing discussions with the State of Florida to negotiate a settlement of the repayment amount. The Company is unable to offer any assurance as to whether or when the requested repayment amount, in whole or in part, will be returned to the State of Florida.

On June 27, 2013, the Company completed a rights offering to holders of the Company's common stock. As a result of the offering, the Company issued 1,199,242 shares of common stock at a price of \$6.25 per share and raised net proceeds of approximately \$7,100,000, net of expenses of approximately \$400,000. The net proceeds of the offering are intended to be used for corporate and working capital purposes, which may include satisfaction of a portion (currently estimated to be \$3,000,000) of the Company's obligation to the PBGC with respect to the Company's pension plan noted above.

On July 3, 2013, the Company's Subscription Fulfillment Services business received notice that a significant customer contract would not be renewed upon its scheduled expiration in February 2015. This customer contract represented approximately 15.6% of revenues for the Company's Subscription Fulfillment Services business for the year ended April 30, 2013 (or 10.8% of the Company's consolidated revenues for that period), and the loss of this revenue following the expiration of the contract in February 2015 could have a material adverse effect on the Company's Subscription Fulfillment Services business, financial condition and results of operations.

Financing Activities

Media Services - The Media Services Credit Facility provides the Media Services business with a revolving credit loan and letter of credit facility of up to \$15,000,000, with availability within that limit based upon the lesser of (i) a percentage of the borrowers' eligible accounts receivable, which may include certain unbilled accounts receivable, or (ii) the recent level of collections of accounts receivable. Subject to certain terms, funds may be borrowed, repaid and re-borrowed at any time. The Media Services Credit Facility requires the borrowers to meet certain covenants, including maintaining a minimum fixed charge coverage ratio. Borrowings under the Media Services Credit Facility are being used for Media Services' working capital needs and general business purposes and, subject to the minimum fixed charge coverage ratio required by the Media Services Credit Facility being at least at a stated level, may also be used to provide payments (based upon a prescribed formula) on certain indebtedness due the borrowing group's parent

that is not a party to the Media Services Credit Facility. At April 30, 2013, the borrowing availability under the Media Services Credit Facility was \$12,895,000, and there was \$620,000 outstanding against this availability. The highest amount borrowed during 2013 was \$6,770,000.

The borrowers' obligations under the Media Services Credit Facility are secured by substantially all of their assets other than real property. The revolving loans under the Media Services Credit Facility may be fluctuating rate borrowings or Eurodollar fixed rate based borrowings or a combination of the two as the borrowers may select. Fluctuating rate borrowings bear interest at a rate which is, at the borrowers' option, either (i) the reserve adjusted daily published rate for one month LIBOR loans plus a margin of 3% or (ii) the highest of two daily published

market rates and the bank lender's base commercial lending rate in effect from time to time, but in any case not less than 3% plus a margin of 2% (that is, not less than 5%). Eurodollar fixed rate based borrowings may be for one, two or six months and bear interest at the reserve adjusted Eurodollar interest rates for borrowings of such durations, plus a margin of 3%, which may be reduced to 2.75% depending on the borrowers' financial condition. The interest rate on outstanding borrowings at April 30, 2013 was 3.20%.

Under the terms of the Media Services Credit Facility, if there is a violation of a covenant and during the continuance of such violation, or if the borrowers do not maintain a prescribed fixed charge coverage ratio, the Media Services companies are barred from repaying indebtedness to or otherwise distributing funds to the borrowing group's parent company and the lender is entitled to terminate the Media Services Credit Facility and seek immediate payment of any outstanding borrowing.

Real Estate - AMREP Southwest had a bank loan scheduled to mature on September 1, 2012 that, at August 13, 2012, had an outstanding principal balance of \$16,214,000. The loan bore fluctuating interest at the annual rate of reserve adjusted 30-day LIBOR plus 3.5%, but not less than 5%, and required that a cash reserve of at least \$500,000 be maintained with the lender to fund interest payments. The loan was secured by a mortgage on certain real property of AMREP Southwest in Rio Rancho, New Mexico and required that the appraised value of the collateral be at least 2.5 times the outstanding principal of the loan. The loan was subject to a number of restrictive covenants including a requirement that AMREP Southwest maintain a minimum tangible net worth and a restriction on AMREP Southwest making distributions and other payments to the Company beyond a stated management fee.

On August 13, 2012, a company (the "New Lender") owned by Nicholas G. Karabots, a significant shareholder of the Company who was then Vice Chairman of the Company's board of directors and of its executive committee, purchased the bank loan and agreed to extend its maturity to December 1, 2012 on substantially its existing terms to accord the parties time to negotiate a longer extension or for AMREP Southwest to identify a possible alternate financing source. In August 2012, another director of the Company purchased a 20% participation in the loan from the New Lender.

AMREP Southwest and the New Lender entered into an agreement effective December 1, 2012 amending the terms of the loan. Under the terms of the loan as now in effect, it matures on December 1, 2017, bears interest monthly at 8.5% per annum and is secured by its original collateral and by additional collateral (the "newly mortgaged land") comprised of the balance of the real property owned by AMREP Southwest in Rio Rancho and by a pledge of the stock of its subsidiary, Outer Rim Investments, Inc., which owns approximately 12,000 acres, for the most part scattered lots, in Sandoval County, New Mexico and which are not currently being offered for sale. The total book value of the real property collateralizing the loan was approximately \$71,963,000 as of April 30, 2013. A sale transaction by AMREP Southwest of the newly mortgaged land for more than \$50,000 or of any AMREP Southwest-owned land other than land zoned and designated as a residential classification for more than \$100,000 requires the approval of the New Lender. Otherwise, the New Lender is required to release the lien of its mortgage on any land being sold by AMREP Southwest in the ordinary course to an unrelated party on terms AMREP Southwest believes to be commercially reasonable and at a price AMREP Southwest believes to be not less than the land's fair market value or, in the case of the newly mortgaged land, its wholesale value, upon receipt of AMREP Southwest's certification to such effect. The loan may be prepaid at any time without premium or penalty except that if the prepayment is in connection with the disposition of AMREP Southwest or substantially all of its assets there is a prepayment premium, initially 5% of the amount prepaid, with the percentage declining by 1% each year. No payments of principal are required until maturity, except that 25% of the net proceeds, as defined, from any sales of real property by AMREP Southwest are required to be applied to the payment of the loan. No new borrowings are permitted under this facility. The requirement to maintain the reserve for interest and the restrictive covenants that applied prior to the amendment for the most part continue to apply, including a covenant restricting AMREP Southwest from making distributions and other payments to the Company beyond a stated management fee, except that there is no longer a requirement regarding the ratio of the appraised value of the collateral to the amount of the loan. At April 30, 2013, the outstanding principal of the loan was \$16,007,000.

At April 30, 2013, the borrowers under both the Media Services Credit Facility and the AMREP Southwest loan from the New Lender were in compliance with the covenants of each facility.

Other notes payable consist of a \$4,320,000 mortgage note payable on a warehouse with a maturity date of February 2018 and an interest rate of 6.35%, and \$31,000 of equipment financing loans with maturity dates through April 2014 and an average interest rate of 7.54%. The amount of Other notes payable due within one year totals \$144,000.

Operating Activities

Receivables from Media Services operations increased from \$40,544,000 at April 30, 2012 to \$47,203,000 at April 30, 2013, primarily due to increased billings at the end of 2013 and the timing of the collection of receivables partially offset by a \$2,000,000 reserve for doubtful accounts receivable from a magazine wholesaler. Refer to Note 2 to the consolidated financial statements included in this annual report on Form 10-K for additional information regarding this reserve for doubtful accounts receivable. Receivables from real estate operations and corporate increased from \$55,000 at April 30, 2012 to \$107,000 at April 30, 2013, primarily due to a receivable from the Company's pension plan resulting from the Company paying certain expenses on the plan's behalf.

Real estate inventory totaled \$75,401,000 at April 30, 2012 compared to \$73,906,000 at April 30, 2013. Inventory in AMREP Southwest's core real estate market of Rio Rancho decreased from \$71,109,000 at April 30, 2012 to \$69,614,000 at April 30, 2013, primarily reflecting an impairment charge of \$1,125,000 on certain take-back lots. The balance of real estate inventory consisted of properties in Colorado. Investment assets decreased from \$11,262,000 at April 30, 2012 to \$10,876,000 at April 30, 2013 as a result of (i) an impairment charge of \$217,000 on certain take-back lots and (ii) an impairment reserve of \$169,000 related to a building in Rio Rancho, New Mexico under contract for sale.

Intangible and other assets increased from \$13,980,000 at April 30, 2012 to \$14,975,000 at April 30, 2013, reflecting the net effect of amortization of these assets and an increase in value of customer contracts and relationships of approximately \$2,048,000 associated with the assets acquired by FulCircle. Property, plant and equipment decreased from \$25,924,000 at April 30, 2012 to \$25,286,000 at April 30, 2013, primarily due to the net effect of depreciation charges, capital expenditures of \$1,049,000 and the acquisition of approximately \$400,000 of assets by FulCircle.

Accounts payable and accrued expenses decreased from \$85,720,000 at April 30, 2012 to \$85,340,000 at April 30, 2013, primarily from the timing of billings and payments to publishers and vendors, as well as lower business volumes.

The unfunded pension liability of the Company's frozen defined benefit pension plan decreased from \$17,677,000 at April 30, 2012 to \$13,805,000 at April 30, 2013, primarily due to Company contributions to the pension plan of \$4,572,000 partially offset by an increase in actuarial liabilities resulting from a reduction in the discount rate. The Company recorded other comprehensive income of \$734,000 in 2013 and other comprehensive loss of \$3,158,000 in 2012, reflecting the change in the unfunded pension liability in each year net of the related deferred tax and unrecognized prepaid pension amounts.

Investing Activities

Capital expenditures for property, plant and equipment were approximately \$1,049,000 and \$1,073,000 in 2013 and 2012, primarily for upgrades related to technology in both years. The Company believes that it has adequate cash flows from operations and financing capability to provide for anticipated capital expenditures in fiscal 2014, most of which are expected to be in the Subscription Fulfillment Services business.

Future Payments Under Contractual Obligations

The table below summarizes significant contractual cash obligations as of April 30, 2013 for the items indicated (in thousands):

Contractual	Total	Less	1-3	3-5
Obligations		than	years	years
		1 vear		

N o t e s payable \$20,978 \$144 \$867 \$ Operating 33,184 12,348 20,813 leases and

other

Changes in laws, regulations, reserve requirements and regulatory policies

Other factors that could cause actual results to differ materially from furesults expressed or implied by such forward-looking statements

-20-

Table of Contents

RISK FACTORS

In addition to the other information contained in or incorporated by reference this proxy statement/prospectus, including the matters addressed under the cap entitled Cautionary Statement Regarding Forward-Looking Statements, shareholders should carefully consider the following risk factors in deciding whe to vote for approval of the merger proposal. Please see the sections entitled WF YOU CAN FIND MORE INFORMATION beginning on page i and Incorporate Certain Documents by Reference beginning on page 87.

Fluctuations in market prices of CVB Financial common stock and adjustment the cash portion of the merger consideration could affect the value of the merger consideration that Valley shareholders receive for their shares of Valley common stock.

Under the terms of the merger agreement, the aggregate number of shares of C Financial common stock issued in the merger is fixed and will not change if market value of CVB Financial common stock changes or for any other reason. market price of CVB Financial common stock may vary from its price on the immediately prior to the first public announcement of the merger, the date of proxy statement/prospectus and the date of Valley s special meeting, for example market price of CVB Financial common stock may fluctuate as a result of a variet factors, including, among other things, changes in CVB Financial s business operations and prospects, regulatory considerations and general market and econo conditions. Many of these factors are beyond the control of CVB Financial. Theref the market value of the shares of CVB Financial common stock that a Va shareholder receives in the merger could decline correspondingly with any decline the market price of CVB Financial common stock prior to and as of the date merger consideration is determined. Because the price of CVB Financial comstock will fluctuate prior to the merger, CVB Financial cannot assure Va shareholders of the market value of the CVB Financial common stock that they receive in the merger.

Further, the amount of cash included in the merger consideration will be reduced f \$23,400,000 by the amount, if any, that Valley s total merger-related transa expenses exceed \$3,500,000 or, without duplication, the amounts by which categories of Valley s merger-related transaction expenses exceed certain spec amounts. The amount of the reduction, if any, will not be determined until she prior to the merger. See THE MERGER Terms of the Merger Merger Consi and Special Dividend.

The amount of the special dividend payable to holders of Valley s common stock not known and will not be determined until shortly prior to the merger.

The merger agreement provides that Valley will declare a one-time cash divided payable to holders of record of outstanding Valley common stock as of the cloudate of the merger. The aggregate amount of the special dividend will be equal to amount by which Valley stangible common equity exceeds the great

(1) \$37,000,000 (or, if Valley has sold a specified loan, \$37,500,000) or (2) amount required to achieve an 8.0% tangible common equity ratio as of the final of the month immediately preceding the closing date of the merger. Based on Val tangible common equity at September 30, 2016, Valley s shareholders would received approximately \$13,700,000 as a special dividend or approximately \$4 cash per share, in addition to the merger consideration.

Generally, Valley s tangible common equity will increase as Valley recognizes are earnings and will decrease as Valley recognizes any net losses. Therefore, the amount of the special dividend may increase or decrease as Valley recognizes net earning net losses. Further, the amount of the special dividend may depend on whether Valles a specified loan in accordance with the merger agreement. The amount of special dividend will not be determined until the final day of the month immediate preceding the merger and, therefore, will not be known at the time that Valleshareholders vote on the merger proposal.

-21-

Table of Contents

Payment of the Special Dividend is subject to regulatory approvals.

Valley s primary source of cash for the special dividend will be a cash dividend its subsidiary, Valley Business Bank. Valley Business Bank is required to obtain paper approval from the Commissioner of Business Oversight of the State of Califor under the California Financial Code and the FDIC under the FDIC s regulations by Valley can pay the special dividend as the amount of the special dividend is in exof the amount that may be paid under California law without prior approval. November 3, 2016, Valley Business Bank filed applications with the Departmer Business Oversight and FDIC to allow it to make a dividend payment to Valley on November 3, 2016, Valley informed the Federal Reserve of its intent to pay special dividend to its shareholders. Valley Business Bank may not receive regulatory approvals required to pay all or part of the special dividend, which correduce the amount of or eliminate the special dividend payable to Vall shareholders.

If the average trading price of CVB common stock falls outside of a specified ran shortly before the anticipated closing date of the merger, CVB or Valley could have the right to terminate the merger agreement, in which case the merger would not completed.

If the volume-weighted average closing price for CVB Financial common stock the ten trading days ending on the fifth trading day prior to the closing date of merger is less than \$13.50 per share, Valley would have the right, but not obligation, to terminate the merger agreement. Conversely, if the volume-weigh average closing price for CVB Financial common stock for the 10 trading days end on the fifth trading day prior to the closing date of the merger is greater than \$20 per share, CVB Financial would have the right, but not the obligation, to terminate merger agreement.

As a result, even if Valley shareholders and banking regulators approve the me agreement and the merger, the merger may not ultimately be completed. Although either CVB Financial or Valley may have the discretion to terminate the me agreement based on the trading price of CVB Financial common stock, neither par obligated to do so.

The market price for CVB Financial common stock may be affected by factors different from those that historically have affected Valley.

Upon completion of the merger, holders of Valley common stock will become hol of CVB Financial common stock. CVB Financial s business differs from the Valley, and accordingly, the results of operations of CVB Financial will be affected some factors that are different from those currently affecting the results of operat of Valley. For a discussion of the business of CVB Financial and some importance factors to consider in connection with its business, see the section entimer INFORMATION ABOUT THE COMPANIES beginning on page 26 a documents incorporated by reference in this proxy statement/prospectus and reference

to under the section entitled WHERE YOU CAN FIND MORE INFORMAT

beginning on page i.

Combining our two companies may be more difficult, costly or time-consuming two expect.

The success of the merger will depend, in part, on CVB Financial s abilisuccessfully combine Valley s organization with its own. If CVB Financial is una achieve this objective, the anticipated benefits of the merger may not be realized for at all or may take longer than expected to be realized and the value of C Financial s common stock could be adversely affected as a result.

CVB Financial and Valley have operated and, until the merger is completed continue to operate, independently. It is possible that the integration process coresult in the loss of key employees or disruption of each company songoing but or inconsistencies in standards, controls, procedures and policies that adversely af our ability to maintain relationships with customers and employees or to achieve anticipated benefits of the merger. As with any merger of banking institutions, the also may be business disruptions that cause us to lose customers or cause customer take their deposits and/or loans out of our banks. As a result of

-22-

Table of Contents

such disruptions or the perceived potential for such disruptions, our competitors to be successful at luring key employees and customers away from us. The success of combined company following the merger may depend in large part on the abilit integrate the two businesses, business models and cultures. If we are not ablintegrate our operations successfully and timely, the expected benefits of the memay not be realized.

Regulatory approvals necessary to complete the merger may not be received, may take longer than expected or may impose conditions that are not presently anticipated or cannot be met.

Before the merger may be completed, various approvals must be obtained from bank regulatory and other governmental authorities. These governmental entities impose conditions on the granting of such approvals. Such conditions or changes the process of obtaining regulatory approvals could have the effect of delay completion of the merger or of imposing additional costs or limitations on C Financial following the merger. The regulatory approvals may not be received at time, may not be received in a timely fashion and may contain conditions on completion of the merger that are not anticipated or cannot be met. It is a condition closing the merger that no regulatory approval shall contain or shall have resulted or reasonably be expected to result in the imposition of, any burdensome condition the combined company following the merger.

The merger agreement may be terminated in accordance with its terms, and the merger may not be completed.

The merger agreement is subject to a number of conditions that must be fulfille order to complete the merger. Those conditions include approval of the mer agreement by Valley shareholders, receipt of requisite regulatory approvals subject certain limitations set forth in the merger agreement, absence of orders prohibi completion of the merger, effectiveness of the registration statement of which proxy statement/prospectus is a part, approval of the shares of CVB Finan common stock to be issued to Valley shareholders for listing on the NASDAQ Gl Select Market, the continued accuracy of the representations and warranties by parties, the absence of a material adverse effect on either CVB Financial or Valley satisfaction of certain financial tests by Valley and the performance by both partie their covenants and agreements contained in the merger agreement, and C Financial s receipt of a legal opinion from its tax counsel. These conditions closing of the merger may not be fulfilled, and accordingly, the merger may no completed. In addition, if the merger is not completed by February 27, 2017, ei CVB Financial or Valley may choose not to proceed with the merger, and the particle of the control of the contr can mutually decide to terminate the merger agreement at any time, before or a Valley shareholders approve the merger. Further, CVB Financial and Valley may 6 to terminate the merger agreement in certain other circumstances. If the me agreement is terminated under certain circumstances, Valley may be required to p termination fee of \$3,500,000 to CVB Financial. Please refer to the section enti THE MERGER AGREEMENT Termination; Termination Fee beginning on

for a description of these circumstances. If the merger agreement is terminated, Va

will not pay the special dividend.

Failure to complete the merger could negatively impact CVB Financial s, Citize Business Bank s and/or Valley s respective businesses, financial condition, rest operations and/or stock prices.

If the merger agreement is terminated and the merger is not completed, the ongo businesses of CVB Financial, Citizens Business Bank and Valley may be adver affected. The market prices of CVB Financial s and Valley s respective common may decline to the extent that the current market price reflects a market assump that the merger will be completed. In addition, CVB Financial and Valley respective reactions to the termination of the merger from custom depositors, investors, vendors and others with whom they deal, and neither C Financial nor Valley would realize any of the anticipated benefits of having complete merger. The expenses of each of CVB Financial and Valley incurred in connect with the merger, such as legal and accounting fees, must be paid even if the mergen not completed and may not be recovered from the other party.

-23-

Table of Contents

Further, if the merger agreement is terminated and the Valley board of directors so another merger or business combination, Valley shareholders cannot be certain Valley will be able to find a party willing to pay the equivalent or gre consideration than that which CVB Financial has agreed to pay in the merger.

CVB Financial is not required to complete the merger if Valley does not meet certain financial measures.

CVB Financial s obligation to complete the merger is subject to a number conditions, including, but not limited to Valley s satisfying certain financial conditions and limits. For example, the merger agreement provides that Valley must meet satisfied the following financial conditions: (1) as of the last day of the month preceding closing date, Valley s closing tangible common equity shall not be less than the g of (a) \$37,000,000 (or, if Valley has completed the sale of one specified less \$37,500,000) or (b) the amount necessary for Valley to achieve a tangible compequity ratio of at least 8.0% as of such date; (2) the allowance for loan loss ratio so not be less than 1.1%; (3) total assets shall not be less than \$410,000,000; (4) average noninterest-beat deposits shall not be less than \$150,000,000; (5) average noninterest-beat deposits shall not be less than \$150,000,000 and (6) Valley s merger-retransaction expenses shall not exceed \$3,500,000 in the aggregate and none of specified categories of merger-related expenses shall exceed certain specified categories of merger-related expenses shall exceed certain specified categories of merger-related expenses shall exceed certain specified complete the merger and the merger might not occur.

Valley will be subject to business uncertainties and contractual restrictions while the merger is pending.

Uncertainty about the effect of the merger on employees and customers may hav adverse effect on Valley and consequently, if the merger occurs, on CVB Financian These uncertainties may impair Valley s ability to attract, retain and motivat personnel until the merger is completed, and could cause customers and others deal with Valley to seek to change existing business relationships with Valley, w could negatively affect its results of operations. Retention of certain employees be challenging while the merger is pending, as certain employees may experie uncertainty about their future roles with CVB Financial. If key employees dep CVB Financial s business following the merger could be harmed and/or Va business would be harmed if the merger is not completed and Valley then continue an independent bank. In addition, the merger agreement restricts Valley from mal certain acquisitions and loans and taking other specified actions until the me occurs without the consent of CVB Financial. These restrictions may prevent Va from pursuing attractive business opportunities that may arise prior to the comple of the merger. See the section entitled THE MERGER AGREEMENT Covena Agreements Conduct of Businesses Prior to the Completion of the Merger beg on page 55 of this proxy statement/prospectus for a description of the restric covenants to which Valley is subject.

Some of the directors and officers of Valley may have interests and arrangement that may have influenced their decisions to support or recommend that you approximately ap

the merger.

The interests of some of the directors and officers of Valley may be different f those of Valley shareholders generally, and directors and officers of Valley may participants in arrangements that are different from, or in addition to, those of Va shareholders. These interests are described in more detail in the section of this pr statement/prospectus entitled THE MERGER Interests of Valley Director Executive Officers in the Merger beginning on page 49.

Valley shareholders will have a significantly reduced ownership percentage and voting interest after the merger and will exercise less influence over managemen

Valley shareholders currently have the right to vote in the election of the boar directors of Valley and on certain other matters affecting Valley. The merger transfer control of Valley to CVB Financial and, indirectly, to the shareholder CVB Financial. When the merger occurs, each Valley shareholder (other t shareholders exercising dissenters—rights with respect to all of their shares of V stock) will become a shareholder of CVB Financial with a percentage ownershi CVB Financial much smaller than such

-24-

Table of Contents

shareholder s percentage ownership of Valley. In the aggregate, Valley cushareholders are expected to own approximately 1.8% of the outstanding share CVB Financial s common stock when the merger is completed (excluding any sthey may already own). Because of this, Valley shareholders will have significal less influence on the management and policies of CVB Financial than they now hon the management and policies of Valley.

The merger agreement contains provisions that may discourage other companies from trying to acquire Valley for greater merger consideration.

The merger agreement contains provisions that may discourage a third party f submitting a business combination proposal to Valley that might result in gre value to Valley shareholders than does the merger. These provisions include a gen prohibition on Valley from soliciting, or, subject to certain exceptions, entering discussions with, any third party regarding any acquisition proposal or offers competing transactions. The members of the board of directors of Valley have ag to vote their shares of Valley common stock in favor of the Valley merger prop and the adjournment proposal and against any alternative transaction. Valley also an obligation to submit the merger proposal to a vote by its shareholders, eve Valley receives a proposal that its board of directors believes is superior to the men unless the merger agreement is validly terminated prior to the special meeting. shareholders that are party to the voting and support agreements described in paragraph beneficially own and are entitled to vote in the aggregate approxima 19% of the outstanding shares of Valley common stock as of the record date addition, Valley may be required to pay CVB Financial a termination fe \$3,500,000 in certain circumstances involving acquisition proposals for compe transactions. For further information, please see the sections entitled THE MER AGREEMENT Voting and Support Agreements beginning on page 69 an MERGER AGREEMENT Termination; Termination Fee beginning on page 67.

The combined company expects to incur substantial expenses related to the merg and CVB Financial is not required to complete the merger if Valley s merger-re expenses exceed certain limits.

The combined company expects to incur substantial expenses in connection with consummation of the merger and the combining of the business, operations, network systems, employees, technologies and policies and procedures of the two company Although CVB Financial and Valley have assumed that a certain level of transact and combination expenses would be incurred, there are a number of factors bey their control that could affect the total amount or the timing of their combinate expenses. Many of the expenses that will be incurred, by their nature, are difficute estimate accurately at the present time. Under the merger agreement, if certainsaction costs exceed an aggregate of \$3.5 million, the amount by which threshold is exceeded could negatively impact Valley is net equity and its abiliment the tangible common equity minimum closing condition contained in the meagreement, in which case CVB Financial could decide not to complete the merger.

-25-

Table of Contents

INFORMATION ABOUT THE COMPANIES

CVB Financial Corp. and Citizens Business Bank

CVB Financial is a California corporation that is registered as a bank hold company under the Bank Holding Company Act of 1956, as amended. A September 30, 2016, CVB Financial had consolidated total assets of approximat \$8.0 billion, total loans and lease finance receivables of approximately \$4.2 bill total deposits of approximately \$6.3 billion, customer repurchase agreement approximately \$0.58 billion and total shareholders—equity of approximately billion. CVB Financial had 771 full-time equivalent employees as of September 2016. CVB Financial s principal executive office is located at 701 N. Haven Av Suite 350, Ontario, California 91764.

CVB Financial provides a wide range of banking services through Citizens Busin Bank, its wholly owned subsidiary. Citizens Business Bank is a Califor state-chartered bank headquartered in Ontario, California, and has been conduct business since 1974, originally under the name Chino Valley Bank.

Citizens Business Bank is an independent community bank that offers a full rang banking services in 42 Business Financial Centers located in the Inland Empire, Angeles County, Orange County, San Diego County, Santa Barbara County, Ven County and the Central Valley areas of California. Citizens Business Bank also eight Commercial Banking Centers. Although able to take deposits, these cen operate primarily as sales offices and focus on business clients and their princip professionals and high-net-worth individuals. These centers are located in Burb Encino, Los Angeles, Newport Beach, Oxnard, Santa Barbara, Torrance and Upl California. Citizens Business Bank also has three trust offices in Ontario, New Beach and Pasadena, California. These offices serve as sales offices for Citiz Business Bank s wealth management, trust and investment products.

Through its network of banking offices, Citizens Business Bank emphase personalized service combined with a full range of banking and trust services businesses, professionals and individuals. Although Citizens Business Bank foot the marketing of its services to small- and medium-sized businesses, a full range banking, investment and trust services are made available to the local consumarket.

Citizens Business Bank offers a standard range of bank deposit products. The include checking, savings, money market and time certificates of deposit for business and personal accounts. Citizens Business Bank s deposit accounts are in by the FDIC up to applicable limits.

Citizens Business Bank provides a full complement of lending products, include commercial, agribusiness, consumer, real estate loans and equipment and vehileasing. Commercial products include lines of credit and other working capfinancing, accounts receivable lending and letters of credit. Agribusiness products loans to finance the operating needs of wholesale dairy farm operations, cattle feed

livestock raisers and farmers. Citizens Business Bank also provides bank-quali lease financing for municipal governments. Financing products include automo leasing and financing, lines of credit, credit cards and home equity loans and line credit. Real estate loans include mortgage and construction loans.

Citizens Business Bank also offers a wide range of specialized services designed its commercial customers, including cash management systems for monitoring of flow, a credit card program for merchants, courier pick-up and delivery, pay services, remote deposit capture, electronic funds transfers by way of domestic international wires and automated clearinghouse, and online account access.

Citizens Business Bank offers financial services and trust services through CitizensTrust division. These services include fiduciary services, mutual fur annuities, 401(k) plans and individual investment accounts.

As a bank holding company, CVB Financial is subject to the supervision of Federal Reserve. It is required to file with the Federal Reserve reports and o information regarding its business operations and the

-26-

business operations of its subsidiaries. As a California state-chartered bank, Citiz Business Bank is subject to supervision, periodic examination and regulation by Department of Business Oversight and by the FDIC as its primary federal regulator.

CVB Financial s principal executive office is located at 701 North Haven Av Suite 350, Ontario, California 91764, telephone number: (909) 980-4030.

CVB Financial common stock is traded on the NASDAQ Global Select Market up the symbol CVBF.

Additional information about CVB Financial and its subsidiaries may be found in documents incorporated by reference into this proxy statement/prospectus. Please see the section entitled WHERE YOU CAN FIND MORE INFORMAT beginning on page i.

Valley

General

Valley is a bank holding company registered under the Bank Holding Company Ac 1956, as amended. As of September 30, 2016, Valley had consolidated total assert approximately \$429.9 million, total loans and lease finance receivable approximately \$309.8 million, total deposits of approximately \$374.4 million shareholders—equity of approximately \$50.7 million. Valley had full-time-equivalent employees as of September 30, 2016.

Valley provides a wide range of banking services through Valley Business Bank wholly-owned subsidiary. Valley Business Bank is a California state-chartered be headquartered in Visalia, California, and has been conducting business since 19 originally under the name Bank of Visalia. Valley Business Bank offers a full rang credit and depository services primarily to small- and medium-sized business Valley Business Bank currently operates from four full-service banking office Visalia, Fresno, Tulare and Woodlake, California. Valley Business Bank s depositing the FDIC up to the maximum limits allowable by law.

For the year ended December 31, 2015:

Net income was \$4,457,211, or \$1.51 per diluted share, an increas \$205,980 or 4.8% from the prior year s \$4,251,231, or \$1.43 per diluted sh

Total assets as of December 31, 2015 were \$421.3 million, a 3.1% increase over December 31, 2014;

Deposits decreased by \$7.1 million to \$352.2 million at December 31, 201 2.0% decrease from December 31, 2014;

Capital increased by 5.9% or \$2,591,024, ending 2015 at \$46,786,172; and

The loan portfolio increased to \$298,513,157 at December 31, 2015, compared to \$264,490,919 at the end of 2014, an increase of 12.9%.

As of and for the period ended September 30, 2016:

Net income for the nine months ended September 30, 2016 was \$3,270,176 \$1.12 per diluted share, an increase of \$293,549 or 9.9% from \$2,976,62 \$1.00 per diluted share for the same period of 2015;

Total assets were \$429.9 million, a 7.0% increase over September 30, 2015:

Deposits increased by \$22.7 million to \$374.4 million, a 6.5% increase of September 30, 2015;

Capital increased by 10.9% or \$5.0 million from September 30, 2015 to \$5.0 million at September 30, 2016; and

-27-

The loan portfolio increased to \$309.8 million, compared to \$289.2 million September 30, 2015, an increase of \$20.6 million or 7.1%.

Services

Valley s focus is providing highly personalized banking services to entreprend businesses, professional firms and nonprofit organizations, along with their ow and key managers. Valley offers a full range of deposit accounts, including personal business/professional checking accounts, savings accounts, attorney-client accounts, money market demand accounts, time certificates of deposit and tide accounts for larger depositors. Valley also provides other customary banking product and services, including, among other things, notary public services and wire transformation of the convenience-oriented services and products Valley offers are direct pay deposits, bank-by-mail services, merchant bank card processing, telephone transformation, and ACH origination. Valley also offers a full complement convenience-oriented services including automated teller machines, remote depositive, courier service for bank transactions and a 24-hour Internet online bank capability to allow clients to obtain account information, transfer funds between accounts, and place stop payment orders.

Valley also provides a full array of lending products tailored to meet the need clients in its service market. These products include business lines of credit, t loans, loans guaranteed in whole or part by the U.S. Small Business Administrat or SBA, and state guaranteed loans, equipment loans, accounts receivable inventory financing, construction loans, commercial real estate loans, letters of credit individual lines of credit, auto and other consumer loans.

There have been no significant changes in the types of services offered by Va since its inception, except in connection with new types of accounts allowed by sta or regulation in recent years. Valley has no present plans regarding a new libusiness requiring the investment of a material amount of its total assets. Mo Valley s business originates from within Tulare and Fresno Counties and is print focused on loans to professional services, manufacturers, building contract developers, retailers, wholesalers and business service companies. Valley s bus based upon performance to date, does not appear to be seasonal. Except as described above, no material portion of Valley s loans is concentrated within a single indust group of related industries, nor is Valley dependent upon a single customer or group of related customers for a material portion of its deposits. Management of Valley unaware of any material effect upon Valley s capital expenditures, earning competitive position as a result of federal, state or local environmental regulation.

Service Areas

Valley s current market areas are Tulare and Fresno counties and surrounding are Central California. Valley attracts the majority of its loan and deposit business of the residents and numerous small- to medium-sized businesses, professional find and service entities located in Central California. Valley does not have direct deal with any foreign sources and has no known risks in any international business.

Competition

The banking and financial services business in California generally, and in Val service area specifically, is highly competitive. The increasingly competi environment is a result primarily of changes in regulation, changes in technology product delivery systems, and the accelerating pace of consolidation among finar services providers.

Valley s business is concentrated in its service area that originates from within T and Fresno Counties, including but not restricted to the following communit Visalia, Fresno, Tulare and Woodlake, California. In order to compete with o financial institutions in its service area, Valley relies principally upon direct personant by officers, directors, employees, and shareholders; and specialized services as courier pick-up and

-28-

delivery of non-cash banking items. In addition, Valley has an experienced lend and marketing officer base. Larger banks may have a competitive advantage becare of higher lending limits and major advertising and marketing campaigns. Larger banks perform services, such as trust services, international banking, discount broken and insurance services, that are not offered by Valley.

Increasing levels of competition in the banking and financial services businesses reduce market share or cause the prices charged for services to fall. Results may d in future periods depending on the nature or level of competition. The increasing competitive environment is a result primarily of sustained low market interest rechanges in regulation, changes in technology and product delivery systems, and accelerating pace of consolidation among financial services providers.

Commercial banks such as Valley compete with savings banks, credit unions, o financial institutions, securities brokerage firms, and other entities for funds. instance, yields on corporate and government debt securities and other commer paper affect the ability of commercial banks to attract and hold deposits. Commer banks also compete for loans with savings banks, credit unions, consumer fina companies, mortgage companies and other lending institutions.

Properties

Valley currently operates out of its full-service headquarters in Visalia and from the branch offices located in Fresno, Tulare and Woodlake, California. Valley owns real property where the Visalia, Tulare and Woodlake offices are located. Valleases the Fresno facility.

-29-

VALLEY SPECIAL MEETING OF SHAREHOLDERS

General

This proxy statement/prospectus is being provided to Valley shareholders as part solicitation of proxies by the Valley board of directors for use at the special mee and at any adjournments or postponements of such meeting. Valley commer mailing this proxy statement/prospectus and the enclosed form of proxy to shareholders entitled to vote at the Valley special meeting on or about [] [], 20[

Date, Time and Place of the Special Meeting

The special meeting will be held at the Visalia Country Club located at 625 N Ranch Road, Visalia California 93291 on [], 2017, at 6:00 p.m. (Pacific Time).

Record Date for the Special Meeting; Shares Entitled to Vote

Only holders of record of Valley common stock at the close of business on [], which is the record date for the special meeting, are entitled to receive notice of an vote at the meeting. On the record date, Valley had 3,002,014 shares of its common stock issued, outstanding and eligible to vote at the special meeting. As of the reddate, Valley had approximately [] holders of its common stock of record.

Quorum

A majority of the shares of Valley common stock issued and outstanding and entito vote on the record date must be represented in person or by proxy at the spe meeting in order for a quorum to be present for purposes of transacting busin Proxies marked as abstaining (including proxies containing broker non-votes) on matter to be acted upon by shareholders will be treated as present at the meeting purposes of determining a quorum but will not be counted as votes cast on s matters. If there is no quorum at the special meeting, the affirmative vote of at least majority of the votes present in person or represented by proxy and entitled to vote the meeting may adjourn the special meeting to another date.

Purposes of the Special Meeting

The special meeting is being held to consider and vote on the following proposals:

1. Approval of the Merger Agreement. To consider and vote upon an Agreement and Plan of Reorganization and Merger, dated September 22, 2016, and transactions contemplated therein pursuant to which (i) Valley will merge and into CVB Financial, the separate corporate existence of Valley will can and CVB Financial will survive and continue to exist as a Califor corporation, (ii) each share of Valley common stock (other than shares a which dissenters rights are properly exercised) will be converted into the

to receive cash and common stock of CVB Financial, the parent compan Citizens Business Bank, in accordance with and as determined by the me agreement and (iii) holders of Valley common stock as of the date of merger would be entitled to receive a special cash dividend. We refer to proposal as the merger proposal.

2. Grant of Discretionary Authority to Adjourn Meeting. To consider and vote upon a proposal to grant discretionary authority to adjourn the special meeting, if necessary, to permit further solicitation of proxies if there are not sufficient votes at the time of the special meeting to approve the terms of the merger agreement. We refer to this proposal as the adjournment proposal.

Recommendation of the Valley Board of Directors

The Valley board of directors recommends that Valley shareholders vote:

FOR the approval of the merger proposal; and

-30-

FOR the approval of the adjournment proposal.

The Valley board of directors unanimously approved the merger agreement the merger and has determined that the merger is in the best interests of Va and its shareholders. See THE MERGER Background of the Merger be on page 37 and THE MERGER Reasons for the Merger and Recommenda the Valley Board of Directors beginning on page 40.

In considering the recommendation of the Valley board of directors with respect to merger, Valley shareholders should be aware that some of the Valley directors executive officers may have interests that are different from, or in addition to, interests of Valley shareholders generally. See THE MERGER Interests of Directors and Officers in the Merger beginning on page 49.

Number of Votes

Each Valley shareholder is entitled to cast one vote, in person or by proxy, for a share held in that shareholder s name on the books of Valley as of the record dathe matters to be submitted to the vote of the shareholders.

Votes Required; Voting Agreements

The votes required for each proposal are as follows:

The merger proposal. The affirmative vote of at least a majority of the share Valley common stock outstanding is required to approve this proposal.

The adjournment proposal. Assuming a quorum is present, the affirmative vote of least a majority of the shares of Valley common stock, present in person represented by proxy and entitled to vote at the special meeting, is required to apprehis proposal.

Valley s directors and executive officers collectively hold, as of the record date for special meeting, 579,122 shares, or approximately 19% of Valley common streligible to vote at the special meeting. Pursuant to voting agreements more for described under the section THE MERGER AGREEMENT Voting and Society Agreements beginning on page 69, each of Valley society, executive office that principal shareholder and other shareholder has agreed to vote his, her of shares of Valley common stock **FOR** approval of the merger agreement at transactions contemplated therein and the merger. A copy of the form of votagreement is attached as Exhibit A to the merger agreement, which is attached to proxy statement/prospectus as **Appendix A** and is incorporated herein by references.

Voting of Proxies

Whether or not you plan to attend the special meeting, we urge you to complete, and date the enclosed proxy card and to return it promptly in the envelope proving Returning the proxy card will not affect your right to attend the special meeting

vote in person. You may also vote over the Internet or by telephone. Instructions all voting methods can be found on the proxy card included with this pr statement/prospectus.

If you properly fill in your proxy card and send it to us in time to vote, your proxy individual(s) named on your proxy card) will vote your shares as you have directed NO INSTRUCTION IS SPECIFIED WITH RESPECT TO A MATTER TO ACTED UPON, THE SHARES REPRESENTED BY THE PROXY WILL VOTED IN FAVOR OF THE PROPOSALS LISTED ON THE PROXY FOLLOWS:

FOR the merger proposal; and

FOR the adjournment proposal.

-31-

Abstentions and Broker Non-Votes

If you hold your shares of Valley common stock in street name (that is, thr broker or other nominee), you must vote your shares through your broker. You shares receive a form from your broker asking how you want to vote your shares. Follow instructions on that form to give voting instructions to your broker. Under the r that govern brokers which are voting with respect to shares held in street r brokers have the discretion to vote such shares on routine, but not on non-rout matters. At the special meeting, none of the proposals to be presented constitu routine matter. THEREFORE, YOUR BROKER MAY NOT VOTE YOUR SHAI FOR ANY OF THE VALLEY PROPOSALS SET FORTH IN THIS P STATEMENT/PROSPECTUS, INCLUDING THE APPROVAL OF THE MERO AGREEMENT AND THE MERGER, WITHOUT YOUR SPECIFIC DIRECTI A broker non-vote occurs when your broker does not vote on a particular pr because the broker does not receive instructions from the beneficial owner and of not have discretionary authority. It is VERY IMPORTANT that you instruct y broker or nominee how to vote your shares. Therefore, if you wish to represented, you must vote by completing the information that is sent to you your broker or nominee.

Revoking Proxies

Valley shareholders who hold their shares in certificate form may revoke their pro at any time before the time their proxies are voted at the special meeting by (i) fi with the Corporate Secretary of Valley, an instrument revoking it or a duly executive proxy bearing a later date or (ii) appearing in person at the special meeting advising the chairman of the Valley board of directors of the holder s intent to verify the special meeting.

Written notices of proxy revocations must be sent so that they will be received be the taking of the vote at the special meeting as follows:

Valley Commerce Bancorp

701 West Main Street

Visalia, California 93291

Attention: Fred P. LoBue, Jr., Corporate Secretary

For shareholders whose shares are held in street name and if you have instructed broker or other nominee to vote your shares, you must follow directions received for your broker or other nominee in order to change those instructions.

Dissenters Rights

Holders of Valley common stock will have dissenters rights with respect t proposal to approve the merger agreement and the merger. After the special meeting

notice will be sent advising shareholders if the merger has been approved. In order perfect dissenters—rights, a shareholder of Valley common stock must d following:

Not vote FOR the merger agreement and the merger;

Make a timely written demand upon Valley for purchase in cash of dissenting shareholder s shares at their fair market value as of September 2016, which demand includes:

the number of the shares held of record by the dissenting shareholder that such holder demands be paid in cash, and

what the dissenting shareholder claims to be the fair market value of his her shares as of September 22, 2016 immediately prior to announcement the terms of the merger;

Have the dissenting shareholder s demand received by Valley within 30 after the date on which the notice of the approval of the merger by outstanding shares is mailed to the shareholder;

-32-

Submit certificates representing the dissenting shareholder s share endorsement in accordance with Section 1302 of the California Corporat Code; and

Comply with such other procedures as are required by Chapter 13 of California Corporations Code.

If dissenters—rights are properly perfected, such dissenter has the right to cash amount equal to the fair market value of its shares of Valley common stock a September 22, 2016, the day of, and immediately prior to, the first purannouncement of the terms of the merger. The Valley board of directors determined the fair market value for each share of Valley common stock to be \$10 as of September 22, 2016 based upon the last reported trading price of Valley common stock on the OTC Pink market on September 22, 2016, the day of, and immediately prior to, the first announcement of the terms of the merger. Please read the second terminal discontinuous prior to, the first announcement of the terms of the merger. Please read the second terminal discontinuous prior to, the first announcement of the terms of the merger. Please read the second terminal discontinuous prior to, the first announcement of the terms of the merger. Please read the second terminal discontinuous prior to, the first announcement of the terms of the merger. Please read the second terminal discontinuous prior to, the first announcement of the terms of the merger. Please read the second terminal discontinuous prior to, the first announcement of the terms of the merger.

If dissenters—rights are perfected and exercised or capable of being perfected exercised with respect to more than 10% of Valley—s outstanding shares of constock, then CVB Financial is not obligated to complete the merger and could termithe merger agreement.

Other Matters

Because it is a special meeting of shareholders, no other matters may be presented consideration by shareholders at the special meeting.

Solicitation of Proxies

The Valley board of directors is soliciting proxies for the special meeting. Valley pay for the cost of solicitation of proxies. In addition to solicitation by mail, Val directors, officers and employees may also solicit proxies from shareholder telephone, facsimile, or in person. Valley will not pay any additional compensation these directors, officers or employees for these activities but may reimburse them reasonable out-of-pocket expenses.

If Valley management deems it advisable, the services of individuals or compa that are not regularly employed by Valley may be used in connection with solicitation of proxies. Arrangements will also be made with brokerage houses other custodians, nominees and fiduciaries to send the proxy materials to benefit owners. Valley will, upon request, reimburse those brokerage houses and custod for their reasonable expenses in so doing.

Attending the Special Meeting

All Valley shareholders as of the record date, or their duly appointed proxies, attend the special meeting.

If you hold your shares of Valley common stock in your name as a shareholder record and you wish to attend the special meeting, please bring your proxy evidence of your stock ownership, such as your most recent account statement, to special meeting. You must also bring valid photo identification.

If your shares of Valley common stock are held in street name in a stock bro account or by a bank or nominee and you wish to attend the special meeting, you re to bring a copy of a bank or brokerage statement to the special meeting reflecting years stock ownership as of the record date. You must also bring valid photo identification

-33-

THE MERGER PROPOSAL

As discussed throughout this proxy statement/prospectus, Valley is asking shareholders to approve the merger proposal. Holders of Valley common stock sho read carefully this proxy statement/prospectus in its entirety, including the appendit for more detailed information concerning the merger agreement and the merger particular, holders of Valley common stock are directed to the merger agreement copy of which is attached as **Appendix A** to this proxy statement/prospectus.

The Valley board of directors unanimously recommends a vote FOR the r proposal.

Each of the directors of Valley has entered into a voting and support agreen with CVB Financial and Valley, pursuant to which they have agreed to vote the merger proposal. For more information regarding the voting and suppagreements, please see the section entitled THE MERGER AGREEMENT and Support Agreements beginning on page 69.

THE ADJOURNMENT PROPOSAL

If there are not sufficient shares of Valley common stock represented to constitute quorum at the special meeting or if the number of shares of Valley common structure to the special meeting of the merger agreement and the transactions contemporated in some sufficient to approve that proposal at the meeting, then the person designated as the proxy holder stated in the proxy card of Valley intends to move adjourn the special meeting in order to enable the Valley board of directors to so additional proxies for approval of the merger proposal.

In this proposal, Valley is asking shareholders to grant discretionary authority to person(s) designated as the proxy holder stated in the proxy card to move to adjourn the special meeting if there are not sufficient shares represented to constitute a quotate the meeting or if the number of shares voting for approval of the merger propose not sufficient to approve that proposal at the meeting. If the shareholders of Valapprove the adjournment proposal, Valley could adjourn the special meeting to a ladate and use the additional time to solicit additional proxies, including the solicitate of proxies from shareholders that have previously voted on the merger proposal Among other things, approval of the adjournment proposal could mean that, every Valley has received proxies representing a sufficient number of votes against approand adoption of the merger proposal, Valley could adjourn the special meet without a vote on the merger proposal and seek to convince the holders of those she to changes their votes to votes in favor of the approval and adoption of the merger proposal.

If the special meeting is adjourned so that the Valley board of directors can so additional proxies to approve the merger proposal, Valley will not be required to any notice to shareholders of the adjourned meeting s place, date and time other an announcement of the place, date and time provided at the special meeting.

Vote Required

At the special meeting, the adjournment proposal requires the affirmative vote of least a majority of the shares voted in person or represented by proxy and entitle vote on the proposal. Abstentions and broker non-votes will have no effect on proposal to adjourn the special meeting, but will be treated as present at the mee for purposes of determining a quorum.

Brokers may not vote on the adjournment proposal without specific instructions f the person who beneficially owns the shares. Thus, shares held by a broker to what you do not give instructions on how to vote will have no effect on the outcome of vote on the adjournment proposal. However, if the number of affirmative votes for the adjournment proposal is a majority of the votes cast, but such votes do constitute a majority of the quorum required to transact business at the spe meeting, then abstentions and broker non-votes will have the same effect as a vagainst the merger proposal.

-34-

Recommendation of the Valley Board of Directors

The Valley board of directors recommends a vote **FOR** granting discret authority to adjourn the special meeting, if necessary, to permit further solicitatio proxies if there are not sufficient votes at the time of the special meeting to apprehe merger proposal.

-35-

THE MERGER

The following is a discussion of the merger and the material terms of the mergerement between CVB Financial and Valley. You are urged to read carefully merger agreement in its entirety, a copy of which is attached as Appendix A to proxy statement/prospectus and incorporated by reference herein. This summary a not purport to be complete and may not contain all of the information about merger agreement that is important to you. This section is not intended to provide with any factual information about CVB Financial, Citizens Business Bank, Valle Valley Business Bank. Such information can be found elsewhere in this prestatement/prospectus and in the public filings CVB Financial makes with the SEC described in the section entitled WHERE YOU CAN FIND MORE INFORMATE beginning on page i.

Terms of the Merger

Transaction Structure

The merger agreement provides for CVB Financial s acquisition of Valley throug merger of Valley with and into CVB Financial, with CVB Financial continuing as surviving corporation.

Immediately following the merger, Valley Business Bank, which is a wholly-ow subsidiary of Valley, will merge with and into Citizens Business Bank, with Citiz Business Bank continuing as the surviving corporation (we sometimes refer to merger between the subsidiary banks as the bank merger.)

Merger Consideration and Special Dividend

In the merger, all outstanding shares of Valley common stock will be converted an aggregate of 1,942,673 shares of CVB Financial common stock and \$23,400,00 cash, subject to potential adjustment depending on Valley s merger-related expe In addition, under the merger agreement, Valley will pay a special dividend to hol of Valley common stock. The amount of the special dividend will depend on Val tangible common equity as of the end of the month immediately prior to the closin the merger and would have been approximately \$13,700,000 if the merger had be completed on October 31, 2016, based on Valley s tangible common equity September 30, 2016. Based on the 3,002,014 shares of Valley common st outstanding and the closing price of CVB Financial common stock of \$16.67 per s reported on the Nasdaq Global Select Market on November 4, 2016 and Val tangible common equity capital as of September 30, 2016, each share of Va common stock would be converted into the right to receive 0.6471 shares of C Financial common stock valued at \$10.79 and \$7.79 in cash and would entitle holder to receive a cash dividend in the estimated amount of \$4.56, having a comb value of approximately \$23.14 per share of Valley common stock. Valley sharehol will receive cash in lieu of any fractional shares of CVB Financial common stock a result of the merger, Valley shareholders would own approximately 1.8% of C Financial s outstanding common stock after the merger is completed, excludin

shares of CVB Financial common stock they may already own.

The value of the merger consideration and the special dividend that Value shareholders will receive in connection with the merger is subject to change. number of shares of common stock that CVB Financial will issue to Vall shareholders is fixed and, therefore, the value of the stock component will chang the market price of CVB Financial common stock changes. The aggregate consideration of \$23,400,000 is subject to potential downward adjustment by amount, if any, that Valley s total merger-related transaction expenses ex \$3,500,000 and, without duplication, the amount or amounts that the change in corpayments and benefits payable to employees exceed \$1,800,000; that cont termination fees exceed \$1,000,000; that legal and professional fees exceed aggregate of \$600,000 or that the cost of a tail policy for Valley s directors and of liability insurance exceeds \$70,000.

Further, the amount of the special dividend that Valley will pay to its shareholdepends on Valley s tangible common equity as of the final date of the nimmediately preceding the merger. The aggregate

-36-

amount of the special dividend will be equal to the amount by which Valley s tar common equity exceeds the greater of (1) \$37,000,000 (or, if Valley has so specified loan, \$37,500,000) or (2) the amount required to achieve an 8.0% tang common equity ratio as of the final day of the month immediately preceding closing date of the merger. Therefore, the amount of the special dividend generally increase as Valley recognizes net earnings and will generally deceas Valley recognizes net losses, if any. Valley shareholders will receive a pro rata por of the stock and cash merger consideration and the special dividend. Based on Val common equity at September 30, 2016, Valley s shareholders would have recapproximately \$13.7 million as a special dividend or approximately \$4.56 cash share, in addition to the merger consideration.

The merger agreement does not permit Valley to issue additional shares of communications and there are no options to purchase common stock outstanding. Therefore, number of shares of Valley common stock outstanding will not change. No fractional shares will be issued. CVB Financial will pay cash in lieu of fractional shares.

Treatment of Valley Stock Options

The merger agreement provides that all outstanding stock options to purchase Va common stock, to the extent not exercised, will be cancelled and the holder of option will be entitled to receive, subject to any required tax withholding, an amount cash, without interest, equal to the amount by which, if any, the value of the share merger consideration to be received by holders of Valley stock exceeds exercise price per share of the option, multiplied by the number of shares of Vastock underlying the option, whether or not vested. As of November 7, 2016, hower all options to purchase Valley common stock had been exercised and there were outstanding options to purchase shares of Valley common stock.

Background of the Merger

Each of CVB Financial and Valley s board of directors and management regular review their respective business strategies, opportunities and challenges as partheir consideration and evaluation of their respective long-term prospects, with goal of enhancing value for their respective shareholders.

For Valley s board of directors, those strategies have included, among other thing business and regulatory environment facing financial institutions generally, as we conditions and ongoing consolidation in the financial services industry. Over the few years, Valley s board of directors has identified that one of the concerns for V is the importance of succession planning, because certain members of execumanagement are approaching retirement age and Valley will also need to find directors to replace those who may plan to retire. Strategic discussion topics by typically involved a review of current and projected market conditions, the result operations of Valley, certain peer group information comparisons, reported me and acquisition activity and selected industry information and analysis provided to board of directors by its financial advisors.

Valley, on a regular basis, has evaluated strategic combinations through acquisition of financial institutions in the Central Valley or the Central Coast area, well as strategic combinations with other financial institutions as means of grow asset size, remaining more competitive and delivering a greater value for Va shareholders. In the normal course of its business Valley has, from time to ti received unsolicited verbal inquiries from various sources regarding possible intering a potential business combination transaction. The general policy of Valley soft directors was to explore those opportunities where Valley would be the succeentity.

In late 2015, Valley discussed a potential strategic transaction where Valley would be the successor entity, but would become part of a larger organization. Through process, Valley recognized that its significant level of capital presented a challe because most acquiring financial institutions were not paying premiums for excapital. Rather, merger consideration was often based on premiums being paid capital ratios between 8% to 9%; with little or no premium attributable to capital excess of that range.

-37-

During this time Valley had conversations with Gary Steven Findley of Gary Ste Findley & Associates, who conducted strategic planning for Valley and its boar directors for the last few years and was familiar with Valley s operation, as well a merger and acquisition environment. During the last few years, Findley had discust the current market conditions and stand-alone value of Valley, as well as poss alternative strategic partners.

In late 2015 Valley was approached by a financial entity about a busin combination. This financial institution was familiar with the Central Va marketplace and was interested in a strategic combination where Valley would acquired by that financial entity for cash and stock. Several conversations occubetween Valley and that other financial institution. When the conversations with financial entity that had approached Valley did not materialize with a value accept to Valley s board of directors, Valley, through its President and Chief Exect Officer Allan W. Stone, made contact with CVB Financial regarding CVB Financial regarding its business in the Central Valley.

Over many years, CVB Financial has considered acquisitions as a means of achiever growth and expanding its market. Most recently and consistent with this strategy February 29, 2016, CVB Financial acquired County Commerce Bank, which based in Oxnard, California and had approximately \$256 million in assets. Previous in 2014, CVB Financial acquired American Security Bank, which was base Newport Beach, California and had approximately \$431 million in assets.

CVB Financial had completed other acquisitions and had established offices in Central Valley over the last several years. In February, 2016, conversation be between Allan W. Stone and Chris Myers, President and CEO of CVB Financial about a strategic combination that would involve both stock of CVB Financial, and a special dividend to Valley shareholders in the amount of the excess capital Valley. Allan W. Stone and representatives of Valley, as well as CVB Financial, in early March to further discuss the strategic combination between CVB Financial and Valley, and preliminary information was disseminated between the parties. But upon those conversations, an additional meeting was held on March 28, 2016 between the parties to discuss the cash, stock and special dividend that would be paid to Valley shareholders as part of the transaction.

Throughout the month of April 2016, additional discussions took place concernstrategic combinations, focusing on the value of CVB Financial shares, the capratios of Valley, the amount of the special dividend and overall timing transaction. Based upon those conversations, a draft letter of intent was submitted CVB Financial to Valley on April 27, 2016, and additional discussions took p between the parties with regard to the letter of intent. The board of directors of Vahad a meeting on May 3, 2016 to further discuss the letter of intent, and based upon conversations during that meeting, additional conversations were held we representatives of CVB Financial, which resulted in a revised letter of intent decomposition of the board of directors of Valley.

The board of directors of Valley met on May 5, 2016 to evaluate the letter of intensity which time Findley was available to discuss the value of the offer. The value of merger consideration offered, combined with the anticipated amount of the specash dividend to be paid to Valley shareholders, was estimated at approximately million. The number of shares being offered by CVB Financial was 1,942,673 sha and the amount of cash being delivered to Valley shareholders by CVB Financial \$23.4 million. There was also discussion with regard to a cap of Valley's profess fees and other transaction costs of Valley identified with the transaction, amountin \$3.5 million.

In the meeting of May 5, 2016, Valley s board of directors discussed the merits of letter of intent and analyzed the price terms, merits, risks and strategic reasons for against the transaction. On May 5, 2016, Valley s board of directors accepted Financial s letter of intent, and authorized and the execution of a nondiscle agreement/confidentiality agreement dated May 4, 2016. Subsequently, on May 2016 CVB Financial was provided with access to an electronic data room, who Valley began to populate with due

-38-

diligence materials. From May through September 15, 2016, Valley and C Financial continued mutual due diligence and negotiated the merger agreement related matters with the help of each institution s legal and financial advisors.

On July 11, 2016, representatives of CVB Financial and Valley met and discussed terms of the due diligence being been conducted and the importance of an increasumount of shareholders—equity to be delivered by Valley as part of the transaction such date, until September 22, 2016, the parties engaged in signific discussion with regard to the terms and conditions of the merger consideration, as as the special dividend and the amount of shareholders—equity to be delivered Valley as of the month prior to closing.

On August 12, 2016, Valley received the first draft of the proposed merger agreem and over the next 40 days the parties, financial advisors and the legal counegotiated the terms of the merger agreement and the related merger documents.

On September 13, 2016, Valley s board of directors convened a special meetin Gary Steven Findley & Associates, Valley s legal counsel, provided a det overview of the merger agreement and related exhibits, including voting agreem and non-solicitation agreements. At that meeting, Vining Sparks reviewed financial aspects of the proposed merger and rendered its verbal opinion, which confirmed in writing as of September 22, 2016, to the effect that, as of that date subject to the procedures followed, assumptions made, matters considered, qualifications and limitations on the review undertaken by Vining Sparks as set f in such opinion, the merger consideration to be received by the holders of Va common stock, inclusive of the special dividend, in the proposed merger was from a financial point of view, to such holders. The board of directors of Va determined that if a few final issues with the draft merger agreement could resolved, most significantly the inclusion of a walk-away collar giving Valley right to terminate the merger agreement if the average trading price of CVB Finar common stock fell below a specified level shortly prior to closing and the condit to and treatment of a potential sale of a certain charged-off loan, they would be position to approve the merger agreement and other related documents.

On September 19, 2016, a meeting was held with representatives of CVB Financial including CVB Financial is President and Chief Executive Officer, Chris Myers, Financial is Chairman of the Board, Ray O Brien, CVB Financial is Chief F Officer, Allen Nicholson, and a representative of CVB Financial is investment by Keefe, Bruyette & Woods, Inc., along with Valley is full board of directors and Steven Findley. At that meeting, there was discussion with regard to the inclusion walk-away collar giving Valley the right to terminate the merger agreement if average trading price of CVB Financial common stock fell below a specified leshortly prior to closing and CVB Financial the right to terminate the merger agreement if the average trading price of CVB Financial common stock rospecified level shortly prior to closing, the conditions to and treatment of a potential of a certain charged-off loan, and the expectations among the respective particle of the condition of the

the merger agreement s inclusion of a walk-away collar and agreed-upon provi concerning the sale of a specified loan.

The CVB Financial board of directors held a meeting on September 21, 2016, which was also attended by CVB Financial is management and its legal and financial advanter a thorough discussion, CVB Financial is board of directors approved the magreement and the other related agreements with Valley and its officers and direct that are contemplated by the merger agreement.

On September 21, 2016, final execution copies of the merger agreement w distributed.

On September 22, 2016, and pursuant to the resolutions adopted by each of C Financial and Valley s board of directors, CVB Financial and Valley entered in merger agreement dated as of September 22, 2016. On September 22, 2016, after close of trading on the NASDAQ Global Select Market, a joint press releannouncing the execution of the merger agreement was issued by CVB Financial Valley.

-39-

Reasons for the Merger and Recommendation of Valley s Board of Directors

Valley s board of directors believes the proposed merger with CVB Financial is and in the best interests of its shareholders, as well as its employees and communities served by Valley. In reaching this conclusion, Valley s board of direction discussed the proposed merger with its senior management and with its financial legal advisors and considered the relative advantages and disadvantages of remain independent rather than entering into the merger. Valley s board of direction unanimously recommend that Valley s shareholders vote in favor of the magreement and consummation of the merger and the other transactions contemple by the merger agreement.

In unanimously approving the merger agreement, the merger and other transact contemplated in the merger agreement, inclusive of the special dividend as bein the best interests of Valley and its shareholders and recommending that Va shareholders vote FOR the merger proposal, Valley s board of directors consulvalley s management, as well as its financial and legal advisors, and consider number of factors, including the following material factors:

Its knowledge of Valley s business, operations, financial condition, quality, earnings, capital and prospects both as an independent organization a possible acquirer executing its strategic plan and as a part of a combi company with CVB Financial, as well as under various other alterna scenarios;

Its understanding of CVB Financial s business, operations, regulatory financial condition, asset quality, earnings, capital and prospects, taking account presentations by senior management of the results of the due diligereview and information from Vining Sparks and Gary Steven Findle Associates;

Its belief that the merger will result in Valley shareholders become shareholders of a premier, locally-operated Southern California commendanking franchise with a diversified revenue stream, strong capital ratio well-balanced loan portfolio and an attractive funding base that has potential to deliver a higher value to Valley s shareholders than alternative the merger;

The complementary nature of the cultures and product mix of Valley and CVB Financial, including strategic focus, target markets and client service, which management believes should facilitate integration and implementation of the transaction;

The expanded possibilities, including organic growth and future acquisit and other strategic transactions, that would be available to the combi company given its larger size, asset base, capital, market capitalization trading liquidity and footprint;

The fact that the combined value of the per share merger consideration special dividend for holders of Valley shareholders in the estimated amour \$23.71 per share represents a premium of approximately 43% over the \$16 last reported trading price of Valley common stock on the OTC Markets Pink market on September 22, 2016 (the day of, and immediately prior to, first announcement of the terms of the merger), and the belief that transaction is likely to provide substantial future value to Valley s shareho as well as the benefit of additional liquidity enjoyed by shareholders NASDAQ-listed security such as the shares of common stock of C Financial;

The importance of dealing with succession planning issues, both from standpoint of the board of directors of Valley, as well as management;

The opinion, dated as of September 22, 2016, delivered to the Valley boar directors by Vining Sparks to the effect that, as of the date of the opinion based upon and subject to the procedures followed, assumptions made, mat considered, and qualifications and limitations on the review undertaker Vining Sparks as set forth in such opinion, the merger consideration to received by the holders of Valley common stock in the proposed mer inclusive of the special dividend, is fair, from a financial point of view, to sholders;

-40-

The fact that CVB Financial s offer, combined with the estimated value special dividend, was in the aggregate significantly higher than the aggre offering price that was discussed with another potential merger partner in 2015;

The familiarity of Valley s management team with CVB Financian management team, and the belief of Valley s management that the management and employees of Valley and CVB Financial possess complementary skills expertise;

Its belief that the potential of combining with a larger company will proadditional products and services to better grow and retain Valley s custo that the combined, more diversified, customer base will improve and diver future revenue sources, and that future earnings prospects will be stronger combined basis;

The effects of the merger on Valley employees, including the retention of a significant number of Valley employees, which would, in turn, increase the retention of Valley customers and the likelihood of success of the combined companies;

Its understanding of the current and prospective environment in which Va and CVB Financial operate, including national and local economic conditi the interest rate environment, increasing operating costs resulting f regulatory initiatives and compliance mandates, the competitive environment for financial institutions generally, and the likely effect of these factor Valley both with and without the proposed transaction;

The reports of Valley s management to Valley s board of directors conc the operations, financial condition and prospects of CVB Financial and expected financial impact of the merger on the combined company, include pro forma assets, earnings, deposits and other financial metrics;

The limited availability of acquisition opportunities that met, or would regoing forward, the strategic goals of Valley, including growth in the curenvironment;

The ability to diversify outside of the Central Valley with the proportransaction between Valley and CVB Financial;

The merger being consistent with Valley s strategic plan, including achieving strong earnings growth and improving customer attraction, retention and service;

The complementary fit of Valley and CVB Financial because of the natur the markets served and the products offered by the two institutions;

The enhancement of the combined company s competitive position expect result from the merger, including that the combined company is expecte have a greater market reach with expanded resources and broader proof offerings in areas currently served by Valley;

The ability of CVB Financial to complete the merger from a financial and regulatory perspective;

The equity interest in the combined company that Valley s existing sharehold will receive in the merger, which will allow such shareholders to continuparticipate in the future success of the combined company;

The cash proceeds to be received by Valley shareholders, including the oportion of the merger consideration and the special dividend;

The greater market capitalization and trading liquidity of CVB Finan common stock, which is listed on NASDAQ Global Select Market, in the ethat Valley shareholders desire to sell the shares of CVB Financial communication to be received by them following completion of the merger;

Its understanding that the merger is intended to qualify as a reorganiz within the meaning of Section 368(a) of the Internal Revenue Code, and such, the merger generally is intended to result in favorable U.S. fed income tax consequences to all of the parties, including Valley s shareho and

-41-

Its review with its independent legal advisor, Gary Steven Findle Associates, of the material terms of the merger agreement, including the bo ability, under certain circumstances, to withdraw its recommendatio Valley s shareholders and to consider another acquisition proposal, subj the potential payment by Valley of a termination fee of \$3.5 million to C Financial and payment of certain expenses, which Valley s board of direct concluded was reasonable in the context of termination fees in compart transactions and in light of the overall terms of the merger agreement.

Valley s board of directors also considered a number of potential risks uncertainties associated with the merger agreement, the merger and the or transactions contemplated by the merger agreement, in connection with deliberation of the proposed transaction, including, without limitation, the following

The possibility that CVB Financial may not be able to retain all officers employees of Valley after the merger;

The potential risk of diverting management attention and resources from operation of Valley s business and towards the completion of the merger;

The restrictions on the conduct of Valley s business prior to the completion of the merger, which are customary for public company merger agreements involving financial institutions, but which, subject to specific exceptions, could delay or prevent Valley from undertaking business opportunities that may arise or any other action it would otherwise take with respect to the operations of Valley absent the pending merger;

The potential risks associated with achieving anticipated cost synergies savings and successfully integrating Valley s business, operation workforce with those of CVB Financial;

The risked that merger-related costs will be greater than expected or we exceed the limits set forth in the merger agreement;

The fact that the interests of certain of Valley s directors and executive officers may be different from, or in addition to, the interests of Valley s other shareholders;

The fact that a portion of the merger consideration consists of a fixed num of shares of CVB Financial common stock;

The fact that, while Valley expects that the merger will be consummated, the can be no assurance that all conditions to the parties—obligations to comple merger agreement will be satisfied, including the risk that necessary regula or shareholder approvals might not be obtained and, as a result, the merger not be consummated;

The risk of potential employee attrition and/or adverse effects on business customer relationships as a result of the pending merger;

The risk that the terms of the merger agreement, including provisions rela to the payment of a termination fee under specified circumstances, altho required by CVB Financial as a condition to its willingness to enter in merger agreement, could have the effect of discouraging other parties might be interested in a transaction with Valley from proposing suc transaction;

The risk that Valley might not satisfy some or all of the financial measures that are conditions to CVB Financial s obligation to complete the merger, in which case CVB Financial could decide not to complete the merger;

The possibility that the merger might not be completed and the impact of a public announcement of the termination of the merger agreement on, among other things, the market price of Valley common stock and Valley operating results, particularly in light of the costs incurred in connection with the transaction; and

The other factors described in this proxy statement/prospectus.

-42-

The foregoing discussion of the information and factors considered by Valley soft directors is not intended to be exhaustive, but includes the material factors considered by Valley soft board of directors. In reaching its decision to approximerger agreement, the merger and the other transactions contemplated by the meagreement, Valley soft board of directors did not quantify or assign any relative we to the factors considered, and individual directors may have given different weight different factors. Valley soft board of directors considered all these factors as a wincluding discussions with, and questioning of Valley soft management and Valindependent financial and legal advisors, and overall considered the factors to favorable to, and to support, its determination.

Valley s board of directors unanimously approved the merger agreemen unanimously recommends that Valley s shareholders vote **FOR** the approvemerger proposal.

The foregoing discussion of the information and factors considered by the Va board of directors is not intended to be exhaustive, but includes the material factors considered by the Valley board of directors. In reaching its decision to approve merger agreement, the merger and the other transactions contemplated by the me agreement, the Valley board of directors did not quantify or assign any relawights to the factors considered, and individual directors may have given diffe weights to different factors. The Valley board of directors considered all these factors as a whole, including discussions with, and questioning of Valley s management Valley s independent financial and legal advisors, and overall considered the fact be favorable to, and to support, its determination.

This summary of the reasoning of Valley s board of directors and other inform presented in this section is forward-looking in nature and, therefore, should be realight of the factors discussed under the section entitled CAUTION STATEMENT REGARDING FORWARD-LOOKING STATEMENTS beginning page 20 of this proxy statement/prospectus.

Opinion of Vining Sparks for Valley s Board of Directors

Valley s board of directors retained Vining Sparks IBG, L.P. (Vining Sparks) of financial advisory and investment banking services. Vining Sparks is a nation recognized investment banking firm with substantial expertise in transactions sint to the proposed transaction and is familiar with Valley and its business. As part of investment banking business, Vining Sparks is regularly engaged in the valuation financial services companies and their securities in connection with mergers acquisitions, private placements and valuations for estate, corporate and of purposes.

On September 22, 2016, Vining Sparks delivered its opinion to Valley that the me consideration and the amount of the special dividend to be received by Va common shareholders in the proposed transaction is fair, from a financial poir view, to Valley s common shareholders. The full text of Vining Sparks opi attached as **Appendix B** to this proxy statement/prospectus and should be read in

entirety.

Vining Sparks opinion was directed to Valley s board of directors and is limited fairness, from a financial point of view, of the merger consideration and the amount the special dividend to be received by Valley common shareholders in the proportion transaction. It did not address Valley s underlying business decision to proceed the proposed transaction or constitute a recommendation to the Valley board directors as to how it should vote on the merger, and does not constitute recommendation to any holder of Valley common stock as to how such shareho should vote in connection with the merger. Vining Sparks did not express any opin as to the fairness of the amount or nature of the compensation to be received in merger by Valley s officers, directors or employees, or class of such persons, reto the compensation to be received by Valley s shareholders.

Vining Sparks opinion was reviewed and approved by Vining Sparks Fa Opinion Committee in conformity with its policies and procedures established un the requirements of Rule 5150 of the Financial Industry Regulatory Authority.

-43-

For purposes of Vining Sparks opinion and in connection with its review oproposed transaction, Vining Sparks has, among other things:

reviewed the terms of the merger agreement made available to Vining Spark

reviewed certain publicly available financial statements, both audited (whavailable) and un-audited, and related financial information of Valley and C Financial, including those included in their respective annual reports for past two years and their respective quarterly reports for the past two years;

reviewed certain internal financial information and financial forecasts related to the business, earnings, cash flows, assets and prospects of Valley furnisto Vining Sparks by Valley management;

held discussions with members of executive and senior management of Va concerning the past and current results of operations of Valley, its cur financial condition and management s opinion of its future prospects;

reviewed reported market prices and historical trading activity of C Financial and Valley common stock;

reviewed certain financial performance and stock market information Valley and CVB Financial and compared such information with similar available for certain other financial institutions the securities of which publicly traded;

reviewed publicly available consensus street estimates of CVB Financial earnings for 2016 through 2018;

reviewed the financial terms of merger and acquisition transactions, to extent publicly available, involving financial institutions and finan institution holding companies that Vining Sparks deemed to be relevant; and

reviewed such other information, financial studies, analyses and investigati as Vining Sparks considered appropriate under the circumstances.

In conducting its review and arriving at its opinion, Vining Sparks has assumed relied, without independent verification, upon the accuracy and completeness of a the financial and other information that has been provided to it by Valley and C

Financial, and their respective representatives, and of the publicly availar information that was reviewed by Vining Sparks. Vining Sparks is not an expert in evaluation of the adequacy of allowances for loan losses and it did not independent verify the adequacy of such allowances. Vining Sparks assumed that the allowances loan losses set forth in the financial statements of CVB Financial and Valley was adequate to cover such losses and complied fully with applicable law, regular policy and sound banking practice as of the date of such financial statements. Vin Sparks did not conduct a physical inspection of any of the properties or facilities. Valley or CVB Financial, did not make any independent evaluation or appraisal of assets, liabilities or prospects of Valley or CVB Financial, was not furnished with such evaluation or appraisal, and did not review any individual credit files.

Vining Sparks relied upon the management of Valley as to the reasonableness of financial and operating forecasts, and projections (and the assumptions and be therefore) provided to or reviewed by Vining Sparks, and Vining Sparks assumed such forecasts and projections reflect the best currently available estimates judgments of Valley management. Valley does not publicly disclose inte management forecasts, projections or estimates of the type furnished to or reviewe Vining Sparks in connection with its analysis of the financial terms of the propo transaction, and such forecasts and estimates were not prepared with a view tow public disclosure. Vining Sparks relied upon the management of CVB Financial a the reasonableness of the publicly available consensus street estimates of Financial earnings. It is understood that these forecasts and estimates referred to at were based on numerous variables and assumptions that are inherently uncertain which may not be within the control of the management of Valley or CVB Financian including without limitation to, the general economic, regulatory and competi conditions. Accordingly, actual results could vary materially from those set fort such forecasts and estimates.

-44-

Vining Sparks opinion is necessarily based on economic, market, and other cond as in effect on, and the information made available to it as of, the date of its opin Events occurring after the date thereof, including but not limited to, changes affect the securities markets, the results of operations or material changes in the asset liabilities of CVB Financial or Valley could materially affect the assumptions use preparing the opinion. Vining Sparks assumed that there were no material change the assets, liabilities, financial condition, results of operations, business or prosp of either Valley or CVB Financial since the date of the last financial statement each entity that were made available to Vining Sparks. Vining Sparks assumed that of the representations and warranties contained in the merger agreement and related agreements are true and correct, that each party to the merger agreement perform all of the covenants required to be performed by each party under sagreement and that the conditions precedent in the merger agreement are not waive

In delivering its opinion to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of directors of Valley, Vining Sparks prepared to the board of the b and delivered to Valley s board of directors written materials containing va analyses and other information. The following is a summary of the material finar analyses performed by Vining Sparks in connection with the preparation of its opin and does not purport to be a complete description of all the analyses performed Vining Sparks. The summary includes information presented in tabular format, w should be read together with the text that accompanies those tables. The preparatio a fairness opinion is a complex process involving various determinations as to most appropriate and relevant methods of financial analysis and the applicatio those methods to the particular circumstances. Therefore, an opinion is not necessary susceptible to partial analysis or summary description. Vining Sparks believes that analyses must be considered as a whole and that selecting portions of such analyses and the factors considered therein, without considering all factors and analyses, co create an incomplete view of the analyses and the processes underlying its opinion its analyses, Vining Sparks made numerous assumptions with respect to indu performance, business and economic conditions, and other matters, many of which beyond the control of Valley, CVB Financial and Vining Sparks. Any estim contained in Vining Sparks analyses are not necessarily indicative of future resu values, which may be significantly more or less favorable than such estima Estimates of values of companies do not purport to be appraisals or necessarily rethe prices at which companies or their securities may actually be sold.

Vining Sparks opinion was based on information available to Vining Sparks the the date of its opinion and conditions as they existed and could be evaluated on date thereof. Vining Sparks reviewed the financial terms of the proposed transaction set forth in the merger agreement and for purposes of the financial analyses describelow, Vining Sparks calculated an implied transaction value of \$72,385,950 \$24.11 per share, which was based on cash consideration of \$23.4 million, issuance of 1,942,673 shares of CVB Financial common stock, a stock price of \$1 per share for CVB Financial (based on the 20 day average closing price end September 21, 2016), and an estimated Special Dividend of \$15.3 million assumiclosing date of February 28, 2017. Per Section 8.2(f) of the merger agreement, Valand CVB Financial have agreed to a trading collar. If the volume-weighted averaging price for CVB Financial common stock for the ten trading days ending or

fifth trading day prior to the closing date of the merger is less than \$13.50, then Va may terminate the merger agreement, or if the volume-weighted average closing p for CVB Financial common stock for the ten trading days ending on the fifth trad day prior to the closing date of the merger is greater than \$20.50, then CVB Financial terminate the merger agreement. A trading collar of \$13.50 to \$20.50 means transaction value would have a minimum value of \$21.63 per share and a maxim value of \$26.16 per share.

-45-

Selected Company Analysis CVB Financial. Vining Sparks used publicly available information to compare selected financial and market trading information for C Financial and a selected group of financial institutions. The CVB Financial peer group consisted of publicly traded California banks with total assets between \$5 billion \$25 billion with a return on assets greater than 0.00%, excluding merger targ While Vining Sparks believes that the companies listed below are similar to C Financial, none of these companies have the same composition, operations, siz financial profile as CVB Financial.

Company	Ticker	City	State
Banc of California, Inc.	BANC	Irvine	CA
Cathay General Bancorp	CATY	Los Angeles	CA
Farmers & Merchants Bank of Long			
Beach	FMBL	Long Beach	CA
Hope Bancorp, Inc.	HOPE	Los Angeles	CA
Opus Bank	OPB	Irvine	CA
PacWest Bancorp	PACW	Beverly Hills	CA
Westamerica Bancorporation	WABC	San Rafael	CA

To perform this analysis, Vining Sparks used financial information as of June 2016, a price of \$17.34 for CVB Financial (the 20 day average closing price ending September 21, 2016) and pricing data for the peer group as of September 21, 20 obtained from SNL Financial LC. The following table sets forth the compara financial and market data:

		Peer Group
	CVB Financial	Median
Total Assets (in millions)	\$ 8,312.3	\$ 8,336.8
Return on Average Assets	1.24%	1.13%
Return on Average Equity	10.13%	9.23%
Equity/Assets	11.93%	12.74%
Loans/Deposits	64.35%	96.57%
Loan Loss Reserve/Gross		
Loans	1.44%	1.16%
Nonperforming		
Assets/Assets	0.53%	0.98%
Efficiency Ratio	44.70%	51.26%
Price/Book Value Per Share	1.89x	1.35x
Price/Tangible Book Value		
Per Share	2.08x	1.76x
Price/Last 12 Months		
Earnings Per Share	17.7x	15.5x

Selected Company Analysis Valley. Vining Sparks used publicly available inform to compare selected financial and market trading information for Valley and a sele group of financial institutions. The Valley peer group consisted of publicly tra

California banks with total assets between \$300 million and \$500 million with a re on assets greater than 0.00%, excluding merger targets. While Vining Sparks belief that the companies listed below are similar to Valley, none of these companies here same composition, operations, size or financial profile as Valley.

Company	Ticker	City	State
AltaPacific Bancorp	ABNK	Santa Rosa	CA
American Riviera Bank	ARBV	Santa Barbara	CA
Bank of Santa Clarita	BSCA	Santa Clarita	CA
Bank of Southern California, N.A.	BCAL	San Diego	CA
Community 1st Bancorp	CFBN	Auburn	CA
New Resource Bank	NWBN	San Francisco	CA
PBB Bancorp	PBCA	Los Angeles	CA
Private Bancorp of America, Inc.	PBAM	La Jolla	CA
Redwood Capital Bancorp	RWCB	Eureka	CA
Suncrest Bank	SBKK	Visalia	CA
United American Bank	UABK	San Mateo	CA

-46-

To perform this analysis, Vining Sparks used financial information as of June 2016, a price of \$16.90 for Valley (the closing price on September 21, 2016) pricing data for the peer group as of September 21, 2016 obtained from SNL Financial. The following table sets forth the comparative financial and market data:

		Peer Group
	Valley	Median
Total Assets (in millions)	\$416.1	\$ 323.5
Return on Average Assets	1.06%	0.76%
Return on Average Equity	9.16%	7.15%
Equity/Assets	11.65%	9.43%
Loans/Deposits	83.56%	87.39%
Loan Loss Reserve/Gross Loans	1.12%	1.15%
Nonperforming Assets/Assets	0.87%	0.30%
Efficiency Ratio	71.17%	67.73%
Price/Book Value Per Share	1.00x	1.03x
Price/Tangible Book Value Per		
Share	1.00x	1.11x
Price/Last 12 Months Earnings		
Per Share	10.1x	13.4x

Stock Trading History. Vining Sparks reviewed the closing per share market prand volumes for CVB Financial common stock and Valley common stock on a dasis from September 21, 2015 to September 21, 2016. CVB Financial is listed trading on NASDAQ under the symbol CVBF. For the period between Septem 2015 and September 21, 2016, the closing price of CVB Financial common stranged from a low of \$14.02 to a high of \$18.77, with an average closing price for period of \$16.80. The 20 day average closing price ending September 21, 2016 \$17.34 per share and the average daily trading volume for CVB Financial was 557 shares.

Valley common stock trades on the OTC Pink Sheets under the ticker symbol V For the period between September 21, 2015 and September 21, 2016, the closing p of Valley common stock ranged from a low of \$15.20 to a high of \$17.00, with average closing price for the period of \$15.85. The closing price on September 2016 was \$16.90 per share and the average daily trading volume for Valley was 1 shares. The transaction value of \$24.11 per share represented a 43% premium of Valley s closing price on September 21, 2016.

Analysis of Selected Financial Institution Transactions. Vining Sparks revie certain publicly available information regarding selected merger and acquisi transactions (the Comparable Transactions) announced from January 1, 2 September 21, 2016 involving 27 Western financial institutions with total assets us \$4 billion and a return on assets over 0.00%. The transactions included in the grare shown in the following chart. This data was obtained from SNL Financial LC.

Acquirer:

Western Alliance Bancorporation

First Interstate BancSystem, Inc.

Heritage Commerce Corp

FNB Bancorp

Heartland Financial USA, Inc.

PBB Bancorp

Northwest Bancorporation, Inc.

American Riviera Bank

Suncrest Bank

California Bank of Commerce

HomeStreet, Inc.

Pacific Premier Bancorp, Inc.

Acquired Company: Bridge Capital Holdings

Absarokee Bancorporation, Inc.

Focus Business Bank America California Bank Premier Valley Bank

First Mountain Bank

Fairfield Financial Holdings Bank of Santa Barbara Sutter Community Bank

Pan Pacific Bank

Orange County Business Bank Security California Bancorp

-47-

Acquired Company: Acquirer: CVB Financial Corp. County Commerce Bank **RBB** Bancorp **TFC Holding Company** Pacific Commerce Bancorp ProAmérica Bank Sierra Bancorp Coast Bancorp State Bank Corp. Country Bank 1st Century Bancshares, Inc. Midland Financial Co. Flathead Bank of Bigfork First Interstate BancSystem, Inc. Treasure State Bank Glacier Bancorp, Inc. Prime Pacific Financial Services Cascade Bancorp California Republic Bancorp Mechanics Bank Central Valley Community Bancorp Sierra Vista Bank Cathay General Bancorp SinoPac Bancorp Commencement Bank Thurston First Bank AltaPacific Bancorp Commerce Bank Suncrest Bank Security First Bank

Vining Sparks reviewed the multiples of transaction value to stated book value, transaction value to tangible book value, transaction value to last twelve more earnings, transaction value to assets and tangible book premium to core deposits calculated high, low, mean and median multiples for the Comparable Transacti These ratios were compared with corresponding transaction ratios for the proportion merger based on the transaction value at the minimum price of \$21.63 per share estimated value of \$24.11 per share, and the maximum value of \$26.16 per share Valley common stock. The results of the analysis are set forth in the following table

		Valley					
	Tran	saction Va	ılue	Comparable Transaction			ıs
Transaction	Min		Max		-		-
Multiples:	\$21.63	\$24.11	\$26.16	Low	Median	Mean	I
Transaction Value							
less Special							
Dividend/Required							
Equity of \$37.5							
Mill.	1.32x	1.52x	1.69x	0.89x	1.30x	1.35x	
Transaction Value							
less Special							ļ
Dividend/Required							ļ
Tang. Equity of							
\$37.5 Mill.	1.32x	1.52x	1.69x	0.89x	1.30x	1.37x	
Transaction							
Value/2015							
Earnings	14.32x	15.97x	17.32x	1.92x	23.56x	26.51x	6
Transaction							
Value/Assets	15.61%	17.40%	18.87%	8.48%	15.70%	15.91%	2
Tangible Premium/							
Core Deposits	4.20%	6.42%	8.27%	(1.58)%	5.91%	6.33%	1

No company or transaction used as a comparison in the above analysis is identically valley or the proposed transaction. Accordingly, an analysis of these results is strictly mathematical. An analysis of the results of the foregoing involves companied considerations and judgments concerning differences in financial and operational contents of Valley and the companies included in the Comparable Transaction

Present Value Analysis. Vining Sparks calculated the present value of theoret future earnings of Valley and compared the transaction value to the calculated pre value of Valley s common stock on a stand-alone basis. Based on projected ear for Valley for 2016 through 2020, discount rates ranging from 10% to 18%, including a residual value, the stand-alone present value of Valley common st indicated an implied range of values per share of \$11.07 to \$24.11.

Discount Rate	18%	16%	14%	12%	10%
Present Value (in					
thousands)	\$33,229	\$38,488	\$45,668	\$ 56,053	\$72,3
Present Value (ner share)	\$ 11.07	\$ 12.82	\$ 15.21	\$ 18.67	\$ 24

-48-

Discounted Cash Flow Analysis. Using a discounted cash flow analysis, Vir Sparks estimated the net present value of the future streams of after-tax cash flow Valley could produce to benefit a potential acquirer, referred to as dividendable income, and added a terminal value. Based on projected earnings for Valley for 2 through 2020, Vining Sparks assumed after-tax distributions to a potential acquirent such that its tier 1 leverage ratio would be maintained at 8.00%. The terminal value Valley was calculated based on Valley s projected 2020 tangible equity, the m price to tangible book multiple paid in the Comparable Transactions and utility discount rates ranging from 10% to 18%. This discounted cash flow analysis indicated implied range of values per share of Valley common stock of \$15.98 to \$20.10.

thousands) \$47,984 \$50,627 \$53,544 \$56,770 \$60,3	Discount Rate	18%	16%	14%	12%	10%
Present Value (per share) \$ 15.98 \$ 16.86 \$ 17.84 \$ 18.91 \$ 20. In the two years prior to the issuance of this opinion, Vining Sparks engage securities and loan sales and trading activity with Valley and CVB Financial an their subsidiary banks for which Vining Sparks was paid commissions or other f which may include mark-ups on the purchase or sale of loans and securities. Purst to the terms of an engagement letter with Valley, Vining Sparks received a fe \$55,000 plus expenses up to \$1,000 upon delivery of its opinion. Vining Sparks not contingent upon consummation of the proposed transaction. In addition, Va	Present Value (in					
In the two years prior to the issuance of this opinion, Vining Sparks engage securities and loan sales and trading activity with Valley and CVB Financial an their subsidiary banks for which Vining Sparks was paid commissions or other f which may include mark-ups on the purchase or sale of loans and securities. Purst to the terms of an engagement letter with Valley, Vining Sparks received a fe \$55,000 plus expenses up to \$1,000 upon delivery of its opinion. Vining Sparks not contingent upon consummation of the proposed transaction. In addition, Va	thousands)	\$47,984	\$50,627	\$ 53,544	\$56,770	\$60,3
securities and loan sales and trading activity with Valley and CVB Financial an their subsidiary banks for which Vining Sparks was paid commissions or other f which may include mark-ups on the purchase or sale of loans and securities. Purse to the terms of an engagement letter with Valley, Vining Sparks received a fe \$55,000 plus expenses up to \$1,000 upon delivery of its opinion. Vining Sparks not contingent upon consummation of the proposed transaction. In addition, Va	Present Value (per share)	\$ 15.98	\$ 16.86	\$ 17.84	\$ 18.91	\$ 20.
their subsidiary banks for which Vining Sparks was paid commissions or other f which may include mark-ups on the purchase or sale of loans and securities. Pursi to the terms of an engagement letter with Valley, Vining Sparks received a fe \$55,000 plus expenses up to \$1,000 upon delivery of its opinion. Vining Sparks not contingent upon consummation of the proposed transaction. In addition, Va	In the two years prior to t	he issuanc	e of this op	oinion, Vin	ing Sparks	engage
which may include mark-ups on the purchase or sale of loans and securities. Purst to the terms of an engagement letter with Valley, Vining Sparks received a fe \$55,000 plus expenses up to \$1,000 upon delivery of its opinion. Vining Sparks not contingent upon consummation of the proposed transaction. In addition, Va	securities and loan sales an	nd trading a	ctivity with	Valley and	l CVB Finaı	ncial an
to the terms of an engagement letter with Valley, Vining Sparks received a fe \$55,000 plus expenses up to \$1,000 upon delivery of its opinion. Vining Sparks not contingent upon consummation of the proposed transaction. In addition, Va	their subsidiary banks for v	which Vinir	ig Sparks w	as paid con	nmissions of	r other f
\$55,000 plus expenses up to \$1,000 upon delivery of its opinion. Vining Sparks not contingent upon consummation of the proposed transaction. In addition, Va	which may include mark-up	os on the pu	rchase or sa	le of loans	and securitie	es. Pursi
not contingent upon consummation of the proposed transaction. In addition, Va	to the terms of an engager	ment letter	with Valley	, Vining S _l	parks receiv	ved a fe
	\$55,000 plus expenses up to	o \$1,000 up	on delivery	of its opinio	on. Vining S	parks
has agreed to indemnify Vining Sparks against certain liabilities and expenses ari	not contingent upon consu	mmation of	the propos	ed transacti	ion. In addit	tion, Va
	has agreed to indemnify Vi	ning Sparks	against cer	tain liabiliti	es and expe	nses ari

out of or incurred in connection with its engagement, including liabilities

Management and Board of Directors of CVB Financial After the Merger

expenses which may arise under the federal securities laws.

Upon completion of the merger, the composition of the boards of directors of C Financial and Citizens Business Bank will remain unchanged. The current direct and executive officers of CVB Financial are expected to continue in their cur positions. Information about the current CVB Financial directors and execut officers can be found in the documents listed under the section entitled WHERE CAN FIND MORE INFORMATION beginning on page i.

Interests of Valley Directors and Executive Officers in the Merger

In considering the recommendation of the Valley board of directors with respect to merger, Valley shareholders should be aware that the directors and executive offi of Valley have certain interests in the merger that may be different from, or in addito, the interests of Valley shareholders generally. The Valley board of directors aware of these interests and considered them, among other matters, in approving merger agreement and the transactions contemplated thereby and in making recommendation that Valley shareholders vote to approve the merger proposal. The interests are described in further detail below.

Stock Ownership

The directors and executive officers of Valley beneficially owned as of the rec date, a total of 579,122 shares of Valley common stock, representing approxima 19% of the total outstanding shares of Valley common stock. Each of the directors three additional executive officers of Valley have entered into a voting and sup agreement pursuant to which, among other things, they have agreed to vote any sh they beneficially own in favor of the merger proposal and the adjournment proportion of the directors and executive officers of Valley will receive the same per share me consideration as the other shareholders of Valley.

Change in Control Payments Under Employment Agreements

Valley is party to agreements with certain of its executive officers that provide severance benefits in the event of certain qualifying events, including in connec with a qualifying termination of employment following a change in control.

-49-

Pursuant to the change in control agreement, dated as of December 28, 2015, betw Valley and Allan W. Stone, President and Chief Executive Officer, upon qualifying termination following a change in control, Mr. Stone will be entitle receive an amount equal to: (i) two times his annual then-current annual base sal (ii) payments equal to two years of his yearly bonus, auto allowance, and country and gym dues; and (iii) continued insurance benefits through age 65. The benefits punder (i) and (ii) will be paid in 24 equal installments.

Pursuant to the change in control agreement, dated as of December 28, 2015, betw Valley and Roy Estridge, Executive Vice President, Chief Operating Officer and C Financial Officer, upon his qualifying termination following a change in control, Estridge will be entitled to receive an amount equal to: (i) two times his and then-current annual base salary; (ii) payments equal to two years of his yearly bo auto allowance, and gym dues; and (iii) continued insurance benefits through 65. The benefits paid under (i) and (ii) will be paid in 24 equal installments.

Pursuant to the change in control agreement, dated as of December 28, 2015, betw Valley and William Kitchen, Executive Vice President and Chief Credit Officer, un his qualifying termination following a change in control, Mr. Kitchen will be entited to receive an amount equal to: (i) his annual then-current annual base salary; (ii) yearly bonus, auto allowance, and gym dues; and (iii) continued insurance benefits one year following the change in control. The benefits paid under (i) and (ii) will paid in 12 equal installments.

Deferred Compensation and Split Dollar Agreements.

In the merger agreement, CVB Financial has agreed to assume Valley s def compensation agreements with certain of its executive officers, including Mes Stone, Estridge and Kitchen. A qualifying termination of the executive officemployment in connection with a change in control may accelerate the vesting participant s account balance, however, the interests of Mr. Stone and Mr. Estrictheir deferred compensation accounts are already fully vested and Mr. Kitchen been offered a position of employment with CVB Financial following the merger therefore, the vesting of his account balance is not expected to accelerate as a result the merger.

CVB Financial has agreed to assume the split dollar agreements, as well as underly bank owned life insurance policies, currently in effect between Valley and cerexecutive officers, including Messrs. Stone, Estridge and Kitchen. The split do agreements provide that, upon such officer s death, the officer s beneficiary will a specified payment from the underlying life insurance proceeds.

Engagement of Allan W. Stone as a Consultant

Citizens Business Bank has agreed to engage Allan W. Stone as a consultant f period of three months, subject to and commencing as of the close of the merger. this period of engagement, Mr. Stone will receive aggregate compensation of \$30, Under the terms of this consulting agreement, Mr. Stone has agreed to assist

advise Citizens Business Bank in connection with (i) organizing and serving co-chair of a regional advisory board with an executive of Citizens Business Ba (ii) making customer retention calls, (iii) making customer introductions introductory calls, (iv) assisting with employee retention, (v) attending region banking events and receptions, (vi) working with executives of Citizens Business Bank on marketing strategies, relationship expansion and cross-sell opportunities, promoting Citizens Business Bank is business, products and services and (viii) matters relating to the foregoing or as mutually agreed by Citizens Business Bank Mr. Stone from time to time.

Even though Mr. Stone s services to the combined banks will continue following merger pursuant to the consulting agreement, he will still be deemed to have be terminated from Valley due to a change in control and, accordingly, he will still entitled to receive the severance payments under his change in control agreement valley, dated December 28, 2015.

-50-

Offer of Employment of to Bill Kitchen

Citizens Business Bank has offered Bill Kitchen, Valley s Executive Vice Pres and Chief Credit Officer, employment following completion of the merger non-executive position.

Summary of Payments to Certain Executive Officers

The following table summarizes certain payments to be received by the execu officers of Valley as a result of the consummation of the transactions under the me agreement. The amounts are calculated as of September 22, 2016, the day of, immediately prior to, the first announcement of the merger and not reflect reducti if any, that may result if the amounts exceed the threshold set forth in section 2800 the Internal Revenue Code in accordance with the executive semployment agree These estimated amounts are based on multiple assumptions that may or may actually occur, including assumptions described in this proxy statement/prospect Some of these assumptions are based on information not currently available and, result, the actual amounts, if any, to be received by an executive officer may different the amounts set forth below.

		Perquisites/				
Name	Cash ⁽¹⁾	Benefits(2)	Total			
Allan W. Stone	\$ 706,542	\$ 138,700	\$845,242			
Roy Estridge	\$ 555,084	\$ 117,000	\$672,084			
William Kitchen	\$ 222,809	\$ 0	\$ 222,809			

- (1) Represents cash amount payable under change in control agreements with res to termination of employee following a change in control.
- (2) Represents estimated cash amount payable on behalf of employee with respect to insurance premium payments under such employee s employment agreement in connection with a change of control.

Summary of Payments of Merger-Related Compensation

The cash amounts to be paid to the executive officers described in the table footnotes above will be paid in a series of monthly payments following the merge specified in the executive s change in control agreement. These amounts will need deducted from Valley s tangible common equity for purposes of determining amount of the special dividend or whether Valley has satisfied the tangible common equity measure that is a condition to CVB Financial s obligation to complete merger to the extent that such payments and any other change in control payment not exceed \$1,800,000 in the aggregate.

Indemnification

Pursuant to the terms of the merger agreement, CVB Financial has agreed indemnify and hold harmless each of Valley's present and former directors and of against any costs or expenses (including reasonable attorneys' fees), judgments, losses, claims, damages or liabilities incurred in connection with any claim, act suit, proceeding or investigation, whether civil, criminal, administrative investigative, arising out of actions or omissions occurring at or prior to the effect time of the merger, including all transactions contemplated by the merger agreement. The merger agreement also provides that either Valley or CVB Financial purchase tail coverage for a period of six years in order to continue providing I insurance, including directors and officers liability insurance, to the office directors of Valley.

Regulatory Approvals Required for the Merger

Completion of the merger is subject to the receipt of all approvals required complete the transactions contemplated by the merger agreement, including merger and the bank merger, from the Federal Reserve, FDIC and the Department Business Oversight and the expiration of any applicable statutory waiting periods,

-51-

in each case subject to the condition that none of the approvals shall contain materially burdensome condition. The merger agreement defines a materially burdensome condition as any condition that would be reasonably likely, following merger, to have a material adverse effect on CVB Financial, Citizens Business Bor Valley (measured on a scale relative to Valley) or to materially restrict or important burden on CVB Financial in connection with the transactions contemple by the merger agreement or with respect to the business or operations of CFinancial. CVB Financial and Valley have agreed to take all actions that necessary, proper and advisable in connection with obtaining all regulatory approand have agreed to fully cooperate with the other in the preparation and filing of regulatory applications and other documents necessary to complete the transaction contemplated by the merger. CVB Financial and Valley have filed applications notifications to obtain these regulatory approvals.

Although the parties currently believe they should be able to obtain all requiregulatory approvals or waivers in a timely manner, they cannot be certain when they will obtain them or, if obtained, whether they will contain any materiburdensome condition to the merger.

FDIC Application Under the Bank Merger Act

Prior approval of the bank merger is required pursuant to Section 18(c) of the Fed Deposit Insurance Act, which we refer to as the Bank Merger Act. Because C Business Bank is a state-chartered bank that is not a member of the Federal Rese System, Citizens Business Bank is required to file its Bank Merger Act applicate with the FDIC. In evaluating an application filed under the Bank Merger Act, FDIC takes into consideration, among other things: (i) the competitive impact of proposed transactions, (ii) financial and managerial resources and future prospect the banks that are party to the merger, (iii) the convenience and needs of communities served by the banks and their compliance with the Communities served by the banks and their compliance with the Community Reinvestment Act, (iv) the banks and effectiveness in combating money-laund activities, and (v) the extent to which the proposed transactions would result in great or more concentrated risks to the stability of the U.S. banking or financial system connection with its review under the Bank Merger Act, the FDIC provides opportunity for public comment on the application for the merger and is authorized hold a public meeting or other proceeding if it determines that would be appropriated.

Transactions approved by the FDIC under the Bank Merger Act generally may not completed until 30 days after the approval of the FDIC is received, during which the Department of Justice may challenge the transaction on antitrust grounds. With approval of the FDIC and the concurrence of the Department of Justice, the wait period may be reduced to no less than 15 days. The commencement of an antitaction would stay the effectiveness of such an approval unless a court specific orders otherwise. In reviewing the merger, the Department of Justice could analy the merger is effect on competition differently than does the FDIC, and, therefore possible that the Department of Justice could reach a different conclusion than FDIC does regarding the merger is effects on competition. A determination be Department of Justice not to object to the merger may not prevent the filing

antitrust actions by private persons or state attorneys general.

California Department of Business Oversight Application

Because Citizens Business Bank and Valley Business Bank are Califor state-chartered banks, the prior approval of the Department of Business Oversight be required under the California Financial Code to merge Valley Business Bank vand into Citizens Business Bank.

In reviewing the merger of Valley Business Bank with and into Citizens Busin Bank, the Department of Business Oversight will take into consideration, among of things: (i) the competitive impact of the merger, (ii) the adequacy of the survividepository corporation is shareholders equity and financial condition, (iii) whet directors and executive officers of the surviving depository institution will satisfactory, (iv) whether the surviving depository corporation will afford reasons promise of successful operation and whether it is reasonable to believe that surviving depository corporation will be operated in a safe and sound manner and

-52-

in compliance with all applicable laws, and (v) whether the merger is fair, just equitable to the disappearing depository corporation and the surviving deposit corporation.

Federal Reserve Approval under the Bank Holding Company Act

CVB Financial is a bank holding company under Section 3 of the BHC Act. Sec 3(a) of the BHC Act generally requires the prior approval of the Federal Reserve any bank holding company to merge with any other bank holding company of acquire direct or indirect ownership or control over more than five percent of voting shares of a bank. CVB Financial expects to file a notice with the Fed Reserve Bank of San Francisco requesting confirmation that the Federal Reserve not require an application under Section 3 of the BHC Act because the merget between CVB Financial and Valley is part of a transaction that involves the merget their subsidiary banks, which is the subject of a separate application under the B Merger Act and CVB Financial will not operate Valley Business Bank but instituted by Sank will merge into Citizens Business Bank immediately after merger.

Applications Relating to the Special Dividend

With respect to the special dividend, Valley Business Bank is required to obtain paper approval from the Commissioner of Business Oversight of the State of Califo under the California Financial Code and the FDIC under the FDIC is regulations to Valley Business Bank can pay the amount of the special dividend to Valley as amount of the special dividend is in excess of the amount that may be paid un California law without prior approval. On November 3, 2016, Valley Business Billed applications with the Department of Business Oversight and FDIC to allow make a dividend payment to Valley and on November 3, 2016, Valley informed Federal Reserve of its intent to pay the special dividend to its shareholders.

Additional Regulatory Approvals, Notices and Filings

Additional notifications, filings and/or applications may be submitted to various of federal and state regulatory authorities and self-regulatory organizations in connect with the merger.

Although CVB Financial and Valley expect to obtain the required regular approvals, there can be no assurances as to whether, or when, these regular approvals will be obtained, the terms and conditions on which the approvals will granted or whether there will be litigation challenging such approvals. There likewise be no assurances that U.S. or state regulatory authorities or other parties not attempt to challenge the merger on antitrust grounds, on the basis of Community Reinvestment Act or for other reasons or, if any such challenge is mas to the result of the challenge.

Accounting Treatment

In accordance with current accounting guidance, the merger will be accounted using the acquisition method. The result of this is that (i) the recorded assets liabilities of CVB Financial will be carried forward at their recorded amou (ii) CVB Financial s historical operating results will be unchanged for the prior pe being reported and (iii) the assets and liabilities of Valley will be adjusted to fair v at the date of the merger. In addition, all identified intangibles will be recorded at value and included as part of the net assets acquired. The amount by which purchase price, consisting of the value of shares of CVB Financial common stoc be issued to former Valley shareholders, the cash consideration, and the cash to paid in lieu of fractional shares and to former option holders, exceeds the fair value the net assets, including identifiable intangibles of Valley at the merger date, will reported as goodwill of CVB Financial. In accordance with current accoun guidance, goodwill is not amortized and will be evaluated for impairment annual Identified intangibles will be amortized over their estimated lives. Further, acquisition method of accounting results in the operating results of Valley be included in the operating results of CVB Financial beginning from the dat completion of the merger.

-53-

Public Trading Markets

CVB Financial common stock is listed on the NASDAQ Global Select Market un the symbol CVBF. Valley common stock is quoted on the OTC Pink market un symbol VCBP. Upon completion of the merger, Valley common stock will trading on the OTC Pink market. CVB Financial common stock issuable in the me will be listed on the NASDAQ Global Select Market.

Exchange of Shares in the Merger

CVB Financial will appoint Computershare Corporate Services as the exchange as to handle the exchange of shares of Valley common stock for shares of CVB Financommon stock and cash. As soon as reasonably practicable after the effective time the merger, the exchange agent will send to each holder of record of Valley common stock at the effective time of the merger who holds shares of Valley common stock at the effective time of the merger who holds shares of Valley common stock certificates for the per share merger consideration the holder is entitle receive under the merger agreement. Upon surrender of stock certificates or be entry shares for cancellation, along with the executed letter of transmittal and of documents described in the instructions, a Valley shareholder will receive the cash any whole shares of CVB Financial common stock such holder is entitled to receive under the merger agreement and cash in lieu of any fractional shares of CVB Financial common stock such holder is entitled to receive. After the effective time, neivalley nor CVB Financial will register any transfers of shares of Valley commotock.

Table of Contents 125

-54-

THE MERGER AGREEMENT

The following is a summary of the material terms and conditions of the men agreement. This summary does not purport to be complete and may not contain a the information about the merger agreement that is important to you. This summary qualified in its entirety by reference to the merger agreement, a copy of which attached as Appendix A to, and incorporated by reference into, this prostatement/prospectus. The rights and obligations of the parties are governed by express terms and conditions of the merger agreement and not by this summary or other information contained in this proxy statement/prospectus. You are urged to the merger agreement carefully and in its entirety before making any decision regarding the merger.

Effects of the Merger

The merger agreement provides for the merger of Valley with and into CVB Financial with CVB Financial as the surviving corporation in the merger. The merger agreement provides that the articles of incorporation and the bylaws of CVB Financial at effect immediately prior to the merger will be the articles of incorporation and bylof the surviving corporation. The merger agreement further provides that, immediate after the merger of Valley and CVB Financial, Valley Business Bank will merge and into Citizens Business Bank, with Citizens Business Bank as the surviving bank

As a result of the merger, there will no longer be any shares of Valley common stauthorized, issued or outstanding. Valley shareholders will only participate in C Financial s future earnings and potential growth through their ownership of Financial common stock. All of the other incidents of direct ownership of Va common stock, such as the right to vote on certain corporate decisions, to e directors and to receive dividends and distributions from Valley, will be extinguis at the effective time of the merger. All of the property, rights, privileges and power CVB Financial and Valley will vest in the surviving corporation, and all claim obligations, liabilities, debts and duties of CVB Financial and Valley will become claims, obligations, liabilities, debts and duties of the surviving corporation.

Effective Time of the Merger

The merger agreement provides that the merger will be consummated on the last of the month in which the last of the closing conditions, including the receipt o regulatory and shareholder approvals and after the expiration of all regulatory was periods, are satisfied or waived, unless another date is selected by mutual agreement of CVB Financial and Valley. The merger will be consummated legally at the time agreement of merger between Valley and CVB Financial, the form of which included as Exhibit B to the merger agreement, has been duly filed with the Secret of State of the State of California. As of the date of this proxy statement/prospect the parties expect that the merger will be effective in the first quarter of 20 However, there can be no assurance as to when or whether the merger will occur.

If the merger is not completed by the close of business on February 27, 2017, merger agreement may be terminated by either Valley or CVB Financial, unless failure of the closing to occur by such date is due to the failure of the party seekin terminate the merger agreement to perform or observe the covenants and agreem of such party set forth in the merger agreement.

For a description of the transaction structure and merger consideration, please section entitled THE MERGER Terms of the Merger beginning on page 36.

Covenants and Agreements

Conduct of Businesses Prior to the Completion of the Merger

Valley has agreed that, prior to the effective time of the merger, it will gener conduct its business in the ordinary course of business consistent with past practi In addition, Valley has agreed to use commercially

-55-

reasonable efforts to maintain and preserve intact its business organization advantageous business relationships and goodwill with governmental entit customers, suppliers, distributors, creditors, lessors, officers and employees business associates; keep available the services of its employees and agents; perf its material obligations under its material contracts; maintain in full force and ef insurance comparable in amount and scope of coverage to that maintained by it as time the merger agreement was signed; maintain its allowance for loan and lo losses in accordance with past practices and methodologies and generally accept accounting principles (except as a result of changes in generally accepted accoun principles or as directed by a governmental entity); charge off all loans and o assets deemed uncollectible or classified as loss in accordance with generally ac accounting principles and applicable law or as directed by a governmental entity); maintain loan classification policies and procedures in accordance with industry practices consistent with past practice. Valley has also agreed not to take any ac that is intended to or would reasonably be expected to adversely affect or material delay any regulatory approvals required for the transactions contemplated in merger agreement or its performance of its covenants and agreements in the me agreement or its consummation of the merger or other transactions contemplated the merger agreement.

In addition to the general covenants above, Valley has agreed that prior to effective time of the merger, subject to specified exceptions, it will not, and will permit any of its subsidiaries to, without the written consent of CVB Financial:

issue or sell additional shares of its capital stock, or securities convertible exchangeable into, or exercisable for, any shares of its capital stock, excupon exercise of stock options outstanding on the date of the metagreement;

make, declare, pay or set aside for payment any dividend or other distribution its capital stock, other than the special dividend and quarterly of dividends consistent with Valley s past practices with respect to timin amount;

adjust, split, combine, redeem, reclassify, purchase or otherwise acquire any shares of its stock or other securities;

amend or modify the material terms of; waive, release or assign any rigunder; terminate, renew or allow to renew automatically; make any payr not then required under; knowingly violate the terms of; or enter into mate contract, lease or regulatory agreement, any restriction on its ability to conits business as it is conducted at the time the merger agreement was entered or any contract governing the terms of Valley common stock, related right.

or any outstanding debt instrument;

sell, transfer, mortgage, lease, encumber or otherwise dispose of any o assets, deposits, business or properties other than in the ordinary cours business and in a transaction that, together with other such transactions, is material to Valley taken as a whole;

acquire (other than by way of foreclosures, in satisfaction of a debt or fiduciary capacity) all or any portion of the assets, business, deposit properties of any other entity, except in the ordinary course of business and transaction that, together with other such transactions, is not material t taken as a whole, and would not reasonably be expected to present a materisk that the completion of the merger will be delayed or the requiregulatory approvals will be more difficult to obtain;

amend Valley s articles of incorporation or bylaws or those of its subsidiar

except as and when required under applicable law or an employee benefit princrease the salary, wages or benefits of any director, officer or employeexcept for ordinary-course, merit-based increases in the base salary employees consistent with past practice, but not more than 2.5% of employee s current base salary; grant, pay or agree to pay any bonus or incentive compensation, except for such bonuses for the year ended Decen 31, 2016 pursuant to Valley s existing 2016 bonus plan in an aggregate amount to exceed 110% of the aggregate amount paid under Valley 2015 bonus plan or pursuant to any retention plan adopted with the concurrence of C Financial;

-56-

adopt any employee benefit plan; grant any new equity award; terminate employment or services of any officer or employee who is party to a chang control agreement other than for cause; enter into any collective bargainin other agreement with a labor organization; or forgive or issue any loans to director, officer or employee;

hire any officer, employee or other service provider, except for a non-execu officer in the ordinary course of business and at an annual base salary no exceed that of the person being replaced;

take, or omit to take, any action that would prevent or impede, or coreasonably be expected to prevent or impede, the merger from qualifying reorganization within the meaning of Section 368(a) of the Code;

incur or guarantee any indebtedness for borrowed money, other than in ordinary course of business, other than advances from the Federal Home I Bank of San Francisco or the Federal Reserve Bank of San Francisco witterm of not more than one week;

enter into any new line of business or make any material change in any b policies and practices with respect to the operation of its business;

make any investment either by contributions to capital, property transfers or purchase of any property or assets of any person or sell or dispose of any such investment or purchase, acquire, sell or dispose of any securities of any type;

settle any action, suit, claim, proceeding, order or investigation consideration not in excess of \$50,000 individually or in excess of \$100, when combined with other settlements;

materially change its interest rates or fees for depository accounts, except determined in good faith to be necessary or advisable based on change market conditions and consistent with its policies;

change its interest rate policy or other risk management policies, procedure practices or fail to follow such policies;

grant or commit to grant any extension of credit, if such extension of credit together with all other credit then outstanding to the same borrower and affiliates, would exceed \$1,000,000 or any extension of credit with a zer below-market rates, other than those specified loans in progress on the dat merger agreement;

renew any extension of credit that would exceed \$1,000,000, except on te which are substantially similar to those of the existing extension of credit which are more favorable to Valley;

sell any real estate, charge off any assets, compromise on any debt or release we exceed \$50,000 in the aggregate, provided that Valley may sell a specified to a party that is not affiliated with the borrower or Valley for \$1,000,00 more;

purchase any loan or loan participation;

securitize any loan or create any special purpose funding or variable interest entity;

invest in any mortgage-backed or mortgage-related securities that would considered high-risk securities or enter into a derivatives transaction;

accept, renew or roll over any brokered deposits or funds obtained through Certificate of Deposit Account Registry Service or similar listing service open any new deposit account that pays an effective yield that exceeds prevailing effective yields provided by the Federal Reserve;

apply for the opening, relocation or closing of any branch office;

make any capital expenditures other than capital expenditures in the ordinand usual course of business consistent with past practice in amounts exceeding \$50,000 individually or \$100,000 in the aggregate;

-57-

pay, loan or advance any amount to, or sell, transfer or lease any proper rights or assets (real, personal or mixed, tangible or intangible) to, or enter any arrangement or agreement with, any of its officers or directors or an their family members, or any affiliates or associates (as defined under Exchange Act) of any of its officers or directors, other than loans originate the ordinary course of business;

make or commit to make any loan or amend the terms of any outstanding to any directors, officers and principal shareholders of Valley or waive rights with respect to any such loan;

change its tax or accounting policies and procedures;

change its fiscal year for tax or accounting purposes;

other than as required by generally accepted accounting principles or governmental entity, reduce any material accrual or reserve, including allowance for loan and lease losses or change the methodology with respecany such reserves;

make any material change in any basic policies and practices with respect loans, deposits and services, liquidity management and cash flow plann marketing, deposit origination, lending, reserves for loan or lease los budgeting, profit and tax planning, personnel practices or any other materials aspect of its business or operations;

grant any power of attorney or similar authority;

take title to any real property without conducting prior thereto environmental investigation, which investigation shall disclose the absence any suspected environmental contamination;

acquire direct or indirect control over any entity, whether by stock purch merger, consolidation or otherwise or make any other investment eithe purchase of securities, contributions to capital, property transfers or purchas any property or assets of any other Person, except, in either instance connection with a foreclosure of collateral or conveyance of such collateral lieu of foreclosure taken in connection with collection of a loan in the ordin course of business consistent with past practice and with respect to loans in to third parties who are not affiliates of Valley;

make or change any material tax elections; change or consent to any change its method of accounting for tax purposes, except as required by applicable law; settle or compromise any material tax liability, claim or assessment; e into any closing agreement, waive or extend any statute of limitations verspect to a material amount of taxes; surrender any right to claim a refund taxes; or file any material amended tax return;

take any action, or omit to take any action, that is intended to or we reasonably be likely to result in any of its representations and warranties in merger agreement becoming untrue in any respect, any of the conditions to merger not being satisfied or delayed, or a violation or breach of any provisof the merger agreement; or

agree to take, or make any commitment to take, any of the foregoing prohibactions.

In addition to the general covenants above, CVB Financial has agreed that, prior to effective time of the merger, subject to specified exceptions, it will not, and will permit any of its subsidiaries to, without the written consent of Valley:

take or omit to take any action that would reasonably be expected to prevent, materially impede or materially delay the merger;

willfully take, or willfully omit to take, any action that is reasonably likely to result in any of the conditions set forth in the merger agreement (other than conditions to CVB Financial s and Citizens Business Bank s obligations) not being satisfied;

willfully take, or willfully omit to take, any action that would preven impede the merger from qualifying as a reorganization within the measurement (Section 368(a) of the Code; and

-58-

except as would not be reasonably expected to have a material adverse effect with respect to CVB Financial and Citizens Business Bank, willfully fail to file all reports with the SEC and any other filings required to be filed with any applicable governmental entity.

Regulatory Matters

CVB Financial and Valley have agreed to promptly prepare this pr statement/prospectus, and CVB Financial has agreed to promptly file with the SE registration statement on Form S-4, of which this proxy statement/prospectus is a p CVB Financial and Valley have each agreed to use their commercially reason efforts to have the Form S-4 declared effective under the Securities Act of 1933 amended, as promptly as practicable after such filing. CVB Financial and Valley B agreed to furnish all information concerning themselves, their affiliates and holders of their capital stock to the other and provide such other assistance cooperation as may be reasonably requested in connection with the preparation, finand distribution of the Form S-4 and this proxy statement/prospectus.

CVB Financial and Valley have agreed to cooperate with each other and use t commercially reasonable best efforts to promptly prepare and file all necess documentation, to effect all applications, notices, petitions and filings and to obtain promptly as practicable all permits, consents, waivers, approvals and authorization all third parties and any governmental entity that are necessary or advisable consummate the transactions contemplated by the merger agreement.

Shareholder Approval

The Valley board of directors has unanimously resolved to recommend to Va shareholders that they adopt and approve the merger agreement (subject to cer exceptions, following the receipt of a superior acquisition proposal (as defibelow)), to submit the merger agreement to Valley shareholders for approval (unthe merger agreement has been terminated in accordance with its terms) and to incits recommendation that Valley shareholders adopt and approve the merger agreement use its reasonable best efforts to obtain from its shareholders a vote adopting merger agreement (subject to certain exceptions, following the receipt of a superacquisition proposal).

NASDAQ Listing

CVB Financial has agreed to use its reasonable commercial efforts to cause the sh of CVB Financial common stock to be issued in the merger to be approved for list on the NASDAQ Global Select Market, subject to official notice of issuance, prior at the effective time of the merger.

Employee Matters

Employees of Valley who become employees of Citizens Business Bank, referred as continuing employees, will be eligible to participate in the employee benefits of Citizens Business Bank on the same terms as such plans and benefits are generated.

offered to employees of Citizens Business Bank in comparable positions. For purport of determining covered employees eligibility and vesting (but not for benefit accounder the employee benefit plans of Citizens Business Bank, for severance benefit and for vacation entitlement (to the extent permitted by applicable law), Citiz Business Bank will recognize such employees—years of service with the Valley s existing benefit plans will be terminated at the time of the merger, exce Valley—s deferred compensation plans. After the effective time of the merger, shall cause Citizens Business Bank to provide a severance benefit to each person was an employee of Valley immediately before the effective time (except for employee who is a party to any agreement providing severance) and whe employment is terminated involuntarily, other than for—Cause—(as defined below Citizens Business Bank within 12 months after the effective time of the merger, severance benefit will consist of a lump-sum severance payment equal to semployee—s regularly scheduled base salary or base wages at the time of

-59-

termination of employment, for a number of weeks equal to two weeks plus one were for each full year of continuous service completed by such employee with Valley Citizens Business Bank at the time of termination of employment, subject applicable tax withholding and subject further to the employee signing a release claims in favor of CVB Financial and Citizens Business Bank and allowing become effective. Cause means the employee spersonal dishonesty, misconduct, breach of fiduciary duty involving personal profit, failure to comply any valid and legal directive of CVB Financial or Citizens Business Bank, failure perform stated duties or violation of any law, rule, regulation (other than traviolations or similar offenses) or order of any governmental entity.

The merger agreement permits Valley to establish a retention program to comperation Valley employees who remain with Valley through the effective time of merger and/or with Citizens Business Bank through the completion of key syst conversions following the merger. The payment will be conditioned upon semployee not resigning prior to the conversion date.

Subject to the requirements of applicable law, Citizens Business Bank has agree take commercially reasonable actions as are necessary to cause the group health maintained by Citizens Business Bank and applicable insurance carriers, third-p administrators and any other third parties, to the extent such group health plan is n available to employees of Valley at the time of the merger, to waive any evidence insurability requirements, waiting periods and any limitations as to preexis medical conditions under the group health plan applicable to employees of Valle the time of the merger and their spouses and eligible dependents (but only to extent that such preexisting condition limitations did not apply or were satisfied up the group health plan maintained by Valley prior to the closing) and to prov employees of Valley at the time of the merger with credit, for the calendar year which the closing occurs, for the amount of any out-of-pocket expenses copayments or deductible expenses that are incurred by them during the calendar in which the merger occurs under a group health plan maintained by Citizens Busi Bank or any of its affiliates (unless such employees of Valley were, when employees by Valley, participants in a Health Maintenance Organization health plan).

Prior to the effective time of the merger, Valley shall terminate its 401(k) plan and other employee benefit plans that CVB Financial may identify, other than Val executive s supplemental compensation agreement, salary continuation agreement split dollar agreements, which CVB Financial will maintain following the mer Valley will provide CVB Financial with evidence of such terminations to satisfaction of CVB Financial.

The merger agreement specifies that none of its provisions confer upon any employed Valley who is employed by Citizens Business Bank after the merger any right respect to continuance of employment or other service. Nor does the mergerement interfere in any way with the right of CVB Financial to terminate employment or other association of any person at any time. The merger agreements are third-party beneficiary right in any person. The terms of the mergerement do not constitute an amendment of, or interfere in any way with the right

CVB Financial and its subsidiaries to amend, terminate or otherwise discontinue, or all CVB Financial employee plans and any other plans, practices or policie CVB Financial in effect from time to time.

Indemnification and Directors and Officers Insurance

From and after the effective time of the merger, CVB Financial has agree indemnify and hold harmless each person who, prior to the effective time, is or w director or officer of Valley against any costs or expenses (including reasons attorneys fees), judgments, fines, losses, claims, damages or liabilities incurred connection with any claim, action, suit, proceeding or investigation, whether c criminal, administrative or investigative, arising out of actions or omissions occur at or prior to the effective time of the merger, including the transactions contemple by the merger agreement. CVB Financial shall also advance expenses as

-60-

incurred to the fullest extent permitted under applicable law, provided that indemnified person to whom expenses are advanced provides an undertaking to resuch advances if it is ultimately determined that he or she is not entitle indemnification.

The merger agreement provides that either Valley or CVB Financial shall purel director s and officer s liability insurance tail coverage that serves to reimbur present and former officers and directors of Valley with respect to claims against so directors and officers arising from facts or events occurring before the medincluding the transactions contemplated by the merger agreement). The price of sinsurance coverage shall not exceed 250% of the annual cost of Valley's cudirector s and officer s liability insurance policy. The insurance coverage shall term of six years following the effective time of the merger or, if less, the longest that may be obtained for an amount equal to 250% of the annual cost of Valleurrent director s and officer s liability insurance policy. Such insurance shall content the same coverage and amounts, and contain terms and conditions no advantageous to the parties entitled to indemnity under the merger agreement, as coverage currently provided by Valley's existing insurance policies.

Special Dividend

The merger agreement provides that Valley shall declare the special dividend pays to holders of Valley common stock as of the closing date of the merger. The aggree amount of the special dividend shall not exceed the amount by which Valley is tarked to specified loan in accordance with the merger agreement, \$37,500,000 (2) the amount of tangible common equity necessary for Valley to achieve a tanged common equity capital ratio of 8.0%. For purposes of determining the amount of special dividend, Valley is tangible common equity excludes the impact of Valley at Common equity excludes the impact of Valley ex

The merger agreement requires Valley to seek, and to cause Valley Business Banseek, any regulatory approvals necessary to permit them to pay the special divide without violation of applicable law. Valley is required to deposit the aggregate among the special dividend with its stock transfer agent at least three business days prior the closing date of the merger.

No Solicitation

The merger agreement provides that none of Valley nor any of its officers, direct and employees shall, and Valley will cause its officers, directors, age representatives, advisors and affiliates not to, initiate, solicit, encourage or knowing facilitate any inquiries or the making of proposals with respect to, or engage in negotiations concerning, or provide any confidential or nonpublic information or to, or have any discussions with, any person relating to, any alternative acquisitions.

proposal (as defined below) or otherwise facilitate any effort to attempt or mak implement an alternative acquisition proposal. However, if at any time after the of the merger agreement and prior to, but not after, obtaining the approval of merger agreement by Valley shareholders, Valley receives an unsolicited bona alternative acquisition proposal and the board of directors of Valley concludes in g faith that such alternative acquisition proposal constitutes, or is reasonably expecte result in, a superior acquisition proposal (as defined below), then Valley and its be of directors may, and Valley may permit representatives to, furnish or cause to furnished nonpublic information and participate in such negotiations or discussion the extent that the board of directors of Valley concludes in good faith (and base the advice of its legal counsel and its financial advisers) that its failure to take s actions would breach or would be more likely than not to breach its fiduciary dutie shareholders under applicable law; provided that prior to providing any s nonpublic information or engaging in any such negotiations, Valley must have ent into a confidentiality agreement with such third party on terms no less favorable Valley than the confidentiality agreement between Valley and CVB Financial, which expressly permits Valley to comply

-61-

with its obligations pursuant to the merger agreement. Valley will immediately contained to the merger agreement. and cause to be terminated any activities, discussions or negotiations conducted be the date of the merger agreement with any persons other than CVB Financial respect to any alternative acquisition proposal and will use its commerci reasonable efforts, to (i) enforce any confidentiality or similar agreement relating t alternative acquisition proposal and (ii) within 10 business days after the date of merger agreement, to request and confirm the return or destruction of any confider information provided to any person (other than CVB Financial and its affilia pursuant to any such confidentiality or similar agreement. Valley must promptly in any event within two business days) advise CVB Financial following receipt of alternative acquisition proposal, any discussions or negotiations that are sought t initiated or continued or any request for nonpublic information or inquiry that we reasonably be expected to lead to any alternative acquisition proposal and substance thereof (including the identity of the person making such alterna acquisition proposal) and keep CVB Financial promptly apprised of any rela developments, discussions and negotiations (including the terms and conditions of such request, inquiry or alternative acquisition proposal, or all amendment proposed amendments thereto) on a current basis (it being understood that for avoidance of doubt that no such communications to CVB Financial will be deeme adverse change of recommendation, as defined below). Valley agrees that it contemporaneously provide to CVB Financial any confidential or nonpu information concerning Valley that it may provide to any other person in connec with any alternative acquisition proposal.

An alternative acquisition proposal, which the merger agreement refers company acquisition proposal, is a tender or exchange offer, proposal for a reconsolidation or other business combination involving Valley or any proposal or to acquire in any manner more than 10% of the voting power in, or more than 10% the fair market value of the business, assets or deposits of, Valley or any purannouncement of a proposed plan or intention to do any of the foregoing or agreements to engage in any of the foregoing, other than the transactions contemple by the merger agreement and any sale of whole loans and securitizations in ordinary course.

A superior acquisition proposal, which the merger agreement refers to as a superior proposal, is an unsolicited bona fide written alternative acquisition prothat the board of directors of Valley concludes in good faith to be more favorable for a financial point of view to its shareholders than the merger and the other transact contemplated by the merger agreement and to be reasonably capable of beconsummated on the terms proposed, (i) after receiving the advice of its financial visors (who shall be Vining Sparks or another nationally recognized investments banking firm), (ii) after taking into account the likelihood of consummation of stransaction on the terms set forth therein and (iii) after taking into account all lefinancial (including the financing terms of any such proposal), regulatory and of aspects of such proposal (including any expense reimbursement provisions conditions to closing) and any other relevant factors permitted under applicable and after taking into account any amendment or modification to the merger agreed to by CVB Financial; provided that for purposes of the definition of superior visions of the definition of superior visions are considered to by CVB Financial; provided that for purposes of the definition of superior visions conditions to closing) and any other relevant for purposes of the definition of superior visions conditions to be considered to by CVB Financial; provided that for purposes of the definition of superior visions conditions to closing) and any other relevant factors permitted under applicable and after taking into account any amendment or modification to the merger agreed to by CVB Financial; provided that for purposes of the definition of superior visions conditions to closing the vision vis

acquisition proposal, the references to more than 10% in the definition of alt acquisition proposal will be deemed to be references to 50%.

None of the members of the board of directors of Valley may, except as expre permitted by the merger agreement, withdraw or materially and adversely modify recommendation that Valley shareholders vote to approve the merger agreement recommend to Valley shareholders an alternative acquisition proposal other than merger, which we refer to as an adverse change of recommendation, or calcommit Valley to enter into any agreement or understanding other than confidentiality agreement referred to above relating to any alternative acquisition proposal made to Valley. Nevertheless, in the event that Valley receives an alternative acquisition proposal that the Valley board of directors concludes in good from constitutes a superior acquisition proposal, the board of directors of Valley may man adverse change of recommendation or terminate the merger agreement as lon Valley gives CVB Financial prior written notice at least five business days betaking such action and during such five-business-day period Valley negotiates in graith with CVB Financial

-62-

to enable CVB Financial to make an improved offer that is at least as favorable to shareholders of Valley as such alternative acquisition proposal.

Under certain circumstances Valley must pay CVB Financial a termination fe \$3,500,000 if it terminates the merger agreement following its receipt of an alterna acquisition proposal or a superior acquisition proposal. For more information, plesee the section entitled THE MERGER AGREEMENT Termination; Termination beginning on page 67.

Representations and Warranties

The merger agreement contains representations and warranties made by Valle CVB Financial relating to a number of matters, including the following:

corporate organization, qualification to do business and corporate power; capitalization; requisite corporate authority to enter into the merger agreement an complete the contemplated transactions; absence of conflicts with governing documents, applicable laws or cer agreements as a result of entering into the merger agreement or completing merger; timely filing of reports; conformity with generally accepted accounting principles of Valley s fina statements; no broker s fees payable in connection with the merger; absence of certain changes or events since December 31, 2015; absence of undisclosed liabilities; compliance with applicable law;

regulatory matters;
inapplicability of takeover laws;
employee benefits matters;
government authorizations;
fairness opinion from financial advisor;
accuracy of Valley information provided in this proxy statement/prospectus
legal proceedings;
material contracts;
environmental matters;
taxes and tax returns;
intellectual property;
properties and assets;
insurance;
accounting and internal controls;
no derivatives;

-63-

loan matters; investment securities; affiliate transactions; operating losses; labor and employment matters; trust activities; and information technology, security and privacy matters. The merger agreement also contains representations and warranties made by CV. Valley relating to a number of matters, including the following: corporate organization, qualification to do business, corporate power subsidiaries; capitalization; requisite corporate authority to enter into the merger agreement an complete the contemplated transactions; absence of conflicts with governing documents, applicable laws or cer agreements as a result of entering into the merger agreement or completing merger; required regulatory consents necessary in connection with the merger;

proper filing of documents with regulatory agencies and the SEC and

accuracy of information contained in the documents filed;

conformity with generally accepted accounting principles and S requirements of CVB Financial s financial statements filed with the SEC;

broker s fees payable in connection with the merger;

absence of certain changes or events since December 31, 2015;

compliance with applicable law;

accuracy of CVB Financial information provided in this prestatement/prospectus;

legal proceedings; and

accounting and internal controls.

Certain of these representations and warranties by Valley and CVB Financial qualified as to materiality or material adverse effect.

Material adverse effect means, with respect to any party, a material adverse effthe business, assets or deposit liabilities, properties, operations, condition (financial otherwise) or results of operations of such party and its subsidiaries taken as a where except that a material adverse effect shall not be deemed to include effects arise of, relating to or resulting from (i) changes in applicable generally accept accounting principles or regulatory accounting requirements generally affecting of companies in the banking industries in which such party and its subsidiaries oper (ii) changes in laws general applicability to companies of similar size in the banking industries in which such party and its subsidiaries operate; (iii) changes in glo national or regional political conditions or general economic or market conditional currency exchanges in prevailing interest rates, credit availability and liquid currency exchange rates, and price levels or trading volumes in the U.S. or fore securities markets) affecting other companies in the banking industries in which separty and its subsidiaries operate; (iv) changes in the credit markets, any downgrain the credit markets or adverse credit

-64-

events resulting in deterioration in the credit markets generally and including char to any previously correctly applied asset marks resulting therefrom; (v) a declin the trading price of CVB Financial s common stock that would not entitle a paterminate the merger agreement in accordance with its terms or a failure, in an itself, to meet earnings projections, but not, in either case, including any underly causes thereof; (vi) any outbreak or escalation of hostilities, or declared or undeclaracts of war or terrorism; or (vii) actions or omissions taken with the prior write consent of the other party or expressly required by the merger agreement, except effects attributable to or resulting from any of the changes, events, conditions or tredescribed in clauses (i), (ii), (iii), (iv) and (vi) shall not be excluded to the externany disproportionate impact they have on such party and its subsidiaries, taken whole, as compared to other companies in the banking industry in which such p and its subsidiaries operate. A material adverse effect also includes, with respectany party, a material adverse effect on the ability of such party to consummate transactions contemplated by the merger agreement on a timely basis.

The representations and warranties in the merger agreement do not survive effective time of the merger, and as described below under the section entitled MERGER AGREEMENT Termination; Termination Fee beginning on page 67 merger agreement is validly terminated, there will be no liability or damages arisunder the representations and warranties of CVB Financial or Valley, or otherwunder the merger agreement, unless CVB Financial or Valley willfully breached merger agreement.

Conditions to the Merger

Conditions to Each Party s Obligations

The respective obligations of each of CVB Financial and Valley to complete merger are subject to the satisfaction of the following conditions:

receipt of the requisite approval of the merger agreement by Valley s shareholders;

the effectiveness of the registration statement on Form S-4, of which this prestatement/prospectus is a part, and the absence of a stop order suspendeffectiveness or proceeding initiated or threatened by the SEC for that purp and

the absence of any order, injunction or decree that would prevent or millegal the completion of the merger.

Conditions to Obligations of CVB Financial

The obligation of CVB Financial to complete the merger is also subject to satisfaction or waiver by CVB Financial of the following conditions:

the accuracy of the representations and warranties of Valley as of the cloudate of the merger, other than, in most cases, those failures to be true correct that would not reasonably be expected to result in a material adverse of Valley or CVB as the surviving corporation in the merger;

performance in all material respects by Valley of the obligations required to performed by it at or prior to the closing date;

receipt by CVB Financial of an opinion of Manatt, Phelps & Phillips, LLP a certain tax matters;

receipt of all required regulatory approvals, none of which shall include materially burdensome condition, and the expiration of all statutory wai periods;

holders of not more than 10% of the outstanding shares of Valley common stock shall have exercised their dissenters rights or have the capacity to do so;

the absence of any material adverse effect with respect to Valley;

receipt by Valley of any required consents, copies of which shall be delive to CVB Financial;

-65-

receipt of resignations from each director of Valley and its subsidiaries;

receipt of voting and support agreements from each of the Valley directors its three executive officers (which, as noted, has been completed);

receipt of noncompetition and non-solicitation agreements from each of the Valley directors and its three executive officers (which, as noted, has been completed);

as of at least three (3) business days prior to the closing date, Valley shall h provided CVB Financial with satisfactory evidence that Valley has satis each of the following minimum financial conditions: (1) as of the last da the month preceding the closing date, Valley s closing tangible con shareholders equity (after giving effect to the special dividend but prior t pre-closing adjustments otherwise required by the merger agreement without the effect of up to \$3,500,000 in Valley s merger-related transa expenses), shall not be less than the greater of (a) \$37,500,000 if Valley completed the sale of a specified nonperforming loan or, \$37,000,000 if Va has not completed the sale the loan or (b) the amount necessary for Valle achieve a tangible common equity ratio of at least 8.0% as of such date; (2) allowance for loan loss ratio, determined as of the final day of the mo immediately preceding the closing date, shall not be less than 1.1%; (3) to assets shall not be less than \$410,000,000 as of the final day of the mo immediately preceding the closing date; (4) average accruing loans for 30-day period ending on the fifth business day prior to closing shall not be than \$295,000,000; and (5) average noninterest-bearing deposits for the 30period ending on the fifth business day prior to closing shall not be less \$150,000,000;

Valley shall have delivered an opinion from Crowe Horwath LLP tha agreement, contract or arrangement to which any employee of Valley is a p will result in the payment of any amount that would not be deductible reason of Section 280G of the Code:

As of at least three business days prior to the closing date, CVB shall hereceived satisfactory evidence that Valley is merger-related transaction cosmot exceed \$3,500,000 and certain categories of transaction costs do not exceed specified amounts, namely, change in control payments and benepayable to employees do not exceed \$1,800,000; contract termination feel not exceed \$1,000,000; legal and professional fees do not exceed an aggregof \$600,000; and Valley is cost of a tail policy for its directors and off liability insurance does not exceed \$70,000;

receipt of a properly executed statement from Valley that meets requirements of the Foreign Investment in Real Property Tax Act; and

Allan W. Stone, President and Chief Executive Officer of Valley, and Citiz Business Bank shall have entered into the consulting agreement (which noted, has been completed).

Conditions to Obligations of Valley

The obligation of Valley to complete the merger is also subject to the satisfactio waiver by Valley of the following conditions:

the accuracy of the representations and warranties of CVB Financial as of closing date of the merger, other than, in most cases, those failures to be and correct that would not reasonably be expected to result in a materadverse effect on CVB Financial;

performance in all material respects by CVB Financial of its obligative required to be performed by it at or prior to the closing date;

the absence of any material adverse effect with respect to CVB Financial;

receipt of all required regulatory approvals and the expiration of all statuwaiting periods; and

the shares of CVB Financial common stock included in the mer consideration shall have been authorized for listing on the NASDAQ Glo Select Market.

-66-

Termination; Termination Fee

The merger agreement may be terminated at any time prior to the effective time of merger, whether before or after approval of the merger agreement by Vashareholders:

by mutual written consent of CVB Financial and Valley;

by either CVB Financial or Valley if a requisite regulatory approval is deand such denial has become final and non-appealable, if a government enadvises CVB Financial or Valley that it will deny a requisite regular approval (or intends to revoke or rescind such an approval) in writing and sedenial becomes unappealable or if a governmental entity of compejurisdiction has issued a final, non-appealable order, injunction or depermanently enjoining or otherwise prohibiting or making illegal consummation of the transactions contemplated by the merger agreement;

by either CVB Financial or Valley if the merger is not completed on or be February 27, 2017, unless the failure of the closing to occur by such date is to the failure of the party seeking to terminate the merger agreement to perfor observe the covenants and agreements of such party set forth in the meagreement;

by either CVB Financial or Valley if there is a breach by the other party of of its covenants, agreements, representations or warranties that wo individually or in the aggregate with other breaches by such party, result in failure of a closing condition of the other party, and such breach is not convithin 30 days following written notice to the party committing the breach the breach, by its nature, cannot be cured within such time (provided that terminating party is not then in material breach of any representation, warrancovenant or other agreement contained in the merger agreement);

by either CVB Financial or Valley if Valley shareholders have not appropriate the merger agreement and the transactions contemplated thereby at the convened Valley special meeting or at any adjournment or postponent thereof, provided that the failure to obtain such shareholder approval was caused by the terminating party s material breach of any of its obligations the merger agreement;

by CVB Financial prior to obtaining Valley shareholder approval, in the exact (a) Valley breaches in any material respect the merger agreement; (b) Valley

the board of directors of Valley submits the merger agreement to Va shareholders without recommending approval or withdraws or advers modifies such recommendation or makes an adverse change recommendation; (c) at any time after the end of 15 business days follow receipt of an alternative acquisition proposal, the board of directors of Va fails to reaffirm its recommendation the shareholders vote to approve merger as promptly as practicable (but in any event within five business of after receipt of any written request to do so by CVB Financial); or (d) a ten offer or exchange offer for outstanding shares of Valley common stoce publicly disclosed (other than by CVB Financial or one of its affiliates) and board of directors of Valley recommends that its shareholders tender to shares in such tender or exchange offer or, within 10 business days after commencement of such tender or exchange offer, the board of director Valley fails to recommend against acceptance of such offer;

by Valley, prior to obtaining Valley shareholder approval, in order to enter a definitive agreement providing for a superior acquisition proposal (as defiabove) (provided that Valley is not in material breach of any of the terms of merger agreement and Valley pays CVB Financial a termination fee in adva of or concurrently with such termination, as described below);

by Valley, by written notice to CVB Financial if that the volume weight average closing price of CVB Financial common stock reported on NASDAQ Global Select Market for the ten trading days ending on the formula trading day prior to closing is less than \$13.50; or

by CVB Financial, by written notice to Valley if the volume-weighted aver closing price of CV B Financial common stock reported on the NASD Global Select Market for the ten trading days ending on the fifth trading prior to closing is greater than \$20.50.

-67-

Valley must pay CVB Financial a termination fee of \$3,500,000 in the event that:

the merger agreement is terminated by Valley in order to enter into a defining agreement providing for a superior acquisition proposal;

CVB Financial terminates the merger agreement due to no comp recommendation; or

any person has made an alternative acquisition proposal, which proposal has been publicly announced, disclosed or proposed and not withdrawn, and (a) thereafter the merger agreement is terminated (i) by either party pursuant to the termination provision for delay or pursuant to the termination provision for no approval by Valley shareholders or (ii) by CVB Financial pursuant to the termination provision for breach and (b) within 18 months after such termination of the merger agreement, an alternative acquisition proposal is consummated or any definitive agreement with respect to an alternative acquisition proposal is entered into (provided that references to 10% in the definition of alternative acquisition proposal are deemed to be references to 50%).

Effect of Termination

If the merger agreement is validly terminated, the merger agreement will become and of no effect, and none of Valley, CVB Financial, any of their respect subsidiaries or any of their officers or directors will have any liability of any na whatsoever under the merger agreement, or in connection with the transactic contemplated by the merger agreement, except that (i) the provisions of the merger agreement relating to confidentiality obligations of the parties, the termination of publicity and certain other technical provisions will continue in effect notwithstand termination of the merger agreement and (ii) neither Valley nor CVB Financial see relieved or released from any liabilities or damages arising out of its willful material breach of any provision of the merger agreement.

The purpose of the termination fee is to compensate CVB Financial for entering the merger agreement, for taking actions to consummate the transactions contemple by the merger agreement and for incurring the costs and expenses related to merger and other losses and expenses, including foregoing the pursuit of or opportunities by CVB Financial. The payment of the termination fee is C Financial s sole and exclusive remedy with respect to termination of the magreement, subject to a willful material breach by Valley or an action for fraud.

Amendments, Extensions and Waivers

The merger agreement may be amended by the parties, by action taken or author by their respective boards of directors, at any time before or after approval of mat presented in connection with the merger by the shareholders of Valley, except after any approval of the transactions contemplated by the merger agreemen Valley shareholders there may not be, without further approval of such sharehold any amendment of the merger agreement that would require further approval of Va shareholders under applicable law.

At any time prior to the effective time of the merger, the parties, by action take authorized by their respective boards of directors, may, to the extent legally allow (i) extend the time for the performance of any of the obligations or other acts of other party, (ii) waive any inaccuracies in the representations and warranties contain the merger agreement or (iii) waive compliance with any of the agreement conditions contained in the merger agreement. Any agreement on the part of a part any extension or waiver must be in writing.

Stock Market Listing

CVB Financial has agreed to apply to have the shares of CVB Financial commutators to be issued in the merger approved for listing on the NASDAQ Global Se Market, which is the principal trading market for

-68-

existing shares of CVB Financial common stock. It is a condition to CVB Financiand Valley s obligations to complete the merger that such approval is obtained, so to official notice of issuance. Following completion of the merger, Valley computed will cease trading and will no longer be quoted on the OTC Pink market.

Fees and Expenses

All fees and expenses incurred in connection with the merger, the merger agreem and the transactions contemplated by the merger agreement, including costs expenses of printing and mailing this proxy statement/prospectus, shall be paid by party incurring such fees or expenses, whether or not the merger is consummated except for the termination fee that is payable solely by Valley in the instant described above.

Voting and Support Agreements

In connection with entering into the merger agreement and as an inducement to willingness of CVB Financial to enter into the merger agreement, each Valley s directors and executive officers executed and delivered to CVB Financial voting and support agreement, which we refer to collectively as the voting and su agreements. Each director and executive officer entered into the voting and su agreement in his capacity as the record or beneficial owner of shares of Valley and in his capacity as a director or executive officer of Valley or as a trustee of any ber plan. The following summary of the voting and support agreements is subject to, qualified in its entirety by reference to, the full text of the voting and supparagreements included as Exhibit A to the merger agreement, which is attached Appendix A to this proxy statement/prospectus.

Pursuant to the voting and support agreements, each shareholder party thereto agree to vote his shares of Valley common stock, as applicable:

in favor of the merger, the merger agreement and the transactions contempl by the merger agreement;

against any action or agreement that would result in a breach in any materespect of any covenant, representation or warranty or any other obligation agreement under the merger agreement; and

except with the prior written consent of CVB or as otherwise contemplate the merger agreement, against, the following actions (other than the merger the transactions contemplated by the merger agreement): (1) any extraordin corporate transactions, such as a merger, consolidation or other busin combination; (2) any sale, lease or transfer of a material amount of Va assets; (3) any change in the majority of the board of directors of Valley

any material change in the present capitalization of Valley; (5) any amendment of Valley s articles of incorporation or bylaws; (6) any other material charmonic Valley s corporate structure or business; or (7) any other action that is into or could reasonably be extended to impede, interfere with delay, postpodiscourage or materially adversely affect the merger.

Each director and the three executive officers of Valley, in his capacity a shareholder of Valley, also agreed not to enter into any agreement or understand with any person or entity to vote or give instructions in any manner inconsistent to the above clauses.

Until the earlier of the termination of the merger agreement or the effective time, edirector agreed not to, directly or indirectly:

sell, transfer, exchange, pledge, assign, hypothecate, encumber, tende otherwise dispose of his shares of Valley common stock or enforce or pe execution of the provisions of any redemption, share purchase or s recapitalization or other agreement with respect to such shares;

enter into any contract, option or other agreement, arrangemen understanding with respect to the transfer of, directly or indirectly, any of shares or any securities convertible into or exercisable for shares, any o capital stock of Valley or any interest in any of the foregoing with any person

-69-

enter into swap or any other agreement or any transaction that transfers, in whole or in part, directly or indirectly, the economic consequence of ownership of the shares; or

take any action that would make any of his representations or warran contained in the voting and support agreement untrue or incorrect in material respect or have the effect of preventing or disabling the shareho from performing the shareholder s obligations under the voting and su agreement.

Each director and the three executive officers of Valley, in his capacity shareholder of Valley, has further agreed to not directly or indirectly initiate, sol induce or knowingly encourage, or knowingly take any action to facilitate the mal of, any inquiry, offer or proposal that constitutes, or could reasonably be expected lead to an alternative acquisition proposal, or participate in any discussion negotiations regarding any alternative acquisition proposal, or furnish, or otherwafford access, to any person (other than CVB Financial) to any information or with respect to Valley or otherwise relating to an alternative acquisition propexcept as permitted by the merger agreement.

The obligations of the shareholders will terminate upon the earlier of consummation of the merger or, if the merger is not consummated, upon termination of the merger agreement.

Noncompetition and Nonsolicitation Agreements

In order for CVB Financial and Citizens Business Bank to have the full beneficial ownership of Valley and the business it conducts, including its goodwill, following effective time of the merger, each director of Valley (including Allan W. Stone, President and Chief Executive Officer of Valley), Roy Estridge, Valley s Executive President and Chief Financial Officer, and Bill Kitchen, Valley s Executive President and Chief Credit Officer, entered into a noncompetition and nonsolicital agreement with Citizens Business Bank, which provides that he will not:

without the written consent of Citizens Business Bank, directly or indirect own, manage, operate, finance, control or participate in the owners management, operation, control or financing of, or be connected as an offidirector, employee, partner, principal, agent, representative or consultar otherwise with, any business or enterprise engaged in any business the competitive with or similar to the financial services provided by Valley with any of the California counties of Fresno, Kern, King, Madera or Tul Financial services means the origination, purchasing, selling and services

commercial, real estate, residential, construction and consumer loans and/or solicitation and provision of deposit services and services related thereto.

without the written consent of Citizens Business Bank, directly or indirectly behalf of any depository institution or its affiliates, (i) solicit or aid in solicitation of any customers or prospective customers for financial servi (ii) solicit or aid in the solicitation of any officers or employees of Valle (iii) induce or attempt to induce any customer, prospective customer, supp distributor, officer or employee of Valley to terminate such pers relationships with Citizens Business Bank; in addition, at the request of C Financial or Citizens Business Bank, each director and executive officer ag to use his best efforts to retain the business of CVB and/or Citizens Business Bank and promote the acquisition of new business by CVB Financial and Citizens Business Bank.

use or disclose Valley s trade secrets. The directors and executive officer agree to deliver all documents, reports, drawings, designs, plans, proposals other tangible evidence of trade secrets to CVB Financial and/or Citiz Business Bank. Trade secrets means the confidential customer and o information of Valley.

The obligations under the noncompetition and nonsolicitation agreements gener end on the second anniversary of the effective time of the merger.

-70-

MATERIAL UNITED STATES FEDERAL INCOME TAX CONSEQUENC OF THE MERGER

The following is a discussion of material U.S. federal income tax consequences of merger to U.S. holders (as defined below) of Valley common stock that exchange the shares of Valley common stock for shares of CVB Financial common stock and on the merger. This discussion does not address any tax consequences arising until the laws of any state, local or foreign jurisdiction, or under any U.S. federal laws of than those pertaining to the income tax, nor does it address any tax consequences arising under the unearned income Medicare contribution tax pursuant to the He Care and Education Reconciliation Act of 2010. This discussion is based upon Code, the regulations promulgated under the Code and court and administration rulings and decisions, all in effect on the date of this proxy statement/prospect These authorities may change, possibly retroactively, and any change could affect accuracy of the statements and conclusions set forth in this discussion.

This discussion addresses only those U.S. holders of Valley common stock that their shares of Valley common stock as a capital asset within the mean Section 1221 of the Code (generally, property held for investment). Importantly, discussion does not address all aspects of U.S. federal income taxation that may relevant to a particular U.S. holder in light of that U.S. holder s individual circumstances or to a U.S. holder that is subject to special treatment under the federal income tax laws, including, without limitation, a U.S. holder that is:

a financial institution;
a tax-exempt organization;
a regulated investment company;
a real estate investment trust;
an S corporation or other pass-through entity (or an investor in an S corporation or other pass-through entity);
an insurance company;
a mutual fund:

a controlled foreign corporation or passive foreign investment company;

a dealer or broker in stocks and securities, or currencies;

a trader in securities that elects to use the mark-to-market method accounting;

a holder of Valley common stock subject to the alternative minimum provisions of the Code;

a holder of Valley common stock that received Valley common stock through a tax qualified retirent plan or otherwise as compensation;

a holder of Valley common stock that has a functional currency other than the U.S. dollar;

a holder of Valley common stock that holds Valley common stock as part hedge, straddle, constructive sale, conversion or other integrated transaction

a person that is not a U.S. holder;

a U.S. holder subject to withholding under the Foreign Account Compliance Act, or FATCA; or

a U.S. expatriate or certain former citizens or long-term residents of the UnStates.

For purposes of this discussion, the term U.S. holder means a beneficial ow Valley common stock that is for U.S. federal income tax purposes (a) an individualized or resident of the United States, (b) a corporation (or any other entity treate a corporation for U.S. federal income tax purposes) organized in or

-71-

under the laws of the United States or any state thereof or the District of Colum (c) a trust if (i) a court within the United States is able to exercise primary supervisorer the administration of the trust and one or more U.S. persons have the authorit control all substantial decisions of the trust or (ii) such trust was in existence August 20, 1996 and has made a valid election to be treated as a U.S. person for federal income tax purposes or (d) an estate, the income of which is includible in g income for U.S. federal income tax purposes regardless of its source.

If an entity or an arrangement treated as a partnership for U.S. federal income purposes holds Valley common stock, the tax treatment of a partner in s partnership generally will depend on the status of the partner and the activities of partnership. We strongly recommend that any entity treated as a partnership for lederal income tax purposes that holds Valley common stock, and any partners in spartnership, consult their own tax advisors.

DETERMINING THE ACTUAL TAX CONSEQUENCES OF THE MERG TO A HOLDER OF VALLEY COMMON STOCK MAY BE COMPLEX A WILL DEPEND IN PART ON THE HOLDER S SPECIFIC SITUATION STRONGLY RECOMMEND THAT EACH HOLDER OF VALL COMMON STOCK CONSULT ITS OWN TAX ADVISOR AS TO THE T CONSEQUENCES OF THE MERGER IN ITS PARTICUL CIRCUMSTANCE, INCLUDING THE APPLICABILITY AND EFFECT THE ALTERNATIVE MINIMUM TAX AND ANY STATE, LOCAFOREIGN OR OTHER TAX LAWS AND OF CHANGES IN THOSE LAWS

Tax Consequences of the Merger Generally

In connection with the filing with the SEC of the registration statement on Form of which this proxy statement/prospectus is a part, CVB Financial s tax cou Manatt, Phelps & Phillips, LLP, which we refer to as Manatt, has rendered opinion to CVB Financial. In addition, the obligations of the parties to complete merger are conditioned on, among other things, CVB Financial s receipt of an or from Manatt dated as of the closing date of the merger, that for U.S. federal inc tax purposes the merger will be treated as a reorganization within the meanin Section 368(a) of the Code. The conditions relating to receipt of such closing opinion may be waived by CVB Financial. CVB Financial does not currently intenwaive the conditions related to its receipt of the closing tax opinion. In addition, ability of Manatt to deliver such closing tax opinion is conditioned on the mer satisfying the continuity-of-proprietary-interest requirement. That require generally will be satisfied if CVB Financial common stock constitutes at least 429 the value of the total merger consideration. The determination by tax counsel a whether the merger will be treated as a reorganization within the mean Section 368(a) of the Code is based on the facts and law existing as of the closing of the merger.

These opinions are and will be subject to customary qualifications and assumptiincluding assumptions regarding the absence of changes in existing facts and completion of the merger strictly in accordance with the merger agreement and

registration statement. In rendering its tax opinions, Manatt relied and will rely u representations and covenants, including those contained in certificates of officer CVB Financial and Valley, reasonably satisfactory in form and substance to Mar and will assume that these representations are true, correct and complete with regard to any knowledge limitation, and that these covenants will be complied wit any of these assumptions or representations are inaccurate in any way, or any of covenants are not complied with, these opinions could be adversely affected. opinions represent Manatt s best legal judgment but have no binding effect or of status of any kind, and no assurance can be given that contrary positions will no taken by the Internal Revenue Service or a court considering the issues. In addit we have not requested, nor do we intend to request, a ruling from the Internal Reve Service as to the U.S. federal income tax consequences of the merger. According there can be no assurances that the Internal Revenue Service will not assert, or the court will not sustain, a position contrary to any of the tax consequences set f below or any of the tax consequences described in the tax opinions. The follow discussion assumes that the merger will be consummated as described in the me agreement and in this proxy statement/prospectus and that CVB Financial will waive the closing opinion condition described above in this paragraph.

-72-

U.S. Holders That Receive a Combination of CVB Financial Common Stock and C as Merger Consideration

If a U.S. holder s adjusted tax basis in Valley common stock surrendered is less the sum of the fair market value of the shares of CVB Financial common stock and amount of cash (other than cash received in lieu of a fractional share of CVB Finar common stock) received by the U.S. holder pursuant to the merger, then the holder will recognize gain in an amount equal to the lesser of (a) the sum of amount of cash (other than cash received in lieu of a fractional share of CVB Finar common stock) and the fair market value of CVB Financial common stock recei minus the adjusted tax basis of the Valley shares surrendered in exchange there and (b) the amount of cash received by the U.S. holder (other than cash receive lieu of a fractional share of CVB Financial common stock). However, if a U.S. hol adjusted tax basis in the Valley shares surrendered is greater than the sum of amount of cash (other than cash received in lieu of a fractional share of CVB Finar common stock) and the fair market value of CVB Financial common stock recei the U.S. holder s loss will not be currently allowed or recognized for U.S. fe income tax purposes. If a U.S. holder of Valley shares acquired different block Valley shares at different times or different prices, it is recommended that the holder consult its tax advisor regarding the manner in which gain or loss should determined for each identifiable block. Except to the extent any cash receive treated as a dividend as discussed below, any recognized gain generally wil long-term capital gain if, as of the effective date of the merger, the U.S. hole holding period with respect to the Valley shares surrendered exceeds one year some cases, if the U.S. holder actually or constructively owns CVB Financial comstock other than CVB Financial common stock received in the transaction, recognized gain could be treated as having the effect of the distribution of a divide under the tests described in Section 302 of the Code, in which case such gain we be treated as dividend income to the extent of such U.S. holder s ratable sha accumulated earnings and profits (as calculated for U.S. federal income tax purpos In such cases, it is recommended that U.S. holders that are corporations consult t tax advisors regarding the potential applicability of the extraordinary divi provisions of the Code.

The aggregate tax basis of CVB Financial common stock received by a U.S. ho that exchanges its Valley shares for a combination of CVB Financial common stand cash as a result of the merger (including any fractional share interests deer received and redeemed for cash as described below) will be the same as the aggregate tax basis of the Valley shares surrendered in exchange therefor, reduced by amount of cash received on the exchange (excluding cash received in lieu fractional share of CVB Financial common stock) plus the amount of any gair dividend income recognized upon the exchange (excluding any gain recognized result of any cash received in lieu of a fractional share of CVB Financial common stock). The holding period of CVB Financial common stock received (including fractional share deemed received and redeemed) will include the holding period of Valley shares surrendered. We recommend that a U.S. holder receiving a combinat of CVB Financial common stock and cash consult its own tax advisor regarding manner in which cash and CVB Financial common stock should be allocated am

the U.S. holder s Valley shares and the manner in which the above rules would in the holder s particular circumstance.

U.S. Holders That Receive Solely Cash Due to Exercise of Dissenters Rights

Upon the proper exercise of dissenters—rights, the exchange of Valley shares sole cash generally will result in recognition of gain or loss by the U.S. holder in amount equal to the difference between the amount of cash received and the U holder—s tax basis in the Valley shares surrendered. The gain or loss recognized w long-term capital gain or loss if, as of the effective date of the merger, the U.S. holdeding period for the Valley shares surrendered exceeds one year. The deductib of capital losses is subject to limitations. In some cases, if a U.S. holder actuall constructively owns CVB Financial common stock after the merger, the cash received under the test forth in Section 302 of the Code, in which case such U.S. holder may he dividend income up to the amount of the cash received. In such cases, it recommended that U.S. holders that are corporations consult their tax advise regarding the potential applicability of the—extraordinary dividend—provisions Code.

-73-

Cash Instead of a Fractional Share

If a U.S. holder receives cash in lieu of a fractional share of CVB Financial commstock, the U.S. holder will be treated as having received a fractional share of CF Financial common stock pursuant to the merger and then as having exchanged fractional share of CVB Financial common stock for cash in a redemption by CF Financial. As a result, the U.S. holder generally will recognize gain or loss equated the difference between the amount of cash received and the U.S. holder is basis fractional share of CVB Financial common stock as set forth above. This gain or generally will be capital gain or loss and will be long-term capital gain or loss if, at the effective date of the merger, the U.S. holder is holding period with respect fractional share (including the holding period of Valley common stock surrende therefor) exceeds one year. The deductibility of capital losses is subject to limitation

The Special Dividend

The special dividend will be paid from the assets of Valley and is not mer consideration. Each recipient of a special dividend will have taxable income to extent of that shareholder s ratable share of the current or accumulated earning profits of Valley.

Backup Withholding

If a U.S. holder is a non-corporate holder of Valley common stock, the U.S. ho may be subject, under certain circumstances, to information reporting and bac withholding on any cash payments that the U.S. holder receives. A U.S. ho generally will not be subject to backup withholding, however, if the U.S. holder:

furnishes a correct taxpayer identification number, certifying that it is subject to backup withholding on IRS Form W-9 or substitute or succesform included in the letter of transmittal that the U.S. holder will receive otherwise complies with all the applicable requirements of the bac withholding rules; or

provides proof that it is otherwise exempt from backup withholding. Any amounts withheld under the backup withholding rules are not an additional and will generally be allowed as a refund or credit against the U.S. holder is federal income tax liability if the U.S. holder timely furnishes the required informato the Internal Revenue Service.

Certain Reporting Requirements

If a U.S. holder that receives CVB Financial common stock in the merge considered a significant holder, such U.S. holder will be required (a) to statement with its U.S. federal income tax return providing certain facts pertinent

the merger, including such U.S. holder s tax basis in, and the fair market value Valley common stock surrendered by such U.S. holder and (b) to retain permarked records of these facts relating to the merger. A significant holder is any shareholder that, immediately before the merger, (y) owned at least 1% (by vot value) of the outstanding stock of Valley or (z) owned Valley securities with a basis of \$1.0 million or more.

HOLDERS OF VALLEY COMMON STOCK ARE STRONG ENCOURAGED TO CONSULT THEIR TAX ADVISORS WITH RESPETO THE APPLICATION OF U.S. FEDERAL INCOME TAX LAWS THEIR PARTICULAR SITUATIONS AS WELL AS ANY T CONSEQUENCES ARISING UNDER THE U.S. FEDERAL ESTATE OR GTAX RULES, OR UNDER THE LAWS OF ANY STATE, LOCAL, FOREIOR OTHER TAXING JURISDICTION OR UNDER ANY APPLICABLE TREATY.

-74-

DISSENTERS RIGHTS OF VALLEY SHAREHOLDERS

Valley shareholders who do not vote in favor of the merger agreement and comply with the requirements set forth in Chapter 13 of the California Gen Corporation Law, which we refer to as the Corporations Code, may deman Valley acquire their shares of Valley common stock for cash at their fair market value of September 22, 2016, the day of, and immediately prior to, the first pulling announcement of the terms of the merger, excluding any appreciation or depreciation consequence of the merger. The Valley board of directors has determined that fair market value of the Valley shares on September 22, 2016 was \$16.55 per shares

Any Valley shareholder wishing to exercise dissenters—rights is urged to consult counsel before attempting to exercise dissenters—rights. Failure to comply strictly all of the procedures set forth in Chapter 13 of the Corporations Code, which consof Sections 1300-1313, may result in the loss of a shareholder—s statutory disserights. In such case, such shareholder will be entitled to receive the mer consideration under the merger agreement.

The following discussion is a summary of Chapter 13 of the Corporations Cowhich sets forth the procedures for Valley shareholders to dissent from the proportions and to demand statutory dissenters—rights of appraisal of their shares und Corporations Code. The following discussion is not a complete statement of provisions of the Corporations Code relating to the rights of Valley shareholder receive payment of the fair market value of their shares and is qualified in its entity reference to the full text of Chapter 13 of the Corporations Code, which is proving its entirety as **Appendix C** to this proxy statement/prospectus.

All references in Chapter 13 of the Corporations Code and in this section shareholder are to the holder of record of the shares of Valley common stock which dissenters rights are asserted. A person having a beneficial interest in the sof Valley common stock held of record in the name of another person, such broker or nominee, cannot enforce dissenters rights directly and must act promp cause the holder of record to follow the steps summarized below properly and timely manner to perfect such person s dissenters rights.

If dissenters—rights are perfected and exercised or are capable of being perfecte exercised with respect to more than 10% of Valley outstanding shares of commutators, then CVB Financial has the option to terminate the merger agreement.

Even though a shareholder who wishes to exercise dissenters—rights may be required take certain actions following receipt of this proxy statement/prospectus to perfect dissenters—rights, if the merger agreement is later terminated and the merge abandoned, no Valley shareholder will have the right to any payment from Valle CVB Financial, other than necessary expenses incurred in proceedings initiate good faith and reasonable attorneys—fees, by reason of having taken that action.

Not Vote For the Merger

Any Valley shareholder who desires to exercise dissenters—rights must not vote the approval of the merger and the merger agreement. If a Valley shareholder return signed proxy without indicating a decision on the proposal, or returns a signed procard approving the proposal, his, her or its shares will be counted as votes in favor the merger proposal and such shareholder will lose any dissenters—rights. Therefore you wish to dissent and you execute and return your proxy card in the accompany form, you must specify that you either disapprove of the proposal or abstain for voting on the proposal.

Notice of Approval by Valley

If the merger is approved by Valley shareholders, Valley is required within 10 of after the approval to send to those Valley shareholders who did not vote **FO** approval of the merger a written notice of

-75-

approval of the merger by Valley s shareholders accompanied by a copy of Sec 1300, 1301, 1302, 1303 and 1304 of the Corporations Code, a statement of the p determined by Valley to represent the fair market value of the dissenting sh immediately prior to the public announcement of the term of the merger agreement September 22, 2016 and a brief description of the procedure to be followed if shareholder desires to exercise dissenters right under the Corporations Code statement of price determined by Valley to represent the fair market valu dissenting shares, as set forth in the notice of approval, will constitute an offe Valley to purchase the dissenting shares at the stated price if the merger is compl and the dissenting shares do not otherwise lose their status as such. Within 30 c after the date of the mailing of the notice of shareholder approval, a dissen shareholder must submit to Valley, or its transfer agent for endorsement as dissen shares, the stock certificates representing the Valley shares as to which s shareholder is exercising dissenters rights. If the dissenting shares are uncertified then such shareholder must provide written notice of the number of shares that shareholder demands Valley purchase within 30 days after the date of the mailin the notice of shareholder approval.

Written Demand for Payment

In addition, to preserve dissenters—rights, a Valley shareholder must make a w demand for the purchase of the shareholder—s dissenting shares and payment to shareholder of the fair market value of Valley common stock within 30 days after date on which the notice of approval is mailed. Simply failing to return a proxy or indicating on your proxy card your disapproval of the merger, does not constitute proper written demand under the Corporations Code. To comply with requirements under the Corporations Code, the written demand must:

specify the shareholder s name and mailing address and the number and cl shares of Valley stock held of record which the shareholder demands Valley purchase;

state that the shareholder is demanding purchase of the shares and paymer their fair market value; and

state the price that the shareholder claims to be the fair market value of the shares immediately prior to the first public announcement of the terms of the merger agreement on September 22, 2016; the statement of fair market value constitutes an offer by the shareholder to sell the shares to Valley at that price.

Any written demands for payment should be sent to Valley Commerce Banc Attention: Allan W. Stone, President and Chief Executive Officer, 701 West M. Street, Visalia, California 93291. Shares of Valley common stock held shareholders who have perfected their dissenters rights in accordance with Chap

of the Corporations Code and have not withdrawn their demands or otherwise their dissenters—rights are referred to as dissenting shares.

Payment of Agreed-Upon Price

If Valley and a dissenting shareholder agree that the shares are dissenting shares agree on the price of the shares, the dissenting shareholder is entitled to receive agreed-upon price with interest at the legal rate on judgments from the date of agreement and will not receive any of the merger consideration with respect to shares. Payment for the dissenting shares must be made within 30 days after the lof the date of that agreement or the date on which all statutory and contract conditions to the merger are satisfied. Payments are also conditioned on the surrer of any certificates representing the dissenting shares.

Determination of Dissenting Shares or Fair Market Value

If Valley denies that shares are dissenting shares or the shareholder fails to agree Valley as to the fair market value of the shares, then, within six months after notic approval of the merger is sent by Valley to its

-76-

shareholders, any shareholder demanding purchase of such shares as dissenting shor any interested corporation may file a complaint in the Superior Court in the pro-California county asking the court to determine whether the shares are dissenshares or to determine the fair market value of the shareholder is shares, or both, of intervene in any action pending on such complaint. If a complaint is not file intervention in a pending action is not made within the specified six-month period dissenter is rights are lost. If the fair market value of the dissenting shares is at the court will determine, or will appoint one or more impartial appraiser determine, such fair market value.

Maintenance of Dissenting Share Status

Except as expressly limited by Chapter 13 of the Corporations Code, holder dissenting shares continue to have all the rights and privileges incident to their sh until the fair market value of their shares is agreed upon or determined. A holder dissenting shares may not withdraw a demand for payment unless Valley consent the withdrawal.

Dissenting shares lose their status as dissenting shares, and dissenting shareholcease to be entitled to require Valley to purchase their shares, upon any of following:

The merger is abandoned;

The shares are transferred before their submission to Valley for the requendorsement;

The dissenting shareholder and Valley do not agree on the status of the sh as dissenting shares or do not agree on the purchase price, but neither Va nor the shareholder files a complaint or intervenes in a pending action wi six months after Valley mails a notice that its shareholders have approved merger; or

With Valley s consent, the dissenting shareholder withdraws the shareholder and for purchase of the dissenting shares.

To the extent that the provisions of Chapter 5 of the Corporations Code (which p conditions on the power of a California corporation to make distributions to shareholders) prevent the payment to any holders of dissenting shares of the market value of the dissenting shares, the dissenting shareholders will be creditors of Valley for the amount that they otherwise would have received in repurchase of their dissenting shares, plus interest at the legal rate on judgments the date of payment, but be subordinate to all other creditors of Valley in liquidation proceeding, with the debt to be payable when permissible under

provisions of Chapter 5 of the Corporations Code.

-77-

COMPARISON OF SHAREHOLDERS RIGHTS

Valley is incorporated under the laws of the State of California, and the right Valley shareholders are governed by the laws of the State of California, in particular the California Corporations Code, and Valley s articles of incorporation and Va bylaws. As a result of the merger, Valley shareholders will receive shares of C Financial common stock and will become CVB Financial shareholders. C Financial is incorporated under the laws of the State of California, and the right CVB Financial shareholders are governed by the laws of the State of California particular, the California Corporations Code and CVB Financial s articl incorporation and CVB Financial s bylaws. Thus, following the merger, the rig Valley shareholders who become CVB Financial shareholders in the merger continue to be governed by the laws of the State of California, but will no longe governed by Valley s articles of incorporation and bylaws, but instead will be gov by CVB Financial s articles of incorporation and CVB Financial s bylaws.

Set forth below is a summary comparison of material differences between the right Valley shareholders under Valley s articles of incorporation and bylaws (left col and the rights of CVB Financial shareholders under CVB Financial s articles incorporation and bylaws (right column). The summary set forth below discu certain of the material differences between the rights of CVB Financial sharehol and Valley shareholders under such documents. Copies of the full text of C Financial s articles of incorporation and CVB Financial s bylaws currently in eff available without charge by following the instructions in the section entitled WF YOU CAN FIND MORE INFORMATION beginning on page i.

Valley

CVB Financial

Authorized Capital Stock

Valley s articles of incorporation states that CVB Financial s articles of incorporation the authorized capital stock of Valley states that the authorized capital stoc consists of 30,000,000 shares of common CVB Financial consists of 225,000 stock, without par value, and 10,000,000 shares of common stock, without shares of preferred stock. As of November value, and 20,000,000 shares 4, 2016, there were 3,002,014 shares of preferred stock, without par value. A Valley common stock outstanding and no September 30, 2016, there w shares of preferred stock outstanding or 108,097,493 shares of CVB Finan designated. Subject to compliance with the common stock outstanding and California Corporations Code, and Valley sshares of preferred stock outstandin articles of incorporation and bylaws, the designated. Subject to compliance Valley board of directors may authorize the the California Corporations Code issuance of additional shares of common CVB Financial s articles of incorpostock and preferred stock.

and bylaws, the CVB Financial boar directors may authorize the issuance

additional shares of common stock preferred stock.

Dividends

The shareholders of Valley are entitled to The shareholders of CVB Financial receive dividends when and as declared by entitled to receive dividends when the board of directors, out of funds legally as declared by the board of directors, available for the payment of dividends, as of funds legally available for provided in the California Corporations payment of dividends, as provided in Code. The California Corporations Code California Corporations Code. provides that a corporation may make a California Corporations Code prov distribution to its shareholders if retained that a corporation may mak earnings immediately prior to the dividend distribution to its shareholder payout is at least equal to the amount of the retained earnings immediately price proposed distribution. In the event that the dividend payout is at least equa sufficient retained earnings are not the amount of the proposed distribut available for the proposed distribution, a In the event that sufficient retai corporation may, nevertheless, make a earnings are not available for distribution if it meets both the quantitative proposed distribution, a corpora solvency and the liquidity tests. In generalay, nevertheless, make a distribution the quantitative solvency

it meets both the quantitative solv and the liquidity

-78-

Valley

corporation equals at least 1.25 times its solvency test requires that the sur liabilities. The liquidity test generally the assets of the corporation equal requires that a corporation have current least 1.25 times its liabilities. assets at least equal to current liabilities or, liquidity test generally requires the if the average of the earnings of the corporation have current assets at 1 corporation before taxes on income and equal to current liabilities or, if before interest expenses for the two average of the earnings of preceding fiscal years was less than the corporation before taxes on income average of the interest expense of the before interest expenses for the corporation for such fiscal years, then preceding fiscal years was less than current assets must equal at least 1.25 times average of the interest expense of current liabilities. In certain circumstances, corporation for such fiscal years, to Valley may be required to obtain the prior current assets must equal at least approval of the Federal Reserve Board to times current liabilities. In cer make capital distributions to shareholders circumstances, CVB Financial ma of Valley.

CVB Financial

test requires that the sum of the assets of the tests. In general, the quantital required to obtain the prior approva the Federal Reserve Board to m capital distributions to shareholder CVB Financial.

Voting Rights

Each outstanding share is entitled to one Each CVB Financial sharehol vote on each matter submitted to a vote of entitled to vote is entitled to one vote shareholders. In the election of directors, each share held of record on all man each shareholder may cumulate votes and submitted to a vote of shareholders give one candidate a number of votes equal the election of directors, e to the number of directors to be elected shareholder may cumulate votes multiplied by the number of votes to which give one candidate a number of votes to which the shareholder s shares are normally equal to the number of directors to entitled or may distribute the shareholder selected multiplied by the numbe votes on the same principle among as many votes to which the shareholder s s candidates as the shareholder thinks fit.

No shareholder is entitled to cumulate votes principle among as many candidate in favor of any candidate or candidates the shareholder thinks fit. unless such candidate s or candidates names have been placed in nomination prior to the No shareholder is entitled to cumu voting and the shareholder has given notice votes in favor of any candidate at the meeting prior to the voting of the candidates unless such candidate shareholder s intention to cumulate the candidates names have been place shareholder s votes. If any one shareholder nomination prior to the voting and has given such notice, this fact shall be shareholder has given notice at announced to all shareholders and proxies meeting prior to the voting of present, who may then cumulate their votes shareholder s intention to cumular for candidates in nomination.

are normally entitled or may distrib the shareholder s votes on the

shareholder s votes. If any

In any election of directors, the candidates shareholders and proxies present, receiving the highest number of votes of the may then cumulate their votes shares entitled to be voted for them, up to candidates in nomination. the number of directors to be elected by such shares, are elected.

shareholder has given such notice, fact shall be announced to

In any election of directors, candidates receiving the highest nun of votes of the shares entitled to voted for them, up to the numbe directors to be elected by such sha are elected.

Qualification of Directors

serve as a member of the board of directors: qualification for directors. C (i) who is a director, officer, employee, or Financial s Nominating and Corp agent for any other financial institution, or Governance Committee Char (ii) who has been or is the assignee of provides that its Nominating anyone who has any understanding with Corporate Governance Committee any other

Valley s bylaws provide that no person shallCVB s bylaws include no spe consider specific factors when

-79-

Valley

person could be called upon to reveal or in including the potential direct any way utilize information obtained as a business background, curr director or will, directly or indirectly, responsibilities, commun attempt to effect or encourage any action of involvement, ownership of C Valley; or (iii) with the exception of Financial s voting securities, know directors who are also employees of Valley, and contacts in the CVB Financ who does not hold as beneficial owner a industry and other industries relevan minimum of \$25,000 fair market value of the its business, ability to work as Valley common stock; provided, however, of an effective group and abilit that the board of directors may permit a commit adequate time to serve person who does not satisfy one or more of director, but none of these factor the qualifications listed above to serve as a dispositive. member of the board of directors following the board s determination that such action will not compromise the business plans or strategic focus of Valley.

CVB Financial

financial institution, pursuant to which that evaluating potential director candidates

Number of Directors

Valley s bylaws state that the authorized CVB Financial s bylaws state that number of directors constituting the board authorized number of direct of directors will be from eight to 15, with constituting the board of directors the exact number within such range to be from seven to 13, with the ex determined from time to time by a number within said range to resolution of the board of directors or by the determined from time to time by shareholders at an annual meeting of resolution of the board of directors. shareholders. The number of directors is number of directors is fixed by fixed by the board of directors at eight, and board of directors at eight, and there there are currently eight members of the currently eight members of the C Valley board of directors.

Financial board of directors.

Election of Directors

Valley s bylaws provide that directors shall CVB Financial s bylaws provide be elected annually by the shareholders at directors shall be elected annually the annual meeting of the shareholders. If, the shareholders at the annual mee for any reason, the annual meeting or an of the shareholders. If, for any rea adjournment thereof is not held or the the annual meeting or an adjournment directors are not elected thereat, then the thereof is not held or the directors directors may be elected at any special not elected thereat, then the direc meeting of the shareholders called and held may be elected at any special meetin for that purpose. The term of office of the the shareholders called and held for directors shall begin immediately after their purpose. The term of office of election and continue until their respective directors shall begin immediately a successors are elected and qualified.

their election and continue until t respective successors are elected

qualified.

Classification of Board of Directors

Valley s bylaws do not provide for aCVB Financial s bylaws do not pr classified board of directors. for a classified board of directors.

Vacancies

the board of directors, not including a vacancy on the board of directors vacancy created by the removal of a be filled by a majority of the remain director, may be filled by approval of the directors, even though less that board or, if the number of directors then in quorum, or by a sole remaining directors office is less than a quorum, by the Any director so elected may hold of unanimous written consent of the directors for the remainder of the full term of then in office, the affirmative vote of a

Valley s bylaws provide that a vacancy on CVB Financial s bylaws provide director in which the vacancy

-80-

Valley

CVB Financial

majority of the remaining directors, or by a occurred until such director s succ sole remaining director. Any director so is elected at an annual or speelected may hold office for the remainder of shareholders meeting. The shareholders the full term of the director in which the may elect a director at any time to vacancy occurred until such director sany vacancy not filled by directors. successor is elected at an annual or special such election by written consent o shareholders meeting. A vacancy created by than to fill a vacancy created by rem the removal of a director may be filled only requires the consent of a majority of by the approval of the shareholders. In outstanding shares entitled to vote. addition, the shareholders may elect a director at any time to fill any vacancy not filled by directors.

Removal of Directors

A director may be removed from office A director may be removed from of without cause by a vote of shareholders without cause by a vote of sharehol holding a majority of the outstanding shares holding a majority of the outstand entitled to vote at an election of directors; shares entitled to vote at an electio however, unless the entire board is directors; however, unless the en removed, an individual director shall not be board is removed, an individual dire removed if the votes cast against removal, shall not be removed if the votes or not consenting in writing to such against removal, or not consentin removal, would be sufficient to elect such writing to such removal, would director if voted cumulatively at an election sufficient to elect such director if voted cumulatively at an election at which the same total number of votes cumulatively at an election at which were cast. In addition, a director may also total number of votes were cast, o be removed from office by the Superior such action is taken by written cons Court of the county in which the principal all shares entitled to vote were vo office is located, at the suit of shareholders and the entire number of direct holding at least 10% of the number of authorized at the time of the direc outstanding shares of any class, in case of most recent election were then be fraudulent or dishonest acts or gross abuse elected. In addition, a director may of authority or discretion with reference to be removed from office by the Supe Valley, in the manner provided by the law.

No reduction of the authorized number of shareholders holding at least 10% of directors shall have the effect of removing number of outstanding shares of any director before his or her term of office class, in case of fraudulent or disho expires.

Court of the county in which principal office is located, at the su acts or gross abuse of authority discretion with reference to corporation, in the manner provided the law.

No reduction of the authorized nun of directors shall have the effec removing any director before his or

term of office expires.

Nomination of Director Candidates by Shareholders

Valley s bylaws permit shareholders who CVB Financial s bylaws pe are entitled to vote for the election of shareholders who are entitled to vote directors to nominate a director for election the election of directors to nomina if written notice is delivered to the president director for election if written notice of the corporation by the later of 21 days delivered to the president of prior to any meeting of shareholders called corporation not less than 10 nor n for the election of directors or ten days after than 60 days prior to any meeting the date that a notice of the meeting is shareholders called for election mailed.

directors. If fewer than 10 days not the meeting is given to sharehold such notice of intention to nomin must be received by the president of corporation not later than the time f in the notice of the meeting for opening of the meeting.

-81-

CVB Financial Advance Notice of Shareholder Proposals

Valley does not have an advance notice CVB Financial does not have provision for the submission of shareholder advance notice provision for proposals at its annual meeting of submission of shareholder proposa shareholders.

its annual meeting of shareholders.

Shareholder Action Without a Meeting

Any action required to be taken by Any action which may be taken at shareholders of Valley must be taken at a annual or special meeting duly called annual meeting or a special shareholders may be taken witho meeting of shareholders. No action may be meeting and without prior notice taken by written consent.

consent in writing, setting forth action so taken, is signed by the nun of shareholders whose affirmative would be required to take such action a meeting at which all shares entitle vote thereon were present and vo Unanimous written consent shal required for the election of director non-vacant positions.

Special Meetings of Shareholders

Valley may call a special shareholders CVB Financial may call a spemeeting at any time upon the request of the shareholders meeting at any time board of directors, the chairman of the the request of the board of direct board, the president, or Valley shareholders chairman of the board, the presiden entitled to cast not less than 10% of the of the CVB Financial sharehold votes at such a meeting.

entitled to cast not less than 10% of votes at such a meeting.

Indemnification of Directors and Officers

Valley s articles of incorporation authorize CVB Financial s articles of incorporation Valley to indemnify its directors, officers, authorize the corporation to indem and agents, through bylaw provisions, or its directors, officers, and age through agreements with such agents, both through bylaw provisions, agreem in excess of the indemnification otherwise with such agents or other persons, permitted by Section 317 of the California of shareholders or disinterested direc Corporations Code to the fullest extent or otherwise, in excess of permitted by applicable law.

indemnification otherwise permitted Section 317 of the Califor

Valley s bylaws provide that Valley willCorporations Code to the fullest ex indemnify any person who was or is a permitted by applicable law. party, or is threatened to be made a party to any threatened, pending or completed

action or proceeding by reason of the fact CVB Financial s bylaws provide that he or she is or was an agent of Valley CVB Financial will indemnify or is or was serving at the request of Valley directors, officers and employees, an as a director, officer, employee or agent of its subsidiaries, who was or is maanother corporation, partnership, joint party or is threatened to be made a p venture, trust or other enterprise, to the to or is otherwise involved i fullest extent permitted by law. Valley has proceeding against expenses, judgme entered into separate indemnification fines, settlements, and other amount agreements with its directors and officers actually and reasonably incurre for such purpose.

connection with any such proceed arising by reason of the fact that such person is or was an agent of corporation. CVB Financial has enter into separate indemnificat agreements with its directors for s purpose.

-82-

Valley **CVB Financial** Amendments to Articles of Incorporation and Bylaws

Valley s articles of incorporation may be CVB Financial s articles of incorporation amended if approved by the board of may be amended by the affirmative directors and approved by the affirmative of the issued and outstanding sha vote of a majority of the outstanding shares entitled to vote. CVB Financial s by represented and voting at a duly held may be amended or repealed by shareholder meeting at which a quorum is board of directors or the affirmative present. Valley s bylaws may be amended or of a majority of the outstanding sh repealed by the board of directors or the entitled to vote. An amendm affirmative vote of a majority of the reducing the number of directors outstanding shares entitled to vote.

fixed-number board or the minin number of directors of variable-number board to a number than five cannot be adopted if the v cast against its adoption at a meet are equal to more than 16-2/3% of outstanding shares entitled to vote.

Dissenters Rights

Valley does not have any capital stock Under the California Corporati outstanding that is listed on a national Code, because CVB Financial s cor securities and, therefore, all outstanding stock is listed on a national security shares of the capital stock of Valley exchange certified by the California generally have dissenters rights with respectCommissioner of the Departmen to a business combination or other Business Oversight, holders of C reorganization requiring their vote. Financial common stock do not h However, in a business combination where dissenters rights with respect Valley shareholders possess more than of business combination or ot 5/6 of the voting power of the surviving reorganization requiring their v corporation, such shareholders do not have unless their shares are subject to transdissenters rights.

restrictions or are exchanged for me consideration other than sol securities listed on a national securexchange certified by the California Commissioner of the Departmen Business Oversight and cash in lie fractional shares.

-83-

BENEFICIAL OWNERSHIP OF VALLEY COMMON STOCK

The following table sets forth certain information as of [], 2016, the record da Valley s special meeting, concerning the beneficial ownership of Valley s outst common stock: (i) by each of Valley s directors; (ii) by each of Valley s excofficers; and (iii) by all directors and executive officers of Valley as a grow Management is not aware of any change in control of Valley that has occurred s January 1, 2016 or of any arrangement that may, at a subsequent date, result change in control of Valley other than the pending merger. As used with respect Valley, the term executive officer means the President and Chief Executive Control of Valley and Chief Credit Officer and the Executive Control of Valley and are specifically excluded from participation be executive officers of Valley and are specifically excluded from participation policy-making by resolution of the Valley board of directors. The information been obtained from Valley, or from information furnished directly by the individual entity named below to Valley.

As of [], 2016, there were 3,002,014 shares of Valley common stock issue outstanding. For purposes of the following table, shares that the beneficial owner the right to acquire within 60 days of [], 2016 are deemed to be issued and outsta and have been treated as outstanding in determining the amount and natur beneficial ownership and in calculating the percentage of ownership of the individuals possessing such interest, but not for any other individuals. As of [], however, no options to acquire shares of Valley common stock were outstanding.

		Shares of
		Common Stock Percent
Name of Beneficial Owners ⁽¹⁾	Position B	eneficially Owned ⁽²⁾ Class ⁽²⁾
Walter A. Dwelle	Chairman an	d
	Director	134,049 4.4
Roy O. Estridge	Executive Vic	e 21,383 0.7
	President, Chie	$\pm f$
	Operating Office	er
	and Chief Financia	al
	Officer	
Donald A. Gilles	Director	51,620 ⁽³⁾ 1.7
Philip R. Hammond, Jr.	Director	94,058 ⁽⁴⁾ 3.1
Russell F. Hurley	Vice Chairman and	d
	Director	61,582 2.0
William R. Kitchen	Executive Vic	e
	President and Chie	ef
	Credit Officer	14,652 0.49
Fred P. LoBue, Jr.	Secretary and	d
	Director	87,921 ⁽⁵⁾ 2.9
Kenneth H. Macklin	Director	39,835 ⁽⁶⁾ 1.3
Barry R. Smith	Director	41,871 ⁽⁷⁾ 1.3

Allan W. Stone	President, Chief Executive Officer		
	and Director	32,151	1.0
All Directors and Executive		579,122	19.2
Officers as a Group (10 in			
number)			

- (1) Except as shown, the address for all persons is c/o Valley Commerce Bancorp, W. Main Street, Visalia California, 93291.
- (2) Subject to applicable community property laws and shared voting and investing power with a spouse, the persons listed have sole voting and investment powith respect to such shares unless otherwise noted.
- (3) Includes 5,048 shares held in the name of his spouse.
- (4) Includes 15,253 shares held in trust with his spouse.
- (5) Includes 4,993 shares held in the name of his spouse.
- (6) Includes 13,159 shares held by Vintage Equipment Profit Sharing Plan.
- (7) Includes 6,737 shares held in trust with his spouse.

-84-

Principal Shareholders

No persons are known to management to own directly or indirectly, five percentage more of Valley soutstanding common stock as of September 30, 2016, exception follows:

Name and Address		
	Amount of	
of Beneficial Owner	Beneficial OwnershipPero	cent of Class
The Banc Fund Company,	257,503	8.58%
LLC		
208 South LaSalle Street		
Chicago, Illinois 60604		

-85-

LEGAL AND TAX OPINIONS

Manatt, Phelps & Phillips, LLP will deliver prior to the effective time of the me its opinion to CVB Financial as to certain U.S. federal income tax consequences of merger. Please see the section entitled MATERIAL UNITED STATES FEDE INCOME TAX CONSEQUENCES OF THE MERGER beginning on page 71 validity of CVB Financial common stock to be issued in connection with the me will be passed upon for CVB Financial by its Manatt, Phelps & Phillips, LLP.

EXPERTS

The consolidated financial statements of CVB Financial as of December 31, 2015 2014, and for each of the years in the three-year period ended December 31, 2015, management s assessment of the effectiveness of internal control over finareporting as of December 31, 2015 have been incorporated by reference herein and the registration statement in reliance upon the report of KPMG LLP, independent registered public accounting firm, incorporated by reference herein, and upon authority of said firm as experts in accounting and auditing.

OTHER MATTERS

Valley s bylaws provide that no matters may be brought before a special sharehol meeting other than as set forth in the notice to shareholders of the special sharehol meeting. Consequently, no matters will be presented for consideration at the special other than as described in this proxy statement/prospectus.

-86-

INCORPORATION OF CERTAIN DOCUMENTS BY REFERENCE

The SEC allows CVB Financial to incorporate certain information into this pr statement/prospectus by reference to other information that has been filed with SEC. The information incorporated by reference is deemed to be part of this pr statement/prospectus, except for any information that is superseded by information this proxy statement/prospectus. The documents that are incorporated by refere contain important information about the companies and you should read this pr statement/prospectus together with any other documents incorporated by reference this proxy statement/prospectus.

This proxy statement/prospectus incorporates by reference the following documthat have previously been filed with the SEC by CVB Financial:

Annual Report on Form 10-K for the year ended December 31, 2015 filed February 29, 2016 and the amendment to such report field on March 24, 20

Proxy Statement for Annual Meeting filed on April 6, 2016;

Quarterly Reports on Form 10-Q for the quarter ended March 31, 2016 filed May 10, 2016, the quarter ended June 30, 2016 filed on August 9, 2016, the quarter ended September 30, 2016 filed on November 9, 2016;

Current Reports on Form 8-K filed on April 13, 2016; April 27, 2016; May 20, 2016; June 29, 2016; and July 29, 2016 (and the amendn thereto filed on August 3, 2016); and

The description of CVB Financial common stock contained in the Registration Statement on Form 8-A, dated June 11, 2001, which registers its common stock under Section 12 of the Securities Exchange Act of 1934, together with any amendments or reports filed with the SEC for the purpose of updating the description.

In addition, CVB Financial is incorporating by reference any documents it may under Section 13(a), 13(c), 14 or 15(d) of the Exchange Act after the date of proxy statement/prospectus and prior to the date of the special meeting of Va shareholders; provided, however, that CVB Financial is not incorporating by refere any information furnished (but not filed), except as otherwise specified herein.

CVB Financial files annual, quarterly and special reports, proxy statements and obusiness and financial information with the SEC. You may obtain the information incorporated by reference and any other materials CVB Financial files with the swithout charge by following the instructions in the section entitled WHERE

CAN FIND MORE INFORMATION beginning on page i of this p statement/prospectus.

Neither CVB Financial nor Valley has authorized anyone to give any informator make any representation about the merger or the respective companies that different from, or in addition to, that contained in this prostatement/prospectus or in any of the materials that have been incorporated this proxy statement/prospectus. Therefore, if anyone does give you informat of this sort, you should not rely on it. If you are in a jurisdiction where offer exchange or sell, or solicitations of offers to exchange or purchase, the securi offered by this proxy statement/prospectus or the solicitation of proxies unlawful, or if you are a person to whom it is unlawful to direct these type activities, then the offer presented in this proxy statement/prospectus does extend to you. The information contained in this proxy statement/prospectus unless information specifically indicates that another date applies.

-87-

Append

Execution Ver

AGREEMENT AND PLAN OF REORGANIZATION AND MERGER

by and between

CVB FINANCIAL CORP.

and

VALLEY COMMERCE BANCORP

September 22, 2016

TABLE OF CONTENTS

ARTICL	E I MERGER	-
1.1	The Merger	
1.2	Effective Time	
1.3	Effects of the Merger	
1.4	Conversion of Stock	
1.5	Valley Stock Options	
1.6	Articles of Incorporation and Bylaws	
1.7	Board of Directors and Officers	
1.8	Bank Merger	
ARTICL	E II ARTICLE II EXCHANGE OF SHARES	
2.1	Delivery of Merger Consideration	
2.2	Exchange Procedures	
ARTICL		
VALLE		
3.1	Corporate Organization	
	Capitalization	
3.3	Authority; No Violation	
3.4	Consents and Approvals	A
3.5	Reports	A
3.6	Financial Statements; Controls	A
3.7	Broker s Fees	A
3.8	Absence of Changes	A
3.9	Compliance with Applicable Law	A
3.10	State Takeover Laws; Rights Agreement	A
3.11	Employee Benefit Plans	A
3.12	Approvals	A
3.13	Opinion	A
3.14	Valley Information	A
3.15	Legal Proceedings	A
3.16	Material Contracts	A
3.17	Environmental Matters	A
3.18	Taxes	A

A-

Table of Contents

3.19 Reorganization

3.20	0 Intellectual Property	A-
3.21	1 Properties	A-
3.22	2 Insurance	A-
3.23	3 Accounting and Internal Controls	A-
3.24	4 Derivatives	A-
3.25	5 Loan Matters	A-

TABLE OF CONTENTS

(Continued)

	Pa
3.26 Community Reinvestment Act Compliance	A-
3.27 Bank Secrecy Act Compliance	A-
3.28 Investment Securities	A-
3.29 Related Party Transactions	A-
3.30 Operating Losses	A-
3.31 Employee and Labor Matters	A-
3.32 Trust Matters	A-
3.33 Information Technology; Security & Privacy	A-
3.34 Accuracy and Completeness of Information Furnished	A-
ARTICLE IV REPRESENTATIONS AND WARRANTIES OF CVB	A-
4.1 Corporate Organization	A-
4.2 Capitalization	A-
4.3 Authority; No Violation	A-
4.4 Consents and Approvals	A-
4.5 Reports	A-
4.6 Financial Statements	A-
4.7 Broker s Fees	A-
4.8 Absence of Changes	A-
4.9 Compliance with Applicable Law	A-
4.10 Approvals	A-
4.11 CVB Information	A-
4.12 Legal Proceedings	A-
4.13 Accounting and Internal Controls	A-
4.14 Reorganization	A-
4.15 Accuracy and Completeness of Information Furnished	A-
ARTICLE V COVENANTS RELATING TO CONDUCT OF BUSINESS	A-
5.1 Conduct of Valley Prior to the Effective Time	A-
5.2 Valley Forbearances	A-
5.3 CVB Forbearances	A-
ARTICLE VI ADDITIONAL AGREEMENTS	A-

Table of Contents 192

6.1 Regulatory Matters

6.2 Reasonable Best Efforts

6.3	Access to Information
6.4	Shareholder Approval
6.5	Nasdaq Listing
6.6	Employee Matters

ii

A-

A-

TABLE OF CONTENTS

(Continued)

		Pa
6.7	Indemnification; Directors and Officers Insurance	A-
6.8	Special Dividend	A-
	No Solicitation	A-
6.10	Takeover Laws	A-
	Schedule Updates	A-
6.12	2 Notification of Certain Matters	A-
6.13	Third-Party Agreements	A-
6.14	Pre-Closing Adjustments	A-
6.15	Shareholder Litigation and Protests	A-
ARTICI	LE VII CONDITIONS PRECEDENT	A-
7.1	Conditions to Each Party s Obligation to Effect the Merger	A-
7.2	Conditions to Obligations of CVB and Citizens Business Bank	A-
7.3	Conditions to Obligations of Valley	A-
ARTICI	LE VIII TERMINATION AND AMENDMENT	A-
8.1	Termination	A-
8.2	Effect of Termination	A-
8.3	Fees and Expenses	A-
8.4	Amendment	A-
8.5	Extension; Waiver	A-
ARTICI	LE IX GENERAL PROVISIONS	A-
9.1	Closing	A-
9.2	Non-survival of Representations, Warranties and Agreements	A-
9.3	Notices	A-
9.4	Interpretation	A-
9.5	Counterparts	A-
9.6	Entire Agreement	A-
9.7	Governing Law; Jurisdiction	A-
9.8	Waiver of Jury Trial	A-
9.9	Publicity	A-
9.10	Assignment; Third-Party Beneficiaries	A-
	Specific Performance	A-

Table of Contents 194

9.12 Valley Disclosure Schedule

Exhibit A	Form of Voting and Support Agreement
Exhibit B	Form of Non-Competition and Non-Solicitation Agreement
Exhibit C	Form of Agreement of Merger
Exhibit D	Form of Bank Merger Agreement

iii

INDEX OF DEFINED TERMS

Section

Acquisition Proposal	6.90
Adverse Change of Recommendation	8.1
Adverse Shareholder Action	S
Advisors	7.2(1
Affiliate	3.60
Affiliated	3.60
Aggregate Cash Amount	1.4(d)
Agreement	Preamb
Agreement of Merger	1
Allowance for Loan Loss Ratio	7.2(n)
Approvals	6.10
Average Non-Interest Bearing Deposits	7.2(n)
Average Accruing Loans	7.2(n)(v)
Bank Merger	1
Bank Merger Agreement	1
BHC Act	3.1
Book-Entry Share	1.4
Business Day	Ģ
California Secretary	1
Cause	6.6
Certificate	1.4
CGCL	1.10
Change of Control Payments	3.16
Citizens Business Bank	1
Closing	Ģ
Closing Balance Sheet	7.2(n)(i
Closing Date	Ģ
Closing Tangible Common Equity	7.2(n)
Closing Tangible Common Equity Minimum	7.2(n)
Code	Recita
Commissioner	3.1(c)
Confidentiality Agreement	6.3
Controlled Group Liability	3.110
Covered Entity	6.6
CRA	3.
CVB	Preamb
CVB Articles	4.10
CVB Average Closing Price	1.4(d)(i
CVB Bylaws	4.10
CVB Capitalization Date	4.2
CVB Common Stock	4.2
CVB Preferred Stock	4.2
CVB SEC Reports	4.50

CVB Stock Plans

D&O Insurance	6.7(
Derivative Transaction	3.2
Determination Date	8.1(f)(ii
Dissenting Share(s)	1.4(
Effective Time	1
Employee Benefit Plan	3.11(
End Date	8.1(b)(i

4.2(

INDEX OF DEFINED TERMS

(Continued)

	Secti
Enforceability Exceptions	3.3
Environmental Laws	3.
ERISA	3.11
ERISA Affiliate	3.11
Exchange Act	3.
Exchange Agent	
Exchange Agent Agreement	
Exchanged Shares	2.2
Excepted Loan	5.2
FDIC	3.1(c)
Federal Reserve	3.1
Form S-4	3
GAAP	3.20
Governmental Entity	3
Indemnified Parties	6.7
Intellectual Property	3.20(e)
Investment Security	3.
IRS	3.
IT Assets	3.20(e)(
Knowledge of CVB	Ó
Knowledge of Valley	Ó
Law	3.30
Lease	3.21
Letter of Transmittal	2.2
Liens	3.30
Loans	3.25
Material Adverse Effect	3
Material Contract	3.16
Materially Burdensome Regulatory Condition	6.10
Measurement Date	7.2(n)(2)
Merger	Recit
Merger Consideration	1.4
Multiemployer Plan	3.11
Multiple Employer Plan	3.11
Nasdaq	4
Non-Competition Agreements	Recit
Notice of Superior Proposal	6.9(c)(ii)
Option Termination Date	1.5
Per Share Cash Amount	1.4(d)(
Per Share Exchange Ratio	1.4(d)
Per Share Merger Consideration Value	1.5(b)

Permitted Encumbrances	3.21
Person(s)	
Pre-Dividend Common Equity	
Previously Disclosed	9.12
Proxy Statement	
Regulatory Agencies	3.5
Regulatory Agreement	3.9

2

INDEX OF DEFINED TERMS

(Continued)

Secti

	Secti
Requisite Regulatory Approvals	7.2
Sarbanes-Oxley Act	3.9
SEC	3
Securities Act	3.2(a)(i
Special Dividend	(
Subsidiary	3.1(c)(t
Support Agreements	Recit
Surviving Corporation	Recit
Takeover Laws	3.10
Tax	3.
Tax Return	3.
Taxes	3.
Total Assets	7.2(n)(1)
Trade Secrets	3.20(e)
Trading Day	1.4(d)(
Transaction Costs	7.2(n)(vi
Treasury Shares	1.40
Valley	Preamb
Valley Acquisition Proposal	8.3(b)(ii)(2), 6.9
Valley Articles	3.10
Valley Audited Financial Statements	3.6
Valley Board Recommendation	
Valley Bylaws	3.10
Valley Capitalization Date	3.2(a)
Valley Closing Shares	1.4(d)(
Valley Common Stock	3.2
Valley Disclosure Schedule	9.12
Valley Filings	3.5
Valley Financial Statements	3.6
Valley Interim Financial Statements	3.6
Valley IT Systems	3.
Valley Leased Properties	3.21
Valley Licensed Intellectual Property	3.20(e)(i
Valley Owned Intellectual Property	3.20(e)(
Valley Owned Properties	3.21
Valley Real Property	3.21
Valley Shareholder Approval	3.3
Valley Shareholder Meeting	6.5
Valley Stock Award	3.2(a)
Valley Stock Option	1.5
Valley Stock Plans	1.5(b)
· and j stock I fallo	1.5(0)

Valley Superior Proposal

Valley Termination Fee	
VBB	
Voting Debt	
Withdrawal Liability	

3

6.9 8.3(b)

3.2(a)(3.11

AGREEMENT AND PLAN OF REORGANIZATION AND MERGER

THIS AGREEMENT AND PLAN OF REORGANIZATION AND MERGER dated as of September 22, 2016 (this <u>Agreement</u>), is by and between CVB Finar Corp., a California corporation (<u>CVB</u>), and Valley Commerce Bancorp, a Califo corporation (<u>Valley</u>).

RECITALS

- A. The respective Boards of Directors of CVB and Valley have determined that it is the best interests of their respective companies and shareholders to consummate the strategic business combination transaction provided for in this Agreement.
- B. On the terms and subject to the conditions set forth in this Agreement, Valley w merge with and into CVB (the <u>Merger</u>), with CVB as the surviving corporation Merger (sometimes hereinafter referred to as the <u>Surviving Corporation</u>).
- C. The parties intend that the Merger shall qualify as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended (the <u>Code</u>), and that this Agreement shall constitute a plan of reorganization for p of Sections 354 and 361 of the Code.
- D. As an inducement for CVB to enter into this Agreement, each member of Valley Board of Directors and certain of its executive officers have simultaneously herewisentered into Voting and Support Agreements (the <u>Support Agreements</u>) and Non-Competition and Non-Solicitation Agreements (the <u>Non-Competition Agreements</u>), each dated as of the date hereof and substantially in the forms attachereto as <u>Exhibit A</u> and <u>Exhibit B</u>, with CVB.
- E. The parties desire to make certain representations, warranties and agreements in connection with the Merger and also to prescribe certain conditions to the Merger.

NOW, THEREFORE, in consideration of the mutual covenants, representations, warranties and agreements contained in this Agreement, and intending to be legally bound hereby, the parties agree as follows:

ARTICLE I

MERGER

1.1 The Merger.

(a) Subject to the terms and conditions of this Agreement, in accordance with the California General Corporation Law (the <u>CGC</u>L), at the Effective Time, Valley merge with and into CVB. CVB shall be the Surviving Corporation in the Merger a shall continue its existence under the Laws of the State of California under the name CVB Financial Corp. As of the Effective Time, the separate corporate existence Valley shall cease.

- (b) CVB may at any time change the structure or method of effecting the combinations contemplated by this Agreement; <u>provided</u>, <u>however</u>, that no such change shall, without the consent of Valley, (i) materially alter or change the amout or kind of the Merger Consideration (as defined below) or the amount of the Specia Dividend, (ii) materially adversely affect the tax consequences of the Merger to shareholders of Valley, or (iii) impede or delay in any material respect the receipt of the Requisite Regulatory Approvals or consummation of the transactions contemplated by this Agreement.
- 1.2 <u>Effective Time</u>. Subject to the terms and conditions of this Agreement, on or before the Closing Date, the parties will execute and cause an agreement of merger substantially the form attached hereto as of <u>Exhibit B</u> (the <u>Agreement of Merger filed with the Secretary of State of the State of California (the <u>California Secretar provided in Section 1103 of the CGCL</u>. The Merger shall become effective on the and at the time (the <u>Effective Time</u>) that the Agreement of Merger has been filed the California Secretary.</u>

- 1.3 <u>Effects of the Merger</u>. At and after the Effective Time, the Merger shall have the effects set forth in the applicable provisions of the CGCL.
- 1.4 <u>Conversion of Stock</u>. At the Effective Time, by virtue of the Merger and witho any further action on the part of Valley or CVB or the shareholders of either of the foregoing:
- (a) <u>Valley Common Stock</u>. Each share of Valley Common Stock (other than shares that are owned by shareholders who have perfected and not withdrawn a demand for dissenters—rights pursuant to Chapter 13 of the CGCL (each, a <u>Dissenting Shares</u> collectively <u>Dissenting Shares</u>)) issued and outstanding immediately prior to the Effective Time shall be converted into the right to receive (i) cash in an amount equation to the Per Share Cash Amount, without interest thereon, and (ii) the number of who shares of CVB Common Stock equal to the Per Share Exchange Ratio (rounded up down to the nearest whole share) (such cash and such shares collectively, the <u>Mediconsideration</u>). At the Effective Time, all shares of Valley Common Stock shall alonger be outstanding and shall automatically be cancelled and shall cease to exist.
- (b) <u>Cancellation of Excluded Shares</u>. Any shares of Valley Common Stock owned Valley as treasury stock or owned, directly or indirectly, by Valley, CVB or any of CVB s Subsidiaries (other than those held in a fiduciary capacity or as a result of opreviously contracted) (<u>Treasury Shares</u>), and, subject to Section 1.4(g), any Dissenting Shares, shall automatically be cancelled and retired and shall cease to eat the Effective Time of the Merger and no consideration shall be issued in exchange therefor.
- (c) <u>No Effect on Outstanding CVB Stock</u>. Each share of CVB Common Stock issue and outstanding immediately prior to the Effective Time shall remain an issued and outstanding share of CVB Common Stock and shall not be affected by the Merger.
- (d) For purposes of this Agreement:
- (i) <u>Aggregate Cash Amount</u> means \$23,400,000, minus, without duplication, (1) amount, if any, that the Transaction Costs exceed \$3,500,000 and (2) the aggregate the amounts, if any, that any component(s) of the Transactions Costs exceed(s) the respective amounts set forth in Section 7.2(1) of the Valley Disclosure Schedule.
- (ii) Valley Closing Shares means the aggregate number of shares of Valley Con Stock issued and outstanding immediately prior to the Effective Time. Notwithstanding the foregoing, the Valley Closing Shares shall not exceed to number of shares outstanding on the Valley Capitalization Date, plus any shares of Valley Common Stock issued or issuable pursuant to duly exercised Valley Stock Options outstanding on the date of this Agreement that are disclosed in Section 3.2(a)(ii) of the Valley Disclosure Schedule.
- (iii) <u>CVB Average Closing Price</u> means the daily closing volume-weighted aver price of CVB Common Stock on the Nasdaq Global Select Market (based on regulary trading) for the ten (10) consecutive Trading Days ending on the fifth (45)

Trading Day before the Closing Date.

- (iv) <u>Per Share Cash Amount</u> means the quotient of the Aggregate Cash Amount divided by Valley Closing Shares.
- (v) <u>Per Share Exchange Ratio</u> means the quotient of 1,942,673 shares of CVB Common Stock divided by the Valley Closing Shares.
- (vi) <u>Trading Day</u> means a day that CVB Common Stock is traded on the NASD capital market.
- (e) <u>Effect of Conversion</u>. All of the shares of Valley Common Stock converted into the right to receive the Merger Consideration pursuant to this Article I shall no long be outstanding and shall automatically be

A-2

cancelled and shall cease to exist as of the Effective Time, and each certificate previously representing any such shares of Valley Common Stock (each, a <u>Certificated and each non-certificated share of Valley Common Stock represented by book-entry (each, a <u>Book-Entry Share</u>) shall thereafter represent only the right to receive the Merger Consideration into which the shares of Valley Common Stock represented such Certificate or Book-Entry Share have been converted pursuant to this Section and Section 2.2(f).</u>

(f) <u>Adjustments</u>. If, between the date of this Agreement and the Effective Time, the outstanding shares of CVB Common Stock shall have been increased, decreased, changed into or exchanged for a different number or kind of shares or securities as result of a reorganization, recapitalization, reclassification, stock dividend, stock spreverse stock split or other similar change in capitalization, an appropriate and proportionate adjustment shall be made to the Per Share Exchange Ratio.

(g) Dissenting Shares.

- (i) No later than ten (10) days following the date that the Valley Shareholder Approis received, Valley or the Surviving Corporation shall provide each record holder of Valley Common Stock entitled to vote on the Merger with a notice including the information set forth in Section 1301(a) of the CGCL.
- (ii) Notwithstanding any provision of this Agreement to the contrary, no Dissenting Shares shall be converted into or represent a right to receive the applicable consideration for such shares set forth in this Agreement, if any, but the holder of s Dissenting Shares shall only be entitled to such dissenters—rights as are granted by Chapter 13 of the CGCL. If a holder of shares of Valley Common Stock who dema that Valley purchase such shares of Valley Common Stock under Chapter 13 of the CGCL shall thereafter effectively withdraw or lose (through failure to perfect or otherwise) such holders—dissenters—rights with respect to such shares of Valley Common Stock, then, as of the occurrence of such withdrawal or loss, each such short Valley Common Stock shall be deemed as of the Effective Time to have been converted into and represent only the right to receive, in accordance with Section 1 the Merger Consideration for such shares set forth in this Article I.
- (iii) Valley shall comply in all respects with the provisions of Chapter 13 of the CC with respect to the Dissenting Shares. Valley shall give CVB (A) prompt notice of written demands for purchase of any such shares of Valley Common Stock pursuan Chapter 13 of the CGCL, attempted withdrawals of such demands and any other instruments served pursuant to Chapter 13 of the CGCL and received by Valley in connection therewith and (B) the opportunity to direct all negotiations and proceedings with respect to purchase of any shares of Valley Common Stock under Chapter 13 of the CGCL. Valley shall not voluntarily make any payment with respect only demands for purchase of Valley Common Stock or offer to settle or settle ar such demands without the prior written consent of CVB.

1.5 Valley Stock Options.

- (a) Valley shall take all actions that may be reasonably necessary or that CVB reasonably considers appropriate to provide that (i) each holder of an outstanding option to purchase Valley common stock granted under any Valley Stock Plan (a Valley Stock Option) (whether or not then vested or exercisable) shall be provided with written notice at least 60 days prior to the Option Termination Date pursuant to which all outstanding Valley Stock Options held by such holder shall become fully vested and may be exercised by such holder on or before the Business Day immediately preceding the Closing (the Option Termination Date) and (ii) to the extent that any such Valley Stock Option is not so exercised prior to, and continues be outstanding immediately prior to, the Option Termination Date, such Valley Stock Options shall be cancelled as of the Effective Time without consideration therefore
- (b) For purposes of this Agreement,
- (i) <u>Valley Stock Plans</u> means Valley s 2007 Equity Incentive Plan and Valley Amended and Restated 1997 Stock Option Plan; and

A-3

- (ii) <u>Per Share Merger Consideration Value</u> means the quotient of (x) the sum of Aggregate Cash Amount plus (2) the product of the Per Share Exchange Ratio multiplied by the CVB Average Closing Price multiplied by the Valley Closing Shares, divided by (y) the Valley Closing Shares.
- (c) At least thirty (30) days prior to the Closing Date ,Valley shall obtain a written acknowledgement and waiver (in form and substance reasonably satisfactory to CV from each holder of a Valley Stock Option in existence at such time (i) confirming number of Valley Stock Options held, (ii) electing to exercise the Valley Stock Option or before the Option Termination Date and (iii) confirming the cancelation of so Valley Stock Options if not exercised prior to the Effective Time and (iv) containing waivers for such other matters as reasonably determined by CVB. Valley shall prova copy of each such acknowledgement and waiver to CVB at least ten (10) Business Days prior to the Closing Date.
- (d) Prior to the Effective Time, the Board of Directors of Valley and the Compensation Committee of the Board of Directors of Valley, as applicable, shall adopt any necessary resolutions to effectuate the provisions of Section 1.5(a) (c).
- 1.6 <u>Articles of Incorporation and Bylaws</u>. At the Effective Time, the Articles of Incorporation of CVB, as amended, as then in effect, will be the Articles of Incorporation of the Surviving Corporation, and the Bylaws of CVB, as amended, then in effect, will be the Bylaws of the Surviving Corporation.
- 1.7 <u>Board of Directors and Officers</u>. From and after the Effective Time, the director and officers of CVB shall consist of the persons serving as directors and officers of CVB immediately prior to the Effective Time and such directors and officers shall each case hold office until their respective successors are duly elected and qualified or their earlier death, resignation or removal.
- 1.8 Bank Merger. On the Closing Date and immediately following the Merger, Val Business Bank (<u>VBB</u>), a California state-chartered bank and a wholly-owned Subsidiary of Valley, will merge (the Bank Merger) with and into Citizens Busi Bank, a California state-chartered bank and a wholly-owned Subsidiary of CVB (<u>Citizens Business Bank</u>). The Bank Merger shall become effective at the time specified in Section 4887 of the California Financial Code. The parties shall cause Bank Merger to become effective immediately after the Effective Time. Citizens Business Bank shall be the surviving entity in the Bank Merger and shall continue corporate existence under the name Citizens Business Bank, and at the effective of the Bank Merger, the separate corporate existence of VBB shall cease and outstanding shares of VBB shall be canceled without consideration. The Bank Mer shall be implemented pursuant to an agreement of merger substantially in the form Exhibit D to this Agreement (the Bank Merger Agreement). In order to obtain the necessary state and federal regulatory approvals for the Bank Merger, the parties hereto shall cause the following to be accomplished prior to the filing of applicatio for regulatory approval: (a) Valley shall cause the Bank Merger Agreement to be d executed by VBB and delivered to Citizens Business Bank, (b) CVB shall cause Citizens Business Bank to adopt the Bank Merger Agreement, CVB, as the sole

shareholder of Citizens Business Bank, shall approve the Bank Merger Agreement CVB shall cause the Bank Merger Agreement to be duly executed by Citizens Business Bank and delivered to VBB. Valley shall cause VBB, and CVB shall cause Citizens Business Bank, to execute such an agreement of merger and such other documents and certificates (in each case in form and substance reasonably satisfact to CVB and Valley) as are necessary to make the Bank Merger effective immediate following the Effective Time.

ARTICLE II

EXCHANGE OF SHARES

2.1 <u>Delivery of Merger Consideration and Special Dividend Amount</u>. At or prior to the Effective Time, CVB shall (a) deposit, or cause to be deposited, with an excharagent, which shall be a bank or trust company selected by CVB and reasonably acceptable to Valley (the <u>Exchange Agent</u>), pursuant to an agreement (the

A-4

Exchange Agent Agreement) entered into prior to the Effective Time, shares of Common Stock issuable pursuant to Section 1.4(a), an amount in cash to make the payments pursuant to Sections 1.4(a) and, to the extent then determinable, any cash payable in lieu of fractional shares pursuant to Section 2.2(f). At least three Busine Days prior to the Effective Time, Valley shall deposit with Valley s stock transfer agent an amount in cash equal to the aggregate amount of the Special Dividend wit instructions to pay the Special Dividend to holders of Valley Common Stock in accordance with Section 6.8.

2.2 Exchange Procedures.

- (a) As soon as reasonably practicable after the Effective Time, the Exchange Agen shall mail to each holder of record of Certificate(s) or Book-Entry Shares which, immediately prior to the Effective Time, represented outstanding shares of Valley Common Stock whose shares were converted into the right to receive the Merger Consideration pursuant to Section 1.4 (<u>Exchanged Shares</u>), (i) a letter of transmit (the contents of which shall be reasonably acceptable to Valley and which shall specify that delivery shall be effected, and risk of loss and title to Certificate(s) or Book-Entry Shares shall pass, only upon delivery of Certificate(s) (or affidavits of loss in lieu of such Certificate(s))) or Book-Entry Shares to the Exchange Agent and shall be substantially in such form and have such other provisions as shall be prescribed by the Exchange Agent Agreement (the <u>Letter of Transmittal</u>) and (ii) instructions for use in surrendering Certificate(s) or Book-Entry Shares in exchange for the Merger Consideration and any dividends or distributions to which such holder is entitled pursuant to Section 2.2(c).
- (b) Upon surrender to the Exchange Agent of its Certificate(s) or Book-Entry Sharaccompanied by a properly completed Letter of Transmittal, a holder of Exchanged Shares will be entitled to receive promptly after such surrender in accordance with Exchange Agent s customary practice, the Merger Consideration in respect of the Exchanged Shares, including any cash in lieu of fractional shares of CVB Common Stock, represented by its Certificate(s) or Book-Entry Shares and any amounts pay pursuant to Section 2.2(c). Until so surrendered, each such Certificate or Book-Entry Shares shall represent after the Effective Time, for all purposes, only the right to receive, without interest, the Merger Consideration and any cash in lieu of fraction shares of CVB Common Stock to be issued or paid in consideration thereof upon surrender of such Certificate or Book-Entry Shares in accordance with, and any dividends or distributions to which such holder is entitled pursuant to, this Article I
- (c) No dividends or other distributions with respect to CVB Common Stock shall be paid to the holder of any unsurrendered Certificate or Book-Entry Shares with respect to the shares of CVB Common Stock represented thereby, in each case unless and until the surrender of such Certificate or Book-Entry Share in accordance with this Article II. Subject to the effect of applicable abandoned property, escheat or similar Laws, following surrender of any such Certificate or Book-Entry Share in accordance with this Article II, the record holder thereof shall be entitled to receive, without interest, (i) the amount of dividends or other distributions with a record date after the Effective Time theretofore payable with respect to the whole shares of CVB Common Stock respect to the control of the shares of CVB Common Stock respect to the control of the shares of CVB Common Stock represented the respect to the case unless and until the surrender of such Certificate or Book-Entry Share in accordance with this Article II, the record holder thereof shall be entitled to receive, without interest, (i) the amount of dividends or other distributions with a record date after the Effective Time theretofore payable with respect to the whole shares of CVB Common Stock represented to the case unless and until the surrender of such case unless an

Stock represented by such Certificate or Book-Entry Share and paid prior to such surrender date, and/or (ii) at the appropriate payment date, the amount of dividends other distributions payable with respect to shares of CVB Common Stock represent by such Certificate or Book-Entry Shares with a record date after the Effective Tim (but before such surrender date) and with a payment date subsequent to the issuance the CVB Common Stock issuable with respect to such Certificate or Book-Entry Shares.

(d) In the event of a transfer of ownership of a Certificate or Book-Entry Shares representing Exchanged Shares that are not registered in the stock transfer records Valley, the shares of CVB Common Stock, the Per Share Cash Amount plus any cain lieu of fractional shares of CVB Common Stock with respect to such Exchanged Shares shall be issued or paid in exchange therefor to a Person other than the Person whose name the Certificate or Book-Entry Shares so surrendered is registered if the Certificate or Book-Entry Shares formerly representing such Exchanged Shares shabe properly endorsed or otherwise be in proper form for transfer and the Person requesting such payment or issuance shall pay any transfer or other similar taxes required

A-5

by reason of the payment or issuance to a Person other than the registered holder of the Certificate or Book-Entry Shares, or establish to the reasonable satisfaction of CVB that the tax has been paid or is not applicable. The Exchange Agent (or, subsequent to the earlier of (x) the twelve (12) month anniversary of the Effective Time and (y) the expiration or termination of the Exchange Agent Agreement, CVI the Surviving Corporation) shall be entitled to deduct and withhold from any cash otherwise payable pursuant to this Agreement to any holder of Exchanged Shares's amounts as the Exchange Agent, CVB or the Surviving Corporation, as the case may be, is required to deduct and withhold under the Code, or any provision of state, los or foreign Tax Law, with respect to the making of such payment. To the extent the amounts are so withheld by the Exchange Agent, CVB or the Surviving Corporation as the case may be, and timely paid over to the appropriate Governmental Entity su withheld amounts shall be treated for all purposes of this Agreement as having been paid to the holder of Exchanged Shares in respect of whom such deduction and withholding was made by the Exchange Agent or CVB, as the case may be.

- (e) After the Effective Time, there shall be no transfers on the stock transfer books Valley of the shares of Valley Common Stock that were issued and outstanding immediately prior to the Effective Time other than to settle transfers of such Valley Common Stock that occurred prior to the Effective Time. If, after the Effective Time Certificates or Book-Entry Shares representing any such shares of Valley Common Stock are presented for transfer to the Exchange Agent, they shall be cancelled and exchanged for the applicable Merger Consideration and any cash in lieu of fraction shares of CVB Common Stock to be issued or paid in consideration therefor in accordance with the procedures set forth in this Article II.
- (f) Notwithstanding anything to the contrary contained in this Agreement, no fractional shares of CVB Common Stock shall be issued upon the surrender of Certificates or Book-Entry Shares for exchange, no dividend or distribution with respect to CVB Common Stock shall be payable on or with respect to any fractional share, and such fractional share interests shall not entitle the owner thereof to vote to any other rights of a shareholder of CVB. In lieu of the issuance of any such fractional share, CVB shall pay to each former shareholder of Valley who otherwise would be entitled to receive such fractional share an amount in cash (rounded to the nearest cent) determined by multiplying (i) the CVB Average Closing Price by (ii) fraction of a share (after taking into account all shares of Valley Common Stock he by such holder at the Effective Time and rounded to the nearest thousandth when expressed in decimal form) of CVB Common Stock to which such holder would otherwise be entitled to receive pursuant to Section 1.4.
- (g) Any portion of the Merger Consideration or the cash for the Special Dividend deposited with the Exchange Agent that remains unclaimed by the shareholders of Valley as of the twelve (12) month anniversary of the Effective Time will be transferred to CVB. In such event, any former shareholders of Valley who have no theretofore complied with this Article II shall thereafter look only to CVB with respect to the Merger Consideration, any cash in lieu of any fractional shares, and a unpaid dividends and distributions on the CVB Common Stock deliverable in respect of each share of Valley Common Stock such shareholder holds as determined pursu

to this Agreement, in each case, without any interest thereon. Notwithstanding the foregoing, none of CVB, the Surviving Corporation, the Exchange Agent or any ot Person shall be liable to any former holder of shares of Valley Common Stock for amount delivered in good faith to a public official pursuant to applicable abandone property, escheat or similar Laws.

(h) In the event that any Certificate shall have been lost, stolen or destroyed, upon making of an affidavit of that fact by the Person claiming such Certificate to be lost stolen or destroyed and, if reasonably required by CVB or the Exchange Agent, the posting by such Person of a bond in such amount as CVB may determine is reasonably necessary as indemnity against any claim that may be made against it we respect to such Certificate, the Exchange Agent will issue in exchange for such lost stolen or destroyed Certificate the applicable Merger Consideration deliverable in respect thereof pursuant to this Agreement.

A-6

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF VALLEY

Except as Previously Disclosed, Valley hereby represents and warrants to CVB as follows:

3.1 Corporate Organization.

- (a) <u>Organization.</u> Valley is a corporation duly incorporated, validly existing and in good standing under the Laws of the State of California. Valley has the requisite corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted and, except as would not reasonabe expected to individually or in the aggregate have a Material Adverse Effect on Valley, is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties assets owned or leased by it makes such licensing or qualification necessary. Valled duly registered with the Board of Governors of the Federal Reserve System (the Federal Reserve) as a bank holding company under the Bank Holding Company of 1956, as amended (the BHC Act).
- (b) <u>Articles and Bylaws.</u> True, complete and correct copies of the Articles of Incorporation of Valley (the <u>Valley Articles</u>), and the Bylaws of Valley (the <u>Valley Articles</u>), as in effect as of the date of this Agreement, have been provided to CVI The Valley Articles and the Valley Bylaws provided to CVB are in full force and effect. True, complete and correct copies of the Articles of Incorporation, Bylaws of other similar charter documents of each Subsidiary of Valley have been have been provided to CVB and are in full force and effect.

(c) Subsidiaries.

(i) Section 3.1(c) to the Valley Disclosure Schedule sets forth a true and complete of all Subsidiaries of Valley as of the date hereof and any material joint ventures, formal partnerships or similar arrangements in which Valley or any of its Subsidian has a limited liability, partnership or other equity interest (and the amount and percentage of any such interest). Section 3.1(c) to the Valley Disclosure Schedule sets forth a list identifying the number and owner of all outstanding capital stock of other equity securities of each Subsidiary of Valley, options, warrants, stock appreciation rights, scrip, rights to subscribe to, calls or commitments of any chara whatsoever relating to, or securities or rights convertible into, shares of any capital stock or other equity securities of such Subsidiary, or agreements by which such Subsidiary may become bound to issue additional shares of its capital stock or other equity securities, or options, warrants, scrip, rights to subscribe to, calls or commitments for any shares of its capital stock or other equity securities and the identity of the parties to any such agreements. Valley does not, directly or indirectly beneficially own any equity securities or similar interests of any entity or any interin a partnership or joint venture of any kind other than the Subsidiaries. There is no person whose results of operations, cash flows, changes in shareholders equity or

financial position are consolidated in the financial statements of Valley other than to Subsidiaries listed in Section 3.1(c) of the Valley Disclosure Schedule.

(ii) Each of Valley s Subsidiaries is a corporation duly incorporated, validly existi and in good standing under the Laws of the states of its incorporation. Each of Val Subsidiaries has the requisite corporate power and authority to own or lease all of i properties and assets and to carry on its business as it is now being conducted and, except as would not reasonably be expected to individually or in the aggregate hav Material Adverse Effect on Valley, is duly licensed or qualified to do business in e jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary. VBB is duly authorized by the Commissioner of the California Department of Business Oversight (the <u>Commissioner</u>) to conduct the business of a commercial bank under the California Financial Code. The deposit accounts of VBB are insured by the Federal Deposit Insurance Corporation (the <u>FDIC</u>) through the Deposit Insurance Fund to the fullest extent permitted by La premiums and assessments required to be paid in connection therewith have been p when due, and no proceedings for the termination of such insurance are pending or the Knowledge of Valley, threatened.

A-7

- (iii) There are no restrictions on the ability of any Subsidiary of Valley to pay dividends or distributions except, in the case of a Subsidiary that is a regulated entifor restrictions on dividends or distributions generally applicable to all such regula entities.
- (iv) As used in this Agreement, the term <u>Subsidiary</u> has the meaning ascribed to Section 2(d) of the BHC Act, except that when such term is used with respect to an entity that is not a bank holding company, the meaning shall nonetheless be deeme apply to such entity.

3.2 Capitalization.

- (a) The authorized capital stock of Valley consists of 30,000,000 shares of commo stock (the <u>Valley Common Stock</u>) and 10,000,000 shares of preferred stock, of 30,000 shares have been designated as Series A Junior Participant Preferred Stock.
- (i) As of, September 21, 2016 (the <u>Valley Capitalization Date</u>), (A) 2,870,977 sl of Valley Common Stock were issued and outstanding and (B) no shares of Valley preferred stock were issued or outstanding.
- (ii) As of the Valley Capitalization Date, no shares of Valley Common Stock were reserved for issuance, except in connection with awards under the Valley Stock Plato purchase no more than 384,570 shares of Valley Common Stock, of which (A) 131,037 shares were reserved and available for issuance upon exercise of outstandivalley Stock Options and (B) 253,533 shares of Valley Common Stock were reserved available for issuance pursuant to future awards under the Valley Stock Plans. shares of Valley Common Stock issued as restricted stock awards under any of the Valley Stock Plans and subject to forfeiture (Valley Stock Awards) are issued o outstanding. As of the Valley Capitalization Date, no shares of Valley s preferred were designated or reserved for issuance other than 30,000 shares of preferred stock designated as Series A Junior Participant Preferred Stock.
- (iii) All of the issued and outstanding shares of Valley Common Stock and all of the capital stock of each Subsidiary of Valley have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights, with no personal liability attaching to the ownership thereof. Other than a cash dividend of \$0.10 per share of Valley Common Stock declared by the Valley Board of Director on August 16, 2016, which cash dividend has a record date of September 6, 2016 a an ex-dividend date of September 2, 2016, there are no outstanding dividends, whe current or accumulated, due or payable on any of the capital stock of Valley. As of date of this Agreement, no bonds, debentures, notes or other indebtedness having t right to vote on any matters on which shareholders of Valley may vote (<u>Voting D</u> are issued or outstanding. There are no contractual obligations of Valley or any Subsidiary of Valley (1) to repurchase, redeem or otherwise acquire any shares of capital stock of Valley or any equity security of Valley or any securities representing the right to purchase or otherwise receive any shares of capital stock or any other equity security of Valley or (2) pursuant to which Valley is or could be required to register shares of Valley capital stock or other securities under the Securities Act o

1933, as amended (the <u>Securities Act</u>). Except for the Support Agreements, ther no voting trusts or other agreements or understandings to which Valley, or, to the Knowledge of Valley, any of their respective officers or directors, is a party with respect to the voting of any Valley Common Stock, Voting Debt or other equity securities of Valley. Except pursuant to this Agreement, and Valley Stock Options. Valley does not have and is not bound by any outstanding subscriptions, options, warrants, calls, rights, commitments or agreements of any character calling for the purchase or issuance of any shares of the capital stock of Valley, Voting Debt of Valley or any other equity securities of Valley. Section 3.2(a) of the Valley Disclosure of Valley Discourse of Valley Discour Schedule sets forth a true and complete list of all Valley Stock Options, Valley Sto Awards or any other equity-based awards, outstanding as of the Valley Capitalizati Date, specifying on a holder-by-holder basis (A) the name of such holder, (B) the number of shares subject to each such award, or the number of Common Stock Options held by such holder, (C) as applicable, the grant date of each such award, as applicable, the vesting schedule of each such award, and (E) the exercise price f each such Valley Stock Option.

A-8

(b) Other than the Valley Stock Options issued under the Valley Stock Plans that a outstanding as of the Valley Capitalization Date and listed in Section 3.2(a) of the Valley Disclosure Schedule, no other equity-based awards or rights are outstanding of the Valley Capitalization Date. Since the Valley Capitalization Date through the date hereof, Valley has not (i) issued or repurchased any shares of Valley Commor Stock, Voting Debt or other equity securities of Valley, other than in connection w exercise of Valley Stock Options in accordance with their terms that were outstand on the Valley Capitalization Date or (ii) issued or awarded any options, stock appreciation rights, restricted shares, restricted stock units, deferred equity units, awards based on the value of Valley capital stock or any other equity-based awards With respect to each grant of Valley Stock Options, (1) each such grant was made accordance with the terms of the applicable Valley Stock Plans and all applicable Laws and (2) each such grant was properly accounted for in accordance with gener accepted accounting principles in the United States applied to banks or banking holding companies for the applicable period(s) (<u>GAAP</u>) in the financial stateme (including the related notes) of Valley in accordance with all applicable Laws. All Valley Stock Options granted by Valley or any of its Subsidiaries have been granted with a per share exercise or reference price at least equal to the fair market value or the underlying stock on the date the option was granted, within the meaning of Sec 409A of the Code and associated U.S. Department of the Treasury guidance, and e Valley Stock Option has a grant date identical to the date on which the Board of Directors of Valley or compensation committee of the Board of Directors of Valley actually awarded such option. From January 1, 2015 through the date of this Agreement, except as set forth in Section 3.2(b) of the Valley Disclosure Schedule Valley has not (A) accelerated the vesting of or lapsing of restrictions with respect any stock-based compensation awards or long-term incentive compensation awards (B) with respect to executive officers of Valley, entered into or amended any employment, severance, change of control or similar agreement (including any agreement providing for the reimbursement of excise taxes under Section 4999 of Code) or (C) adopted or amended any stock bonus or compensation plan.

3.3 Authority; No Violation.

(a) Valley has full corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated hereby. VBB has ful corporate power and authority to execute and deliver the Bank Merger Agreement to consummate the transactions contemplated thereby. The execution and delivery this Agreement and the consummation of the transactions contemplated hereby have been duly and validly approved and this Agreement duly adopted unanimously by Board of Directors of Valley. As of the date of this Agreement, the Board of Director of Valley has unanimously determined that the Merger, on the terms and condition set forth in this Agreement, is in the best interests of Valley and its shareholders are has directed that this Agreement and the transactions contemplated hereby be submitted to Valley s shareholders for approval at a duly held meeting of such shareholders and has adopted a resolution to the foregoing effect. The execution are delivery of the Bank Merger Agreement and the consummation of the transactions contemplated hereby have been duly and validly approved and the Bank Merger Agreement has been duly adopted unanimously by the Board of Directors of VBB

by Valley as the sole shareholder of VBB. Except for the approval of this Agreement and the transactions contemplated hereby by the affirmative vote of a majority of a the outstanding Valley Common Stock (the <u>Valley Shareholder Approval</u>) no of corporate proceedings on the part of Valley or VBB are necessary to approve this Agreement or to consummate the Merger, the Bank Merger or the other transaction contemplated hereby. This Agreement has been duly and validly executed and delivered by Valley and (assuming due authorization, execution and delivery by Constitutes the valid and binding obligation of Valley, enforceable against Valley in accordance with its terms (except as may be limited by bankruptcy, insolvency, fraudulent transfer, moratorium, reorganization or similar Laws of general applicability relating to or affecting the rights of creditors generally and subject to general principles of equity and Section 8(b)(6)(D) of the Federal Deposit Insurance Act, 12 U.S.C. Section 1818(b)(6)(D) (as applicable) (the <u>Enforceability Exception</u>

(b) Except as set forth in Section 3.3(b) of the Valley Disclosure Schedule, neither execution and delivery of this Agreement by Valley, nor the consummation by Val as the case may be, of the Merger, or the other transactions contemplated hereby, n compliance by Valley with any of the terms or provisions of this

A-9

Agreement, nor the consummation by VBB of the Bank Merger, will (i) violate an provision of Valley s or VBB s Articles of Incorporation or Bylaws, or (ii) assum that the consents, approvals and filings referred to in Section 3.4 are duly obtained and/or made, (A) violate any law, statute, rule, regulation, judgment, order, injunct or decree issued, promulgated or entered into by or with any Governmental Entity (each, a <u>Law</u>) applicable to Valley or any of its Subsidiaries, or any of its prope assets or (B) violate, conflict with, result in a breach of any provision of or the loss any benefit under, constitute a default (or an event that, with notice or lapse of time both, would constitute a default) under, result in the termination of or a right of termination or cancellation under, accelerate the performance required by, or result the creation of any liens, pledges, charges, claims and security interests and similar encumbrances (<u>Lien</u>s), upon any of the respective properties or assets of Valley VBB under, any of the terms, conditions or provisions of any note, bond, mortgage indenture, deed of trust, license, Lease, franchise, permit, agreement, bylaw or other instrument or obligation to which Valley or VBB is a party or by which it or any or properties or assets is bound.

3.4 Consents and Approvals. Except for (a) the filing by CVB with the Securities a Exchange Commission (<u>SEC</u>) of a registration statement on Form S-4 (or such of applicable form) (the <u>Form S-4</u>) that includes a proxy statement in definitive for relating to the meeting of Valley s shareholders to be held in connection with this Agreement and the transactions contemplated by this Agreement (the <u>Proxy</u> Statement), and declaration of effectiveness of the Form S-4, (b) filings of applications and notices with the Commissioner pursuant to the California Financia Code and approval of or non-objection to such applications, filings and notices wit respect to the Bank Merger, (c) the filing of a bank merger application with the FD pursuant to the Bank Merger Act of 1960, as amended, (d) the filing of the Agreem of Merger with the California Secretary, (e) the filing of the Bank Merger Agreement with the California Secretary and the Commissioner, (f) Valley s filing of a notice concerning the Merger with FINRA, (g) such filings and approvals as are required be made or obtained under the securities or blue sky Laws of various states in connection with the issuance of the shares of CVB Common Stock pursuant to the Agreement, and (h) the filing by CVB with the Federal Reserve under the BHC Ac for such approvals, waiver or non-objection as may be necessary for CVB to acqui Valley, no consents or approvals of or filings or registrations with any foreign, federal or state banking or other regulatory, self-regulatory or enforcement authorities or a courts, administrative agencies or commissions or other governmental authorities of instrumentalities (each a Governmental Entity), are necessary in connection wit consummation by Valley of the Merger or VBB of the Bank Merger and the other transactions contemplated by this Agreement.

3.5 Reports.

(a) Each of Valley and VBB have timely filed all reports, registrations, statements certifications, together with any amendments required to be made with respect ther (collectively, <u>Valley Filings</u>) that they were required to file since January 1, 201 (i) the Federal Reserve, (ii) the FDIC, (iii) the Commissioner and any other state banking or other state regulatory authority, (iv) the U.S. Small Business

Administration, (v) the SEC, (vi) any other federal, state or foreign regulatory authority and (vii) any applicable industry self-regulatory organizations (collective Regulatory Agencies) and with each other applicable Governmental Entity, and other reports and statements required to be filed by them since January 1, 2013, including any report or statement required to be filed pursuant to the Laws, rules or regulations of the United States, any state, any foreign entity, or any Regulatory Agency or other Governmental Entity, have paid all fees and assessments due and payable in connection therewith, and there are no violations or exceptions in any surport or statement that are unresolved as of the date hereof. As of their respective dates, each of such Valley Filings (i) complied in all material respects with all Law and regulations enforced or promulgated by the Governmental Entity with which it was filed (or was amended so as to be in compliance promptly following discovery any such noncompliance) and (ii) did not contain any untrue statement of a material fact. Valley has made available to CVB true and correct copies of all such Valley Filings.

(b) Valley is in compliance in all material respects with the applicable listing and corporate governance rules and regulations of the FINRA with respect to the quota of Valley Common Stock on the OTC Pink.

A-10

3.6 Financial Statements; Controls.

- (a) Valley has delivered to CVB correct and complete copies of (i) the audited consolidated balance sheets of Valley as of December 31, 2013, 2014 and 2015, an the related audited consolidated statements of income, shareholders equity and ca flows for the years ended December 31, 2013, 2014 and 2015 (Valley Audited Financial Statements) and (ii) an unaudited balance sheet of Valley as of August 2016, and the related unaudited statements of income, shareholders equity and ca flows for the eight months ended August 31, 2016 (the Valley Interim Financial Statements and, together with Valley Audited Financial Statement, the Valley <u>Financial Statements</u>). Valley has also delivered to CVB true, correct and comple copies of each management letter or other letter delivered to Valley by Crowe Horwath LLP in connection with Valley Audited Financial Statements or relating t any review of the internal controls of Valley since January 1, 2013. The Valley Financial Statements (i) fairly present in all material respects the consolidated financial condition of Valley, as of the respective dates indicated and its results of operations and statements of cash flows, for the respective periods then ended, sub in the case of the Valley Interim Financial Statements, to normal recurring adjustments; (ii) have been prepared in accordance with GAAP and/or applicable regulatory accounting principles or banking regulations consistently applied (except otherwise indicated therein); (iii) set forth as of the respective dates indicated adeq reserves for loan losses and other contingencies; and (iv) are based upon the books and records of Valley. The books and records of Valley and each of its Subsidiarie have been maintained in all material respects in accordance with GAAP and any of applicable legal and accounting requirements and reflect only actual transactions. A of the date hereof, Crowe Horwath LLP has not resigned (or informed Valley that intends to resign) or been dismissed as independent public accountants of Valley a result of or in connection with any disagreements with Valley on a matter of accounting principles or practices, financial statement disclosure or auditing scope procedure.
- (b) Neither Valley nor any of its Subsidiaries has incurred any liability or obligatio any nature whatsoever (whether absolute, accrued, contingent, determined, determinable or otherwise and whether due or to become due), except for (i) those liabilities that are reflected or reserved against on the consolidated balance sheet of Valley included in the Valley Financial Statements (including any notes thereto), (i liabilities incurred in the ordinary course of business consistent with past practice since June 30, 2016 which have been Previously Disclosed or (iii) in connection withis Agreement and the transactions contemplated hereby. Except for those liabiliti described in subsections (i) through (iii) of the immediately preceding sentence, the is no basis for the asserting against Valley or any liability, obligation or claim that may reasonably be expected to have a Material Adverse Effect.
- (c) Since January 1, 2013, (i) neither Valley nor any of its Subsidiaries, nor to the Knowledge of Valley, any director, officer, employee, auditor, accountant or representative of Valley or any of its Subsidiaries, has received or otherwise obtain knowledge of, any complaint, allegation, assertion or claim, whether written or ora regarding the accounting or auditing practices, procedures, methodologies or methodologies.

of Valley or any of its Subsidiaries or their respective internal accounting controls relating to periods after January 1, 2013, including any complaint, allegation, asser or claim that Valley has engaged in questionable accounting or auditing practices, a (ii) no attorney representing Valley or any of its Subsidiaries, whether or not employed by Valley or any of its Subsidiaries, has reported evidence of a violation securities Laws, breach of fiduciary duty or similar violation relating to periods aft January 1, 2013, by Valley or any of its Subsidiaries, or any of their respective officers, directors, employees or agents to the Valley Board of Directors or the VB Board of Directors or any respective committee thereof or to any director or officer Valley or any of its Subsidiaries.

(d) Neither Valley nor any of its Subsidiaries is a party to, or has any commitment become a party to, any joint venture, off-balance sheet partnership or any similar contract or arrangement (including any contract or arrangement relating to any transaction or relationship between or between Valley or any of its Subsidiaries on one hand, and any unconsolidated Affiliate, including any structured finance, speci purpose or limited purpose entity or Person, on the other hand, or any off-balance sheet arrangement), where the result, purpose or intended effect of such contract of arrangement is to avoid disclosure of any material transaction involving, or

A-11

material liabilities of, Valley in Valley Audited Financial Statements or Valley
Unaudited Financial Statements. For purposes of this Agreement, <u>Affiliate</u> and
<u>Affiliate</u>d mean, with respect to any Person, any other Person that directly or
indirectly controls, is controlled by or is under common control with such first Pers
For the purposes of this definition, control (including, the terms controlling,
by and under common control with), as applied to any Person, means the posse
directly or indirectly of the power to direct or cause the direction of the management
and policies of that Person, whether through the ownership of voting securities by
contract or otherwise.

- 3.7 <u>Broker s Fees</u>. Neither Valley, VBB nor any of their respective officers, direct employees or agents has utilized any broker, finder or financial advisor or incurred any liability for any broker s fees, commissions or finder s fees in connection wit Merger, the Bank Merger or any other transactions contemplated by this Agreement
- 3.8 Absence of Changes. Except as set forth in Section 3.8 of the Valley Disclosure Schedule, since December 31, 2015: (1) Valley and each of its Subsidiaries have each conducted its respective business only in the ordinary and usual course of its busin-(2) no damage, destruction or other casualty loss (whether or not covered by insurance) that may involve a loss of more than \$100,000 has been experienced by Valley or any of its Subsidiaries; (3) no direct or indirect redemption, purchase or other acquisition by Valley or any of its Subsidiaries of any equity securities or any declaration, setting aside or payment of any dividend or other distribution on or in respect of its respective capital stock, whether consisting of money other personal property, real property or other things of value; (4) neither Valley nor any of its Subsidiaries has taken any of the actions that Valley has agreed not to take or perm the Valley Subsidiaries to take from the date hereof through the Effective Time pursuant to Section 5.2; (5) and no event, change or development or combination of changes or developments has occurred that have had or would reasonably be expec to have either individually or in the aggregate, a Material Adverse Effect on Valley As used in this Agreement, the term <u>Material Adverse Effect</u> means, with respec any party, a material adverse effect on (a) the business, assets or deposit liabilities, properties, operations, condition (financial or otherwise) or results of operations of such party and its Subsidiaries taken as a whole; <u>provided</u>, <u>however</u>, that, with resp to this clause (a), a Material Adverse Effect shall not be deemed to include effect arising out of, relating to or resulting from (A) changes after the date hereof in applicable GAAP or regulatory accounting requirements generally affecting other companies in the banking industries in which such party and its Subsidiaries opera-(B) changes after the date hereof in Laws general applicability to companies of sim size in the banking industries in which such party and its Subsidiaries operate, (C) changes after the date hereof in global, national or regional political conditions or general economic or market conditions (including changes in prevailing interest ra credit availability and liquidity, currency exchange rates, and price levels or trading volumes in the United States or foreign securities markets) affecting other compan in the banking industries in which such party and its Subsidiaries operate, (D) chan after the date hereof in the credit markets, any downgrades in the credit markets, or adverse credit events resulting in deterioration in the credit markets generally and including changes to any previously correctly applied asset marks resulting therefre

(E) a decline in the trading price of CVB Common Stock or a failure, in and of itse to meet earnings projections, but not, in either case, including any underlying cause thereof, (F) any outbreak or escalation of hostilities, or declared or undeclared acts war or terrorism, or (G) actions or omissions taken with the prior written consent of the other party or expressly required by this Agreement, except that effects attribute to or resulting from any of the changes, events, conditions or trends described in clauses (A), (B), (C), (D) and (F) shall not be excluded to the extent of any disproportionate impact they have on such party and its Subsidiaries, taken as a whas compared to other companies in the banking industry in which such party and its Subsidiaries operate; or (b) the ability of such party to consummate the transactions contemplated by this Agreement on a timely basis.

3.9 Compliance with Applicable Law.

(a) Except as set forth in Section 3.9 of the Valley Disclosure Schedule, each of Valley and each of its Subsidiaries holds, and has at all times since December 31, 2013 held, all licenses, franchises, permits and

A-12

authorizations which are necessary for the lawful conduct of their respective businesses and ownership of their respective properties, rights and assets under and pursuant to applicable Law (and have paid all fees and assessments due and payable connection therewith) and, to the Knowledge of Valley, no suspension or cancellat of any such necessary license, franchise, permit or authorization is threatened. Exc as set forth in Section 3.9 of the Valley Disclosure Schedule, Valley and each of its Subsidiaries has complied in all material respects with, and is not in material defau or violation of, (i) any applicable Law, including all Laws related to data protection privacy, the USA PATRIOT Act, the Bank Secrecy Act, the Equal Credit Opportunity Act and Regulation B, the Fair Housing Act, the Community Reinvestment Act, th Fair Credit Reporting Act, the Truth in Lending Act and Regulation Z, the Home Mortgage Disclosure Act, the Fair Debt Collection Practices Act, the Electronic Fu Transfer Act, the Dodd-Frank Wall Street Reform and Consumer Protection Act, a regulations promulgated by the Consumer Financial Protection Bureau, the U.S. Si Business Administration, the Interagency Policy Statement on Retail Sales of Nondeposit Investment Products, the SAFE Mortgage Licensing Act of 2008, the I Estate Settlement Procedures Act and Regulation X, and any other Law relating to bank secrecy, discriminatory lending, financing or leasing practices, money laundering prevention, Sections 23A and 23B of the Federal Reserve Act and Regulation W, the Sarbanes-Oxley Act of 2002 (the <u>Sarbanes-Oxley Act</u>), and a agency requirements relating to the origination, sale and servicing of loans, and (ii) any posted or internal privacy policies relating to data protection or privacy, including the protection of personal information, and neither Valley nor any of its Subsidiarie has any Knowledge of, nor have any of them received from a Governmental Entity since January 1, 2012, notice of, any material defaults or violations of any applications Law relating to Valley or any of its Subsidiaries.

- (b) Valley and each of its Subsidiaries has properly administered in all material respects all accounts for which it acts as a fiduciary, including accounts for which it serves as a trustee, agent, custodian, personal representative, guardian, conservator investment advisor, in accordance with the terms of the governing documents and applicable Law. None of Valley or any of its Subsidiaries, or any director, officer of employee of Valley or any of its Subsidiaries, has committed any breach of trust or fiduciary duty with respect to any such fiduciary account, and the accountings for exact fiduciary account are true and correct and accurately reflect the assets of such fiduciary account.
- (c) Neither Valley nor any of its Subsidiaries is subject to any cease-and-desist ord or enforcement action issued by, nor is Valley or any of its Subsidiaries a party to a written agreement, or consent agreement or memorandum of understanding with, or party to any commitment letter or similar undertaking with, or subject to any capital directive by, or since January 1, 2013 adopted any board resolutions at the request any Governmental Entity (each a Regulatory Agreement), nor has Valley or any Subsidiaries been advised since January 1, 2013 and prior to the date hereof by any Governmental Entity that it is considering issuing, initiating, ordering or requesting any such Regulatory Agreement.

(d) Neither Valley nor any of its Subsidiaries, nor to the Knowledge of Valley, any their respective directors, officers, agents, employees or any other Persons acting of their behalf, (i) has violated the Foreign Corrupt Practices Act, 15 U.S.C. § 78dd-1 seq., as amended, or any other similar applicable foreign, federal or state legal requirement, (ii) has made or provided, or caused to be made or provided, directly indirectly, any payment or thing of value to a foreign official, foreign political part candidate for office or any other Person while knowing or having a reasonable beli that the Person will pay or offer to pay the foreign official, party or candidate, for t purpose of influencing a decision, inducing an official to violate their lawful duty, securing an improper advantage, or inducing a foreign official to use their influence affect a governmental decision, (iii) has paid, accepted or received any unlawful contributions, payments, expenditures or gifts, (iv) has violated or operated in noncompliance with any export restrictions, money laundering Law, anti-terrorism Law or regulation, anti-boycott regulations or embargo regulations or (v) is current subject to any United States sanctions administered by the Office of Foreign Asset Control of the U.S. Department of the Treasury.

A-13

3.10 State Takeover Laws; Rights Agreement.

- (a) No business combination, fair price, affiliate transaction, moratorium share, takeover or interested shareholder Law or other similar anti-takeover regulation (collectively, the <u>Takeover Laws</u>) is applicable to this Agreement or transactions contemplated hereby.
- (b) Neither Valley nor any of its Subsidiaries has any shareholder rights plan, poi pill or similar plan or arrangement in effect.

3.11 Employee Benefit Plans.

- (a) Section 3.11(a) of the Valley Disclosure Schedule sets forth a true, complete an correct list of each employee benefit plan, program, policy, practice, or other arrangement providing benefits to any current or former employee, officer or direc of Valley or any of its Subsidiaries or any beneficiary or dependent thereof that is sponsored or maintained by Valley or any of its Subsidiaries to which Valley or an its Subsidiaries contributes or is obligated to contribute on the date hereof, whether not written, including, without limitation, any employee welfare benefit plan within the meaning of Section 3(1) of the Employee Retirement Income Security Act of 1974, as amended (<u>ERISA</u>), any employee pension benefit plan within the mean Section 3(2) of ERISA (whether or not such plan is subject to ERISA) and any equ purchase plan, option, equity bonus, phantom equity or other equity plan, profit sharing, bonus, retirement (including compensation, pension, health, medical or lif insurance benefits), supplemental retirement, deferred compensation, excess benefit incentive compensation, severance, change in control or termination pay, agreemen providing for Change of Control Payments, hospitalization or other medical or den life or other insurance (including any self-insured arrangements), supplemental unemployment, salary continuation, sick leave or other leave of absence benefits, short- or long-term disability, or vacation benefits plan or any other agreement or policy or other arrangement providing employee benefits, employment-related compensation, fringe benefits or other benefits (whether qualified or nonqualified, funded or unfunded) (each an Employee Benefit Plan).
- (b) With respect to each Employee Benefit Plan, Valley has delivered or made available to CVB a true, correct and complete copy of: (i) each writing constituting part of such Employee Benefit Plan, including, without limitation, all plan docume benefit schedules, contracts and notices, and a written summary of any such Emplo Benefit Plan that is not otherwise documented in writing; (ii) the most recent Annu Report (Form 5500 Series) and accompanying schedule; (iii) all investment policy statements or guidelines, delegations and charters related to any Employee Benefit Plan; (iv) each trust agreement, insurance contract, group annuity contract or other funding mechanism; (v) each agreement with any services provider, investment manager or investment advisor; (vi) the current summary plan description and any material modifications thereto; (vii) the most recent annual financial report; (viii) the most recent actuarial report; (viii) the most recent determination letter from the IRS; and (x) all correspondence with the IRS, the U.S. Department of Labor, the Pensio Benefit Guaranty Corporation or any other Governmental Entity. Except as

specifically provided in the foregoing documents delivered or made available to CY and, except as provided in this Agreement, there are no amendments to any Employ Benefit Plan that have been adopted or approved nor has Valley or any of its Subsidiaries undertaken to make any such amendments or to adopt or approve any new Employee Benefit Plan. No Employee Benefit Plan is maintained outside the jurisdiction of the United States, or covers any employee residing or working outside the United States.

(c) Each Employee Benefit Plan intended to qualify under Section 401(a) of the Co and each related trust intended to qualify under Section 501(a) of the Code has received a favorable determination or opinion letter from the IRS with respect to easuch Employee Benefit Plan as to its qualified status under the Code, including all amendments to the Code effected by the Tax Reform Act of 1986 and subsequent legislation for the most recent cycle applicable to such qualified plan pursuant to Revenue Procedure 2005-66 (as amended or otherwise revised by subsequent IRS guidance), any such letter has not been revoked (nor has revocation been

A-14

threatened) and no fact or event has occurred since the date of such letter or letters from the IRS that could reasonably be expected to adversely affect the qualified sta of any such Employee Benefit Plan or the exempt status of any such trust.

- (d) With respect to each Employee Benefit Plan, Valley and each of its Subsidiarie has complied in all material respects, and is now in substantial compliance with all provisions of ERISA, the Code and all Laws and regulations applicable to such Employee Benefit Plans and each Employee Benefit Plan has been administered in material respects in accordance with its terms. There is not now, nor do any circumstances exist that could reasonably be expected to give rise to, any requirem for the posting of security with respect to any Employee Benefit Plan or the imposition of any lien on the assets of Valley or any of its Subsidiaries under ERIS or the Code. Neither Valley nor any of its Subsidiaries has engaged in a transaction with respect to any applicable Employee Benefit Plan that, assuming the taxable period of such transaction expired as of the date hereof, could subject Valley or any its Subsidiaries to a tax or penalty imposed by either Section 4975 of the Code or Section 502(i) of ERISA.
- (e) All contributions required to be made to any Employee Benefit Plan by applical Law or regulation or by any plan document or other contractual undertaking, and a premiums due or payable with respect to insurance policies funding any Employee Benefit Plan, for any period through the date hereof have been timely made or paid full or, to the extent not required to be made or paid on or before the date hereof, h been accrued on Valley Financial Statements to the extent required under GAAP. Each Employee Benefit Plan that is an employee welfare benefit plan under Sectio 3(1) of ERISA is either (i) funded through an insurance company contract and is not welfare benefit fund with the meaning of Section 419 of the Code or (ii) unfunded
- (f) (i) No Employee Benefit Plan is a multiemployer plan within the meaning of Section 3(37) or 4001(a)(3) of ERISA (a <u>Multiemployer Plan</u>) or a plan that has or more contributing sponsors at least two of whom are not under common control within the meaning of Section 4063 of ERISA (a <u>Multiple Employer Plan</u>); (ii) of Valley nor any of its ERISA Affiliates has, at any time during the last six years, contributed to or been obligated to contribute to any Multiemployer Plan or Multip Employer Plan; (iii) neither Valley nor any of its ERISA Affiliates has incurred an Withdrawal Liability that has not been satisfied in full; and (iv) no Employee Bene Plan is subject to Title IV or Section 302 of ERISA or to Sections 412 or 430 of the Code. ERISA Affiliate means, with respect to any entity, trade or business, any entity, trade or business that is a member of a group described in Section 414(b), (c (m) or (o) of the Code or Section 4001(b)(1) of ERISA that includes the first entity trade or business, or that is a member of the same controlled group as the first e trade or business pursuant to Section 4001(a)(14) of ERISA. Withdrawal Liability means liability to a Multiemployer Plan as a result of a complete or partial withdra from such Multiemployer Plan, as those terms are defined in Part I of Subtitle E of Title IV of ERISA.
- (g) There does not exist, nor, to the Knowledge of Valley, do any circumstances ex that could reasonably be expected to result in, any Controlled Group Liability that

would be a liability of Valley or any of its ERISA Affiliates following the Closing. Without limiting the generality of the foregoing, neither Valley nor any of its ERIS Affiliates, has engaged in any transaction described in Section 4069 or Section 420 or 4212 of ERISA. Controlled Group Liability means any and all liabilities (i) u Title IV of ERISA, (ii) under Section 302 of ERISA, (iii) under Sections 412 and 4971 of the Code, (iv) as a result of a failure to comply with the continuation cover requirements of Section 601 et seq. of ERISA and Section 4980B of the Code, (v) result of a failure to comply with the group health care coverage requirements of sections 4980D or 4890H of the Code and (vi) under corresponding or similar provisions of other applicable Laws or regulations.

(h) Except as set forth in Section 3.11(h) of the Valley Disclosure Schedule, neithe Valley nor any of its Subsidiaries has any liability for life, health, medical or other welfare benefits to former employees or beneficiaries or dependents thereof, excep for health continuation coverage as required by Section 4980B of the Code or Part Title I of ERISA and at no expense to Valley or any of its Subsidiaries. Valley and each of its Subsidiaries has reserved the right to amend, terminate or modify at any time all plans or arrangements providing for post-retirement welfare benefits.

A-15

- (i) Except as set forth in Section 3.11(i) of the Valley Disclosure Schedule, there are no pending or threatened claims (other than routine undisputed claims for benefits the ordinary course), lawsuits or arbitrations pertaining to any Employee Benefit P which have been asserted, threatened or instituted, and to the Knowledge of Valley set of circumstances exists which may reasonably give rise to a claim or lawsuit against Valley or any of its Subsidiaries, the Employee Benefit Plans, any fiduciari thereof with respect to their duties to the Employee Benefit Plans or the assets undeany of the Employee Benefit Plans. Valley has not taken any corrective action or many filing under any voluntary correction program of the IRS, the U.S. Department Labor or any other Governmental Entity with respect to any Employee Benefit Plan and Valley has no knowledge of any plan defect that would qualify for correction under any such program. No audit or other proceeding by a Governmental Entity is pending or, to Knowledge of Valley, threatened with respect to any Employee Beneflan.
- (j) Except as set forth in Section 3.11(j) of the Valley Disclosure Schedule, each Employee Benefit Plan that is or was a nonqualified deferred compensation plan within the meaning of Section 409A of the Code and associated U.S. Department of the Treasury guidance at all times since January 1, 2005, has been and is being operated in compliance with, and has been and is in documentary compliance with Section 409A of the Code and IRS regulations and guidance thereunder. No compensation payable by Valley has been reportable as nonqualified deferred compensation in the gross income of any individual or entity, and subject to an additional tax, as a result of the operation of Section 409A of the Code and no arrangement exists with respect to a nonqualified deferred compensation plan that would result in income inclusion under Section 409A(b) of the Code.
- (k) Except as set forth in Section 3.11(k) of the Valley Disclosure Schedule, neither the execution and delivery of this Agreement nor the consummation of the transactions contemplated hereby, either alone or together with any other event or events, will (i) result in any payment (including, without limitation, severance, cha in control, forgiveness of indebtedness or otherwise) becoming due under any Employee Benefit Plan, whether or not such payment is contingent, (ii) increase an payments or benefits otherwise payable under any Employee Benefit Plan, (iii) res in the acceleration of the time of payment, vesting or funding of any benefits including, but not limited to, the acceleration of the vesting and exercisability of ar equity awards, whether or not contingent, (iv) result in any limitation on the right of Valley to amend, merge, terminate or receive a reversion of assets from any Emplo Benefit Plan or related trust, (v) require the funding of any trust or other funding vehicle or (vi) limit or restrict the right of Valley or, after the consummation of the transactions contemplated hereby, the Surviving Corporation, to merge, amend or terminate any of the Employee Benefit Plans. There is no agreement, contract or arrangement to which Valley is a party that could, individually or collectively, resu in the payment of any amount that would not be deductible by reason of Section 28 of the Code, as determined without regard to Section 280G(b)(4) of the Code. No Employee Benefit Plan provides for the gross-up or reimbursement of Taxes under Section 4999 or 409A of the Code, or otherwise.

- (l) Each individual who renders services to Valley who is classified by Valley as having the status of an independent contractor or other non-employee status for an purpose (including for purposes of taxation and tax reporting and participation und Employee Benefit Plans) is properly so characterized.
- 3.12 <u>Approvals</u>. As of the date of this Agreement, Valley knows of no reason why regulatory approvals from any Governmental Entity required for the consummation the transactions contemplated by this Agreement should not be obtained on a timel basis.
- 3.13 Opinion. The Board of Directors of Valley has received the opinion of Vining Sparks IBG, LP that, as of the date hereof, and based upon and subject to the factor and assumptions set forth therein, the Merger Consideration and the amount of the Special Dividend to be paid to the holders of Valley Common Stock in the Merger fair, from a financial point of view, to such holders.
- 3.14 <u>Valley Information</u>. The information relating to Valley that is provided by Va or its representatives for inclusion in the Proxy Statement and Form S-4, or in any application, notification or other document filed

A-16

with any other Regulatory Agency or other Governmental Entity in connection with the transactions contemplated by this Agreement, will not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances in which they are made, not misleading. The portions of the Form S-4 and the Proxy Statement relating to Valle and other portions within the reasonable control of Valley will comply with the provisions of the Security Act and the Securities Exchange Act of 1934, as amende (the <u>Exchange Act</u>), and the rules and regulations thereunder.

3.15 <u>Legal Proceedings</u>. Except as set forth in Section 3.15 of the Valley Disclosur Schedule, neither Valley nor any of its Subsidiaries is a party to any legal, administrative, arbitration, investigatory or other proceeding (including any investigation, action, or proceeding with respect to Taxes) pending or, to the Knowledge of Valley, is any of the foregoing proceedings threatened, or which Va has reason to believe may be threatened, against or affecting Valley or any of its Subsidiaries or any of their respective current or former directors or officers, or ma involve a claim or claims asserting a liability of \$50,000 individually, or \$100,000 more in the aggregate, or may otherwise restrict the conduct of business by Valley. Section 3.15 of the Valley Disclosure Schedule, includes, with respect to each matter identified, if applicable, the case title, the court, the court file number, the d filed, the law firm representing Valley or any of its Subsidiaries and such other information as may be reasonably requested by CVB. Except as set forth on Sectio 3.15 of the Valley Disclosure Schedule, there is no (i) outstanding judgment, order writ, injunction or decree, stipulation or award of any Governmental Entity or by arbitration, against, or, to the Knowledge of Valley, affecting Valley or any of its Subsidiaries or their respective assets or business that (A) has had or may have a Material Adverse Effect, (B) requires any payment by, or excuses an obligation of third party to make any payment to, Valley or any of its Subsidiaries of an amount exceeding \$50,000, or (C) has the effect of prohibiting any business practice of, or acquisition, retention or disposition of property by Valley or any of its Subsidiaries would apply to CVB or any of its Affiliates after the Merger, or (ii) legal, administrative, arbitration, investigatory or other proceeding pending or, to the Knowledge of Valley that has been threatened, or which Valley has reason to belie may be threatened, against or affecting any director, officer, employee, agent or representative of Valley or any of its Subsidiaries, in connection with which any su Person has or may have rights to be indemnified by Valley or any of its Subsidiaries. In addition thereto, neither Valley nor any of its Subsidiaries is subject any cease and desist or consent order or directive or a party to any written agreeme or memorandum of understanding with any Governmental Entity that restricts the conduct of its business, or in any manner relates to its business, capital adequacy, credit or compliance policies or its management.

3.16 Material Contracts.

(a) Section 3.16 of the Valley Disclosure Schedule lists each agreement, contract, arrangement, commitment or understanding (whether written or oral) to which Val or any of its Subsidiaries is a party to, bound by or subject to (each, a <u>Material Contract</u>): (i) that is a material contract within the meaning of Item 601(b)(10)

SEC s Regulation S-K; (ii) that contains a non-compete or client or customer non-solicit requirement or any other provisions that restricts the conduct of, or the manner of conducting, any line of business of Valley, VBB or any of their respecti Affiliates (or, upon consummation of the Merger or the Bank Merger, of the Surviv Corporation, Citizens Business Bank or any of their respective Affiliates); (iii) that obligates Valley, VBB or any of their respective Affiliates (or, upon consummation the Merger, of CVB, Citizens Business Bank or any of their respective Affiliates) t conduct business with any third party on an exclusive or preferential basis; (iv) tha requires referrals of business or requires Valley or any of its Subsidiaries or any of their respective Affiliates to make available investment opportunities to any Person a priority or exclusive basis; (v) that relates to the incurrence of indebtedness by Valley or any of its Subsidiaries (other than deposit liabilities, trade payables, fede funds purchased, advances and loans from the Federal Reserve or the Federal Hom Loan Bank and securities sold under agreements to repurchase, in each case incurre in the ordinary course of business consistent with past practice) including any sale leaseback transactions, capitalized leases and other similar financing transactions; that grants any right of first refusal, right of first offer or similar right with respect any assets, rights or properties of Valley or any of its Subsidiaries; (vii) that limits payment of dividends by Valley or any of its Subsidiaries; (viii) that relates to a joi venture,

A-17

partnership, limited liability company agreement or other similar agreement or arrangement with any third party, or to the formation, creation or operation, management or control of any partnership or joint venture with any third parties, except in each case that relate to merchant banking investments by Valley or any o Subsidiaries in the ordinary course of business; (ix) that relates to an acquisition, divestiture, merger or similar transaction and which contains representations, covenants, indemnities or other obligations (including indemnification, earn-out other contingent obligations) that are still in effect; (x) that provides for payments be made by Valley or any of its Subsidiaries or any of its successors upon or as a result of a change in control thereof (<u>Change of Control Payments</u>); (xi) that wa negotiated and entered into on an arm s-length basis; (xii) that provides for the guarantee or indemnification by Valley or any of its Subsidiaries of any Person, except for contracts entered into in the ordinary course of business providing for customary and immaterial indemnification; (xiii) that is a consulting agreement or data processing, software programming or licensing contract involving the paymen more than \$50,000 per annum (other than any such contracts which are terminable Valley or any of its Subsidiaries on 60 days or less notice without any required payment or other conditions, other than the condition of notice); (xiv) that grants to Person any right in Valley Owned Intellectual Property or grants to Valley or any of its Subsidiaries a license to Valley Licensed Intellectual Property (excluding licens to shrink-wrap or click-wrap software), in each case that involves the payment or more than \$50,000 per annum or is material to the conduct of the businesses of Va or any of its Subsidiaries; (xv) to which any Affiliate, officer, director, employee o consultant of such party or any of its Subsidiaries is a party or beneficiary; (xvi) th would prevent, materially delay or materially impede Valley s ability to consumm the Merger or the other transactions contemplated hereby any of its Subsidiaries ability to consummate the Bank Merger; (xvii) that contains a put, call or similar ri pursuant to which Valley or any of its Subsidiaries could be required to purchase o sell, as applicable, any equity interests of any Person or assets; (xviii) that involves payment, on a one-time basis or over the life of the term of the agreement, of \$50,0 or more or is more than one year of in length and is not terminable by Valley or an its Subsidiaries without penalty; (xix) any other agreement of any other kind which involves future payments or receipts or performances of services or delivery of iter requiring payment of \$50,000 or more to or by Valley or any of its Subsidiaries or not terminable without penalty, other than payments made under or pursuant to loa agreements, participation agreements and other agreements for the extension of cre each in the ordinary course of their business; or (xx) that is otherwise not entered in in the ordinary course of business or that is material to Valley or its financial condition or results of operations on a consolidated basis. Valley has Previously Disclosed true, correct and complete copies of each Material Contract.

(b) (i) Each Material Contract is a valid and legally binding agreement of Valley or Subsidiary of Valley, as the case may be, and, to the Knowledge of Valley, the counterparty or counterparties thereto, is enforceable in accordance with its terms (subject to the Enforceability Exceptions) and is in full force and effect, (ii) Valley and each of its Subsidiaries has duly performed all obligations required to be performed by it prior to the date hereof under each Material Contract, (iii) neither Valley nor any of its Subsidiaries, nor any counterparty or counterparties, is in brea

of any provision of any Material Contract, and (iv) no event or condition exists that constitutes, after notice or lapse of time or both, will constitute, a breach, violation default on the part of Valley or any of its Subsidiaries under any such Material Contract or provide any party thereto with the right to terminate such Material Contract. Section 3.16(b) of the Valley Disclosure Schedule sets forth a true and complete list of (x) all Material Contracts pursuant to which consents or waivers are may be required and (y) all notices which are required to be given, in each case, proto the performance by Valley or any of its Subsidiaries of this Agreement and the consummation of the Merger and the other transactions contemplated hereby.

3.17 Environmental Matters. (a) Valley and each of its Subsidiaries is in compliance and has at all times in the past complied, in all material respects with any federal, so or local Law, regulation, order, decree, permit, authorization, common Law or ager requirement relating to: (i) the protection or restoration of the environment, health safety as it relates to hazardous substance handling or exposure or the protection of natural resources; (ii) the handling, use, presence, disposal, release or threatened release of, or exposure to, any hazardous substance; or (iii) noise, odor, wetlands, indoor air, pollution, contamination or any injury to Persons or property from exposure to any Hazardous Substance (collectively, Environmental Laws); (b) the are no

A-18

kind, pending, or threatened, by any Person, court, agency, or other Governmental Entity or any arbitral body, against Valley or any of its Subsidiaries relating to liability under any Environmental Law and, to the Knowledge of Valley, there is no reasonable basis for any such proceeding, claim, action or investigation; (c) there a no agreements, orders, judgments or decrees by or with any court, regulatory agence or other Governmental Entity, or any agreements, indemnities or settlements with a Person that impose any liabilities or obligations under, relating to or in respect of a Environmental Law; (d) to Knowledge of Valley, there are, and have been, no rele of any hazardous substances, wastes or other environmental conditions at any prop (currently or formerly owned, operated, or otherwise used by Valley or any of its Subsidiaries) under circumstances which could reasonably be expected to result in liability to or claims against Valley or any of its Subsidiaries relating to any Environmental Law; and (e) there are no reasonably anticipated future events, conditions, circumstances, practices, plans or legal requirements (in each case of Valley or any of its Subsidiaries) that could reasonably be expected to give rise to obligations or liabilities under any Environmental Law. Valley has made available CVB all asbestos surveys and reports, mold surveys and reports, lead surveys and reports, reports on environmental exposure, underground tank removal reports and Phase I and Phase II environmental reports (environmental assessments) during the past five years with respect to any properties owned or leased by it any of its Subsidiaries. For purposes of this Agreement, Hazardous Substance shall include is not limited to, (i) any petroleum or petroleum products, natural gas, or natural ga products, radioactive materials, asbestos, mold, urea formaldehyde foam insulation transformers or other equipment that contains dielectric fluid containing levels of polychlorinated biphenyls (PCBs), and radon gas; (ii) any chemicals, materials, wa or substances defined as or included in the definition of hazardous substances, hazardous materials, extremely hazardous wastes, hazardous wastes, hazardous wastes, toxic substances, toxic pollutants, contaminants, or 1

proceedings, claims, actions, or, to the Knowledge of Valley, investigations of any

hazardous wastes, nazardous materials, extremely nazardous wastes, rest hazardous wastes, toxic substances, toxic pollutants, contaminants, or p words of similar import, under any Environmental Laws; and (iii) any other chemic material, waste or substance which is in any way regulated as hazardous or toxic by any federal, state or local government authority, agency or instrumentality, including mixtures thereof with other materials, and including any regulated building material such as asbestos and lead.

3.18 <u>Taxes</u>. Valley and each of its Subsidiaries (i) have prepared in good faith and duly and timely filed (taking into account any extension of time within which to fil all Tax Returns (as defined below) required to be filed by any of them and all such filed Tax Returns are complete and accurate; and (ii) have paid all Taxes (as define below) that are required to be paid or that Valley is obligated to withhold from amounts owing to any employee, creditor or third party, except with respect to mat contested in good faith and for which adequate reserves have been established and reflected on the financial statements of Valley. None of the Tax Returns or matters currently under any audit, suit, proceeding, examination or assessment by the U.S. Internal Revenue Service (<u>IRS</u>) or the relevant state, local or foreign Tax author Valley has not received written notice from any Tax authority that an audit, suit, proceeding, examination or assessment in respect of such Tax Returns or matters pertaining to Taxes are pending or threatened. No deficiencies have been asserted or

assessments made against Valley or any of its Subsidiaries that have not been paid resolved in full. Since December 31, 2011, no claim has been made against Valley any of its Subsidiaries by any Tax authorities in a jurisdiction where such Valley o any such Subsidiary, as the case may be, does not file Tax Returns that Valley or a of its Subsidiaries is or may be subject to taxation by that jurisdiction. None of Val or any of its Subsidiaries is, and during the past five years neither Valley nor any o Subsidiaries has ever been, a United States real property holding corporation wi the meaning of Section 897 of the Code. No liens for Taxes exist with respect to ar of the assets of Valley or any of its Subsidiaries, except for liens for Taxes not yet and payable or that are under good faith contest as disclosed on Section 3.18 of the Valley Disclosure Schedule. Neither Valley nor any of its Subsidiaries has entered into any closing agreements, private letter rulings, technical advice memoranda or similar agreement or rulings with any Tax authority, nor have any been issued by a Tax authority, in each case that have any continuing effect after the Closing Date. Neither Valley nor any of its Subsidiaries (A) has been a member of an affiliated, combined, consolidated or unitary Tax group for purposes of filing any Return, other than, for purposes of filing, affiliated, combined, consolidated or unit Tax Returns, a group of which Valley was the common parent, (B) has any liability for Taxes of any Person under Treasury Regulations Section 1.1502-6 (or any simi provision of state, local or foreign law) other than Taxes of members of the consolidated

A-19

group of which Valley is the common parent, (C) is a party to or bound by any Tax sharing or allocation agreement or has any other current or potential material contractual obligation to indemnify any other Person with respect to Taxes other th Taxes of members of the consolidated group of which Valley is the common paren (D) has, or has ever had, a permanent establishment in any country other than the country of its organization, or (E) has granted to any Person any power of attorney that is currently in force with respect to any Tax matter. None of Valley, any of its Subsidiaries nor any person act on their behalf has agreed or is required to make ar adjustments after the Closing Date pursuant to Section 481(a) of the Code or any similar provisions of state, local or foreign Law by reason of a change in accounting method. To the Knowledge of Valley, no taxing authority has proposed any such adjustment, nor is any application pending with any taxing authority requesting permission for any changes in accounting methods that relate to its business or operations. Neither Valley nor any of its Subsidiaries has participated in any liste transactions within the meaning of Treasury Regulations Section 1.6011-4(b). Va has made available to CVB true and correct copies of the United States federal inco Tax Returns filed by Valley and each of its Subsidiaries for each of the fiscal years ended December 31, 2011, 2012, 2013, 2014 and 2015. Valley has not been a distributing corporation or controlled corporation (i) in any distribution occu during the last 30 months that was purported or intended to be governed by Section 355 of the Code (or any similar provision of state, local or foreign Law) or (ii) in a distribution that could otherwise constitute part of a plan or series of related transactions (within the meaning of Section 355(e) of the Code) of which the Men are a part. As used in this Agreement, (i) the term <u>Tax</u> (including, with correlati meaning, the term <u>Taxes</u>) includes all United States federal, state, local and fore income, profits, franchise, gross receipts, environmental, customs duty, capital stoc severances, stamp, payroll, sales, employment, unemployment, disability, use, property, withholding, excise, production, value added, occupancy and other taxes, duties or assessments of any nature whatsoever, together with all interest, penalties and additions imposed with respect to such amounts and any interest in respect of s penalties and additions, and (ii) the term <u>Tax Return</u> includes all returns and rep (including elections, declarations, disclosures, schedules, estimates and information returns) required to be supplied to a Tax authority relating to Taxes.

3.19 <u>Reorganization</u>. Neither Valley nor any of its Subsidiaries has taken or agreed take any action, and is not aware of any fact or circumstance, that would prevent or impede, or could reasonably be expected to prevent or impede, the Merger from qualifying as a reorganization within the meaning of Section 368(a) of the Code

3.20 Intellectual Property.

(a) Valley and each of its Subsidiaries (A) solely owns (beneficially, and of record where applicable), free and clear of all Liens, other than Permitted Encumbrances a non-exclusive licenses entered into in the ordinary course of business, all right, title and interest in and to its respective Valley Owned Intellectual Property, and (B) to Knowledge of Valley, has valid and sufficient rights and licenses to all of Valley Licensed Intellectual Property. Valley Owned Intellectual Property is subsisting an to the Knowledge of Valley, valid and enforceable.

- (b) To the Knowledge of Valley, the operations of Valley and each of its Subsidiar as presently conducted (including, for avoidance of doubt, Valley s and its Subsidiars of Valley Licensed Intellectual Property) do not infringe, dilute, misappropriate otherwise violate the Intellectual Property rights of any third Person, and no Person has asserted in writing that Valley or any of its Subsidiaries has infringed, diluted, misappropriated or otherwise violated any third Person s Intellectual Property right To the Knowledge of Valley, no third Person has infringed, diluted, misappropriate or otherwise violated any of Valley s or any of its Subsidiaries rights in Valley C Intellectual Property.
- (c) Valley has taken reasonable measures to protect (A) its rights in its Valley Own Intellectual Property and (B) the confidentiality of all Trade Secrets that are owned used or held by Valley or any of its Subsidiaries and to the Knowledge of Valley, so Trade Secrets have not been used, disclosed to or discovered by any Person except pursuant to appropriate non-disclosure agreements which have not been breached. The Knowledge of Valley, no Person has gained unauthorized access to Valley is on of its Subsidiaries. IT Assets since January 1, 2012.

A-20

- (d) Valley s IT Assets operate and perform substantially as required by Valley and each of its Subsidiaries in connection with their respective businesses and have not materially malfunctioned or failed within the past two years. Valley and each of its Subsidiaries have implemented reasonable backup, security and disaster recovery technology and procedures consistent with industry best practices. Valley and each its Subsidiaries are each compliant with all applicable Laws, rules and regulations, their own privacy policies and commitments to their respective customers, consum and employees, concerning data protection and the privacy and security of personal data and the nonpublic personal information of their respective customers, consum and employees.
- (e) For purposes of this Agreement,
- (i) <u>Intellectual Property</u> means any and all: (i) trademarks, service marks, brand names, collective marks, Internet domain names, logos, symbols, slogans, designs other indicia of origin, together with all translations, adaptations, derivations and combinations thereof, all applications, registrations and renewals for the foregoing and all goodwill associated therewith and symbolized thereby; (ii) patents and patentable inventions (whether or not reduced to practice), all improvements theret and all invention disclosures and applications therefor, together with all divisions, continuations, continuations-in-part, revisions, renewals, extensions, reexamination and reissues in connection therewith; (iii) confidential proprietary business information, trade secrets and know-how, including processes, schematics, business and other methods, technologies, techniques, protocols, formulae, drawings, prototypes, models, designs, unpatentable discoveries and inventions (<u>Trade Secr</u> (iv) copyrights in published and unpublished works of authorship (including databatic) and other compilations of information), and all registrations and applications there and all renewals, extensions, restorations and reversions thereof; and (v) other intellectual property rights.
- (ii) <u>IT Assets</u> means, with respect to any Person, the computers, computer softw firmware, middleware, servers, workstations, routers, hubs, switches, data, data communications lines, and all other information technology equipment, and all associated documentation owned by such Person or such Person s Subsidiaries.
- (iii) <u>Valley Licensed Intellectual Property</u> means the Intellectual Property owner third Persons that is used in or necessary for the operation of the respective business of Valley and each of its Subsidiaries as presently conducted.
- (iv) <u>Valley Owned Intellectual Property</u> means Intellectual Property owned or purported to be owned by Valley or any of its Subsidiaries.

3.21 Properties.

(a) Valley and each of its Subsidiaries (a) has good and insurable title to all the properties and assets owned by Valley or any of its Subsidiaries, as the case may b including, but not limited to, any automated teller machines (the <u>Valley Owned Properties</u>), free and clear of all Liens of any nature whatsoever, except (i) statuted

Liens securing payments not yet due, (ii) Liens for real property Taxes not yet due payable, (iii) easements, rights of way, and other similar encumbrances that do not adversely affect the value or affect the use of the properties or assets subject thereto affected thereby or otherwise impair business operations at such properties as bank facilities and (iv) such imperfections or irregularities of title or Liens as do not materially affect the use of the properties or assets subject thereto or affected therefor otherwise materially impair business operations at such properties (collectively, Permitted Encumbrances), and (b) is the lessee or subleassee of all leaseholds less or subleased by Valley or any of its Subsidiaries (the Valley Real Property), free a collectively with the Valley Owned Properties, the Valley Real Property), free a clear of all Liens of any nature whatsoever, except for Permitted Encumbrances, are in possession of the properties purported to be leased thereunder, and each such lease is valid without default thereunder by the lessee or subleassee or, to the Knowledge Valley, the lessor. Neither Valley nor any of its Subsidiaries owns, or is in the

A-21

process of foreclosing (whether by judicial process or by power of sale) or otherwi in the process of acquiring title to, except pursuant to foreclosures which are pending in the ordinary course of business consistent with past practice, any real property of premises on the date hereof in whole or in part. Section 3.21(a)-(a) of the Valley Disclosure Schedule contains a complete and correct list of all the Valley Owned Properties. Section 3.21(a)-(b) of the Valley Disclosure Schedule contains a complete and correct list of all Valley Leased Properties and together with a list of all applications or subleases and the name of the lessor or sublessor (each, a <u>Lease</u>).

- (b) All buildings, structures, improvements and fixtures on the Valley Real Propert and the equipment located thereon are in good operating condition and repair, ordin wear and tear excepted, and conform to all applicable Laws. All tangible properties Valley or any of its Subsidiaries that are material to the business, financial condition results of operations of Valley and each of its Subsidiaries are in a good state of maintenance and repair, except for ordinary wear and tear, and are adequate for the conduct of the business of Valley and each of its Subsidiaries as presently conducted.
- (c) To the Knowledge of Valley, the buildings, driveways and all other structures a improvements upon the Valley Owned Properties are all within the boundary lines such property or have the benefit of valid easements and there are no encroachmen thereon that would affect the use thereof. To the Knowledge of Valley, there are no outstanding requirements or recommendations by any insurance company that has issued a policy covering the Valley Owned Properties, or by any board of fire underwriters or other body exercising similar functions, requiring or recommendin any repairs or work to be done on any such property.
- (d) Each of the leases for the Valley Leased Property is valid and existing and in further force and effect, and no party thereto is in default and no notice of a claim of default by any party has been delivered to Valley or any of its Subsidiaries, or is now pending, and there does not exist any event that with notice or the passing of time, both, would constitute a default or excuse performance by any party thereto, provide that with respect to matters relating to any party other than Valley or any of its Subsidiaries, the foregoing representation is based on the Knowledge of Valley.
- (e) As to Valley and each of its Subsidiaries, none of the Valley Real Property has been condemned or otherwise taken by any Governmental Entity and, to the Knowledge of Valley, no condemnation or taking is threatened or contemplated an none thereof is subject to any claim, contract or Law which might adversely affect use or value for the purposes now made of it. To the Knowledge of Valley, none of premises or properties of Valley or any of its Subsidiaries is subject to any current interests of third parties or other restrictions or limitations that would impair or be inconsistent with the current use of such property by Valley or any of its Subsidiar.
- (f) Valley has delivered to CVB true, accurate and complete copies of each of the following to the extent in the possession or control of Valley and in any way relate any of the Valley Real Property: (i) title commitments together with legible copies all underlying exceptions, (ii) title policies, (iii) environmental reports, (iv) zoning reports and zoning letters, and (v) licenses and permits.

- (g) Since January 1, 2015, VBB has not applied for or received permission to open any additional branch or operate at any other location.
- 3.22 <u>Insurance</u>. Section 3.22 of the Valley Disclosure Schedule lists all insurance policies and bonds maintained by Valley and each of its Subsidiaries. Except as set forth on Section 3.22 of the Valley Disclosure Schedule, (i) Valley and each of its Subsidiaries is, and at all times within five (5) years hereof each has been, insured with insurers and has insurance coverage adequate to insure against all risks normal insured against by companies in similar businesses and of comparable size, (ii) neither Valley nor any of its Subsidiaries is in default under any policy of insurance or bord such that it could be cancelled, and all such insurance policies and bonds maintained by Valley and each of its Subsidiaries are in full force and effect and, except for expirations in the ordinary course, will remain so through and after the Closing, and (iii) Valley and each of its Subsidiaries

A-22

have filed claims with, or given notice of claims to, its insurers with respect to all material matters and occurrences for which it believes it has coverage. Valley has furnished CVB with true and correct copies of all insurance policies and bonds identified on Section 3.22 of the Valley Disclosure Schedule, including all amendments and supplements thereto.

3.23 Accounting and Internal Controls.

- (a) The records, systems, controls, data and information of Valley and its Subsidiar are recorded, stored, maintained and operated under means (including any electron mechanical or photographic process, whether computerized or not) that are under t exclusive ownership and direct control of Valley or accountants (including all mea of access thereto and therefrom). Valley has devised and maintain internal control over financial reporting (within the meaning of Rules 13a-15(f) and 15d-15(f) under the Exchange Act). Such internal control over financial reporting is effective in providing reasonable assurance regarding the reliability of financial reporting and preparation of financial statements in accordance with GAAP and includes policies and procedures that (i) pertain to the maintenance of records that in reasonable deta accurately and fairly reflect the transactions and dispositions of the assets of Valley (ii) provide reasonable assurance that transactions are recorded as necessary to perpreparation of financial statements in accordance with GAAP, and that receipts and expenditures of Valley are being made only in accordance with authorizations of management and directors of Valley, and (iii) provide reasonable assurance regard prevention or timely detection of unauthorized acquisition, use or disposition of Valley s assets that could have a material effect on its financial statements. Valley Previously Disclosed, based on its most recent evaluation prior to the date hereof, t its auditors and the audit committee of its Board of Directors: (A) any significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting and (B) any fraud, whether or not material, that involves management or other employees who have a significant role in its internal controls over financial reporting.
- (b) Since January 1, 2013, (A) neither Valley nor, to the Knowledge of Valley, any director, officer, auditor, accountant or representative of Valley or any of its Subsidiaries has received or otherwise had or obtained knowledge of any material complaint, allegation, assertion or written claim regarding the accounting or auditing practices, procedures, methodologies or methods (including with respect to loan loan reserves, write-downs, charge-offs and accruals) of Valley or any of its Subsidiaries its respective internal accounting controls, including any material complaint, allegation, assertion or written claim that Valley or any of its Subsidiaries has engain questionable accounting or auditing practices, and (B) no attorney representing Valley or any of its Subsidiaries, whether or not employed by it, has reported evided of a material violation of securities Laws, breach of fiduciary duty or similar violation by it or any of its officers or directors to its board of directors or any committee thereof or to any of its directors or officers.
- 3.24 <u>Derivatives</u>. Except as set forth in Section 3.24 of the Valley Disclosure Schedule, neither Valley nor any of its Subsidiaries is a party to nor has agreed to

enter into an exchange-traded or over-the-counter equity, interest rate, foreign exchange or other swap, forward, future, option, cap, floor or collar or any other contract that is not included on the balance sheet and is a derivative contract (including various combinations thereof) or owns securities that are referred to generically as structured notes, high risk mortgage derivatives, notes, or capped floating rate mortgage derivatives (each, a Derivative Trans All Derivative Transactions, whether entered into for the account of Valley or any its Subsidiaries or for the account of a customer of Valley or any of its Subsidiaries were entered into in the ordinary course of business and in accordance with pruden banking practice and applicable Laws and other policies, practices, and procedures employed by Valley and each of its Subsidiaries is and with counterparties believe be financially responsible at the time, and are legal, valid and binding obligations of Valley or any of its Subsidiaries, as the case may be, enforceable against it in accordance with their terms (except as such enforcement may be limited by the Enforceability Exceptions). Valley and each of its Subsidiaries have duly performe all material respects all of its obligations thereunder to the extent required, and, to Knowledge of Valley, there are no breaches, violations or defaults or allegations or assertions of such by any party thereunder. The financial position of

A-23

Valley on a consolidated basis under or with respect to each such Derivative Transaction has been reflected in all material respects in the books and records of Valley in accordance with GAAP.

3.25 Loan Matters.

- (a) Each loan, loan agreement, note or borrowing arrangement (including leases, or enhancements, commitments, guarantees and interest-bearing assets) in which Vall or any of its Subsidiaries is a creditor (collectively, <u>Loans</u>) currently outstanding evidenced by notes, agreements or other evidences of indebtedness that are true, genuine and what they purport to be, (ii) to the extent secured, has been secured by valid Liens which have been perfected and (iii) to the Knowledge of Valley, is a le valid and binding obligation of the obligor named therein, enforceable in accordance with its terms (subject to the Enforceability Exceptions). The notes or other credit of security documents with respect to each such outstanding Loan were in compliance with all applicable Laws at the time of origination or purchase by Valley or any of Subsidiaries.
- (b) Each outstanding Loan was solicited and originated, and is and has been administered and, where applicable, serviced, and the relevant Loan files are being maintained in accordance with the relevant notes or other credit or security docume and Valley s written underwriting standards, in each case with all applicable requirements of applicable Law and government programs.
- (c) None of the agreements pursuant to which Valley or any of its Subsidiaries has sold or is servicing (i) Loans or pools of Loans or (ii) participations in Loans or po of Loans contains any obligation to repurchase such Loans or interests therein or to pursue any other form of recourse against Valley.
- (d) Valley has Previously Disclosed to CVB all claims for repurchases by Valley or any of its Subsidiaries of Loans that were sold to third parties by Valley or any of i Subsidiaries that are outstanding or threatened (in writing), in each case, as of the chereof or the Closing Date.
- (e) Section 3.25(e) of the Valley Disclosure Schedule sets forth a list of (i) each Lo that as of August 31, 2016 and as of the fifth (5th) Business Day immediately preceding the Closing Date (A) was contractually past due 90 days or more in the payment of principal and/or interest, (B) was on non-accrual status, (C) was classif as substandard, doubtful, loss, classified, criticized, credit risk asse watch list, impaired or special mention (or words of similar import) by Va of its Subsidiaries, or any Governmental Entity (D) a specific reserve allocation existed in connection therewith or (E) was required to be accounted for as a trouble debt restructuring in accordance with ASC 310-40, (ii) each Loan that as of Augus 31, 2016 and as of the fifth (5th) Business Day immediately preceding the Closing Date had an outstanding balance and/or unfunded commitment of \$25,000 or more and that as of such date as to which (A) a reasonable doubt exists as to the timely future collectability of principal and/or interest, whether or not interest is still accru or the Loans are less than 90 days past due or (B) the interest rate terms have been

reduced and/or the maturity dates have been extended subsequent to the agreement under which the Loan was originally created due to concerns regarding the borrow ability to pay in accordance with such initial terms, and (iii) each asset of Valley th as of August 31, 2016 and as of the fifth (5th) Business Day immediately preceding Closing Date was classified as other real estate owned, other repossessed asset an asset to satisfy Loans, and the book value thereof as of such date. For each loan identified in accordance with the immediately preceding sentence, Section 3.25(e) the Valley Disclosure Schedule sets forth the outstanding balance, including accrue and unpaid interest, on each such Loan and the identity of the borrower thereunder of August 31, 2016 and as of the fifth (5th) Business Day immediately preceding the Closing Date.

(f) Section 3.25(f) of the Valley Disclosure Schedule sets forth a list of all Loans as the date of this Agreement by Valley or any of its Subsidiaries to any directors, officers and principal shareholders (as such terms are defined in Regulation O of the Federal Reserve (12 C.F.R. Part 215) of Valley or any of its Subsidiaries. There are employee, officer, director or other Affiliate Loans on which the borrower is paying rate other than that reflected in the note or other relevant credit or security agreement or on which the borrower is paying a rate which was not in compliance with Regulation O, and all such Loans are and were originated in compliance with all applicable Laws.

A-24

- (g) Section 3.25(g) of the Valley Disclosure Schedule sets forth a list of all Loans (providing for zero rate, 1.23% interest rate or any other below-market rate of interest or below-market fees or other teaser rates or fees or their equivalents, (ii) as to valley or VBB, as applicable, has waived its right to collect interest or (iii) providing for an interest rate that is not consistent with Valley s or VBB s, as applicable, we policies with respect to Loan pricing in effect as of the date of this Agreement.
- (h) Neither Valley nor any of its Subsidiaries is now, nor has it ever been, since January 1, 2013 subject to any fine, suspension, settlement or other contract or other administrative agreement or sanction by, or any reduction in any loan purchase commitment from, any Governmental Entity or agency relating to the origination, or servicing of mortgage or consumer Loans or Loans guaranteed by any governmental agency.
- (i) Since January 1, 2013, Valley and each of its Subsidiaries has complied with, as all documentation in connection with the origination, processing, underwriting and credit approval of any mortgage loan originated by Valley or any of its Subsidiarie satisfied: (1) all applicable Laws with respect to the origination, insuring, purchase sale, pooling, servicing, subservicing, loan modification, loss mitigation or filing o claims in connection with such mortgage loans, including, to the extent applicable, Laws relating to real estate settlement procedures, consumer credit protection, trutl lending Laws, usury limitations, fair housing, transfers of servicing, collection practices, equal credit opportunity and adjustable rate mortgages, in each case applicable as of the time of such origination, processing, underwriting or credit approval; (2) the responsibilities and obligations relating to such mortgage loans se forth in any agreement or understanding between Valley and any Agency, loan investor or insurer; (3) the applicable rules, regulations, guidelines, procedures, handbooks and other requirements of any Agency, loan investor or insurer, in each case applicable as of the time of such origination, processing, underwriting or cred approval; and (4) the terms and provisions of any mortgage or other collateral documents and other loan documents with respect to each such mortgage loan; in e case applicable as of the time of such origination, processing, underwriting or cred approval.
- (j) Since January 1, 2013, no customer of Valley or any of its Subsidiaries has indicated in writing to Valley or any of its Subsidiaries that it has terminated or intends to terminate its relationship with Valley or any of its Subsidiaries for poor performance, poor loan service or concern with respect to Valley s any of its Subsidiaries compliance with Laws.
- (k) Since January 1, 2013, neither Valley nor any of its Subsidiaries has engaged in and, to the Knowledge of Valley, no third-party vendors (including outside law firm and other third-party foreclosure services providers) used by Valley or any of its Subsidiaries has engaged in, directly or indirectly, (1) any foreclosures in violation any applicable Law, including but not limited to the Servicemembers Civil Relief or in breach of any binding Regulatory Agreement or (2) the conduct referred to as robo-signing or any other similar conduct of approving or notarizing documents relating to mortgage loans that do not comply with any applicable Law.

- (1) Since January 1, 2013, neither Valley nor any of its Subsidiaries has foreclosed upon, managed or taken a deed or title to, any real estate (other than single-family residential properties) without complying with all applicable FDIC environmental diligence standards (including FDIC Bulletin FIL-14-93, and update FIL-98-2006) foreclosed upon, managed or taken a deed or title to, any such real estate if the environmental assessment indicates the liabilities under Environmental Laws are likely in excess of the asset s value.
- 3.26 Community Reinvestment Act Compliance. VBB is in compliance with the applicable provisions of the Community Reinvestment Act of 1977 (<u>CRA</u>) and regulations promulgated thereunder and has received a CRA rating of satisfactory higher in its most recently completed exam, and Valley has no Knowledge that VB compliance under the CRA should constitute grounds for either the denial by any Governmental Entity of any application to consummate the transactions contempla by this Agreement or the imposition of a materially burdensome condition in connection with the approval of any such application, or the existence of any fact of circumstance or set of facts or circumstances which would reasonably be expected result in VBB having its

A-25

current rating lowered. Neither Valley nor any of its Subsidiaries has received a written notice from any third-party group representing community interests raising concerns or objections with respect to its policies or practices in its relations or practices with customers, vendors or clients.

- 3.27 <u>Bank Secrecy Act Compliance</u>. Neither Valley nor any of its Subsidiaries has been advised of any supervisory concerns regarding its compliance with the Bank Secrecy Act (31 U.S.C. §5322, et seq.) or related state or federal anti-money laundering Laws, including (i) those provisions of the United States Code providin penalties for the laundering of monetary instruments (18 U.S.C. §1956) or engagin monetary transactions in property derived from specified unlawful activity (18 U.S §1957) and (ii) any Know Your Customer regulations, guidelines or supervisory policies and examination requirements.
- 3.28 <u>Investment Securities</u>. Except as set forth on Section 3.28 of the Valley Disclosure Schedule, neither Valley nor any of its Subsidiaries has (a) held any Investment Securities on December 31, 2015, (b) has or acquired, sold or disposed any Investment Security since December 31, 2015, or (c) has held any Investment Security until maturity since December 31, 2015. For purposes of this Agreement, <u>Investment Security</u> means any equity security or debt security as defined in Accounting Standards Codification Topic 320.
- 3.29 <u>Related Party Transactions</u>. Except for Loans set forth in Section 3.25(f) of th Valley Disclosure Schedule, for ordinary course bank deposit, trust and asset management services on arms length terms, and compensation as defined in It of the SEC s Regulation S-K, there are no transactions or series of related transact agreements, arrangements or understandings, nor are there any currently proposed transactions or series of related transactions, between Valley or any of its Subsidiar on the one hand, and any current or former director or executive officer (as defined 3b-7 under the Exchange Act) of Valley or any of its Subsidiaries or any Pers who beneficially owns (as defined in Rules 13d-3 and 13d-5 of the Exchange Act) or more of Valley Common Stock (or any of such Person s immediate family men or Affiliates) on the other hand.
- 3.30 Operating Losses. Section 3.30 of the Valley Disclosure Schedule sets forth a Operating Loss (as defined below) that has occurred at Valley or any of its Subsidiaries during the period after December 31, 2015, to the date of this Agreement. Except as set forth on Section 3.30 of the Valley Disclosure Schedule, its Knowledge, no event has occurred, and no action has been taken or omitted to be taken by any employee of Valley or any of its Subsidiaries that has resulted in the incurrence by Valley or any of its Subsidiaries of an Operating Loss or that might reasonably be expected to result in the incurrence by Valley or any of its Subsidiar of an Operating Loss after the date hereof, which, net of any insurance proceeds payable in respect thereof, exceeds, or would exceed \$25,000 individually or when aggregated with all other Operating Losses, \$50,000 during such period. For purpo of this Agreement, Operating Loss means any individual loss resulting from cas shortages, lost or misposted items, disputed clerical and accounting errors, forged checks, payment of checks over stop payment orders, counterfeit money, wire

transfers made in error, theft, robberies, defalcations, check kiting, fraudulent use credit cards or electronic teller machines, civil money penalties, fines, litigation, claims, arbitration awards or other similar acts or occurrences

3.31 Employee and Labor Matters.

- (a) Section 3.31(a) of the Valley Disclosure Schedule contains a complete and corr list, as of the date of this Agreement, of the name of each employee, job description job location, title, current annual base salary, other compensation and wage and ho exemption status of Valley and each of its Subsidiaries and a summary of all Control or commitments by Valley to increase the compensation or to modify the condition or terms of employment including all agreements regarding Change of Control Payments. All persons who have been treated as independent contractors by Valley tax purposes have met the criteria to be so treated under applicable Law.
- (b) Neither Valley nor any of its Subsidiaries is, nor at any time since January 1, 20 were either of them, a party to or bound by any labor or collective bargaining agreement and, to the Knowledge of Valley, there

A-26

are no organizational campaigns, petitions or other activities or proceedings of any labor union, workers council or labor organization seeking recognition of a collection bargaining unit with respect to, or otherwise attempting to represent, any of the employees of Valley or any of its Subsidiaries or compel Valley or any of its Subsidiaries to bargain with any such labor union, workers council or labor organization. There are no labor related controversies, strikes, slowdowns, walkout other work stoppages pending or, to the Knowledge of Valley, threatened (in writing and neither Valley nor any of its Subsidiaries has experienced any such labor relate controversy, strike, slowdown, walkout or other work stoppage since January 1, 2012. Neither Valley nor any of its Subsidiaries is a party to, or otherwise bound b any consent decree with, or citation by, any Governmental Entity relating to employees or employment practices. Valley and each of its Subsidiaries is in mater compliance with all applicable Laws relating to labor, employment, termination of employment or similar matters, including but not limited to Laws relating to discrimination, disability, classification of workers, labor relations, hours of work, payment of wages and overtime wages, pay equity, immigration, workers compensation, working conditions, employee scheduling, occupational safety and health, family and medical leave, and employee terminations, and has not engaged any unfair labor practices or similar prohibited practices. Except as set forth in Sec 3.31(b) of the Valley Disclosure Schedule, there are no complaints, lawsuits, arbitrations, administrative proceedings, or other proceedings of any nature pendin or, to the Knowledge of Valley, threatened against Valley brought by or on behalf any applicant for employment, any current or former employee, any Person allegin be a current or former employee, any class of the foregoing, or any Governmental Entity, relating to any such Law, or alleging breach of any express or implied contri of employment, wrongful termination of employment, or alleging any other discriminatory, wrongful or tortious conduct in connection with the employment relationship.

- (c) No executive or group of employees has informed Valley of his, her or their int to terminate employment.
- (d) No Person has claimed, or to Knowledge of Valley has valid reason to claim, the any employee or former employee of Valley (i) is in violation of any material terms any employment agreement, confidentiality agreement, non-competition agreement any restrictive covenant with such Person; (ii) has improperly disclosed or utilized trade secret, confidential or proprietary information or documentation belonging to such Person in connection with their employment; or (iii) has interfered in the employment relationship with such Person and any of its present or former employ in violation of any Law or enforceable agreement between such Person and the applicable employee.
- (e) Valley has made available to CVB prior to the date of this Agreement a copy of material written policies and procedures related to the employees of Valley and eac of its Subsidiaries and a written description of all material unwritten policies and procedures related to the employees of Valley and each of its Subsidiaries.

(f) All employees of Valley and each of its Subsidiaries are authorized to work in t United States of America and a Form I-9 has been properly completed and retained with regard to each such employee.

3.32 <u>Trust Matters</u>. Neither Valley nor any of its Subsidiaries exercises any trust powers, including, but not limited to, trust administration, and neither it nor any predecessor has exercised such trust powers for a period from and after January 1, 2013. The term trusts as used in this Section 3.32 includes (i) any and all comm or other trusts between an individual, corporation or other entities and Valley and e of its Subsidiaries or a predecessor, as trustee or co-trustee, including, any pension other qualified or nonqualified employee benefit plans, compensation, testamentary inter vivos, and charitable trust indentures; (ii) any and all decedents estates when Valley or any of its Subsidiaries, or a predecessor is serving or has served as a co-executor or sole executor, personal representative or administrator, administrator de bonis non, administrator de bonis non with will annexed, or in any similar fiduc capacity; (iii) any and all guardianships, conservatorships or similar positions when Valley or any of its Subsidiaries, or a predecessor is serving or has served as a co-grantor or a sole grantor or a conservator or co-conservator of the estate, or in a similar fiduciary capacity; and (iv) any and all agency and/or custodial accounts and/or similar arrangements, including plan administrator for employee

A-27

benefit accounts, under which Valley or any of its Subsidiaries, or a predecessor is serving or has served as an agent or custodian for the owner or other party establish the account with or without investment authority.

3.33 <u>Information Technology; Security & Privacy</u>. All information technology and computer systems and services (including software, information technology and telecommunication hardware and other equipment) relating to the transmission, storage, maintenance, organization, presentation, generation, processing or analysis data and information, whether or not in electronic format, used in or necessary to the conduct of Valley s business on a consolidated basis (collectively, Valley IT Sys have been properly maintained by technically competent personnel, in accordance with standards set by the manufacturers or otherwise in accordance with standards prudent in the industry (including strong passwords), to ensure proper operation, monitoring and use. Valley IT Systems are in material compliance with regulatory standards and guidelines as required by applicable Law. Valley has commercially reasonable disaster recovery plans, procedures and facilities for its business and ha taken commercially reasonable steps to safeguard Valley IT Systems. Valley IT Systems are in good working condition to effectively perform all information technology operations necessary to conduct consolidated business. Neither Valley any of its Subsidiaries has experienced since January 1, 2012 any material disruption to, or material interruption in, its conduct of its business attributable to a defect, bu breakdown, cyber or security breach or other failure or deficiency of Valley IT Systems. Each of Valley and its Subsidiaries has taken reasonable measures to profor the backup and recovery of the data and information necessary to the conduct of businesses (including such data and information that is stored on magnetic or optic media in the ordinary course) without material disruption to, or material interruption in, the conduct of their respective businesses. Neither Valley nor any of its Subsidiaries, is in breach of any material contract related to any Valley IT Systems. To the Knowledge of Valley, Valley and each of its Subsidiaries has at al times complied in all material respects with all applicable legal requirements (including but not limited to all regulatory standards and guidelines) relating to privacy, data protection and the collection and use of personal information gathere accessed in the course of the operations of Valley and each of its Subsidiaries. Valley and each of its Subsidiaries has at all times complied in all material respects with a rules, policies and procedures established by Valley and its Subsidiaries, as applica from time to time with respect to the foregoing. No claims are pending and, to its Knowledge, no claims have been asserted or threatened against Valley or any of its Subsidiaries or are likely to be asserted or threatened against Valley or any of its Subsidiaries by any Person alleging a violation of such Person s privacy, personal confidentiality rights under any such Laws, policies or procedures. The consumma of the Merger, the Bank Merger and the other transactions contemplated hereby wi not breach or otherwise cause any violation of any such Laws, policies or procedures. With respect to all personal information described herein, Valley and e of its Subsidiaries has each taken all steps reasonably necessary (including implementing and monitoring compliance with measures with respect to technical physical security) to protect the information in a manner consistent with the Laws, policies or procedures referred to herein. To the Knowledge of Valley, there has be

Table of Contents 256

no unauthorized access to or other misuse of that information.

3.34 Accuracy and Completeness of Information Furnished. The representations ar warranties made by Valley hereby or any other document delivered pursuant to the terms of this Agreement contain no statements of material fact which are untrue or misleading, or omit to state any material fact which is necessary under the circumstances to prevent the statements contained herein or therein from being misleading. Valley represents and warrants to CVB that the statements contained in this Article III are true, correct and complete as of the date of this Agreement, subj to and as qualified by the Valley Disclosure Schedule. The Valley Disclosure Schedule and the contents thereof including all documents delivered or provided in connection therewith, shall be confidential in accordance with Section 6.3(c) of this Agreement. Valley has made a good faith effort to ensure that the disclosure on each schedule of the Valley Disclosure Schedule corresponds to the section referenced herein. However, for purposes of the Valley Disclosure Schedule, any item disclos on any schedule therein is deemed to be similarly disclosed with respect to all schedules under which such item may be relevant as and to the extent that it is reasonably clear on the face of such schedule that such item applies to such other schedule.

A-28

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF CVB

Except as Previously Disclosed, CVB hereby represents and warrants to Valley as follows:

4.1 Corporate Organization.

- (a) CVB is a corporation duly incorporated, validly existing and in good standing under the Laws of the State of California. Citizens Business Bank is a commercial bank duly formed and validly existing under the Laws of the State of California. Exof CVB and Citizens Business Bank has the requisite corporate power and authorit own or lease all of its properties and assets and to carry on its business as it is now being conducted, and except as would not reasonably be expected, individually or the aggregate, to have a Material Adverse Effect on CVB, is duly licensed or qualitor do business in each jurisdiction in which the nature of the business conducted by or the character or location of the properties and assets owned or leased by it make such licensing or qualification necessary. CVB is duly registered as a bank holding company under the BHC Act.
- (b) True, complete and correct copies of the articles of incorporation of CVB (the <u>CVB Articles</u>) and bylaws of CVB (the <u>CVB By</u>laws), as in effect as of the of this Agreement, have previously been publicly filed by CVB and made available to Valley. The CVB Articles and CVB Bylaws made available to Valley are in full for and effect.
- (c) Citizens Business Bank (i) is duly incorporated validly existing and in good standing under the Laws of its jurisdiction of organization and (ii) has the requisite corporate (or similar) power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted and, except as woul not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect on CVB, is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary. The deposit accounts of Citizens Business Bank are insure by the FDIC through the Deposit Insurance Fund to the fullest extent permitted by Law, all premiums and assessments required to be paid in connection therewith hav been paid when due, and no proceedings for the termination of such insurance are pending or, to the Knowledge of CVB, threatened.

4.2 <u>Capitalization</u>.

(a) The authorized capital stock of CVB consists of (i) 225,000,000 shares of communications, no par value per share (the <u>CVB Common Stock</u>), of which, as of June 30 (the <u>CVB Capitalization Date</u>), 107,946,952 were issued and outstanding, and (i) 20,000,000 shares of preferred stock, no par value per share (<u>CVB Preferred Stock</u>) none of which were outstanding as of the CVB Capitalization Date. As of the CVB

Capitalization Date, 1,124,078 shares of CVB Common Stock were authorized for issuance upon exercise of options issued pursuant to employee and director stock plans of CVB or a Subsidiary of CVB in effect as of the date of this Agreement (th <u>CVB Stock Plans</u>). All of the issued and outstanding shares of CVB Common S have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights, with no personal liability attaching to the ownership ther As of the date of this Agreement, no Voting Debt of CVB is issued or outstanding. Except pursuant to this Agreement and the options described in this Section 4.2, C does not have and is not bound by any outstanding subscriptions, options, warrants calls, rights, commitments or agreements of any character calling for the purchase issuance of any shares of CVB Common Stock, CVB Preferred Stock, Voting Deb CVB or any other equity securities of CVB or any securities representing the right purchase or otherwise receive any shares of CVB Common Stock, CVB Preferred Stock, Voting Debt of CVB or other equity securities of CVB. There are no contractual obligations of CVB or any of its Subsidiaries (i) to repurchase, redeem otherwise acquire any shares of capital stock of CVB or any equity security of CVI its Subsidiaries or any securities representing the right to purchase or otherwise receive any shares of capital stock or any other equity security of CVB or its Subsidiaries or (ii) pursuant to

A-29

which CVB or any of its Subsidiaries is or could be required to register shares of C capital stock or other securities under the Securities Act. There are no voting trusts other agreements or understandings to which CVB, any Subsidiary of CVB or, to the Knowledge of CVB, any of their respective officers or directors, is a party with respect to the voting of any CVB Common Stock, CVB Preferred Stock, Voting Do or other equity securities of CVB. The shares of CVB Common Stock to be issued pursuant to the Merger will be duly authorized and validly issued and, at the Effect Time, all such shares will be fully paid, nonassessable and free of preemptive right with no personal liability attaching to the ownership thereof.

(b) All of the issued and outstanding shares of capital stock or other equity owners interests of Citizens Business Bank are owned by CVB, directly or indirectly, free clear of any material Liens, and all of such shares or equity ownership interests are duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights. Citizens Business Bank does not have or is not bound by any outstanding subscriptions, options, warrants, calls, commitments or agreements of character calling for the purchase or issuance of any shares of capital stock or any other equity security of Citizens Business Bank or any securities representing the r to purchase or otherwise receive any shares of capital stock or any other equity security of Citizens Business Bank.

4.3 Authority; No Violation.

- (a) CVB has full corporate power and authority to execute and deliver this Agreem and to consummate the transactions contemplated hereby. The execution and deliver of this Agreement and the consummation of the Merger or the other transactions contemplated hereby have been duly, validly and unanimously approved and this Agreement duly adopted by the Board of Directors of CVB and the Board of Director of CVB has determined that the Merger, on the terms and conditions set forth in the Agreement, is advisable and in the best interests of CVB its shareholders. This Agreement has been duly and validly executed and delivered by CVB and (assuming due authorization, execution and delivery by Valley) constitutes the valid and bind obligation of each of CVB, enforceable against CVB and Citizens Business Bank is accordance with its terms (subject to the Enforceability Exceptions). The Board of Directors of Citizens Business Bank has unanimously adopted and CVB, as the sol shareholder of Citizens Business Bank, has duly adopted, the Bank Merger Agreement.
- (b) Neither the execution and delivery of this Agreement, nor the consummation by CVB or Citizens Business Bank, as applicable, of the Merger, the Bank Merger or other transactions contemplated hereby, nor compliance with any of the terms or provisions of this Agreement, will (i) violate any provision of the CVB Articles, C Bylaws or similar chartering documents of CVB s Subsidiaries, or (ii) assuming the consents, approvals and filings referred to in Section 4.4 are duly obtained and/made, (A) violate any Law applicable to CVB, any of its Subsidiaries or any of the respective properties or assets or (B) violate, conflict with, result in a breach of any provision of or the loss of any benefit under, constitute a default (or an event that, who notice or lapse of time, or both, would constitute a default) under, result in the

termination of or a right of termination or cancellation under, accelerate the performance required by, or result in the creation of any Lien upon any of the respective properties or assets of CVB or any of its Subsidiaries under, any of the terms, conditions or provisions of any note, bond, mortgage, indenture, deed of true license, lease, franchise, permit, agreement, bylaw or other instrument or obligation which CVB or any of its Subsidiaries is a party or by which any of them or any of their respective properties or assets is bound except, with respect to clause (ii), for such violation, conflict, breach, default, termination, cancellation, acceleration or creation as would not reasonably be expected, individually or in the aggregate, to he a Material Adverse Effect on CVB or Citizens Business Bank.

4.4 <u>Consents and Approvals</u>. Except for (a) any applicable filing with the Nasdaq Stock Market (<u>Nasdaq</u>), (b) the filing with the SEC of the Form S-4 and declara effectiveness of the Form S-4, (c) the filings of applications and notices with the Commissioner pursuant to the California Financial Code and approval of or non-objection to such applications, filings and notices with respect to the Bank Merger, (d) the filing of a bank merger application with the FDIC pursuant to the Bank Merger Act of 1960, as amended, (e) the filing of the

A-30

Agreement of Merger with the California Secretary, (f) the filing of the Bank Merger Agreement with the California Secretary and the Commissioner, (g) Valley is filing notice concerning the Merger with FINRA, (h) such filings and approvals as are required to be made or obtained under the securities or in blue sky. Laws of various in connection with the issuance of the shares of CVB Common Stock pursuant to the Agreement, and (i) the filing by CVB with the Federal Reserve under the BHC Act such approvals, wavier or non-objection as may be necessary for CVB to acquire Valley, no consents or approvals of or filings or registrations with any Government Entity are necessary in connection with the consummation by CVB or Citizens Business Bank of the Merger or the Bank Merger, as applicable, and the other transactions contemplated by this Agreement, No consents or approvals of or filing or registrations with any Governmental Entity are necessary in connection with the execution and delivery by CVB of this Agreement.

4.5 Reports.

- (a) CVB and each of its Subsidiaries have timely filed all reports, registration statements, proxy statements and other materials, together with any amendments required to be made with respect thereto, that they were required to file since Januar 1, 2013 with the Regulatory Agencies and each other applicable Governmental Ent and all other reports and statements required to be filed by them since January 1, 20 including any report or statement required to be filed pursuant to the Laws, rules or regulations of the United States, any state, any foreign entity, or any Regulatory Agency or other Governmental Entity, and have paid all fees and assessments due apayable in connection therewith, and there are no material violations or exceptions any such material report or statement that are unresolved as of the date hereof.
- (b) An accurate and complete copy of each final registration statement, prospectus, report, schedule and definitive proxy statement filed with or furnished to the SEC I CVB pursuant to the Securities Act or the Exchange Act since January 1, 2013 (the CVB SEC Reports) is publicly available. No such CVB SEC Report, at the time furnished or communicated (and, in the case of registration statements and proxy statements, on the dates of effectiveness and the dates of the relevant meetings, respectively), contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statement made therein, in light of the circumstances in which they were made, not misleadir except that information filed as of a later date (but before the date of this Agreement shall be deemed to modify information as of an earlier date. As of their respective dates, all CVB SEC Reports complied as to form in all material respects with the published rules and regulations of the SEC with respect thereto.
- (c) CVB is in compliance in all material respects with the applicable listing and corporate governance rules and regulations of the Nasdaq.
- 4.6 <u>Financial Statements</u>. The financial statements of CVB and its Subsidiaries included (or incorporated by reference) in the CVB SEC Reports (including the related notes, where applicable) (i) have been prepared from, and are in accordance with, the books and records of CVB and its Subsidiaries; (ii) fairly present in all

material respects the consolidated results of operations, cash flows, changes in shareholders equity and consolidated financial position of CVB and its Subsidiari for the respective fiscal periods or as of the respective dates therein set forth (subje in the case of unaudited statements to recurring year-end audit adjustments normal nature and amount); (iii) complied as to form, as of their respective dates of filing the SEC, in all material respects with applicable accounting requirements and with published rules and regulations of the SEC with respect thereto; and (iv) have been prepared in accordance with GAAP consistently applied during the periods involve except, in each case, as indicated in such statements or in the notes thereto. The bo and records of CVB and its Subsidiaries have been maintained in all material respe in accordance with GAAP and any other applicable legal and accounting requirement and reflect only actual transactions. As of the date hereof, KPMG LLP has not resigned (or informed CVB that indicated it intends to resign) or been dismissed as independent public accountants of CVB as a result of or in connection with any disagreements with CVB on a matter of accounting principles or practices, financia statement disclosure or auditing scope or procedure.

A-31

- 4.7 <u>Broker s Fees</u>. Neither CVB nor any of its Subsidiaries nor any of their respect officers or directors have employed any broker, finder or financial advisor or incur any liability for any broker s fees, commissions or finder s fees in connection wit Merger or any other transactions contemplated by this Agreement, other than to Keefe, Bruyette & Woods, Inc.
- 4.8 <u>Absence of Changes</u>. Since December 31, 2015, no event, change or developm or combination of changes or developments have occurred that has had or would reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect on CVB.
- 4.9 Compliance with Applicable Law. CVB and each of its Subsidiaries hold, and have at all times since January 1, 2013 held, all licenses, franchises, permits and authorizations which are necessary for the lawful conduct of their respective businesses and ownership of their respective properties, rights and assets under and pursuant to applicable Law (and have paid all fees and assessments due and payable connection therewith) and, to the Knowledge of CVB, no suspension or cancellation of any such licenses, franchise, permit or authorization is threatened in writing, excin each case where the failure to hold such license, franchise, permit or authorization such fees or assessments or such suspensions or cancellations would not reasonable expected to have, individually or in the aggregate, a Material Adverse Effect on CVB and each of its Subsidiaries have complied with, and are not in default or violation of any, applicable Law relating to CVB or any of its Subsidiaries that wo reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect on CVB.
- 4.10 <u>Approvals</u>. As of the date of this Agreement, CVB knows of no reason why a regulatory approvals from any Governmental Entity required for the consummation the transactions contemplated by this Agreement should not be obtained on a timel basis.
- 4.11 CVB Information. The information relating to CVB and its Subsidiaries that i provided by CVB or its representatives for inclusion in the Form S-4, or in any application, notification or other document filed with any other Regulatory Agency other Governmental Entity in connection with the transactions contemplated by thi Agreement, will not contain any untrue statement of a material fact or omit to state material fact necessary to make the statements therein, in light of the circumstance which they are made, not misleading. The portions of the Proxy Statement relating CVB and its Subsidiaries and other portions within the reasonable control of CVB its Subsidiaries will comply in all material respects with the provisions of the Exchange Act and the rules and regulations thereunder.
- 4.12 <u>Legal Proceedings</u>. Except as Previously Disclosed, there is no suit, action, investigation, claim, proceeding or review pending, or to the Knowledge of CVB, threatened against or affecting it or any of its Subsidiaries or any of the current or former directors or executive officers of it or any of its Subsidiaries and there are n facts or circumstances that would reasonably be expected to result in any claims against CVB or any of its Subsidiaries that would reasonably be expected to have,

either individually or in the aggregate, a Material Adverse Effect on CVB. There is outstanding injunction, order, writ, award, judgment, settlement, arbitration ruling, decree or regulatory restriction imposed upon or entered into by CVB, any of its Subsidiaries or the assets of it or any of its Subsidiaries that would reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect CVB.

4.13 Accounting and Internal Controls.

(a) The records, systems, controls, data and information of CVB and its Subsidiaries are recorded, stored, maintained and operated under means (including any electron mechanical or photographic process, whether computerized or not) that are under the exclusive ownership and direct control of CVB or its Subsidiaries or accountants (including all means of access thereto and therefrom). CVB and its Subsidiaries had devised and maintain internal control over financial reporting (within the meaning Rules 13a-15(f) and 15d-15(f) under the Exchange Act). Such internal control over financial reporting is effective in providing reasonable

A - 32

assurance regarding the reliability of financial reporting and the preparation of financial statements in accordance with GAAP and includes policies and procedure that (i) pertain to the maintenance of records that in reasonable detail accurately an fairly reflect the transactions and dispositions of the assets of CVB, (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparat of financial statements in accordance with GAAP, and that receipts and expenditur of CVB are being made only in accordance with authorizations of management and directors of CVB, and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of CVB s assets the could have a material effect on its financial statements. CVB has designed and implemented disclosure controls and procedures (within the meaning of Rules 13a-15(e) and 15d-15(e) of the Exchange Act) to ensure that material information relating to CVB and its Subsidiaries is made known to its management by others within those entities as appropriate to allow timely decisions regarding required disclosure and to make the certifications required by the Exchange Act and Section 302 and 906 of the Sarbanes-Oxley Act and such disclosure controls and procedure are effective.

- (b) CVB s management has completed an assessment of the effectiveness of its interport over financial reporting in compliance with the requirements of Section 404 the Sarbanes-Oxley Act for the year ended December 31, 2015, and such assessme concluded that such controls were effective. CVB has previously disclosed, based of its most recent evaluation prior to the date hereof, to its auditors and the audit committee of its board of directors (A) any significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting a (B) any fraud, whether or not material, that involves management or other employers who have a significant role in its internal controls over financial reporting.
- (c) Since January 1, 2013, except as Previously Disclosed, (A) neither CVB nor an its Subsidiaries nor, to the Knowledge of CVB, any director, officer, auditor, accountant or representative of it or any of its Subsidiaries has received or otherwishad or obtained knowledge of any material complaint, allegation, assertion or writt claim regarding the accounting or auditing practices, procedures, methodologies or methods (including with respect to loan loss reserves, write-downs, charge-offs and accruals) of CVB or any of its Subsidiaries or their respective internal accounting controls, including any material complaint, allegation, assertion or written claim the CVB or any of its Subsidiaries has engaged in questionable accounting or auditing practices, and (B) no attorney representing CVB or any of its Subsidiaries, whether not employed by it or any of its Subsidiaries, has reported evidence of a material violation of securities Laws, breach of fiduciary duty or similar violation by it or any of its officers or directors to its board of directors or any committee thereof or to an of its directors or officers.
- 4.14 <u>Reorganization</u>. Neither CVB nor Citizen Business Bank has taken or agreed take any action, or is aware of any fact or circumstance, that would prevent or imperor could reasonably be expected to prevent or impede, the Merger from qualifying reorganization within the meaning of Section 368(a) of the Code.

4.15 <u>Accuracy and Completeness of Information Furnished</u>. The representations ar warranties made by CVB hereby or any other document delivered pursuant to the terms of this Agreement contain no statements of material fact which are untrue or misleading, or omit to state any material fact which is necessary under the circumstances to prevent the statements contained herein or therein from being misleading.

ARTICLE V

COVENANTS RELATING TO CONDUCT OF BUSINESS

5.1 <u>Conduct of Valley Prior to the Effective Time</u>. Except as Previously Disclosed, expressly contemplated by or permitted by this Agreement, as required by applicable Law, or with the prior written consent of CVB, during the period from the date of the Agreement to the earlier of the Effective Time or the termination of this Agreement accordance with Article VIII, Valley shall, and shall cause each of its Subsidiaries (i) conduct its business in the ordinary course consistent with past practice in all material respects, (ii) use commercially reasonable efforts to maintain and preserve intact its business organization and advantageous

A-33

business relationships, and goodwill with Governmental Entities, customers, suppl distributors, creditors, lessors, officers and employees and business associates, kee available the services of its present employees and agents, and perform its material obligations under all Material Contracts, (iii) maintain in full force and effect insurance comparable in amount and scope of coverage to that now maintained by (iv) maintain its allowance for loan and lease losses in accordance with past practic and methodologies and GAAP (providing however, that any changes in practices of methodology shall be attributable solely to changes in GAAP or as directed by a Governmental Entity), (v) charge off all Loans and other assets, or portions thereof deemed uncollectible or classified as loss in accordance with GAAP and applica Law or as directed by a Governmental Entity, (vi) maintain loan classification poliand procedures in accordance with industry best practices consistent with past practice, and (vii) take no action that is intended to or would reasonably be expected to adversely affect or materially delay the ability of any of Valley, CVB or Citizen Business Bank to obtain any necessary approvals of any Governmental Entity requ for the transactions contemplated hereby or to perform its covenants and agreemen under this Agreement or to consummate the transactions contemplated hereby.

- 5.2 <u>Valley Forbearances</u>. During the period from the date of this Agreement to the earlier of the Effective Time or the termination of this Agreement in accordance with Article VIII, except as Previously Disclosed, as expressly contemplated or permitted by this Agreement, or as required by applicable Law, Valley shall not, without the prior written consent of CVB and shall not permit any of its Subsidiaries to:
- (a) (i) issue, sell or otherwise permit to become outstanding, or dispose of or encumber or pledge, or authorize or propose the creation of, any additional shares of its capital stock, or securities convertible or exchangeable into, or exercisable for, a shares of its capital stock, or any options, warrants or other rights of any kind to acquire any shares of such capital stock or such convertible or exchangeable securi or receive a cash payment based on the value of any shares of such capital stock, or (ii) permit any additional shares of its capital stock, or securities convertible or exchangeable into, or exercisable for, any shares of its capital stock, or any options warrants or other rights of any kind to acquire any shares of such capital stock or so convertible or exchangeable securities or receive a cash payment based on the valuany shares of such capital stock, to become be subject to new grants, in each case except as required pursuant to the exercise or settlement of Valley Stock Options outstanding on the date hereof in accordance with the terms of the applicable Valle Stock Plan and the agreement governing the applicable Valley Stock Option, in each case as in effect on the date hereof.
- (b) Make, declare, pay or set aside for payment any dividend on or in respect of, or declare or make any distribution on any shares of its capital stock or directly or indirectly adjust, split, combine, redeem, reclassify, purchase or otherwise acquire, any shares of its stock or other securities other than (i) quarterly cash dividends consistent, with respect to timing and amount, with Valley s past practices and polin effect as of the date of this Agreement, declared on or prior to the final day of the month immediately preceding the Closing Date with a record date that precedes the Closing Date and (ii) the Special Dividend.

- (c) Amend or modify the material terms of, waive, release or assign any rights under terminate, renew or allow to renew automatically, make any payment not then required under, knowingly violate the terms of or enter into (i) any Material Contract Lease, Regulatory Agreement, any contract that would be a Material Contract if it were in existence on the date hereof or other binding obligation that is material to Valley, taken as a whole, (ii) any restriction on the ability of Valley or any of its Subsidiaries to conduct its business as it is presently being conducted or (iii) any contract governing the terms of Valley Common Stock or rights associated therewing any other outstanding capital stock or any outstanding instrument of indebtedness
- (d) Sell, transfer, mortgage, lease, guarantee, encumber, license, let lapse, cancel, abandon or otherwise create any Lien on or otherwise dispose of or discontinue and its assets, deposits, business or properties (other than sales pursuant to Section 5.2(which Section 5.2(p) will exclusively govern such sales), except for sales, transfers mortgages, leases, guarantees, encumbrances, licenses, lapse, cancellation, abandonments or other dispositions or discontinuances in the ordinary course of business and in a transaction that, together with other such transactions, is not mate to Valley taken as a whole.

A-34

- (e) Acquire (other than by way of foreclosures or acquisitions of control in a fiduci or similar capacity or in satisfaction of debts previously contracted in good faith, in each case in the ordinary course of business) all or any portion of the assets, business deposits or properties of any other entity, except in the ordinary course of business in a transaction that, together with other such transactions, is not material to it, take as a whole, and would not reasonably be expected to present a material risk that the Closing Date will be materially delayed or that the Requisite Regulatory Approvals will be more difficult to obtain.
- (f) Amend Valley Articles or Valley Bylaws, or similar governing documents of art of its Subsidiaries.
- (g) Except as and when required under applicable Law or (subject to Section 6.6) t terms of any Employee Benefit Plan in effect as of the date hereof (i) increase in an manner the salary, wages, welfare or other benefits of any of the current or former directors, officers, employees or other service providers of Valley or any of its Subsidiaries in effect as of June 30, 2016, except for ordinary course merit-based increases in the base salary of employees (other than directors or executive officers individuals who are party to an employment agreement or change of control agreement with, Valley or its Subsidiaries) on the anniversary of the employee s h date and consistent with past practice but not to exceed two and one-half percent (2.5%) of the employee s base salary, (ii) grant, pay or agree to pay any bonus or incentive compensation except as described on Section 5.2(g)(ii) of the Valley Disclosure Schedule, (iii) become a party to, establish, amend, alter a prior interpretation of in a manner that enhances rights or increases costs, commence participation in, terminate or commit itself to the adoption of any Employee Benefit Plan or plan that would be an Employee Benefit Plan if in effect as of the date here (iv) grant (or commit to grant) any new equity award, (v) grant, pay or increase (or commit to grant, pay or increase) any severance, retirement or termination pay, oth than in connection with termination of employment pursuant to any Employee Ben Plan in effect as of the date hereof and in the ordinary course of business consisten with past practice, (vi) accelerate the payment or vesting of, or lapsing of restrictio with respect to, any stock-based compensation, long-term incentive compensation, deferred compensation or any bonus or other incentive or deferred compensation, (cause the funding of any rabbi trust or similar arrangement or take any action to fu or in any other way secure the payment of compensation or benefits under any Employee Benefit Plan, (viii) terminate the employment or services of any officer employee who is party to a change in control agreement other than for cause, (ix) enter into any collective bargaining or other agreement with a labor organization, (forgive or issue any loans to any current or former officer, employee or director of Valley or any Subsidiary of Valley or (xi) hire any officer, employee or other servi provider except for non-executive positions in the ordinary course of business and an annual base salary not to exceed that of the Person being replaced.
- (h) Take or omit to take, any action that would prevent or impede, or could reasonable expected to prevent or impede, the Merger from qualifying as a reorganization within the meaning of Section 368(a) of the Code.

- (i) Incur or guarantee any indebtedness for borrowed money, other than overnight advances from the San Francisco Federal Reserve Bank or the Federal Home Loan Bank of San Francisco (or term advances with a term of not more than one week) i the ordinary course of business, or assume, guarantee, endorse or otherwise as an accommodation become responsible for the obligations of any other Person.
- (j) Enter into any new line of business or make any material change in any basic policies and practices with respect to pricing of loans, deposits and services, liquid management and cash flow planning, marketing, deposit origination, lending, budgeting, profit and tax planning, personnel practices or any other material aspect Valley s business or operations, except as required by Law, requested by any Regulatory Agency or in the ordinary course of business consistent with past pract and Valley s polices in effect as of the date of this Agreement, including adjustme in interest rates generally.
- (k) (i) Make any investment either by contributions to capital, property transfers or purchase of any property or assets of any Person or sell or dispose of any such investment or (ii) purchase, acquire, sell or dispose of securities of any type.

A-35

- (1) Enter into any settlement, compromise or similar agreement with respect to, any action, suit, claim, proceeding, order or investigation to which Valley or any of its Subsidiaries is or becomes a party after the date of this Agreement, which settleme compromise, agreement or action, suit, claim, proceeding, order or investigation th is settled in an amount and for consideration not in excess of \$50,000 individually \$100,000 in the aggregate and that would not (i) impose any material restriction on business of Valley or any of its Subsidiaries or (ii) create adverse precedent for cla that are reasonably likely to be material to Valley or any of its Subsidiaries.
- (m) Other than as determined to be necessary or advisable by Valley in the good far exercise of its discretion based on changes in market conditions, alter materially its interest rate or pricing fee or fee pricing policies with respect to depository account of any of its Subsidiaries or waive any material fees with respect thereto.
- (n) Except as required by applicable Law or by a Regulatory Agency, (i) implemer adopt any material change in its interest rate and other risk management policies, procedures or practices or (ii) fail to follow in all material respects, Valley s existi policies or practices with respect to managing its exposure to interest rate and othe risk.
- (o) Grant or commit to grant any extension of credit that (i) provides for zero rate, 1.23% interest rate or any other below-market rate of interest, below-market fees of other teaser rates or fees or their equivalents other than those pending loans listed Section 5.1(o) of the Valley Disclosure Schedule or (ii) would, alone or together we all other credit then committed or outstanding to the same Person and all Affiliated Persons, exceed \$1,000,000 (including, if applicable, those pending loans listed in Section 5.1(o) of the Valley Disclosure Schedule), unless, in the case of any extension of credit described in clause (ii) only, Valley provides written notice to CVB describing the pertinent terms of the proposed extension of credit, which shall include all related credit write-ups, and CVB does not object to such extension of credit withree (3) Business Days of its receipt of such notice and any other information related to the extension of credit requested by CVB.
- (p) Sell any real estate owned, charge-off any assets, make any compromises on de release any collateral on loans or commit to do any of the foregoing, if such sale, charge-off, compromise or release would exceed \$50,000 in the aggregate.
- (q) Renew or commit to renew any extension of credit that would exceed \$1,000,00 unless either (i) the extension of credit is renewed on terms substantially similar to those of the existing extension of credit or are more favorable to Valley than those the existing extension of credit and the characteristics of the borrower have not substantially changed or (ii) Valley provides written notice to CVB describing the pertinent terms of the renewal or commitment, which shall include all related credi write-ups, and CVB does not object to such renewal or commitment within three (3 Business Days of its receipt of such notice and any other information related to the proposed renewal or commitment requested by CVB.

- (r) Sell, or commit to sell, or purchase or commit to purchase, any Loan or participation in any extension of credit, or make, acquire a participation in or reacq an interest in a participation sold of any extension of credit, or renew or extend the maturity of any participation in any extension of credit, provided however, Valley may sell all (but not some) of the Excepted Loan for a sale price of \$1,000,000 or greater to a party that is not an Affiliate of Valley, an Affiliate or immediate family member of any of Valley s officers or directors, a borrower under the Excepted Loan any Affiliate or immediate family member of any such borrower. Excepted Loan means the loan specified on Section 5.2(r) of the Valley Disclosure Schedule.
- (s) Enter into any securitizations of any Loans or create any special purpose fundin or variable interest entity other than on behalf of clients.
- (t) Invest in any mortgage-backed or mortgage related securities that would be considered high-risk securities under applicable regulatory pronouncements or einto any Derivatives Transaction.

A-36

- (u) Accept, renew or roll over any brokered deposits or funds obtained through the Certificate of Deposit Account Registry Service or similar listing service or open a new deposit account that pays an effective yield that exceeds the prevailing effective yields provided by the Federal Reserve.
- (v) Make application for the opening, relocation or closing of any, or open, relocate close any, branch office, loan production office or other office or operations facility
- (w) Make any capital expenditures other than capital expenditures in the ordinary a usual course of business consistent with past practice in amounts not exceeding \$50,000 individually or \$100,000 in the aggregate.
- (x) Pay, loan or advance any amount to, or sell, transfer or lease any properties, rig or assets (real, personal or mixed, tangible or intangible) to, or enter into any arrangement or agreement with, any of its officers or directors or any of their famil members, or any Affiliates or associates (as defined under the Exchange Act) of an of its officers or directors, other than Loans originated in the ordinary course of business and, in the case of any such arrangements or agreements relating to compensation, fringe benefits, severance or termination pay or related matters, only otherwise permitted pursuant to this Section 5.2(r).
- (y) Make or commit to make any Loan or amend the terms of any Loan outstanding the date hereof to any directors, officers and principal shareholders (as such terms defined in Regulation O of the Federal Reserve (12 C.F.R. Part 215)) of Valley or Subsidiary or waive any rights with respect to any such Loan.
- (z) Change its tax or accounting policies and procedures or any method or period of accounting unless required by GAAP or a Governmental Entity.
- (aa) Change its fiscal year for tax or accounting purposes.
- (bb) Other than as required by GAAP or any Governmental Entity, reduce any material accrual or reserve, including its allowance for loan and lease losses, any contingency reserve, litigation reserve, or tax reserve, or change the methodology which such accounts generally have been maintained in accordance with past practices.
- (cc) Make any material change in any basic policies and practices with respect to loans, deposits and services, liquidity management and cash flow planning, market deposit origination, lending, reserves for loan or lease losses, budgeting, profit and planning, personnel practices or any other material aspect of its business or operati
- (dd) Grant any Person a power of attorney or similar authority.
- (ee) Take title to any real property without conducting prior thereto an environment investigation, which investigation shall disclose the absence of any suspected environmental contamination.

(ff) (i) Acquire direct or indirect control over any Person, whether by stock purchas merger, consolidation or otherwise, or (ii) make any other investment either by purchase of securities, contributions to capital, property transfers or purchase of an property or assets of any other Person, except, in either instance, in connection with foreclosure of collateral or conveyance of such collateral in lieu of foreclosure take in connection with collection of a loan in the ordinary course of business consistent with past practice and with respect to loans made to third parties who are not Affili of Valley.

(gg) Make or change any material Tax elections, change or consent to any change its method of accounting for Tax purposes (except as required by applicable Tax Law), take any position on any Tax Return filed on or after the date of this Agreem that differs from past practices, settle or compromise any Tax

A-37

liability, claim or assessment, enter into any closing agreement, waive or extend ar statute of limitations with respect to a material amount of Taxes, surrender any rigl claim a refund for Taxes, or file any material amended Tax Return.

- (hh) Take any action or omit to take any action that is intended to or would reasonabe likely to result in (i) any of Valley s representations and warranties set forth in Agreement being or becoming untrue in any respect at any time at or prior to the Effective Time, (ii) any of the conditions to the Merger set forth in Article VII not being satisfied or delayed, or (iii) a violation or breach of any provision of this Agreement, except as may be required by applicable Law.
- (ii) Agree to take, make any commitment to take, or adopt any resolutions of its Bo of Directors in support of, any of the actions prohibited by this Section 5.2.

5.3 CVB Forbearances.

- (a) Except as expressly permitted by this Agreement or with the prior written consor of Valley (which consent shall not be unreasonably withheld), during the period from the date of this Agreement to the earlier of the Effective Time or the termination of this Agreement in accordance with Article VIII, CVB shall not, and shall not perm any of its Subsidiaries to, except as may be required by applicable Law or policies imposed by any Governmental Entity:
- (i) take, or omit to take, any action that would reasonably be expected to prevent, materially impede, materially impair, or materially delay the consummation of the transactions contemplated by this Agreement including adversely affecting the abil of the parties to obtain the Requisite Regulatory Approvals of any Governmental Entity required to complete the transactions contemplated hereby, the Approvals, of the Valley Shareholder Approval or materially increase the period of time necessary obtain such Requisite Regulatory Approvals or the Valley Shareholder Approval;
- (ii) willfully take, or willfully omit to take, any action that is reasonably likely to result in any of the conditions to the Merger set forth in Section 7.1 or Section 7.3 being satisfied;
- (iii) willfully take, or willfully omit to take, any action that would prevent or imped or could reasonably be expected to prevent or impede, the Merger from qualifying reorganization within the meaning of Section 368(a) of the Code; and
- (iv) except as would not be reasonably expected to have a Material Adverse Effect with respect to CVB and Citizens Business Bank, willfully fail to file all Parent SE Reports and any other filings required to be filed with any applicable Governmenta Entity.

ARTICLE VI

ADDITIONAL AGREEMENTS

6.1 Regulatory Matters.

(a) CVB and Valley shall reasonably promptly prepare and shall use their commercially reasonable efforts to file with the SEC, within forty-five (45) days of the date of this Agreement, the Form S-4, which shall include the Proxy Statement a prospectus. Each of CVB and Valley shall use its commercially reasonable effort have the Form S-4 declared effective under the Securities Act as promptly as practicable after such filing, and Valley shall thereafter mail or deliver the Proxy Statement to its shareholders. CVB shall also use its reasonable best efforts to obta all necessary state securities Law or blue sky permits and approvals required to out the transactions contemplated by this Agreement, and Valley shall furnish all information concerning Valley and the holders of Valley Common Stock as may be reasonably requested in connection with any such action.

A-38

(b) The parties shall reasonably cooperate with each other and use their respective commercially reasonable efforts to promptly prepare and file all necessary documentation, to effect all applications, notices, petitions and filings, to obtain as promptly as practicable all permits, consents, approvals and authorizations of all th parties and Governmental Entities that are necessary or advisable to consummate the Merger, the Bank Merger and the other transactions contemplated by this Agreeme as soon as reasonably possible, and to comply with the terms and conditions of all such permits, consents, approvals, and authorizations of all such third parties or Governmental Entities. CVB shall use its commercially reasonable efforts to make initial requisite regulatory filings within thirty (30) days of this Agreement (other t any notice to the Federal Reserve under its regulations, which will be filed in accordance with the timing contemplated by such regulations or the request of the staff of the Federal Reserve). Valley and CVB shall have the right to review in advance and, to the extent practicable, each will consult the other on, in each case subject to applicable Laws, all the non-confidential information relating to Valley CVB (excluding any confidential financial information relating to individuals), as t case may be, and any of their respective Subsidiaries, that appear in any filing mad with, or written materials submitted to, any third party or any Governmental Entity connection with the transactions contemplated by this Agreement. In exercising the foregoing right, each of the parties shall act reasonably and as promptly as practical The parties shall consult with each other with respect to the obtaining of all permit consents, approvals and authorizations (collectively the Approvals) of all third provals and Governmental Entities necessary or advisable to consummate the Merger, the Bank Merger and the other transactions contemplated by this Agreement and each party will keep the other reasonably apprised of the status of matters relating to suc Approvals and the completion of the Merger, the Bank Merger and the other transactions contemplated by this Agreement. Each party shall consult with the oth in advance of any meeting or conference with any Governmental Entity in connect with the Merger, the Bank Merger and the other transactions contemplated by this Agreement.

(c) Each of CVB and Valley shall, upon request, furnish to the other all information concerning itself, its Subsidiaries, directors, officers and shareholders and such oth matters as may be reasonably necessary or advisable in connection with the Proxy Statement, the Form S-4 or any other statement, filing, notice or application made or on behalf of CVB, Valley or any of their respective Subsidiaries to any Governmental Entity in connection with the Merger and the other transactions contemplated by this Agreement. Each of CVB and Valley agrees, as to itself and i Subsidiaries, that none of the information supplied or to be supplied by it for inclusion or incorporation by reference in (i) the Form S-4 will, at the time the Form S-4 and each amendment or supplement thereto, if any, becomes effective under the Securi Act, contain any untrue statement of a material fact or omit to state any material fa required to be stated therein or necessary to make the statements therein not misleading and (ii) the Proxy Statement and any amendment or supplement thereto will, at the date of mailing to shareholders and at the time of Valley s meeting of i shareholders to consider and vote upon approval of this Agreement, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, in the light of the

circumstances under which such statement was made, not misleading. Each of CVI and Valley further agrees that if it becomes aware that any information furnished b would cause any of the statements in the Form S-4 or the Proxy Statement to be fall or misleading with respect to any material fact, or to omit to state any material fact necessary to make the statements therein not false or misleading, to promptly inforthe other party thereof and to take appropriate steps to correct the Form S-4 or the Proxy Statement, as applicable.

(d) Notwithstanding the foregoing, nothing contained herein shall be deemed to require CVB or any of its Subsidiaries to take any action, or commit to take any action, or agree to any condition or restriction, in connection with obtaining the foregoing permits, consents, approvals and authorizations of Governmental Entities that would reasonably be likely, in each case following the Effective Time (but regardless when the action, condition or restriction is to be taken or implemented), have a Material Adverse Effect on CVB or Citizens Business Bank (measured on a scale relative to Valley) or materially restrict or impose a material burden on CVB any of its Subsidiaries (including, after the Effective Time, Valley and its Subsidiaries) in connection with the transactions contemplated hereby or with respect to the business or operation of CVB, Citizens Business Bank or any of their respect Subsidiaries (including, after the Effective Time, Valley and its Subsidiaries) (a Materially Burdensome Regulatory Condition).

A-39

- (e) Each of CVB and Valley shall promptly advise the other upon receiving any communication from any Governmental Entity the consent or approval of which is required for consummation of the Merger and the other transactions contemplated this Agreement that causes such party to believe that there is a reasonable likelihoo that any Requisite Regulatory Approval will not be obtained or that the receipt of a such approval may be materially delayed.
- 6.2 <u>Reasonable Best Efforts</u>. Subject to the terms and conditions of this Agreement each of Valley and CVB agrees to cooperate with the other and use its commercial reasonable efforts in good faith to take, or cause to be taken, all actions, and to do, cause to be done, all things necessary, proper or desirable, or advisable on its part under this Agreement or under applicable Laws to consummate and make effective Merger, the Bank Merger and the other transactions contemplated hereby as prompas practicable, including the satisfaction of the conditions set forth in Article VII hereof.

6.3 Access to Information.

(a) Upon reasonable notice and subject to applicable Laws, Valley shall afford to the officers, employees, accountants, counsel, advisors, agents and other representative of CVB, reasonable access, during normal business hours during the period prior to the Effective Time or the termination of this Agreement in accordance with its term to all its properties, books, contracts, commitments, personnel and records, and, during such period, Valley shall, and shall cause its Subsidiaries to, make available CVB (i) a copy of each report, schedule, registration statement and other document filed or received by it or any of its Subsidiaries during such period pursuant to the requirements of federal securities Laws or federal or state banking Laws (other tha reports or documents that Valley or its Subsidiaries are not permitted to disclose un applicable Law), (ii) all other information concerning its and its Subsidiaries busi properties and personnel as CVB may reasonably request and (iii) access to the necessary information (including Valley s own good faith estimates as available a third-party reports, if any, commissioned by Valley at CVB s request) in order to prepare a good faith estimate of the potential impact of Sections 280G and 4999 of Code with respect to amounts potentially payable to senior executives of Valley an its Subsidiaries in connection with the consummation of the transactions contemplated in the consummation of the consummatio by this Agreement. Upon the reasonable request of Valley, CVB shall furnish such reasonable information about its and its Subsidiaries businesses as is reasonably relevant to Valley and its shareholders in connection with the Merger, the Bank Merger and the other transactions contemplated by this Agreement. Neither Valley any of its Subsidiaries shall be required to provide access to or to disclose information to the extent such access or disclosure would jeopardize the attorney-client privileg of such party or its Subsidiaries or contravene any Law or binding agreement enter into prior to the date of this Agreement. The parties shall make appropriate substitu disclosure arrangements under circumstances in which the restrictions of the preceding sentence apply. In addition to the foregoing, no later than the twenty-fift (25th) calendar day of each month, Valley shall provide CVB with a listing of all no and renewed loans and loan modifications, loan payoffs and loan purchases with a balance of \$100,000 or greater that were completed or made by VBB during the

preceding month. CVB and Citizens Business Bank shall use commercially reason efforts to minimize any interference with Valley s regular business operations dur any such access to Valley s property, books and records.

(b) As soon as reasonably practicable after they become available, but in no event more than twenty-five (25) days after the end of each calendar month ending after the date hereof, Valley will furnish to CVB, (i) consolidated financial statements (including balance sheets, statements of operations and stockholders equity) of it VBB (to the extent available) as of and for such month then ended, (ii) consolidating financial statements (including balance sheets, statements of operations and stockholders equity) of it and VBB (to the extent available) as of and for such most then ended, (iii) Valley s and VBB s internal management reports showing actual financial performance against plan, and (iv) to the extent permitted by applicable L any reports provided to its Valley s and VBB s Boards of Directors or any committee of relating to the financial performance and risk management of Valley or VB

A-40

- (c) All nonpublic information and materials provided pursuant to this Agreement s be subject to the provisions of the Confidentiality Agreement entered into between parties dated May 4, 2016 (the <u>Confidentiality Agreement</u>).
- (d) No investigation by a party hereto or its representatives shall affect or be deemed to modify or waive any representations, warranties or covenants of the other party forth in this Agreement.

6.4 Shareholder Approval. Valley agrees to take, in accordance with applicable La and Valley Articles and Valley Bylaws, all action necessary to convene as soon as practicable after the Form S-4 is declared effective (but in no event later than sixty (60) days after the Form S-4 is declared effective), a special meeting or meetings o shareholders duly called and held for such purposes (the Valley Shareholder Mee to consider and to obtain the Valley Shareholder Approval. Subject to Section 6.9(b)-(c), the Board of Directors of Valley shall at all times prior to and during su special meeting recommend such approval and shall use its reasonable best efforts solicit such approval by its shareholders (the <u>Valley Board</u> <u>Recommendation</u>). Without limiting the generality of the foregoing, unless this Agreement has terminated in accordance with its terms, this Agreement and the Merger shall be submitted to Valley s shareholders at Valley Shareholder Meeting whether or not (x) the Board of Directors of Valley shall have effected an Adverse Change of Recommendation or (y) any Company Acquisition Proposal shall have been publicly proposed or announced or otherwise submitted to Valley or any of it advisors. CVB may require Valley to adjourn, delay or postpone Valley Sharehold Meeting once for a period not to exceed thirty (30) calendar days (but prior to the o that is two (2) Business Days prior to the End Date) to solicit additional proxies necessary to obtain Valley Shareholder Approval. Valley shall not, without the prid written consent of CVB, adjourn or postpone Valley Meeting; provided that Valley may, without the prior written consent of CVB, adjourn or postpone the Valley Shareholder Meeting (A) if on the date on which the Valley Shareholder Meeting i originally scheduled, Valley has not received proxies representing a sufficient num of shares of Valley Common Stock to obtain the Valley Shareholder Approval, Va shall adjourn the Valley Shareholder Meeting until such date as shall be mutually agreed upon by Valley and CVB, which date shall not be less than five (5) Busines Days nor more than ten (10) Business Days after the date of adjournment, and subj to the terms and conditions of this Agreement shall continue to use all reasonable be efforts, together with its proxy solicitor, to assist in the solicitation of proxies from shareholders relating to the Valley Shareholder Approval, (B) after consultation with CVB, if the failure to adjourn or postpone the Valley Shareholder Meeting would reasonably be expected to be a violation of applicable Law for the distribution of a required supplement or amendment to the Proxy Statement, or (C) after consultation with Valley, for a single period not to exceed ten (10) Business Days, to solicit additional proxies if necessary to obtain the Valley Shareholder Approval. Once Valley has established the record date for determining shareholders of Valley entit to vote at the Valley Shareholder Meeting, Valley shall not change such record dat establish a different record date for the Valley Shareholder Meeting without the pri written consent of CVB, unless required to do so by applicable Law, the Valley Articles or the Valley Bylaws.

6.5 <u>Nasdaq Listing</u>. Prior to the Closing Date, CVB shall file with Nasdaq any required notices or forms with respect to listing the shares of CVB Common Stock be issued in the Merger.

6.6 Employee Matters.

(a) Except as otherwise provided in this Agreement or any Employee Benefit Plan, Employee Benefit Plans of Valley will be discontinued and employees of Valley we become employees of Citizens Business Bank on the Closing Date (each such employee, a <u>Covered Employee</u>) shall become eligible for the employee benefit of Citizens Business Bank on the same terms as such plans and benefits are general offered from time to time to employees of Citizens Business Bank in comparable positions with Citizens Business Bank. Effective as of a date no later than the day immediately preceding the Closing Date, Valley shall file all required reports with respect to the Valley Business Bank 401(k) Plan (the <u>Valley 401(k) Plan</u>) include without limitation a Form 5500 for 2015, and shall terminate the Valley 401(k) Plan and any other Employee Benefit Plans identified by CVB to Valley and shall province CVB with evidence that the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated (effective no limitation to the Valley 401(k) Plan has been terminated to the Valley 401(k) Plan has b

A-41

than the day immediately preceding the Closing Date) pursuant to resolutions of Valley Board of Directors. The form and substance of such resolutions shall be subto the review and reasonable approval of CVB. Valley shall also take such other actions in furtherance of terminating the Valley 401(k) Plan as CVB may reasonab require. For purposes of determining such Covered Employees eligibility and ves (but not for benefit accruals) under the employee benefit plans of Citizens Business Bank and entitlement to severance benefits and vacation entitlement (to the extent permitted by applicable Law), Citizens Business Bank shall recognize such employ years of service with Valley beginning on the date such employees commenced employment with Valley through the Closing Date. Citizens Business Bank shall ta any actions necessary to allow participants in the Valley 401(k) Plan who become eligible to participate in the Citizens Business Bank 401(k) Plan to make rollover contributions (but not including the rollover of participant loans in accordance with the terms and conditions of the Citizens Business Bank 401(k) Plan.

- (b) Valley s Executive Supplemental Compensation Agreements, Salary Continua Agreements and Split Dollar Arrangements set forth in Section 6.6(b) of the Valley Disclosure Schedule shall continue in effect and shall be maintained by the Survivi Corporation following the Closing Date.
- (c) Subject to the requirements of applicable Law, CVB shall cause Citizens Busin Bank to take such commercially reasonable actions as are necessary to cause the gr health plan maintained by Citizens Business Bank or an Affiliate thereof, and applicable insurance carriers, third party administrators and any other third parties, the extent such group health plan is made available to employees of Valley on the Closing Date, to (i) waive any evidence of insurability requirements, waiting period and any limitations as to preexisting medical conditions under the group health pla applicable to employees of Valley on the Closing Date and their spouses and eligib dependents (but only to the extent that such preexisting condition limitations did no apply or were satisfied under the group health plan maintained by Valley prior to the Closing) and (ii) provide employees of Valley on the Closing Date with credit, for calendar year in which the Closing occurs, for the amount of any out-of-pocket expenses and copayments or deductible expenses that are incurred by them during calendar year in which the Closing occurs under a group health plan maintained by Citizens Business Bank or any of its Affiliates (unless such employees of Valley w when employed by Valley, participants in a Health Maintenance Organization heal plan).
- (d) Without limiting the generality of Section 9.10, the provisions of this Section 6 are solely for the benefit of the parties to this Agreement, and no current or former employee, independent contractor or any other individual associated therewith shall regarded for any purpose as a third-party beneficiary of this Agreement. In no ever shall the terms of this Agreement be deemed to (i) establish, amend, or modify any Employee Benefit Plan, or any plan, program, agreement or arrangement maintaine or sponsored by CVB, Valley or any of their respective Affiliates; (ii) alter or limit ability of CVB or any of its Subsidiaries (including, after the Closing Date, the Surviving Corporation and its Subsidiaries) to amend, modify or terminate any Employee Benefit Plan, employment agreement or any other benefit or employment

plan, program, agreement or arrangement after the Closing Date; or (iii) confer upon any current or former employee, independent contractor or other service provider a right to employment or continued employment or continued service with CVB or a of its Subsidiaries (including, following the Closing Date, the Surviving Corporation and its Subsidiaries), or constitute or create an employment or other agreement with any employee, independent contractor or other service provider.

(e) For a period of one (1) year following the Closing Date, CVB shall cause Citize Business Bank to provide a lump sum severance payment to each Covered Employ (other than any Covered Employee having an agreement providing for severance) vis terminated by Citizens Business Bank other than for Cause in an amount equal to such employee s regularly scheduled base salary or base wages at the time of termination of employment for a period equal to (i) two weeks plus (ii) one week feeach full year of continuous service completed by such employee with Valley and Citizens Business Bank at the time of termination of employment (without proration for partial years), subject to applicable tax withholding and subject further the employee signing a release of claims in favor of CVB and Citizens Business Bank allowing it to become effective. For

A-42

purposes of this Section 6.6(e), <u>Cause</u> shall mean the employee s personal disher willful misconduct, breach of fiduciary duty involving personal profit, failure to comply with any valid and legal directive of CVB, Citizens Business Bank, Valley and/or VBB, failure to perform stated duties, violation of any law, rule or regulatio (other than traffic violations or similar offenses) or order of any Governmental Ent

6.7 <u>Indemnification</u>; <u>Directors and Officers Insurance</u>.

- (a) From and after the Effective Time, CVB shall indemnify and hold harmless each present and former director and officer of Valley and its Subsidiaries (in each case only when acting in such capacity) (collectively, the <u>Indemnified Parties</u>) against costs or expenses (including reasonable attorneys fees), judgments, fines, losses, claims, damages or liabilities incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative arising out of actions or omissions occurring at or prior to the Effective Time, including the transactions contemplated by this Agreement, to the extent they are indemnified on the date hereof, to the fullest extent permitted under applicable Law and CVB shall also advance expenses as incurred to the fullest extent permitted un applicable Law; <u>provided</u> that the Indemnified Party to whom expenses are advance provides an undertaking to repay such advances if it is ultimately determined that sendemnified Party is not entitled to indemnification.
- (b) Each of CVB and Valley shall use commercially reasonable efforts to cause the persons serving as officers and directors of Valley immediately prior to the Effecti Time to be covered by a directors and officers liability insurance tail policy obta from Valley s current insurance carrier or, if CVB can obtain a policy on more favorable pricing, an insurance carrier identified by CVB (the <u>D&O Insurance</u>), substantially the same coverage and amounts containing terms and conditions which are generally not less advantageous than Valley s current policy with respect to ac omissions occurring prior to the Effective Time that were committed by such office and directors in their capacity as such for a period of six (6) years provided, however that in no event shall the cost of the D&O Insurance policy be more than two hund and fifty percent (250%) of the current amount expended on an annual basis by Va for its current policy; provided, further, that if the D&O Insurance cannot be obtain as provided by this Section 6.7(b), CVB shall use its commercially reasonable effo to obtain as much comparable insurance as is available; provided, further, that office and directors of Valley may be required to make application and provide customar representations and warranties to an insurance carrier for the purpose of obtaining D&O Insurance.
- (c) Any Indemnified Party wishing to claim indemnification under Section 6.7(a), upon learning of any claim, action, suit, proceeding or investigation described above will promptly notify CVB thereof; <u>provided</u> that failure to so notify will not affect obligations of CVB or under Section 6.7(a) unless and to the extent that CVB is actually and materially prejudiced as a consequence.
- (d) The provisions of this Section 6.7 are intended to be for the benefit of, and shall enforceable by, each director or officer of Valley and its Subsidiaries and his or he

heirs and representatives and shall be binding on the successors and assigns of CV.

(e) In the event that either CVB or any of its successors or assigns (i) consolidates with or merges into any other person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers all or substantially all of its properties and assets to any person, then, and in each such caproper provision shall be made so that the successors and assigns of CVB shall assume the obligations set forth in this Section 6.7.

6.8 <u>Special Dividend</u>. Valley shall, in accordance with Applicable Law, declare a one-time cash dividend payable to holders of record of outstanding Valley Commo Stock as of the Closing Date in an aggregate amount that shall not exceed the amount by which the Pre-Dividend Common Equity exceeds the greater of (a) the Closing Tangible Common Equity Minimum or (b) the amount of tangible common equity capital necessary for Valley to achieve a tangible common equity capital ratio of 8. (determined in accordance with GAAP,

A-43

regulatory accounting requirements and Valley s policies and procedures as of the of this Agreement) as of the final day of the month immediately preceding the Clos (the <u>Special Dividen</u>d), but only if and to the extent that such dividend would no cause Valley to fail to satisfy any condition to closing set forth in Section 7.2. As u herein, Pre-Dividend Common Equity shall mean Valley s tangible common shareholders equity calculated in accordance GAAP, regulatory accounting requirements and Valley s policies and procedures as of the date of this Agreement calculated as of the final day of the month immediately preceding the Closing, with giving effect to the Special Dividend, but prior to any pre-closing adjustments effectuated in accordance with Section 6.14 and the Transaction Costs. Valley shall seek, and shall cause VBB to seek, any approvals, consents or waivers necessary to enable Valley to pay the Special Dividend within 15 days of the date of this Agreement, and each of Valley and VBB shall have received, any such approvals, consents or waivers from any Governmental Entities necessary to enable Valley to the Special Dividend without violation of Applicable Law on the Closing Date. Th amount of the Special Dividend payable on each share of Valley Common Stock outstanding as of the record date for the Special Dividend shall be equal to the quotient of the aggregate amount of the Special Dividend divided by the number of shares of Valley Common Stock outstanding on the Closing Date. Valley shall dep the aggregate amount of the Special Dividend with its stock transfer agent at least three (3) Business Days prior to the Closing and shall instruct its stock transfer age to pay the pro rata amount of the Special Dividend to holders of record of outstand Valley Common Stock as of the Closing Date.

6.9 No Solicitation.

- (a) Valley agrees that none of it or any of its officers, directors and employees will and it will cause its and its officers, directors, agents, representatives, advisors and Affiliates not to, initiate, solicit, encourage or knowingly facilitate any inquiries or making of proposals with respect to, or engage in any negotiations concerning, or provide any confidential or nonpublic information or data to, or have any discussion with, any Person relating to, any Acquisition Proposal made by a Person (other that CVB or Citizens Business Bank) to Valley or any of its officers, directors, agents, representatives, advisors and Affiliates (a <u>Valley Acquisition Proposal</u>) or other facilitate any effort to attempt or make or implement a Valley Acquisition Proposa
- (b) Notwithstanding anything to the contrary contained in this Agreement, if at any time after the date hereof and prior to, but not after, obtaining the Valley Sharehold Approval, Valley receives a bona fide Valley Acquisition Proposal and such Valley Acquisition Proposal did not result from a breach of this Section 6.9, and the Board Directors of Valley concludes, in good faith, that such Valley Acquisition Proposal constitutes, or is reasonably expected to result in, a Valley Superior Proposal, then Valley and its Board of Directors may, and may permit its representatives to, furnisor cause to be furnished nonpublic information and participate in such negotiations discussions to the extent that the Board of Directors of Valley concludes in good fa (and after conferring with outside legal counsel and its financial advisors) that the failure to take such action would breach or would be more likely than not to result breach of its fiduciary duties to Valley s shareholders; provided that prior to provide

any nonpublic information permitted to be provided pursuant to the foregoing provided or engaging in any negotiations, it shall have entered into a confidentiality agreement with such third party on terms no less restrictive in the aggregate to the counterpart than those contained in the Confidentiality Agreement and which expressly permit Valley to comply with its obligations pursuant to this Section 6.9. Subject to the foregoing and Section 6.9(c) below, Valley will immediately cease and cause to be terminated any activities, discussions or negotiations conducted before the date of Agreement with any persons other than CVB with respect to any Valley Acquisition Proposal and will use its commercially reasonable efforts, subject to applicable La to (i) enforce any confidentiality or similar agreement relating to a Valley Acquisit Proposal and (ii) within three (3) Business Days after the date hereof, request and confirm the return or destruction of any confidential information provided to any Person (other than CVB and its Affiliates) pursuant to any such confidentiality or similar agreement). Valley will promptly (and in any event within two (2) Business Days) advise CVB following receipt of any Valley Acquisition Proposal, any discussions or negotiations are sought to be initiated or continued or any request for nonpublic information or inquiry that would reasonably be expected to lead to any Valley Acquisition Proposal and the substance thereof (including the identity of the Person making such Valley Acquisition Proposal), and will keep CVB promptly apprised of any related developments,

A-44

discussions and negotiations (including the terms and conditions of any such reque inquiry or Valley Acquisition Proposal, or all amendments or proposed amendment thereto) on a current basis (it being understood that for the avoidance of doubt that such communications to CVB shall be deemed an Adverse Change of Recommendation). Valley agrees that it shall contemporaneously provide to CVB a confidential or nonpublic information concerning Valley that may be provided to a other Person in connection with any Valley Acquisition Proposal which has not previously been provided to CVB.

- (c) (i) None of the Board of Directors of Valley or any committee thereof shall: (A except as expressly permitted by, and after compliance with, Section 6.9(c)(ii)(B) hereof, make any Adverse Change of Recommendation; or (B) cause or permit Valto enter into any letter of intent, memorandum of understanding, agreement in principle, acquisition agreement, merger agreement or other agreement (other than confidentiality agreement referred to in Section 6.9(b) entered into in compliance values of Section 6.9(b)) relating to any Valley Acquisition Proposal made to Valley.
- (ii) Notwithstanding anything to the contrary contained in this Agreement, at any tiprior to, but not after, obtaining the Valley Shareholder Approval, the Board of Directors of Valley may make an Adverse Change of Recommendation or terminate this Agreement pursuant to Section 8.1(d) if Valley receives a Valley Acquisition Proposal that is not withdrawn and the Board of Directors of Valley concludes in g faith that such Valley Acquisition Proposal constitutes a Valley Superior Proposal; provided that:
- (1) The Board of Directors of Valley concludes in good faith (and based on the advof outside legal counsel) that failure to take such actions would be more likely than not to result in a violation of its fiduciary duties under applicable Law;
- (2) Valley provides CVB prior written notice at least five (5) Business Days prior taking such action, which notice shall state that the Board of Directors of Valley has received a Valley Superior Proposal and, absent any revision to the terms and conditions of this Agreement, the Board of Directors of Valley has resolved to effer an Adverse Change of Recommendation or to terminate this Agreement pursuant to Section 8.1(d), as applicable, which notice shall specify the basis for such Adverse Change of Recommendation or termination, including the material terms of Valley Superior Proposal (a Notice of Superior Proposal) (it being understood for the avoidance of doubt that such Notice of Superior Proposal shall not be deemed an Adverse Change of Recommendation);
- (3) during such five (5)-Business Day period (and, with respect to any modification to any Valley Acquisition Proposal, an additional five (5) Business Day periods), Valley negotiates in good faith with CVB (to the extent that CVB wishes to negotiate to enable CVB to make an improved offer that is favorable to the shareholders of Valley so that such Valley Acquisition Proposal would cease to constitute a Valley Superior Proposal; and

- (4) at the end of such five (5)-Business Day period(s) (or such earlier time that CV advises Valley that it no longer wishes to negotiate to amend this Agreement), the Board of Directors of Valley, after taking into account any modifications to the ter of this Agreement and the Merger agreed to by CVB after receipt of such notice, continues to believe that such Valley Acquisition Proposal constitutes a Valley Superior Proposal.
- (d) As used in this Agreement, <u>Acquisition Proposal</u> means a tender or exchange proposal for a merger, consolidation or other business combination involving the p or any of its Subsidiaries or any proposal or offer to acquire in any manner more the 10% of the voting power in, or more than 10% of the fair market value of the busine assets or deposits of, the party or any of its Subsidiaries or any public announcement of a proposed plan or intention to do any of the foregoing or any agreements to enging any of the foregoing, other than the transactions contemplated by this Agreement and any sale of whole loans and securitizations in the ordinary course. As used in the Agreement, <u>Valley Superior Proposal</u> means an unsolicited bona fide written Valequisition Proposal that did not otherwise result from a breach of this

A-45

Section 6.9, that the Board of Directors of Valley concludes in good faith to be more favorable from a financial point of view to its shareholders than the Merger and the other transactions contemplated hereby and to be reasonably capable of being consummated on the terms proposed, (i) after receiving the advice of its financial advisors (who shall be Vining Sparks IBG, LP or another a nationally recognized investment banking firm), (ii) after taking into account the likelihood of consummation of such transaction on the terms set forth therein and (iii) after taking into account all legal, financial (including the financing terms of any such proposal regulatory and other aspects of such proposal (including any expense reimburseme provisions and conditions to closing) and any other relevant factors permitted unde applicable Law, and after taking into account any amendment or modification to the Agreement agreed to by CVB; provided that for purposes of the definition of Valley Acquisition Proposal shall be deemed to be references to 50%.

- 6.10 <u>Takeover Laws</u>. No party will take any action that would cause the transaction contemplated by this Agreement to be subject to requirements imposed by any Takeover Law and each of them will take all necessary steps within its control to exempt (or ensure the continued exemption of) the transactions contemplated by th Agreement from, or if necessary challenge the validity or applicability of, any applicable Takeover Law, as now or hereafter in effect. If any Takeover Laws becomes applicable to this Agreement or the transactions contemplated hereby or thereby, including the Merger, the parties shall take all reasonable action necessary ensure that the transactions contemplated by this Agreement, including the Merger may be consummated as promptly as practicable on the terms contemplated hereby and otherwise to minimize the effect of such Takeover Law on this Agreement or t transactions contemplated hereby, including the Merger.
- 6.11 Schedule Updates. Not later than the twenty-fifth (25th) day of each calendar month between the date of this Agreement and the Closing Date, and at least seven Business Days prior to the Closing, Valley shall provide to CVB a supplemental Valley Disclosure Schedule reflecting any required changes thereto between the da of this Agreement and the Closing Date which would have been required to be set forth or described in such disclosure schedule or which is necessary to correct any information in any Valley representation or warranty or such disclosure schedule which has been rendered inaccurate thereby, with Schedule 3.25(e) to be updated we respect to Loans as of the last Business Day of the last calendar month prior to the Closing Date. Delivery of such supplemental disclosure schedules shall not cure a breach or modify a representation or warranty of this Agreement or for purposes of determining the satisfaction of any conditions to consummation of the transactions contemplated by this Agreement, or otherwise affect the respective rights, obligation representations, warranties, covenants or agreements of the parties to this Agreement.
- 6.12 <u>Notification of Certain Matters</u>. In addition to the covenants set forth in Section 6.3(b) herein, Valley and CVB will give prompt notice to the other of any fact, everor circumstance known to it that (a) is reasonably likely, individually or taken toge with all other facts, events and circumstances known to it, to result in any Material Adverse Effect with respect to it or (b) would cause or constitute a breach of any o

representations, warranties, covenants or agreements contained herein that reasona could be expected to give rise, individually or in the aggregate, to a failure of a condition in Article VII.

6.13 Third-Party Agreements.

- (a) The parties shall use commercially reasonable efforts to obtain (i) the consents waivers required to be obtained from any third parties in connection with the Merg and the other transactions contemplated hereby (in such form and content as mutua agreed by the parties) promptly after the date of this Agreement and (ii) the cooperation of such third parties to effect a smooth transition in accordance with the parties timetable at or after the Effective Time. Valley shall cooperate with CVB minimizing the extent to which any contracts to which Valley is a party will contin in effect following the Effective Time, in addition to complying with the prohibition Section 5.2.
- (b) Without limiting the generality of Section 6.13(a), Valley shall use commercial reasonable efforts to provide data processing, item processing and other processing support or outside contractors to assist CVB in

A-46

performing all tasks reasonably required to result in a successful conversion of the data and other files and records of Valley to CVB s production environment, in su manner sufficient to ensure that a successful conversion will occur at the time (on after the Effective Time) mutually agreed by the parties, subject to any applicable Laws, including Laws regarding the exchange of information and other Laws regarding competition. Among other things, Valley shall:

- (i) reasonably cooperate with CVB to establish a mutually agreeable project plan to effectuate the conversion;
- (ii) use its commercially reasonable efforts to have Valley s outside contractors continue to support both the conversion effort and its ongoing needs until the conversion can be established;
- (iii) provide, or use its commercially reasonable efforts to obtain from any outside contractors, all data or other files and layouts reasonably requested by CVB for use planning the conversion, as soon as reasonably practicable;
- (iv) provide reasonable access to Valley s personnel and facilities and, with the co of its outside contractors, its outside contractors personnel and facilities, to enable conversion effort to be completed on schedule; and
- (v) give notice of termination, conditioned upon the completion of the transactions contemplated by this Section 6.13(b), of the contracts of outside data, item and oth processing contractors or other third-party vendors to which Valley are bound whe directed to do so by CVB.
- (c) CVB agrees that all actions taken pursuant to this Section 6.13 shall be taken in manner intended to minimize disruption to the customary business activities of Value of
- (d) Valley shall use its reasonable best efforts to obtain the consents, approvals or waivers from the agreements set forth in Section 6.13(d) of the Valley Disclosure Schedule and lessor estoppel certificates, in a form reasonably satisfactory to CVB all of Valley Leased Property not earlier than thirty (30) calendar days nor later that five (5) calendar days prior to the Closing Date.
- 6.14 <u>Pre-Closing Adjustments</u>. In addition to the adjustments otherwise required by this Agreement, at or before the Effective Time, Valley shall make such additional accounting entries or adjustments, including charge-offs of loans, as CVB shall dir in compliance with applicable Laws, in order to implement its plans following the Closing or to reflect expenses and costs related to the Merger; <u>provided</u>, <u>however</u>, Valley shall not be required to take such additional actions more than one (1) Busin Day prior to the Closing or prior to the time CVB agrees in writing that all of the conditions to its obligations to close as set forth in Article VII have been satisfied waived, and (ii) based upon consultation with counsel and accountants for Valley, such additional adjustment shall (A) require any filing with any Governmental Ent or (B) violate any Law applicable to Valley. Any such pre-closing adjustments that CVB shall direct shall be disregarded for purposes of determining the satisfaction of

the conditions set forth in Section 7.2(j) hereof, the Closing Tangible Common Equity, and any as reductions in the Aggregate Cash Amount.

6.15 Shareholder Litigation and Protests. Valley shall promptly advise CVB orally and in writing of any shareholder litigation or community-based protests against Valley or its directors relating to this Agreement, the Merger or any of the other transactions contemplated hereby and thereby (an _Adverse Shareholder Action) shall keep CVB fully informed regarding any such shareholder litigation, including providing all relevant documentation. Valley shall consult with CVB and give good faith consideration of its comments and advice and give CVB the opportunity to participate in the defense or settlement of any such litigation, provided that CVB sl pay its own expenses, subject to applicable Law. No settlement in connection with such Adverse Shareholder Action shall be agreed to without CVB s prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed

A-47

ARTICLE VII

CONDITIONS PRECEDENT

- 7.1 <u>Conditions to Each Party</u> <u>s Obligation to Effect the Merger</u>. The respective obligations of the parties to effect the Merger shall be subject to the satisfaction at prior to the Effective Time of the following conditions:
- (a) Shareholder Approval. The Valley Shareholder Approval shall have been obtain
- (b) <u>Form S-4</u>. The Form S-4 shall have become effective under the Securities Act a no stop order suspending the effectiveness of the Form S-4 shall have been issued no proceedings for that purpose shall have been initiated or threatened by the SEC.
- (c) <u>No Injunctions or Restraints; Illegality</u>. No order, injunction or decree issued by any court or agency of competent jurisdiction or other Law preventing or making illegal the consummation of the Merger or any of the other transactions contemplate by this Agreement shall be in effect.
- 7.2 <u>Conditions to Obligations of CVB and Citizens Business Bank</u>. The obligations CVB to effect the Merger are also subject to the satisfaction, or waiver by CVB, at prior to the Effective Time, of the following conditions:
- (a) Representations and Warranties. The representations and warranties of Valley s forth in this Agreement shall be true and correct as of the date of this Agreement as as of the Effective Time as though made on and as of the Effective Time (except the representations and warranties that by their terms speak specifically as of the date of this Agreement or another date shall be true and correct as of such date); provided, however, that no representation or warranty of Valley (other than the representation and warranties set forth in (i) Sections 3.1(a), 3.2(b), 3.7, 3.8, 3.10, 3.11, 3.15, 3.16 3.18, 3.23, 3.24, 3.25, 3.26, 3.28, 3.31, and 3.33 which shall be true and correct in material respects, and (ii) Section 3.2(a), 3.3(a), and 3.3(b), which shall be true and correct in all respects) shall be deemed untrue or incorrect for purposes hereunder consequence of the existence of any fact, event or circumstance inconsistent with s representation or warranty, unless such fact, event or circumstance, individually or taken together with all other facts, events or circumstances inconsistent with any representation or warranty of Valley has had or would reasonably be expected to re in a Material Adverse Effect on Valley or the Surviving Corporation; provided furt that for purposes of determining whether a representation or warranty is true and correct for purposes of this Section 7.2(a) any qualification or exception for, or reference to, materiality (including the terms material, materially, in all mat Material Adverse Effect or similar terms or phrases) in any such representation or warranty shall be disregarded; and CVB shall have received a certificate signed on behalf of Valley by the Chief Executive Officer or the Chief Financial Officer of Valley to the foregoing effect.
- (b) <u>Performance of Obligations of Valley</u>. Valley shall have performed in all mater respects all obligations required to be performed by it under this Agreement at or p

to the Effective Time; and CVB shall have received a certificate signed on behalf of Valley by the Chief Executive Officer or the Chief Financial Officer of Valley to steffect.

- (c) <u>Tax Opinion</u>. CVB shall have received an opinion of Manatt, Phelps & Phillips LLP, dated the Closing Date and based on facts, representations and assumptions described in such opinion, to the effect that the Merger will qualify as a reorganiz within the meaning of Section 368(a) of the Code. In rendering such opinion, Mana Phelps & Phillips, LLP, will be entitled to receive and rely upon customary certificates and representations of officers of CVB and Valley.
- (d) <u>Regulatory Approvals</u>. (i) All consents, registrations, approvals, permits and authorizations required to be obtained prior to the Effective Time by Valley or CV (or any of its Subsidiaries) from the Federal Reserve, the FDIC, the Commissioner and Nasdaq and (ii) any other regulatory approvals set forth in Sections 3.4 and 4.4 the failure of which to be obtained would reasonably be expected to have a Materia Adverse Effect

A-48

on CVB or Valley, in each case required to consummate the transactions contemple by this Agreement, including the Merger and the Bank Merger, shall have been obtained and shall remain in full force and effect and all statutory waiting periods i respect thereof shall have expired (all such approvals and the expiration of all such waiting periods being referred to as the Regulatory Approvals), and no such consents, registrations, approvals, permits and authorizations shall contain any Materially Burdensome Regulatory Condition.

- (e) <u>No Material Adverse Effect</u>. Since the date hereof, no event shall have occurred circumstance arisen that, individually or taken together with all other facts, circumstances or events, has had or is reasonably likely to have a Material Adverse Effect with respect to Valley.
- (f) <u>Dissenting Shareholders</u>. Holders of not more than ten percent (10%) of the outstanding shares of Valley Common Stock shall have duly exercised their dissen rights under Section 1300 of the CGCL or otherwise have the capacity to exercise such dissenters rights in accordance with Section 1300 of the CGCL.
- (g) <u>Third Party Consents</u>. Valley shall have obtained each of the consents listed in Schedule 3.16(b) of the Valley Disclosure Schedule and any consents of the type required to be identified in Schedule 3.16(b) of the Valley Disclosure Schedule but were not so identified as of the date of this Agreement. A copy of each such consershall have been delivered to the CVB.
- (h) <u>Directors Resignation</u>s. CVB shall have received the written resignation of ea director of Valley and each of its Subsidiaries (in such director s capacity as a director as of the Effective Time.
- (i) <u>Agreements of Directors and Certain Officers and Employees</u>. CVB shall have received (i), as of the date of this Agreement, a Non-Competition Agreement in the form of <u>Exhibit B</u> from each member of the Valley Board of Directors and each of officers of Valley listed on <u>Attachment 1</u> to <u>Exhibit B</u> and no action shall have bee taken by any such person to rescind, terminate or breach any such agreement as of Closing Date and (ii) a release in a form determined by CVB from each officer of Valley who is entitled to receive a Change of Control Payment.
- (j) <u>Closing Balance Sheet</u>. Valley shall have delivered the Closing Date Balance Sh to CVB at least three (3) Business Days prior to the Closing Date.
- (k) <u>Financial Measures</u>. As of at least three (3) Business Days prior to the Closing Date, CVB shall have received satisfactory evidence that each of the following minimum financial threshold amounts has been met: (i) the Closing Tangible Common Equity as of the final day of the month immediately preceding the Closin Date shall not be less than the greater of (A) the Closing Tangible Common Equity Minimum or (B) the amount necessary for Valley to achieve a tangible common equity ratio of at least 8.0% as of such date (determined in accordance with GAAP regulatory accounting requirements and Valley s policies and procedures as of the of this Agreement); (ii) the Allowance for Loan Loss Ratio, determined as of the file.

day of the month immediately preceding the Closing Date, shall not be less than 1. (iii) Total Assets shall not be less than \$410,000,000 as of the final day of the month immediately preceding the Closing Date; (iv) Average Accruing Loans shall not be less than \$295,000,000; and (v) Average Non-Interest Bearing Deposits shall not be less than \$150,000,000.

- (1) <u>Transaction Costs</u>. The Transaction Costs shall not exceed \$3,500,000 and none the components of the Transaction Costs listed in Section 7.2(1) of the Valley Disclosure Schedule shall exceed the respective amounts specified therein.
- (m) Expense Report; Transaction Costs. Valley shall have caused its attorneys, accountants, auditors and investment bankers which rendered services to Valley in connection with the transactions contemplated by this Agreement (collectively, the <u>Advisors</u>) to have submitted to Valley estimates of their fees and expenses for

A-49

all services rendered in any respect in connection with the transactions contemplate hereby and their reasonable estimates of the amounts of the fees and expenses they expect to incur up to and including the Closing Date at least five (5) Business Days prior to the Closing Date. Valley shall have prepared and submitted to CVB no late than five (5) Business Days prior to the Closing Date a final calculation of all Transaction Costs, certified by Valley s Chief Financial Officer with support for a items therein in a form and substance reasonable to CVB. Based on such summary Valley shall have prepared and submitted a final calculation of such fees and expenses, and Valley, shall have accrued and paid the amount of such fees and expenses as calculated above. Valley shall have caused the Advisors to submit thei final bills for such fees and expenses to Valley for services rendered prior to the Closing Date and such Advisors shall have released Valley, CVB and the Surviving Corporation from liability, or shall have advised them in writing that, upon paymer full of such amounts, they shall have no liability for any fees or expenses to such Advisors.

- (n) For purposes of this Agreement,
- (i) Allowance for Loan Loss Ratio means the ratio of (x) the amount of Valley allowance for loan losses as of such date, determined in accordance with GAAP, regulatory accounting requirements and Valley s policies and procedures as of the of this Agreement to (y) Valley s accrual loans (net of fees) and non-accrual loans such date.
- (ii) <u>Average Non-Interest Bearing Deposits</u> means the average of Valley s total non-interest bearing deposits over the 30-day period ending on the Measurement D
- (iii) <u>Closing Balance Sheet</u> means the a balance sheet of Valley as of the Measu Date in a form and substance reasonably acceptable to CVB, the completeness and accuracy of which shall be certified by Valley s Chief Financial Officer.
- (iv) <u>Measurement Date</u> means the date that is five (5) Business Days prior to the Closing Date.
- (v) <u>Closing Tangible Common Equity</u> means Valley s tangible common shareh equity calculated in accordance GAAP, regulatory accounting requirements and Valley s policies and procedures as of the date of this Agreement, after giving effect the Special Dividend but prior to any pre-closing adjustments effectuated in accordance with Section 6.14 and the Transaction Costs.
- (vi) <u>Closing Tangible Common Equity Minimum</u> means \$37,500,000, if Valley completed the sale of the entire Excepted Loan as of the final day of the month immediately preceding the Closing Date as permitted by Section 5.2(r) or, \$37,000,000, if Valley has not completed the sale of all of the entire Excepted Loa of the final day of the month immediately preceding the Closing Date as permitted Section 5.2(r).

- (vii) <u>Average Accruing Loans</u> means the average of Valley s gross Loans, net of deferred fees and discounts, less Valley s Loans on non-accrual, determined in accordance with GAAP, regulatory accounting requirements and Valley s policies procedures as of the date of this Agreement, for the 30-day period ending on the Measurement Date.
- (viii) Total Assets means Valley s total assets, determined in accordance with C
- (ix) <u>Transaction Cos</u>ts means the following expenses, costs and fees paid or to be in connection with consummation of the transactions described herein: (1) any contract termination fees payable to vendors as to which CVB has provided its agreement to terminate as of the Effective Time or such other time as may be appropriate including the costs and expenses of terminating data processing license and contracts indicated in Section 7.2(1) of the Valley Disclosure Schedule, (2) Change in Control Payments, which shall not exceed the amount indicated in Section 7.2(1) of the Valley Disclosure Schedule, (3) Valley s legal, advisory and other professional fees and professional expenses rendered solely in connection with the transaction

A-50

contemplated by this Agreement, up to the amounts indicated in Section 7.2(1) of the Valley Disclosure Schedule, (4) Valley accounting and valuation fees and expensed rendered solely in connection with the transactions contemplated by this Agreement (5) D&O Insurance premiums and costs, and (6) Valley accounts of printing and matthe Proxy Statement and soliciting the Valley Shareholder Approval.

- (o) <u>280G Opinion</u>. Valley shall have delivered to CVB the written confirmation, to dated as of a date not earlier than five (5) Business Days prior to the Closing Date, Crowe Horwath LLP or another nationally recognized accounting firm reasonably acceptable to CVB, that no agreement, contract or arrangement to which any employee of Valley is a party will result in the payment of any amount that would be deductible by reason of Section 280G of the Code, as determined without regard Section 280G(b)(4) of the Code.
- (p) <u>FIRPTA Certificate</u>. Valley shall have delivered to CVB a properly executed statement from Valley meeting the requirements of Treasury Regulations Sections 1.1445-2(c)(3) and 1.897-2(h)(1), dated as of the Closing Date.
- (q) <u>Consulting Agreement.</u> The chief executive officer of Valley, Allan W. Stone, shall have entered into a consulting agreement with Citizens Business Bank in the form of Schedule 7.2(q) to this Agreement.
- 7.3 <u>Conditions to Obligations of Valley</u>. The obligation of Valley to effect the Mer is also subject to the satisfaction or waiver by Valley at or prior to the Effective Timof the following conditions:
- (a) Representations and Warranties. The representations and warranties of CVB se forth in this Agreement shall be true and correct as of the date of this Agreement at as of the Effective Time as though made on and as of the Effective Time (except the representations and warranties that by their terms speak specifically as of the date this Agreement or another date shall be true and correct as of such date); provided, however, that no representation or warranty of CVB (other than the representations and warranties set forth in (i) Sections 4.1(a), 4.3(a), 4.3(b) and 4.7, which shall be true and correct in all material respects, and (ii) Section 4.8, which shall be true and correct in all respects) shall be deemed untrue or incorrect for purposes hereunder consequence of the existence of any fact, event or circumstance inconsistent with s representation or warranty, unless such fact, event or circumstance, individually or taken together with all other facts, events or circumstances inconsistent with any representation or warranty of CVB has had or would reasonably be expected to res in a Material Adverse Effect on CVB; provided, further, that for purposes of determining whether a representation or warranty is true and correct for purposes of this Section 7.3(a), any qualification or exception for, or reference to, materiality (including the terms material, materially, in all material respects, Effect or similar terms or phrases) in any such representation or warranty shall be disregarded; and Valley shall have received a certificate signed on behalf of CVB I the Chief Executive Officer or the Chief Financial Officer of CVB to the foregoing effect.

- (b) <u>Performance of Obligations of CVB</u>. CVB shall have performed in all material respects all of its obligations required to be performed by it under this Agreement a prior to the Effective Time, and Valley shall have received a certificate signed on behalf of CVB by its Chief Executive Officers or the Chief Financial Officers to su effect.
- (c) <u>Regulatory Approvals</u>. All Requisite Regulatory Approvals shall have been obtained and shall remain in full force and effect.
- (d) <u>No Material Adverse Effect</u>. Since the date hereof, no event shall have occurred circumstance arisen that, individually or taken together with all other facts, circumstances or events, has had or is reasonably likely to have a Material Adverse Effect with respect to CVB.
- (e) <u>Listing</u>. The shares of CVB Common Stock to be issued in the Merger shall have been authorized for listing on the Nasdaq, subject to official notice of issuance.

A-51

ARTICLE VIII

TERMINATION AND AMENDMENT

- 8.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Effect Time, whether before or after approval of the matters presented in connection with Merger by the shareholders of Valley or CVB:
- (a) <u>Mutual Consent</u> by mutual consent of Valley and CVB in a written instrumer authorized by the Boards of Directors of Valley and CVB;
- (b) <u>Either Party</u> by either Valley or CVB;
- (i) No Regulatory Approval (A) if any Governmental Entity that must grant a Requisite Regulatory Approval has (1) denied a Requisite Regulatory Approval an such denial has become final and nonappealable, or (2) advised CVB and/or Valley writing that it will not grant (or intends to rescind or revoke if previously approved any such Requisite Regulatory Approval and such denial has become final and nonappealable, or (B) any Governmental Entity of competent jurisdiction shall have issued a final and nonappealable order, injunction or decree permanently enjoining otherwise prohibiting or making illegal the consummation of the transactions contemplated by this Agreement;
- (ii) <u>Delay</u> if the Merger shall not have been consummated on or before February 2017 (the <u>End Date</u>); provided, that the right to terminate this Agreement pursuathis Section 8.1(b)(ii) shall not be available to any party whose failure to perform cobserve the covenants and agreements of such party set forth in this Agreement resulted in the failure of the Merger to be consummated by the End Date;
- (iii) <u>Breach</u> if there shall have been a breach of any of the covenants or agreement any of the representations or warranties set forth in this Agreement on the part of Valley, in the case of a termination by CVB, or on the part of CVB, in the case of termination by Valley, which breach, either individually or in the aggregate with or breaches by such party, would result in, if occurring or continuing on the Closing Date, the failure of the conditions set forth in Section 7.2 or 7.3, as the case may be and which is not cured within thirty (30) days following written notice to the party committing such breach or by its nature or timing cannot be cured within such time period (provided that the terminating party is not then in material breach of any representation, warranty, covenant or other agreement contained herein); or
- (iv) No Valley Shareholder Approval if the Valley Shareholder Approval shall n have been obtained at the Valley Shareholder Meeting duly convened therefor or a any adjournment or postponement thereof at which a vote on the adoption of this Agreement was taken; provided, however, that no party may terminate this Agreen pursuant to this Section 8.1(b)(iv) if such party has breached in any material respectany of its obligations under this Agreement, in each case in a manner that caused the failure to obtain the Valley Shareholder Approval at the Valley Shareholder Meetin or at any adjournment or postponement thereof (it being agreed by the parties that

failure to obtain the Valley Shareholder Approval where there is no such breach sh not constitute a breach in and of itself;

(c) No Valley Recommendation by CVB, at any time prior to such time as the Va Shareholder Approval is obtained, in the event (A) Valley shall have breached in a material respect Section 6.9; (B) Valley or the Board of Directors of Valley shall have breathed in a submitted this Agreement to its shareholders without a recommendation for approver or otherwise withdraws or materially and adversely modifies (or discloses its intensity of the withdraw or materially and adversely modify) its recommendation as contemplated by Section 6.9(c), or recommends to its shareholders a Valley Acquisition Proposal other than the Merger (an Adverse Change of Recommendation); (C) at any time the end of fifteen (15) Business Days following receipt of an Acquisition Proposal Valley, the Board of Directors of Valley shall have failed to reaffirm its Valley Board Recommendation as promptly as practicable (but in any event within five (5) Business Days) after receipt of any written request to do so by CVB; or (D) a tender offer or exchange offer for outstanding shares of Valley

A-52

Common Stock shall have been publicly disclosed (other than by CVB or an Affilia of CVB) and the Board of Directors of Valley recommends that its shareholders ten their shares in such tender or exchange offer or, within ten (10) Business Days afte the commencement of such tender or exchange offer, the Board of Directors of Valfails to definitively recommend against acceptance of such offer;

- (d) <u>Valley Superior Proposal</u> by Valley, prior to such time as the Valley Shareho Approval is obtained, in order to enter into a definitive agreement providing for a Valley Superior Proposal; <u>provided</u> that (i) Valley is not in material breach of any the terms of this Agreement, and (ii) Valley Termination Fee is paid to CVB in advance of or concurrently with such termination in accordance with Section 8.3(b)
- (e) <u>Burdensome Condition</u>. By CVB if any Regulatory Approval includes, or will in be issued without, the imposition of a Materially Burdensome Regulatory Condition or
- (f) Trading Collar.
- (i) By Valley, upon written notice to CVB within the two (2) Business Days follow the Determination Date (as defined below), in the event of all of the following:
- (1) CVB does not have the right to terminate this Agreement pursuant to Sections 8.1(b)(iii) or 8.1(c) of this Agreement, or CVB has the right to terminate this Agreement pursuant to Sections 8.1(b)(iii) or 8.1(c) of this Agreement and does no exercise such right; and
- (2) The CVB Average Closing Price is less than \$13.50.
- (ii) By CVB, upon its written notice to Valley within the two (2) Business Days following the Determination Date, in the event of all of the following:
- (1) Valley does not have the right to terminate this Agreement pursuant to Section 8.1(b)(iii) of this Agreement, or Valley has the right to terminate this Agreement pursuant to Section 8.1(b)(iii) of this Agreement and does not exercise such right;
- (2) The CVB Average Closing Price is greater than \$20.50.
- (iii) For purposes of Section 7.1(f), <u>Determination Date</u> means the fifthth() Tradin Day immediately preceding the anticipated Closing Date.
- (iv) If CVB declares or effects a stock dividend, reclassification, recapitalization, forward or reverse stock split, or similar transaction between the date of this Agreement and the Determination Date, the prices for the CVB Common Stock use to determine the CVB Average Closing Price shall be appropriately adjusted.
- 8.2 <u>Effect of Termination</u>. In the event of termination of this Agreement by either Valley or CVB as provided in Section 8.1, this Agreement shall forthwith become void and have no effect and none of Valley, CVB, any of its Subsidiaries or any of

officers or directors of any of them shall have any liability of any nature whatsoever under this Agreement, or in connection with the transactions contemplated by this Agreement, except that (i) Sections 6.3(c), 8.2, 8.3, and 9.3 through 9.11 shall survany termination of this Agreement, and (ii) neither Valley nor CVB shall be relieved or released from any liabilities or damages arising out of its willful and material breach of any provision of this Agreement; provided, however, no party hereto shall seek to obtain, or be entitled to, any recovery for consequential, indirect or punitive damages against any other party or any of such party s Subsidiaries, or any of their respective directors, officers, employees, attorneys, agents, advisers, managers, or Affiliates in connection with the transactions contemplated hereby.

A-53

8.3 Fees and Expenses.

(a) All fees and expenses incurred in connection with the Merger, this Agreement, the transactions contemplated by this Agreement (including costs and expenses of printing and mailing the Proxy Statement) shall be paid by the party incurring such fees or expenses, whether or not the Merger is consummated, except as otherwise provided in Section 8.3(b).

(b) Valley Termination Fee.

- (i) In the event that this Agreement is terminated by Valley pursuant to Section 8.1 (Valley Superior Proposal) or CVB pursuant to Section 8.1(c) (No Valley Recommendation), then Valley shall pay CVB a fee, in immediately available funcin the amount of \$3,500,000.00 (the <u>Valley Termination Fee</u>) by wire transfer to account specified by CVB promptly, but in any event prior to or concurrently with termination pursuant to Section 8.1(d) or no later than two (2) Business Days after date of termination pursuant to Section 8.1(c).
- (ii) In the event that any Person shall have made a Valley Acquisition Proposal, where proposal has been publicly announced, disclosed or proposed and not withdrawn, a
- (1) hereafter this Agreement is terminated:
- (a) by either party pursuant to Section 8.1(b)(ii) (Delay), or Section 8.1(b)(iv) (No Valley Shareholder Approval) but only if one or more of the following is true: (A) Valley has effected an Adverse Recommendation Change, (B) Valley has failed to hold the Valley Shareholders Meeting or otherwise breached any provision of Sect 6.9 or (C) Valley has held the Valley Shareholders Meeting and failed to obtain the Valley Shareholder Approval; or
- (b) by CVB pursuant to Section 8.1(b)(iii) (Breach by Valley), and
- (2) within eighteen (18) months after such termination of this Agreement, a Valley Acquisition Proposal shall have been consummated or any definitive agreement wirespect to a Valley Acquisition Proposal shall have been entered into (<u>provided</u> that for purposes of the foregoing, the term <u>Valley Acquisition Proposal</u> shall have to meaning assigned to such term in Section 6.9(d) except that the references to monthan 10% in the definition of Valley Acquisition Proposal shall be deemed to be references to 50%);

then Valley shall pay CVB Valley Termination Fee by wire transfer to an account specified by CVB prior to the earlier of the execution of a definitive agreement wit respect to, or the consummation of, such Valley Acquisition Proposal. In no event shall Valley be obligated to pay CVB Valley Termination Fee on more than one occasion or if Valley has paid any damages for breach and in no event shall Valley obligated to pay CVB any damages if Valley has paid Valley Termination Fee.

(c) <u>Liquidated Damages</u>. Valley and CVB acknowledge that the agreements contain this Section 8.3 are an integral part of the transactions contemplated by this Agreement, and that, without these agreements, neither party would enter into this Agreement. The amounts payable by Valley pursuant to Section 8.3(b) constitute liquidated damages and not a penalty and shall be the sole monetary remedy of CV in the event of termination of this Agreement under such applicable section. In the event that Valley fails to pay when due any amounts payable under this Section 8.3 then (i) the party failing to so pay shall reimburse the other party for all costs and expenses (including disbursements and reasonable fees of counsel) incurred in connection with the collection of such overdue amount, and (ii) the party failing to pay shall pay to the other party interest on such overdue amount (for the period commencing as of the date that such overdue amount was originally required to be paid and ending on the date that such overdue amount is actually paid in full) at a reper annum equal to the prime rate published in *The Wall Street Journal* on the date such payment was required to be made.

A-54

- 8.4 <u>Amendment</u>. This Agreement may be amended by the parties, by action taken authorized by their respective Boards of Directors, at any time before or after approof the matters presented in connection with the Merger by the shareholders of Vall-provided, <u>however</u>, that after any approval of the transactions contemplated by this Agreement by such shareholders, there may not be, without further approval of such shareholders, any amendment of this Agreement that requires further approval und applicable Law. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- 8.5 Extension; Waiver. At any time prior to the Effective Time, the parties, by activaken or authorized by their respective Boards of Directors, may, to the extent legal allowed, (a) extend the time for the performance of any of the obligations or other of the other party, (b) waive any inaccuracies in the representations and warranties contained in this Agreement or (c) waive compliance with any of the agreements of conditions contained in this Agreement. Any agreement on the part of a party to an such extension or waiver shall be valid only if set forth in a written instrument sign on behalf of such party, but such extension or waiver or failure to insist on strict compliance with an obligation, covenant, agreement or condition shall not operate waiver of, or estoppel with respect to, any subsequent or other failure.

ARTICLE IX

GENERAL PROVISIONS

- 9.1 <u>Closing</u>. On the terms and subject to conditions set forth in this Agreement, the closing of the Merger (the <u>Closing</u>) shall take place at 10:00 a.m., Pacific Time, offices of Manatt, Phelps & Phillips, LLP, counsel to CVB, on the last Business Date of the calendar month in which the last to be satisfied of the conditions set forth in Article VII is satisfied (other than those conditions that by their nature are to be satisfied or waived at the Closing but subject to the satisfaction or waiver of those conditions), which date is on or after February 1, 2017, or such other date as determined by mutual agreement of the parties (the <u>Closing Date</u>).
- 9.2 <u>Non-survival of Representations</u>, Warranties and Agreements. This Article IX the agreements of Valley and CVB contained in Section 6.7 shall survive the consummation of the Merger. All other representations, warranties, covenants and agreements set forth in this Agreement shall terminate immediately upon the consummation of the Merger.
- 9.3 <u>Notices</u>. All notices and other communications in connection with this Agreem shall be in writing and shall be deemed given if delivered personally, sent via facsimile (with confirmation), mailed by registered or certified mail (return receipt requested) or delivered by an express courier (with confirmation) to the parties at t following addresses (or at such other address for a party as shall be specified by lik notice):
- (a) if to CVB or Citizens Business Bank, to:

CVB Financial Corp.

701 North Haven Avenue

Ontario, California 91764

Attention: Christopher D. Myers, President & Chief

Executive Officer

Facsimile: (909) 481-2120

with a copy (which shall not constitute notice) to:

Manatt, Phelps & Phillips, LLP

One Embarcadero Center

San Francisco, California 94111

Attention: Craig D. Miller, Esq.

David Gershon, Esq.

Facsimile: (415) 291-7474

A-55

(b) if to Valley, to:

Valley Commerce Bancorp

701 West Main Street

Visalia, California 93291

Attention: Allan W. Stone, President and

Chief Executive Officer

Facsimile: (559) 740-0747

with a copy (which shall not constitute notice) to:

Gary Steven Findley & Associates

3808 East La Palma Avenue

Anaheim, California 92807

Attention: Gary Steven Findley, Esq.

Facsimile: (714) 630-7919

9.4 <u>Interpretation</u>. When a reference is made in this Agreement to Articles, Section Exhibits or Schedules, such reference shall be to an Article or Section of or Exhibi Schedule to this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the includes or including are used in this Agreement, they shall words include, to be followed by the words without limitation. References to the date hereof mean the date of this Agreement. As used in this Agreement, the phrase Knowled <u>Valley</u> and similar language means the actual knowledge, after reasonable investigation, of any of Valley s officers listed on Section 9.4 of the Valley Disclo Schedule, and the phrase Knowledge of CVB and similar language means the ac knowledge, after reasonable investigation, of the Chief Executive Officer and Chie Financial Officer of CVB. For purposes of this definition, a reasonable investigat by a Person shall mean a review by the officer of such Person of documents and/or such other information that is reasonably related to the performance of the job functions or oversight responsibilities of such officer. As used in this Agreement, Person or Persons means any individual, bank, corporation (including not-for joint-stock company, general or limited partnership, limited liability company, join venture, estate, business trust, trust, association, organization, Governmental Entity other entity of any kind or nature. All schedules and exhibits hereto shall be deeme part of this Agreement and included in any reference to this Agreement. As used in this Agreement, Business Day means Monday through Friday of each week, exc

legal holiday recognized as such by the United States federal government or any day on which banking institutions in the State of California are authorized or obligated close. If any term, provision, covenant or restriction contained in this Agreement is held by a court or a federal or state regulatory agency of competent jurisdiction to be invalid, void or unenforceable, the remainder of the terms, provisions and covenant and restrictions contained in this Agreement shall remain in full force and effect, as shall in no way be affected, impaired or invalidated. If for any reason such court or regulatory agency determines that any provision, covenant or restriction is invalid, void or unenforceable, it is the express intention of the parties that such provision, covenant or restriction be enforced to the maximum extent permitted.

9.5 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts (including by facsimile or other electronic means), all of which shall be considered one and the same agreement and shall become effective when counterparts have be signed by each of the parties and delivered to the other party, it being understood the each party need not sign the same counterpart.

9.6 Entire Agreement. This Agreement (including the documents and the instrumer referred to in this Agreement), together with the Confidentiality Agreement, constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, between the parties with respect to the subjematter of this Agreement, other than the Confidentiality Agreement.

A-56

- 9.7 Governing Law; Jurisdiction. This Agreement shall be governed by and construin accordance with the Laws of the State of California, without giving effect to its principles of conflicts of Laws. The parties hereto agree that any suit, action or proceeding brought by either party to enforce any provision of, or based on any materiarising out of or in connection with, this Agreement or the transactions contemplate hereby shall be brought in any federal or state court located in the State of Californ Each of the parties hereto submits to the jurisdiction of any such court in any suit, action or proceeding seeking to enforce any provision of, or based on any matter arising out of, or in connection with, this Agreement or the transactions contemplate hereby and hereby irrevocably waives the benefit of jurisdiction derived from preservor future domicile or otherwise in such action or proceeding. Each party hereto irrevocably waives, to the fullest extent permitted by Law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding any such court or that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum.
- 9.8 Waiver of Jury Trial. Each party hereto acknowledges and agrees that any controversy that may arise under this Agreement, and in respect of the transactions contemplated hereby, is likely to involve complicated and difficult issues, and therefore each party hereby irrevocably and unconditionally waives any right such party may have to a trial by jury in respect of any legal action, directly or indirectly arising out of, or relating to, this Agreement or any documents referred to in this Agreement, or the transactions contemplated by this Agreement. Each party certificand acknowledges that (a) no representative, agent or attorney of any other party herepresented, expressly or otherwise, that such other party would not, in the event of litigation, seek to enforce the foregoing waiver, (b) each party understands and has considered the implications of this waiver, (c) each party makes this waiver voluntarily, and (d) each party has been induced to enter into this Agreement by, among other things, the mutual waivers and certifications in this Section 9.8.
- 9.9 <u>Publicity</u>. Neither Valley nor CVB shall, and none of Valley nor CVB shall per any of their respective Subsidiaries to, issue or cause the publication of any press release or other public announcement with respect to, or otherwise make any public statement, or, except as otherwise specifically provided in this Agreement, any disclosure of nonpublic information to a third party, concerning, the transactions contemplated by this Agreement without the prior consent (which shall not be unreasonably withheld, conditioned, or delayed) of CVB, in the case of a proposed announcement, statement or disclosure by Valley or VBB, or of Valley, in the case a proposed announcement, statement or disclosure by CVB or Citizens Business Bank; <u>provided</u>, <u>however</u>, that any of either Valley or VBB, on the one hand, or CV or Citizens Business Bank, on the other, may, without the prior consent of the othe party (but after prior consultation with the other party to the extent practicable und the circumstances) issue or cause the publication of any press release or other publicannouncement to the extent required by Law or by the applicable rules and regulated of the Nasdaq.
- 9.10 <u>Assignment: Third-Party Beneficiaries</u>. Neither this Agreement nor any of the rights, interests or obligations under this Agreement shall be assigned by either of the state of the s

parties without the prior written consent of the other party (which shall not be unreasonably withheld or delayed). Any purported assignment in contravention her shall be null and void. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by each of the parties and their respective successors and permitted assigns. Except for Section 6.7, which is intended to benefit each Indemnified Party and his or her heirs and representatives, nothing in this Agreement, expressed or implied, is intended to confer upon any person, other than the parties hereto or their respective successors, any rights, remedies, obligations or liabilities under or by reason of this Agreement.

9.11 Specific Performance. The parties agree that irreparable damage would occur the event that any of the provisions of this Agreement were not performed in accordance with their specific terms. It is accordingly agreed that, except as specifically set forth Section 8.3(c), each party acknowledges that money damages would be an insufficient remedy for any breach of this Agreement by such party and that such party would cause the other party hereto irreparable harm and that, accordingly, each party also agrees that in the event of any breach or threatened breach of this Agreement by such party, the other party shall be entitled to equitable relief without the

A-57

requirement of posting a bond or other security, including in the form of injunction and specific performance of the terms hereof, this being in addition to any other remedies to which such party is entitled at Law or in equity.

9.12 <u>Valley Disclosure Schedule</u>.

- (a) Before entry into this Agreement, Valley delivered to CVB a schedule (a <u>Vall Disclosure Schedule</u>) which sets forth, among other things, items the disclosure of which is necessary or appropriate either in response to an express disclosure requirement contained in a provision hereof or as an exception to one or more representations or warranties contained in Article III or to one or more covenants contained herein; <u>provided</u>, <u>however</u>, that notwithstanding anything in this Agreement to the contrary, the mere inclusion of an item as an exception to a representation or warranty shall not be deemed an admission that such item represents a material exception or material fact, event or circumstance or that such item has had or would be reasonably likely to have a Material Adverse Effect.
- (b) For purposes of this Agreement, when used with respect to Valley, <u>Previously Disclosed</u> means, for Valley, information set forth by Valley in the applicable sec of the Valley Disclosure Schedule or any other section the Valley Disclosure Schedule (so long as it is reasonably clear from the context that the disclosure in such other paragraph of the Valley Disclosure Schedule is also applicable to the section of this Agreement in question) and, when used with respect to CVB, means information concerning CVB set forth in the CVB SEC Reports.

[Signature page follows]

A-58

IN WITNESS WHEREOF, the parties have caused this Agreement to be execute by their respective officers thereunto duly authorized as of the date first above writ

CVB FINANCIAL CORP.

By: /s/ E. Allen NicholsonName: E. Allen NicholsonTitle: Executive Vice President and Chief Financial Officer

VALLEY COMMERCE BANCORP

By: /s/ Allan W. Stone Name: Allan W. Stone

Title: President & Chief Executive

Officer

<u>EXHIBI</u>

VOTING AND SUPPORT AGREEMENT

This VOTING AND SUPPORT AGREEMENT (this <u>Agreement</u>) is made and edinto as of September 22, 2016 by and among CVB Financial Corp., a California corporation (<u>CVB</u>), and the shareholder of Valley Commerce Bancorp, a California corporation (<u>Valley</u>), that is a signatory to this Agreement (the <u>Shareholder</u>), solely for purposes of the last sentence of <u>Section 7</u>, Valley.

Recitals

A. CVB and Valley have entered into that certain Agreement and Plan of Merger a Reorganization (and as it may be amended, the <u>Merger Agreement</u>), dated as of date of this Agreement, pursuant to which Valley will merge (the <u>Merger</u>) with CVB, whereupon each share of Valley s common stock (<u>Valley Common Stock</u> be converted into the right to receive the consideration set forth in the Merger Agreement.

B. As a condition to its willingness to enter into the Merger Agreement, CVB has required that the Shareholder, solely in his or her capacity as a shareholder and beneficial owner or record holder of Valley Common Stock, enter into, and the Shareholder has agreed to enter into, this Agreement.

NOW, THEREFORE, in consideration of the foregoing, for good and valuable consideration, the parties hereby agree as follows:

- 1. <u>Representations and Warranties of the Shareholder</u>. The Shareholder hereby represents and warrants to CVB as follows:
- (a) <u>Authority; No Violation</u>. The Shareholder has all necessary power and authority enter into and perform all of the Shareholder s obligations hereunder. The execution delivery and performance of this Agreement by the Shareholder will not violate any other agreement to which the Shareholder is a party, including any voting agreement shareholders agreement, trust agreement, voting trust or similar arrangement. This Agreement has been duly and validly executed and delivered by the Shareholder (at the Shareholder s spouse, if the spouse s consent or agreement is necessary to votor refrain from voting or selling the Shares (as defined below) or to agree or commute do any of the foregoing) and constitutes a valid and binding agreement of the Shareholder and such spouse, enforceable against the Shareholder and the Shareholder s spouse in accordance with its terms.
- (b) Ownership of Shares. The Shareholder is the beneficial owner or record holder the number of shares of Valley Common Stock indicated under the Shareholder s on the signature page hereto (the <u>Existing Shares</u>, and together with any shares of Valley Common Stock acquired by the Shareholder after the date of this Agreement the <u>Shares</u>) and, as of the date of this Agreement, the Existing Shares constitute shares of Valley Common Stock owned of record or beneficially by the Shareholder

With respect to the Existing Shares, subject to applicable community property laws the Shareholder has sole voting power and sole power to issue instructions with respect to the matters set forth in <u>Section 2</u> of this Agreement, sole power of disposition, sole power to demand appraisal rights and sole power to engage in activate forth in <u>Section 2</u> of this Agreement, with no restrictions on the voting rights, rights of disposition or otherwise, subject to applicable laws and the terms of this Agreement.

2. Voting Agreement and Agreement Not to Transfer.

(a) From the date of this Agreement until the Termination Date (as defined in Section 8 below) (the <u>Support Period</u>), except as set forth in Section 6 below, th Shareholder hereby agrees to vote all of the Shares (i) in favor of the Merger, the Merger Agreement and the transactions contemplated by this Agreement at any meeting of shareholders of Valley (and at any adjournment or postponement there and in connection with any

A-1

action of the shareholders of Valley taken by written consent, (ii) against any actio agreement that would result in a breach in any material respect of any covenant, representation or warranty or any other obligation or agreement of Valley under the Agreement or the Merger Agreement, and (iii) except with the prior written conser-CVB or as otherwise contemplated in or permitted by the Merger Agreement, again the following actions (other than the Merger and the transactions contemplated by Merger Agreement): (A) any extraordinary corporate transactions, such as a merge consolidation or other business combination involving Valley; (B) any sale, lease of transfer of a material amount of the assets of Valley; (C) any change in the majorit the board of directors of Valley; (D) any material change in the present capitalizati of Valley; (E) any amendment of Valley s articles of incorporation or bylaws; (F) other material change in Valley s corporate structure or business; or (G) any other action that is intended or could reasonably be extended to impede, interfere with delay, postpone, discourage or materially adversely affect the Merger. During the Support Period, the Shareholder shall not enter into any agreement or understanding with any Person or entity to vote or give instructions after the Termination Date in manner inconsistent with <u>clauses (i)</u>, <u>(ii)</u> or <u>(iii)</u> of the preceding sentence.

- (b) During the Support Period, the Shareholder will not, directly or indirectly (i) se transfer, exchange, pledge, assign, hypothecate, encumber, tender or otherwise disof (collectively, a <u>Transfer</u>), or enforce or permit execution of the provisions of redemption, share purchase or sale, recapitalization or other agreement with Valley any other Person or enter into any contract, option or other agreement, arrangemen understanding with respect to the Transfer of, directly or indirectly, any of the Shar or any securities convertible into or exercisable for Shares, any other capital stock Valley or any interest in any of the foregoing with any Person, (ii) enter into swap any other agreement or any transaction that transfers, in whole or in part, directly of indirectly, the economic consequence of ownership of the Shares, (iii) take any act that would make any of the Shareholder s representations or warranties contained this Agreement untrue or incorrect in any material respect or have the effect of preventing or disabling the Shareholder from performing the Shareholder s obliga under this Agreement; provided however, that this Agreement shall not prohibit the Shareholder from transferring and delivering any of the Shares to Valley to effect t exercise of an option to purchase Valley Common Stock.
- 3. <u>Cooperation</u>. Except as set forth in Section 6 below, during the Support Period to Shareholder agrees that the Shareholder will not directly or indirectly (i) initiate, solicit, induce or knowingly encourage, or knowingly take any action to facilitate to making of, any inquiry, offer or proposal which constitutes, or could reasonably be expected to lead to, a Valley Acquisition Proposal, or (ii) participate in any discussions or negotiations regarding any Valley Acquisition Proposal, or furnish, otherwise afford access, to any Person (other than CVB) to any information or data with respect to Valley relating to a Valley Acquisition Proposal.
- 4. <u>No Solicitation</u>. Except as set forth in Section 6 below, during the Support Perio the Shareholder shall not, and shall not permit any attorney or other representative retained by the Shareholder to, directly or indirectly, (a) take any of the actions specified in Section 6.9(a) of the Merger Agreement that Valley has agreed not to

take, (b) agree to release, or release, any Person from any obligation under any existing standstill agreement or agreement relating to Valley or (c) participate in, directly or indirectly, a solicitation of proxies (as such terms are used in the respect to the voting of the vote, or seek to advise or influent any Person with respect to the voting of, any shares of Valley Common Stock, other than to recommend that shareholders of Valley vote in favor of the adoption and approval of the Merger Agreement and the Merger. Except as set forth in Section 6 below, the Shareholder agrees immediately to cease and cause to be terminated any activities, discussions or negotiations conducted before the date of this Agreement with any Persons other than CVB with respect to any possible Valley Acquisition Proposal and will take all necessary steps to inform any representative retained by Shareholder of the obligations undertaken by the Shareholder pursuant to this Section 4.

5. <u>Notice of Acquisitions: Proposals Regarding Prohibited Transactions</u>. The Shareholder hereby agrees to notify CVB promptly (and in any event within two (2) Business Days) in writing of the number of any additional

A-2

shares of Valley Common Stock or other securities of Valley of which the Shareholder shall cause Valley to comply with the provisions of Sections 6.9(a) an 6.9(b) of the Merger Agreement as if he or she were Valley.

- 6. <u>Shareholder Capacity</u>. The Shareholder is entering this Agreement in the Shareholder s capacity as the record or beneficial owner of the Shares, and not in her capacity as a director of Valley or as a trustee of any Valley benefit plan. Nothin this Agreement shall be deemed in any manner to limit the discretion of the Shareholder to take any action, or fail to take any action, in his or her capacity as a director of Valley or as a trustee of any benefit plan of Valley, that may be required the Shareholder in the exercise of the Shareholder s duties and responsibilities as a director of Valley or as a trustee of any benefit plan of Valley.
- 7. <u>Stop Transfer Order</u>. In furtherance of this Agreement, the Shareholder hereby authorizes and instructs Valley to enter a stop transfer order with respect to all of the Shares for the Support Period. The Valley agrees that it shall comply with such sto transfer instructions.
- 8. <u>Termination</u>. This Agreement and the obligations of the Shareholder hereunder shall terminate upon the first to occur of (a) the Effective Time of the Merger or (b) the termination of the Merger Agreement in accordance with its terms (the <u>Termination Date</u>).
- 9. Specific Performance. The Shareholder acknowledges that it is a condition to the willingness of Parent to enter into the Merger Agreement that Shareholder execute deliver this Agreement and that it will be impossible to measure in money the damages to CVB if Shareholder fails to comply with the obligations imposed by th Agreement and agrees that irreparable injury will result to CVB in the event of a breach of any of the provisions of this Agreement and that CVB may have no adequate remedy at law with respect thereto. Accordingly, in the event of a materia breach of this Agreement, and in addition to any other legal or equitable remedy C may have, the Shareholder agrees that the entry of a preliminary injunction and a permanent injunction (including, without limitation, specific performance) by a conof competent jurisdiction, to restrain the violation or breach thereof by the Shareho or any Affiliates, agents, or any other persons acting for or with the Shareholder in capacity whatsoever, is an appropriate remedy for any such breach and that the Shareholder will not oppose the granting of such relief on the basis that CVB has a adequate remedy at law. The Shareholder further agrees that the Shareholder will r seek, and agrees to waive any requirement for, the securing or posting of a bond in connection with CVB s seeking or obtaining such equitable relief. Such injunctive other equitable remedies are cumulative and shall be CVB s sole remedy under the Agreement, unless CVB shall have sought and been denied by a court of competen jurisdiction injunctive or other equitable remedies and such denial is other than by reason of violation of this Agreement by CVB, the Shareholder submits to the jurisdiction of such court in any such action.

10. Miscellaneous.

- (a) <u>Definitional Matters</u>.
- (i) All capitalized terms used but not defined in this Agreement shall have the respective meanings set forth in the Merger Agreement.
- (ii) The descriptive headings herein are inserted for convenience of reference only are not intended to be part of or to affect the meaning or interpretation of this Agreement.
- (b) <u>Entire Agreement</u>. This Agreement together the Merger Agreement constitute t entire agreement between the parties hereto with respect to the subject matter hered and supersedes all other prior agreements and understandings, both written and ora among the parties, or any of them, with respect to the subject matter hereof.

A-3

- (c) <u>Parties in Interest</u>. This Agreement shall be binding upon and inure solely to the benefit of each party hereto and their respective successors, assigns, heirs, executor administrators and other legal representatives. Nothing in this Agreement, express implied, is intended to confer upon any other person any rights or remedies of any nature whatsoever under or by reason of this Agreement.
- (d) <u>Certain Events</u>. The Shareholder agrees that this Agreement and the obligations hereunder shall attach to the Shares and shall be binding upon any Person to which legal ownership of the Shares shall pass, whether by operation of law or otherwise, including, without limitation, the Shareholder s heirs, executors, guardians, administrators, trustees or successors. Notwithstanding any Transfer of such Share the Shareholder, the Shareholder or, as applicable, the Shareholder s heirs, execute guardians, administrators, trustees or successors, shall remain liable for the performance of all obligations under this Agreement.
- (e) <u>Assignment</u>. This Agreement shall not be assigned without the prior written consent of the other party hereto, and any purported assignment without such consentable be null and void.
- (f) <u>Modifications</u>; <u>Waivers</u>. This Agreement shall not be amended, altered or modifin any manner whatsoever, except by a written instrument executed by the parties hereto. No provision in this Agreement may be waived, except by means of a writing signed by the party against whom the waiver is sought to be enforced. The failure of any party hereto to enforce at any time any provision of this Agreement shall not be construed to be a waiver of such provision, nor in any way to affect the validity of Agreement or any part hereof or the right of any party thereafter to enforce each an every such provision. No waiver of any breach of this Agreement shall be held to constitute a waiver of any other or subsequent breach.
- (g) Governing Law. This Agreement and the legal relations between the parties her shall be governed by and construed in accordance with the laws of the state of California, without regard to the conflict of laws rules thereof. The state or federal courts located within the state of California shall have exclusive jurisdiction over a and all disputes between the parties hereto, whether in law or equity, arising out of relating to this Agreement and the agreements, instruments and documents contemplated hereby and the parties consent to and agree to submit to the jurisdicti of such courts. Each of the parties hereby waives and agrees not to assert in any su-is not personally subject to the jurisdiction of such courts, (ii) such party and such party s property is immune from any legal process issued by such courts, or (iii) a litigation or other proceeding commenced in such courts is brought in an inconvenforum. The parties hereby agreed that mailing of process or other papers in connec with any such action or proceeding in the manner provided in Section 10(k) of this Agreement or in other manner as may be permitted by law, shall be valid and sufficient services thereof and hereby waive any objections to services accomplished in the manner herein provided.

- (h) <u>Reliance on Counsel and Other Advisors</u>. The Shareholder has consulted with s legal, financial, technical or other experts as the Shareholder deems necessary or desirable before entering into this Agreement.
- (i) <u>Validity</u>. The invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement each of which shall remain in full force and effect.
- (j) <u>Counterparts and Execution</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed to be an original, but all of which shall constitute one and the same agreement. Signatures sent by facsimile shall have the same force as manual signed originals.

A-4

(k) <u>Notices</u>. Any notices or other communications required or permitted hereunder shall be in writing and shall be deemed duly given upon (i) transmitter s confirmat of a receipt of a facsimile transmission, (ii) confirmed delivery by a standard overnight carrier or (iii) the expiration of five (5) business days after the day when mailed by certified or registered mail, postage prepaid, addressed at the following addresses (or at such other address as the parties hereto shall specify by like notice)

If to CVB:

CVB Financial Corp.

701 North Haven Avenue

Ontario, California 91764

Attention: Christopher D. Myers, President & Chief Executive Officer

Facsimile: (909) 481-2120

with a copy (which shall not constitute notice) to:

Manatt, Phelps & Phillips, LLP

One Embarcadero Center

San Francisco, California 94111

Attention: Craig D. Miller, Esq.

Facsimile: (415) 291-7474

If to Valley, to:

Valley Commerce Bancorp

701 W. Main St.

Visalia, CA 93291 Attention: Allan Stone, President and Chief Executive Officer

Facsimile:

with a copy (which shall not constitute notice) to:

Gary Steven Findley and Associates

3808 East La Palma Avenue

Anaheim, California 92807

Attention: Gary Steven Findley, Esq. Facsimile: (714) 630-7010

If to the Shareholder, to the address noted on the signature page hereto.

[SIGNATURES APPEAR ON THE IMMEDIATELY FOLLOWING PAGE]

A-5

IN WITNESS WHEREOF, the parties hereto have executed this Support Agreeme as of the date first above written.

	CVB FINANCIAL CORP.
	By:
	Title:
	VALLEY COMMERCE BANCORP
	By:
	Title:
	SHAREHOLDER:
	Name:
	Number of Shares:
	Address for Notices:
	1 Name:
¹ If spousal signature required	

[Signature Page to Voting and Support Agreement]

Exhib

NONCOMPETITION AND NONSOLICITATION AGREEMENT

This NONCOMPETITION AND NONSOLICITATION AGREEMENT (this <u>Agreement</u>) dated as of September 22, 2016 is entered into by and between Citi Business Bank, a California banking corporation (<u>CB</u>B), and <u>[](Shareholder</u>

RECITALS

- A. CVB Financial Corp., a California corporation (<u>CVB</u>), and parent corporatio CBB, and Valley Commerce Bancorp, a California corporation (<u>Valley</u>), and pare corporation of Valley Commerce Bank, a California banking corporation (<u>Bank</u> entered into that certain Agreement and Plan of Reorganization and Merger, dated of September 22, 2016 (the <u>Merger Agreement</u>) which, among other things, contemplates the merger of Valley with and into CVB (the <u>Merger</u>) followed immediately by the merger of Bank into CBB (the <u>Bank Merger</u>).
- B. Shareholder is a beneficial owner of Valley common stock and a director and/or executive officer of Valley and Bank.
- C. Shareholder, as a shareholder of Valley, is or will be entitled to receive substant consideration consisting of CVB s common stock and payments of cash in connec with the transactions contemplated by the Merger Agreement.
- D. As an inducement to CVB to enter into the Merger Agreement and to consumm the Bank Merger, Shareholder agrees to refrain from competing with and using tracsecrets or soliciting customers or employees of Bank in accordance with the terms hereof.

NOW, THEREFORE, in consideration of the premises and respective representation warranties and covenants, agreements and conditions contained herein and in the Merger Agreement, and intending to be legally bound hereby, Shareholder and CB agree as follows:

ARTICLE I

CERTAIN DEFINITIONS

Except as otherwise provided herein, each capitalized term shall have the meaning given to such term in the Merger Agreement. As used in this Agreement, the following terms shall have the meanings set forth:

<u>Customer</u> means any Person with whom Bank has an existing relationship for Financial Services (as defined below) from the date of execution of the Merger Agreement until immediately prior to the Effective Time of the Merger.

<u>Enterprise</u> means the provision of Financial Services conducted by Bank at any from the date of execution of the Merger Agreement until immediately prior to the Effective Time of the Merger.

<u>Financial Institution</u> means a depository institution as that term is defined in Section 348.2, and any parent, Subsidiary or Affiliate thereof or any other corporat or other business entity primarily engaged in the origination or purchase of commercial or consumer loans and leases.

<u>Financial Services</u> means (i) the origination, purchasing, selling and servicing o commercial, real estate, residential, construction and consumer loans and/or (ii) the solicitation and provision of deposit services and services related thereto.

<u>Prospective Customer</u> means any Person with whom Bank has actively pursued relationship to provide Financial Services at any time between the date of execution the Merger Agreement and the Effective Time of the Merger.

Trade Secrets means all secrets and other confidential information, ideas, knowlknow-how, techniques, secret processes, improvements, discoveries, methods, inventions, sales, financial information, Customers, lists of Customers and Prospec Customers, plans, concepts, strategies or products, as well as all documents, report drawings, designs, plans and proposals otherwise pertaining to same or relating to business and properties of Bank of which Shareholder has acquired, or may hereaft acquire, knowledge and possession as a shareholder, director, officer or employee Bank or as a result of the transactions contemplated by the Merger Agreement, provided however, notwithstanding any other provisions of this Agreement to the contrary, Trade Secrets shall not include any (i) information which is or has been available from a third party who learned the information independently and is not of was not bound by a duty or agreement of confidentiality with respect to such information; or (ii) information readily ascertainable from public, trade or other nonconfidential sources (other than as a result, directly or indirectly, of a disclosure other dissemination in contravention of a confidentiality agreement).

ARTICLE II

ACKNOWLEDGMENTS BY SHAREHOLDER

Shareholder acknowledges that:

- 2.1 CVB would not enter into the Merger Agreement unless Shareholder agrees no enter into an activity that is competitive with or similar to the Enterprise in violatic of this Agreement and that, accordingly, this Agreement is a material inducement f CVB to enter into and to carry out the terms of the Merger Agreement. Accordingly Shareholder expressly acknowledges that he or she is entering into this Agreement with CBB to induce CVB to enter into and carry out the terms of the Merger Agreement.
- 2.2 By virtue of Shareholder s position with Bank, Shareholder has developed considerable expertise in the business operations of Bank and has access to Trade Secrets. Shareholder recognizes that CVB and CBB would be irreparably damaged and its substantial investment in Bank materially impaired, if Shareholder were to enter into an activity that is competitive with or similar to the Enterprise in violation of the terms of this Agreement, if Shareholder were to disclose or make use of any Trade Secrets in violation of the terms of this Agreement, or if Shareholder were to solicit Customers, Prospective Customers or employees of Bank in violation of the terms of this Agreement. Accordingly, Shareholder expressly acknowledges that he she is voluntarily entering into this Agreement and that the terms and conditions of this Agreement are fair and reasonable to Shareholder in all respects.

ARTICLE III

NONCOMPETITION AND NONSOLICITATION

3.1 Noncompetition. From the date of this Agreement and for the period ending on expiration of the twenty-four (24) months after the Effective Time of the Merger (t Applicable Period), Shareholder shall not, directly or indirectly, without the priod written consent of CBB own, manage, operate, control, finance or participate in the ownership, management, operation, control or financing of, or be connected as an officer, director, employee, partner, principal, agent, representative, consultant or otherwise with, any business or enterprise engaged in any business that is competit with or similar to the Enterprise within any of the California Counties of Fresno, K King, Madeira or Tulare (the Territory). Notwithstanding the above, Shareholde not be deemed to be engaged directly or indirectly in any business in contravention the

B-2

immediately preceding sentence, if (a) Shareholder participates in any such businessolely (i) as an officer or director of CVB or CBB or (ii) as a passive investor in up 5% of the equity securities or 10% of the debt securities of a company or partnersh provided such securities are publicly traded or (b) Shareholder is employed by a business or enterprise that is engaged primarily in a business other than that which competitive with or similar to the Enterprise and Shareholder does not apply in any manner his or her expertise at such business or enterprise to that part of such business or enterprise that is competitive with or similar to the Enterprise.

- 3.2 <u>Nonsolicitation</u>. During the Applicable Period, Shareholder shall not, directly of indirectly, without the prior written consent of CBB, on behalf of any Financial Institution, (i) solicit or aid in the solicitation of any Customers or Prospective Customers for Financial Services, or (ii) solicit or aid in the solicitation of any office or employees of Bank, or (iii) induce or attempt to induce immediately any Person who is a Customer, Prospective Customer, supplier, distributor, officer or employee Bank as of the date hereof or immediately prior to the Effective Time of the Merge terminate such person is relationships with, or to take any action that would be disadvantageous to CBB as the surviving corporation in the Bank Merger. Shareholder, upon the reasonable request of CVB or CBB, shall also agree to use her or her best efforts to retain the business of CVB and/or CBB and promote the acquisition of new business by CVB and/or CBB.
- 3.3 <u>Trade Secrets</u>. Without limiting the generality of the foregoing and at all times after the date hereof, other than for the benefit of Bank, or as otherwise approved by CVB or CBB, and, after the Effective Time of the Merger (as such term is defined the Merger Agreement), other than for the benefit of the CVB and/or CBB or as otherwise approved by CVB or CBB, Shareholder (i) shall make no use of the Trade Secrets, or any other part thereof, (ii) shall not disclose the Trade Secrets, or any pathereof, to any other Person, and (iii) shall deliver, on and after the Effective Time the Merger, upon the request of CBB, all documents, reports, drawings, designs, plans, proposals and other tangible evidence of Trade Secrets, now possessed or hereafter acquired by Shareholder, to CVB and/or CBB.
- 3.4 Exceptions. Notwithstanding any provision of this Agreement to the contrary, Shareholder may disclose or reveal any information, whether including in whole or part any Trade Secrets, that:
- (a) Shareholder is required to disclose or reveal under any applicable Law, provide Shareholder makes a good faith request that the confidentiality of the Trade Secret preserved and, to the extent not prohibited by applicable Laws, gives CBB prompt notice of such requirement in advance of such disclosure.
- (b) Shareholder is otherwise required to disclose or reveal by any Governmental Agency, as defined below, provided Shareholder makes a good faith request that the confidentiality of the Trade Secrets be preserved and, to the extent not prohibited by applicable Laws, gives CBB prompt notice of such requirement is advance of each disclosure;

(c) Upon the opinion of Shareholder s counsel, Shareholder is compelled to disclo else stand liable for contempt or suffer other censure or penalty imposed by any Governmental Entity, provided Shareholder makes a good faith request that the confidentiality of the Trade Secrets be preserved and, to the extent not prohibited b applicable Laws, gives CBB prompt notice of such requirement in advance of such disclosure.

In furtherance thereof, Shareholder understands that nothing contained in this Agreement prohibits or limits Shareholder from filing a charge or complaint with or reporting possible violations of federal, state or local law to any government agence entity, including but not limited to the Securities and Exchange Commission, the Department of Justice, the Equal Employment Opportunity Commission, the Feder Deposit Insurance Corporation or the California Department of Business Oversight (each a <u>Government Agency</u>), or making other disclosures protected under the whistleblower provisions of any law or regulation. Shareholder further understands that this Agreement does not prohibit or limit Shareholder s cooperation with any Government Agency nor does it preclude or limit Shareholder s right to receive an award for information provided to any Government Agency or as a result of any clifiled with such Government Agency. Shareholder acknowledges that prior notice to CBB or authorization by CBB is not required for the above-referenced protected activities.

B-3

ARTICLE IV

INDEPENDENCE OF OBLIGATIONS

The covenants of Shareholder set forth in this Agreement shall be construed as independent of any other agreement or arrangement between Shareholder, on the o hand, and CBB on the other, and the existence of any claim or cause of action by Shareholder against Bank, CBB, CVB, or any of their respective Affiliates (or the existence of any claim or cause of action by CBB or CVB against Shareholder, as a case may be), shall not constitute a defense to the enforcement of such covenants against Shareholder, or against CBB or CVB, as the case may be.

ARTICLE V

GENERAL

- 5.1 Amendments; Waivers. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto. No provision in this Agreement may be waived, except by means of a writing signed by the party against whom the waiver is sought to be enforced. The failure of any party hereto t enforce at any time any provision of this Agreement shall not be construed to be a waiver of such provision, nor in any way to affect the validity of this Agreement on any part hereof or the right of any party thereafter to enforce each and every such provision. No waiver of any breach of this Agreement shall be held to constitute a waiver of any other or subsequent breach.
- 5.2 <u>Integration</u>. This Agreement constitutes the entire agreement between the partic pertaining to the subject matter hereof and supersedes all prior agreements and understandings of the parties in connection therewith.

5.3 Termination.

- (a) This Agreement shall terminate automatically without further action in the ever that the Merger Agreement is terminated prior to the Effective Time of the Merger
- (b) Unless sooner terminated under subsection (a) of this Section 5.3, the obligation of Shareholder under this Agreement shall terminate only on the mutual agreement Shareholder, on the one hand, and CBB on the other hand.
- 5.4 Specific Performance. Shareholder acknowledges and agrees that irreparable injury will result to CBB in the event of a breach of any of the provisions of this Agreement and that CBB may have no adequate remedy at law with respect thereto Accordingly, in the event of a material breach of this Agreement, and in addition to any other legal or equitable remedy CBB may have, Shareholder agrees that the en of a preliminary injunction and a permanent injunction (including, without limitative specific performance) by a court of competent jurisdiction, to restrain the violation breach thereof by Shareholder or any Affiliates, agents, or any other persons acting or with Shareholder in any capacity whatsoever, is an appropriate remedy for any second control of the provisions of this Agreement and that CBB may have no adequate remedy at law with respect thereto accordingly.

breach and that Shareholder will not oppose the granting of such relief on the basis that CBB has an adequate remedy at law. Shareholder submits to the jurisdiction of such court in any such action. Shareholder agrees that he or she will not seek, and agrees to waive any requirement for, the securing or posting of a bond in connection with CBB is seeking or obtaining such equitable relief. In addition, after discussing matter with Shareholder, CBB shall have the right to inform any third party that CB reasonably believes to be, or to be contemplating, participating with Shareholder or receiving from Shareholder assistance in violation of this Agreement, of the terms this Agreement and the rights of CBB hereunder, and that participation by any such persons with Shareholder in activities in violation of Shareholder is agreement with CBB set forth in this Agreement may give rise to claims by CBB against such third party in addition to any other remedy to which they may be entitled at law or in equity.

B-4

5.5 Severability. If any provision of this Agreement shall be held by a court of competent jurisdiction to be unreasonable as to duration, activity or subject, it shall deemed to extend only over the maximum duration, range of activities or subjects to which such provision shall be valid and enforceable under applicable Law. If an provisions shall, for any reason, be held by a court of competent jurisdiction to be invalid, illegal or unenforceable, such invalidity, illegality or unenforceability shall not affect any other provision of this Agreement, but this Agreement shall be construed as if such invalid, illegal or unenforceable provision had never been contained herein.

5.6 <u>Notices</u>. Any notice or communication required or permitted hereunder, shall be deemed to have been given if in writing and (a) delivered in person, (b) delivered be confirmed facsimile transmission, (c) sent by overnight carrier, postage prepaid wi return receipt requested or (d) mailed by certified or registered mail postage prepaid with return receipt requested, addressed as follows:

If to CBB, addressed to:

Citizens Business Bank

701 North Haven Avenue

Ontario, California 91764

Attention: Christopher D. Myers

Fax No.: (909) 483-7199

With a copy addressed to:

Manatt, Phelps & Phillips, LLP

One Embarcadero Center, 30th Floor

San Francisco, California 94111

Attention: Craig D. Miller.

Fax No: (415) 291-7474

If to Shareholder, addressed to:

Fax No: ()

or at such other address and to the attention of such other person as a party may provide by notice to the other in accordance with this Section 5.6. Any such notice communication shall be deemed received on the date delivered personally or delive by confirmed facsimile transmission or on the next Business Day after it was sent by overnight carrier, postage prepaid with return receipt requested or on the third Business Day after it was sent by certified or registered mail, postage prepaid with return receipt requested.

- 5.7 <u>Waiver of Breach</u>. Any failure or delay by CBB in enforcing any provision of the Agreement shall not operate as a waiver thereof. The waiver by CBB of a breach of any provision of this Agreement by Shareholder shall not operate or be construed a waiver of any subsequent breach or violation thereof. All waivers shall be in writing and signed by the party to be bound.
- 5.8 <u>Assignment</u>. This Agreement may be assignable by CBB only in connection w a sale of all or substantially all of its assets or a merger or reorganization in which not the surviving corporation. Any attempted assignment in violation of this prohibition shall be null and void.

B-5

- 5.9 <u>Binding Effect; Benefit to Successors</u>. This Agreement shall be binding upon Shareholder and upon Shareholder s successor and representatives and shall inure the benefit of CBB and its successors, representatives and assigns.
- 5.10 <u>Governing Law</u>. This Agreement and the legal relations between the parties slee governed by and construed in accordance with the laws of the State of California applicable to contracts between California parties made and performed in this State.
- 5.11 <u>Headings</u>. The descriptive headings of the several Articles and Sections of thi Agreement are inserted for convenience only and do not constitute a part of this Agreement.
- 5.12 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, a of which shall be considered one and the same agreement and shall become effecti when one or more counterparts have been signed by each party hereto and delivere each party hereto. Facsimiles containing original signatures shall be deemed for all purposes to be originally signed copies of the documents which are the subject of sfacsimiles.

[SIGNATURES APPEAR ON THE IMMEDIATELY FOLLOWING PAGE]

B-6

IN WITNESS WHEREOF, the parties to this Agreement have duly executed this Agreement as of the day and year first above written.

CITIZENS BUSINESS BANK

By: Title:

SHAREHOLDER

(Signature)

(Type or Print Shareholder Name)

B-7

Attachment 1 to Exhib

Officers Signing Noncompetition And Nonsolicitation Agreements

- 1. Allan W. Stone
- 2. Roy O. Estridge
- 3. Bill Kitchen

Exhib

AGREEMENT OF MERGER

THIS AGREEMENT OF MERGER, dated as of [] (this <u>Agreement</u>), is made entered into by and between, CVB Financial Corp., a California corporation (<u>CV</u> and Valley Commerce Bancorp, a California corporation (<u>Valley</u>).

WHEREAS, the Boards of Directors of CVB and Valley have approved, and deem advisable and in the best interests of CVB, Valley and their respective shareholders that CVB and Valley consummate the business transaction provided for in this Agreement in which Valley would merge with and into CVB (the <u>Merger</u>) as contemplated in that certain Agreement and Plan of Reorganization and Merger da as of September 22, 2016 by and among CVB and Valley (the <u>Plan of Merger</u>), providing, among other things, for the execution and filing of this Agreement and to consummation of the Merger.

NOW, THEREFORE, in consideration of the promises and mutual agreements contained in this Agreement, the parties to this Agreement hereby agree that Valley shall be merged with and into CVB in accordance with the provisions of the laws of the State of California upon the terms and subject to the conditions set forth as follows:

Section 1. The Merger.

- (a) <u>Effective Time</u>. The Merger shall become effective on the date and at the time this Agreement of Merger is filed with the California Secretary of State (the <u>Effectime</u>).
- (b) Effect of the Merger. At the Effective Time, Valley shall be merged with and in CVB and the separate corporate existence of Valley shall cease. CVB shall be the surviving corporation (the <u>Surviving Corporation</u>) in the Merger. The Surviving Corporation shall thereupon succeed, without other transfer, to all rights and properties of, and shall be subject to all the debts and liabilities of, Valley and the separate existence of Surviving Corporation as a California corporation, with all its purposes, objects, rights, powers, privileges and franchises, shall continue unaffect and unimpaired by the Merger.
- (c) <u>Name of Surviving Corporation</u>. The name of the Surviving Corporation shall be CVB Financial Corp.
- (d) <u>Articles of Incorporation and Bylaws</u>. From and after the Effective Time and unthereafter amended as provided by law, the Articles of Incorporation and Bylaws of CVB as in effect immediately prior to the Effective Time shall be and continue to the Articles of Incorporation and Bylaws of the Surviving Corporation.
- (e) <u>Board of Directors and Officers</u>. The directors and officers of CVB at the Effectime will be the directors and officers of the Surviving Corporation until they are

removed or their successors are elected and qualified.

(f) <u>Further Actions</u>. Valley shall execute and deliver any documents and instrumen and shall take all actions, as requested by the Surviving Corporation, necessary or desirable to evidence or carry out the Merger.

Section 2. Treatment of Shares.

(a) Shares of Valley.

(i) At the Effective Time, by virtue of the Merger, and without any action on the part of the holders of Valley common stock, each share of Valley common stock issued and outstanding immediately prior to the Effective Time (other than shares that are dissenting shares—within the meaning of Chapter 13 of the California General Corporation Law) shall converted into the right to receive (i) [] shares of CVB common stock, together with any cash in lieu of fractional shares, and (ii) \$[] cash

C-1

- (ii) Any shares of Valley common stock owned by Valley as treasury stock or own directly or indirectly, by Valley, CVB or any of CVB s subsidiaries (other than the held in a fiduciary capacity or as a result of debts previously contracted) shall automatically be cancelled and retired and shall cease to exist at the Effective Time the Merger and no consideration shall be issued in exchange therefor.
- (iii) Shares of Valley common stock that are dissenting shares within the meaning Chapter 13 of the California General Corporation Law will not be converted as described in Section 3(a)(i), but from and after the Effective Time will represent on the right to receive such value as may be determined under Chapter 13 of the California General Corporation Law.
- (iv) At the Effective Time, the stock transfer books of Valley will be closed and no transfer of Valley common stock theretofore outstanding will thereafter be made.
- (b) <u>Shares of CVB</u>. All shares of CVB common stock issued and outstanding immediately prior to the Effective Time shall remain outstanding and unaffected by the Merger.

Section 3. Termination and Amendment.

- (a) <u>Termination</u>. This Agreement shall terminate prior to the Effective Time in the event that the Plan of Merger shall be terminated as provided therein.
- (b) Amendment. This Agreement may be amended by CVB and Valley at any time prior to the Effective Time without the approval of the shareholders of CVB or Va with respect to any of its terms except any change in its principal terms, the terms relating to the form or amount of consideration to be delivered to Valley sharehold in the Merger or as may otherwise be required by the Plan of Merger or by law. Th Agreement may not be amended, except by an instrument in writing signed on beh of each of the parties hereto.
- Section 4. Rules of Construction. Descriptive headings as to the contents of particular sections are for convenience only and do not control or affect the meaning, construction or interpretation of this Agreement. Each use herein of the plural includes singular and vice versa, in each case as the context requires or as it is otherwise appropriate. The word or is used in the inclusive sense. Any and all documents instruments referred to herein are incorporated herein by reference hereto as thoughfully set forth herein verbatim. If there is any conflict between the terms of this Agreement and the terms of the Plan of Merger, the terms of the Plan of Merger are control.
- Section 5. <u>Counterparts</u>. This Agreement may be signed in any number of counterparts, each of which shall be deemed an original, and all of which shall be deemed but one and the same instrument.

[Continued and to be signed on following page]

C-2

IN WITNESS WHEREOF, the parties have duly executed this Agreement of Merg as of the date first written above.

CVB FINANCIAL CORP

By: []

By: []

VALLEY COMMERCE BANCORP

By: []

By: []

C-3

Exhib

AGREEMENT OF MERGER

THIS AGREEMENT OF MERGER, dated as of [], 2017 (this Merger Agreemer made and entered into by and between Citizens Business Bank, a California banking corporation (<u>Citizens</u>), and Valley Business Bank, a California banking corpora (<u>Valley</u>).

WHEREAS, the Boards of Directors of Citizens and Valley have approved, and de it advisable and in the best interests of Citizens, Valley and their respective shareholders, that Citizens and Valley consummate the business transaction provid for herein in which Valley would merge with and into Citizens (the <u>Merger</u>) as contemplated in that Agreement and Plan of Reorganization and Merger dated as of September 22, 2016 by and between CVB Financial Corp. and Valley Commerce Bancorp (the <u>Plan of Merger</u>), providing, among other things, for the execution filing of this Merger Agreement and the consummation of the Merger.

NOW, THEREFORE, in consideration of the promises and mutual agreements contained in this Merger Agreement, Citizens and Valley hereby agree that Valley shall be merged with and into Citizens in accordance with the provisions of the law of the State of California upon the terms and subject to the conditions set forth as follows:

Section 1. The Merger.

- (a) <u>Effective Time</u>. The Merger shall become effective on the date and at the time this Merger Agreement, as certified by the California Secretary of State, is filed with California Department of Business Oversight pursuant to Section 4887(b) of the California Financial Code (the <u>Effective Time</u>).
- (b) Effect of the Merger. At the Effective Time, Valley shall be merged with and ir Citizens and the separate corporate existence of Valley shall cease. Citizens shall be the surviving corporation (the <u>Surviving Corporation</u>) in the Merger. The Surviv Corporation shall thereupon succeed, without other transfer, to all rights and properties of, and shall be subject to all the debts and liabilities of, Valley and the separate existence of Surviving Corporation as a California corporation, with all its purposes, objects, rights, powers, privileges and franchises, shall continue unaffect and unimpaired by the Merger.
- (c) <u>Name of Surviving Corporation</u>. The name of the Surviving Corporation shall be Citizens Business Bank.
- (d) <u>Articles of Incorporation and Bylaws</u>. From and after the Effective Time and unthereafter amended as provided by law, the Articles of Incorporation and Bylaws o Citizens as in effect immediately prior to the Effective Time shall be and continue be the Articles of Incorporation and Bylaws of the Surviving Corporation.

- (e) <u>Board of Directors</u>. The directors and officers of Citizens at the Effective Time will be the directors and officers of the Surviving Corporation until they are remov or their successors are elected and qualified.
- (f) <u>Offices</u>. All branch offices of Citizens and Valley that were in lawful operation immediately prior to the Effective Time shall continue to be the branch offices of t Surviving Corporation upon consummation of the Merger, subject to the opening c closing of any offices which may be authorized by the Surviving Corporation and applicable regulatory authorities after the date of this Agreement.

Section 3. Treatment of Shares.

- (a) <u>Shares of Valley</u>. At the Effective Time, by virtue of the Merger, and without a action on the part of the holders of common stock of Valley (<u>Valley Common Stock</u> each share of Valley Common Stock issued and outstanding immediately prior to t Effective Time shall be cancelled without consideration.
- (b) <u>Shares of Citizens</u>. All shares of common stock of Citizens issued and outstand immediately prior to the Effective Time shall remain outstanding.

Section 4. Termination and Amendment.

- (a) <u>Termination</u>. Notwithstanding the approval of this Merger Agreement by the shareholders of Valley or Citizens, this Merger Agreement shall terminate forthwit prior to the Effective Time in the event that the Plan of Merger shall be terminated therein provided.
- (b) <u>Amendment</u>. This Merger Agreement may be amended by Citizens and Valley any time prior to the Effective Time without the approval of the shareholders of Valley or Citizens with respect to any of its terms except any change in its principaterms, the terms relating to the form or amount of consideration to be delivered to Valley shareholders in the Merger or as may otherwise be required by the Plan of Merger or by law. This Merger Agreement may not be amended, except by an instrument in writing signed on behalf of each of the parties hereto.
- (c) <u>Counterparts</u>. This Merger Agreement may be signed in any number of counterparts, each of which shall be deemed an original, and all of which shall be deemed but one and the same instrument.

[Continued and to be signed on following page]

D-2

IN WITNESS WHEREOF, the parties have duly executed this Merger Agreement of the date first written above.

CITIZENS BUSINESS BANK	VALLEY BUSINESS BANK
By: [] []	By: [] []
By: [], Secretary	By: [], Secretary
	D 2

APPENDI

September 22, 2016

Board of Directors

Valley Commerce Bancorp

701 West Main Street

Visalia, California 93291

Members of the Board:

You have requested our opinion as to the fairness, from a financial point of view the holders of the outstanding shares of common stock of Valley Commerce Banc Visalia, California (Valley) of the consideration (the Merger Consideration received by Valley in the merger (the Merger) of Valley with and into CVB Fin Corp., Ontario, California (CVB) pursuant to the Agreement and P Reorganization and Merger by and between CVB and Valley (the Agreement).

Pursuant to the terms of the Agreement, each share of Valley Common Stock iss and outstanding immediately prior to the Effective Time shall be converted into right to receive (i) cash in an amount equal to the Per Share Cash Amount, with interest thereon, and (ii) the number of whole shares of CVB Common Stock equa the Per Share Exchange Ratio (rounded up or down to the nearest whole share) (s cash and such shares collectively, the Merger Consideration). The Per Shar Amount means the quotient of the Aggregate Cash amount of \$23.4 million minus amount, if any, that Transaction Costs exceed \$3.5 million divided by Valley Clo Shares. The Per Share Exchange Ratio means the quotient of 1,942,673 shares of C Common Stock divided by Valley Closing Shares. Additionally, Valley shall declar one-time cash dividend payable to holders of record of outstanding Valley Com Stock as of the Closing Date in an aggregate amount that shall not exceed the amount by which the Closing Tangible Common Equity exceeds \$37,500,000, subject adjustment as indicated in Section 7.2(n)(vi) (the Special Dividend). All capit items used in this letter shall have the meanings ascribed to them in the Agreem The terms of the Merger are more fully set forth in the Agreement.

For purposes of this opinion and in connection with our review of the proportransaction, we have, among other things:

- 1. Reviewed the terms of the draft of the Agreement dated September 20, 2016;
- Reviewed certain publicly available financial statements, both audited (wl available) and un-audited, and related financial information of Valley CVB, including those included in their respective annual reports for the two years and their respective quarterly reports for the past two years;
- 3. Reviewed certain internal financial information and financial forecasts related to the business, earnings, cash flows, assets and prospects of Valley furnisto us by Valley;

Vining-Sparks Community Bank Advisory Group

1601 W. 38th Street Suite 207 Austin, Texas 78731

(512) 495-9890 Fax (512) 495-9894

Member FINRA/SIPC

B-1

Board of Directors

Valley Bancorp

September 22, 2016

Page 2

- 4. Reviewed publicly available research reports and analyst earnings estimates CVB as provided by CVB;
- 5. Held discussions with members of executive and senior management of Va concerning the past and current results of operations of Valley, its cur financial condition and management s opinion of its future prospects;
- Reviewed reported market prices and historical trading activity of CVB Valley common stock and reviewed the Trading Collar for CVB as define Section 8.1(f) of the Agreement;
- 7. Reviewed the financial terms of merger and acquisition transactions, to extent publicly available, involving financial institutions and finan institution holding companies that we deemed to be relevant; and
- 8. Reviewed such other information, financial studies, analyses investigations, as we considered appropriate under the circumstances.

In giving our opinion, we have assumed and relied, without independent verificat upon the accuracy and completeness of all of the financial and other information has been provided to us by Valley and CVB, and their respective representatives, of the publicly available information that was reviewed by us. We are not exper the evaluation of allowances for loan losses and have not independently verified s allowances. We assumed that the aggregate allowance for loan losses set forth in financial statements of CVB and Valley is adequate to cover such losses and comp fully with applicable law, regulatory policy and sound banking practice as of the of such financial statements. We were not retained to and we did not condu physical inspection of any of the properties or facilities of Valley or CVB, did make any independent evaluation or appraisal of the assets, liabilities or prospect Valley or CVB, were not furnished with any such evaluation or appraisal, and did review any individual credit files. Our opinion is necessarily based on econor market, and other conditions as in effect on, and the information made available t as of, the date hereof. Accordingly, it is important to understand that altho subsequent developments may affect its opinion, we do not have any obligatio further update, revise, or reaffirm our opinion. We express no opinion on matters

legal, regulatory, tax or accounting nature or the ability of the Merger, as set fort the Agreement, to be consummated. No opinion is expressed as to whether alternative transaction might be more favorable to holders of Valley s common than the Merger.

Vining Sparks IBG, L.P. (Vining Sparks), as part of its investment banking but is regularly engaged in the valuation of banks and bank holding companies, thrifts thrift holding companies, and various other financial services companies connection with mergers and acquisitions and valuations for other purposes rendering this fairness opinion, we have acted on behalf of the Board of Director Valley and will receive a fee for our services, which is payable upon delivery of opinion.

Vining Sparks opinion as expressed herein is limited to the fairness, from a final point of view, of the Merger Consideration to be received by the holders of Valcommon stock in the Merger and does not address Valley a underlying bus decision to proceed with the Merger. We have been retained on behalf of the Boar Directors of Valley, and our opinion does not constitute a recommendation to director of Valley as to how such director should vote with respect to the Agreem In rendering this opinion, we express no opinions with respect to the amount or na of any compensation to any officers, directors, or employees of Valley or CVB, or class of such persons relative to the consideration to be received by the holders of common stock of Valley in the transaction or with respect to the fairness of any scompensation.

B-2

Board of Directors

Valley Bancorp

September 22, 2016

Page 3

In the two years prior to the issuance of this opinion, Vining Sparks engage securities and loan sales and trading activity with Valley and CVB and/or to subsidiary banks for which Vining Sparks was paid commissions or other fees, who may include mark-ups on the purchase or sale of loans and securities.

Except as hereinafter provided, this opinion may not be disclosed, communical reproduced, disseminated, quoted or referred to at any time, to any third party of any manner or for any purpose whatsoever without our prior written consent, who consent will not be unreasonably withheld, based upon review by us of the contertany such public reference, which shall be satisfactory to us in our reasonation judgment. This letter is addressed and directed to the Board of Directors of Vallet your consideration of the Merger and is not intended to be and does not constitute recommendation to any shareholder as to how such shareholder should vote we respect to the Merger. This opinion was approved by the fairness opinion committed of Vining Sparks.

Subject to the foregoing and based on our experience as investment bankers, activities as described above, and other factors we have deemed relevant, we are of opinion as of the date hereof that the Merger Consideration to be received by holders of Valley common stock is fair, from a financial point of view.

Sincerely,

VINING SPARKS IBG, L.P.

B-3

APPENDI

§ 1300. Reorganization or short-form merger; dissenting shares; corporate purchase at fair market value; definitions

- (a) If the approval of the outstanding shares (Section 152) of a corporation is requifor a reorganization under subdivisions (a) and (b) or subdivision (e) or (f) of Section 1201, each shareholder of the corporation entitled to vote on the transactio and each shareholder of a subsidiary corporation in a short-form merger may, by complying with this chapter, require the corporation in which the shareholder hold shares to purchase for cash at their fair market value the shares owned by the shareholder which are dissenting shares as defined in subdivision (b). The fair mark value shall be determined as of the day of, and immediately prior to, the first announcement of the terms of the proposed reorganization or short-form merger, excluding any appreciation or depreciation in consequence of the proposed reorganization or short-form merger, as adjusted for any stock split, reverse stock split, or share dividend that becomes effective thereafter.
- **(b)** As used in this chapter, dissenting shares means shares to which all of the following apply:
- (1) That were not, immediately prior to the reorganization or short-form merger, list on any national securities exchange certified by the Commissioner of Corporations under subdivision (*o*) of Section 25100, and the notice of meeting of shareholders that the reorganization summarizes this section and Sections 1301, 1302, 1303 and 1304; provided, however, that this provision does not apply to any shares with respect to which there exists any restriction on transfer imposed by the corporation by any law or regulation; and provided, further, that this provision does not apply that any shares where the holder of those shares is required, by the terms of the reorganization or short-form merger, to accept for the shares anything except:

 (A) shares of any other corporation, which shares, at the time the reorganization or
- short-form merger is effective, are listed on any national securities exchange certifity the Commissioner of Corporations under subdivision (*o*) of Section 25100;
- (B) cash in lieu of fractional shares described in the foregoing subparagraph (A); o (C) any combination of the shares and cash in lieu of fractional shares described in foregoing subparagraphs (A) and (B).
- (2) That were outstanding on the date for the determination of shareholders entitled vote on the reorganization and (A) were not voted in favor of the reorganization or (B) if described in paragraph (1), were voted against the reorganization, or were he of record on the effective date of a short-form merger; provided, however, that subparagraph (A) rather than subparagraph (B) of this paragraph applies in any cas where the approval required by Section 1201 is sought by written consent rather that a meeting.
- (3) That the dissenting shareholder has demanded that the corporation purchase at their fair market value, in accordance with Section 1301.

- (4) That the dissenting shareholder has submitted for endorsement, in accordance v Section 1302.
- (c) As used in this chapter, dissenting shareholder means the recordholder of dissenting shares and includes a transferee of record.

§ 1301. Notice to holders of dissenting shares in reorganizations; demand for purchase; time; contents

(a) If, in the case of a reorganization, any shareholders of a corporation have a right under Section 1300, subject to compliance with paragraphs (3) and (4) of subdivisition (b) thereof, to require the corporation to purchase their shares for cash, that corporation shall mail to each of those shareholders a notice of the approval of the reorganization by its outstanding shares (Section 152) within 10 days after the date that approval, accompanied by a copy of Sections 1300, 1302, 1303, and 1304 and section, a statement of the price determined by the corporation to represent the fair market value of the dissenting shares, and a brief description of the procedure to be followed if the shareholder desires to exercise the shareholder is right under those sections.

C-1

The statement of price constitutes an offer by the corporation to purchase at the pri stated any dissenting shares as defined in subdivision (b) of Section 1300, unless the lose their status as dissenting shares under Section 1309.

- (b) Any shareholder who has a right to require the corporation to purchase the shareholder s shares for cash under Section 1300, subject to compliance with paragraphs (3) and (4) of subdivision (b) thereof, and who desires the corporation to purchase shares shall make written demand upon the corporation for the purchase those shares and payment to the shareholder in cash of their fair market value. The demand is not effective for any purpose unless it is received by the corporation or a transfer agent thereof (1) in the case of shares described in subdivision (b) of Section 1300, not later than the date of the shareholders meeting to vote upon the reorganization, or (2) in any other case, within 30 days after the date on which the notice of the approval by the outstanding shares pursuant to subdivision (a) or the notice pursuant to subdivision (h) of Section 1110 was mailed to the shareholder.
- (c) The demand shall state the number and class of the shares held of record by the shareholder which the shareholder demands that the corporation purchase and shall contain a statement of what the shareholder claims to be the fair market value of the shares as determined pursuant to subdivision (a) of Section 1300. The statement of market value constitutes an offer by the shareholder to sell the shares at that price.

§ 1302. Submission of share certificates for endorsement; uncertificated securities

Within 30 days after the date on which notice of the approval by the outstanding shares or the notice pursuant to subdivision (h) of Section 1110 was mailed to the shareholder, the shareholder shall submit to the corporation at its principal office of the office of any transfer agent thereof, (a) if the shares are certificated securities, the shareholder is certificated representing any shares which the shareholder demands the corporation purchase, to be stamped or endorsed with a statement that the share are dissenting shares or to be exchanged for certificates of appropriate denomination of the number of shares which the shareholder demands that the corporation purchase to the number of shares which the shareholder demands that the corporation purchase transfers of the dissenting shares on the books of the corporation new certificates, initial transaction statement, and other written statements issued therefor shall bear a like statement, together with the name of the original dissenting holder of the shares.

§ 1303. Payment of agreed price with interest; agreement fixing fair market value; filing; time of payment

(a) If the corporation and the shareholder agree that the shares are dissenting share and agree upon the price of the shares, the dissenting shareholder is entitled to the agreed price with interest thereon at the legal rate on judgments from the date of the agreement. Any agreements fixing the fair market value of any dissenting shares as between the corporation and the holders thereof shall be filed with the secretary of corporation.

(b) Subject to the provisions of Section 1306, payment of the fair market value of dissenting shares shall be made within 30 days after the amount thereof has been agreed or within 30 days after any statutory or contractual conditions to the reorganization are satisfied, whichever is later, and in the case of certificated securities, subject to surrender of the certificates therefor, unless provided otherwise by agreement.

§ 1304. Action to determine whether shares are dissenting shares or fair mark value; limitation; joinder; consolidation; determination of issues; appointmen appraisers

(a) If the corporation denies that the shares are dissenting shares, or the corporation and the shareholder fail to agree upon the fair market value of the shares, then the shareholder demanding purchase of such shares as dissenting shares or any interest corporation, within six months after the date on which notice of the approval by the outstanding shares (Section 152) or notice pursuant to subdivision (h) of Section 152 was mailed to the shareholder, but not thereafter, may file a complaint in the supericourt of the proper county praying the court

C-2

to determine whether the shares are dissenting shares or the fair market value of the dissenting shares or both or may intervene in any action pending on such a compla

- **(b)** Two or more dissenting shareholders may join as plaintiffs or be joined as defendants in any such action and two or more such actions may be consolidated.
- (c) On the trial of the action, the court shall determine the issues. If the status of the shares as dissenting shares is in issue, the court shall first determine that issue. If the fair market value of the dissenting shares is in issue, the court shall determine, or suppoint one or more impartial appraisers to determine, the fair market value of the shares.

§ 1305. Report of appraisers; confirmation; determination by court; judgmen payment; appeal; costs

- (a) If the court appoints an appraiser or appraisers, they shall proceed forthwith to determine the fair market value per share. Within the time fixed by the court, the appraisers, or a majority of them, shall make and file a report in the office of the cl of the court. Thereupon, on the motion of any party, the report shall be submitted to the court and considered on such evidence as the court considers relevant. If the confinds the report reasonable, the court may confirm it.
- (b) If a majority of the appraisers appointed fail to make and file a report within 10 days from the date of their appointment or within such further time as may be allow by the court or the report is not confirmed by the court, the court shall determine the fair market value of the dissenting shares.
- (c) Subject to the provisions of Section 1306, judgment shall be rendered against the corporation for payment of an amount equal to the fair market value of each dissenshare multiplied by the number of dissenting shares which any dissenting sharehold who is a party, or who has intervened, is entitled to require the corporation to purchase, with interest thereon at the legal rate from the date on which judgment we entered.
- (d) Any such judgment shall be payable forthwith with respect to uncertificated securities and, with respect to certificated securities, only upon the endorsement and delivery to the corporation of the certificates for the shares described in the judgment.
- (e) The costs of the action, including reasonable compensation to the appraisers to fixed by the court, shall be assessed or apportioned as the court considers equitable but, if the appraisal exceeds the price offered by the corporation, the corporation shappy the costs (including in the discretion of the court attorneys—fees, fees of experwitnesses and interest at the legal rate on judgments from the date of compliance we Sections 1300, 1301 and 1302 if the value awarded by the court for the shares is me than 125 percent of the price offered by the corporation under subdivision (a) of Section 1301).

§ 1306. Prevention of immediate payment; status as creditors; interest

To the extent that the provisions of Chapter 5 prevent the payment to any holders of dissenting shares of their fair market value, they shall become creditors of the corporation for the amount thereof together with interest at the legal rate on judgme until the date of payment, but subordinate to all other creditors in any liquidation proceeding, such debt to be payable when permissible under the provisions of Chapter 5.

§ 1307. Dividends on dissenting shares

Cash dividends declared and paid by the corporation upon the dissenting shares aft the date of approval of the reorganization by the outstanding shares (Section 152) a prior to payment for the shares by the corporation shall be credited against the total amount to be paid by the corporation therefor.

C-3

§ 1308. Rights of dissenting shareholders pending valuation; withdrawal of demand for payment

Except as expressly limited in this chapter, holders of dissenting shares continue to have all the rights and privileges incident to their shares, until the fair market value their shares is agreed upon or determined. A dissenting shareholder may not withdra demand for payment unless the corporation consents thereto.

§ 1309. Termination of dissenting share and shareholder status

Dissenting shares lose their status as dissenting shares and the holders thereof ceas be dissenting shareholders and cease to be entitled to require the corporation to purchase their shares upon the happening of any of the following:

- (a) The corporation abandons the reorganization. Upon abandonment of the reorganization, the corporation shall pay on demand to any dissenting shareholder who has initiated proceedings in good faith under this chapter all necessary expens incurred in such proceedings and reasonable attorneys fees.
- (b) The shares are transferred prior to their submission for endorsement in accorda with Section 1302 or are surrendered for conversion into shares of another class in accordance with the articles.
- (c) The dissenting shareholder and the corporation do not agree upon the status of shares as dissenting shares or upon the purchase price of the shares, and neither file complaint or intervenes in a pending action as provided in Section 1304, within six months after the date on which notice of the approval by the outstanding shares or notice pursuant to subdivision (h) of Section 1110 was mailed to the shareholder.
- (d) The dissenting shareholder, with the consent of the corporation, withdraws the shareholder s demand for purchase of the dissenting shares.

§ 1310. Suspension of right to compensation or valuation proceedings; litigation of shareholders approval

If litigation is instituted to test the sufficiency or regularity of the votes of the shareholders in authorizing a reorganization, any proceedings under Sections 1304 1305 shall be suspended until final determination of such litigation.

§ 1311. Exempt shares

This chapter, except Section 1312, does not apply to classes of shares whose terms and provisions specifically set forth the amount to be paid in respect to such shares the event of a reorganization or merger.

§ 1312. Right of dissenting shareholder to attack, set aside or rescind merger or reorganization; restraining order or injunction; conditions

(a) No shareholder of a corporation who has a right under this chapter to demand payment of cash for the shares held by the shareholder shall have any right at law of equity to attack the validity of the reorganization or short-form merger, or to have to reorganization or short-form merger set aside or rescinded, except in an action to to whether the number of shares required to authorize or approve the reorganization have been legally voted in favor thereof; but any holder of shares of a class whose terms and provisions specifically set forth the amount to be paid in respect to them in the event of a reorganization or short-form merger is entitled to payment in accordance with those terms and provisions or, if the principal terms of the reorganization are approved pursuant to subdivision (b) of Section 1202, is entitled to payment in accordance with the terms and provisions of the approved reorganization.

(b) If one of the parties to a reorganization or short-form merger is directly or indirectly controlled by, or under common control with, another party to the reorganization or short-form merger, subdivision (a) shall not

C-4

apply to any shareholder of such party who has not demanded payment of cash for such shareholder is shares pursuant to this chapter; but if the shareholder institutes action to attack the validity of the reorganization or short-form merger or to have the reorganization or short-form merger set aside or rescinded, the shareholder shall not thereafter have any right to demand payment of cash for the shareholder is shares pursuant to this chapter. The court in any action attacking the validity of the reorganization or short-form merger or to have the reorganization or short-form merger set aside or rescinded shall not restrain or enjoin the consummation of the transaction except upon 10 days prior notice to the corporation and upon a determination by the court that clearly no other remedy will adequately protect the complaining shareholder or the class of shareholders of which such shareholder is a member.

(c) If one of the parties to a reorganization or short-form merger is directly or indirectly controlled by, or under common control with, another party to the reorganization or short-form merger, in any action to attack the validity of the reorganization or short-form merger or to have the reorganization or short-form merger set aside or rescinded, (1) a party to a reorganization or short-form merger which controls another party to the reorganization or short-form merger shall have burden of proving that the transaction is just and reasonable as to the shareholders the controlled party, and (2) a person who controls two or more parties to a reorganization shall have the burden of proving that the transaction is just and reasonable as to the shareholders of any party so controlled.

§ 1313. Conversions deemed to constitute a reorganization; application of chapter

A conversion pursuant to Chapter 11.5 (commencing with Section 1150) shall be deemed to constitute a reorganization for purposes of applying the provisions of th chapter, in accordance with and to the extent provided in Section 1159.

C-5

PART II

INFORMATION NOT REQUIRED IN PROSPECTUS

Item 20. Indemnification of Directors and Officers

The following summary is qualified in its entirety by reference to the complete text. CVB Financial s articles of incorporation and bylaws.

CVB Financial is subject to the California General Corporation Law (the CC which provides a detailed statutory framework covering indemnification of any off or other agent of a corporation who is made or threatened to be made a party to legal proceeding by reason of his or her services on behalf of such corporation.

With respect to indemnification, the CGCL provides that to the extent any offi director or other agent of a corporation is successful on the merits in defense legal proceeding to which such person is a party or is threatened to be made a part reason of his or her service on behalf of such corporation or in defense of any cla issue, or matter therein, such agent shall be indemnified against expenses actually reasonably incurred by the agent in connection therewith, but does not req indemnification in any other circumstance. The CGCL also provides that a corpora may indemnify any agent of the corporation, including officers and directors, aga expenses, judgments, fines, settlements and other amounts actually and reasons incurred in a third-party proceeding against such person by reason of his or services on behalf of the corporation, provided the person acted in good faith and manner he or she reasonably believed to be in the best interests of such corporat The CGCL further provides that in derivative suits a corporation may indemnify a person against expenses incurred in such a proceeding, provided such person a in good faith and in a manner he or she reasonably believed to be in the best inter of the corporation and its shareholders. Indemnification is not available in deriva actions (i) for amounts paid or expenses incurred in connection with a matter th settled or otherwise disposed of without court approval or (ii) with respect to mai for which the agent shall have been adjudged to be liable to the corporation unless court shall determine that such person is entitled to indemnification.

The CGCL permits the advancing of expenses incurred in defending any proceed against a corporate agent by reason of his or her service on behalf of the corporate upon the giving of a promise to repay any such sums in the event it is later determentate that such person is not entitled to be indemnified. Finally, the CGCL provides that indemnification provided by the statute is not exclusive of other rights to which the seeking indemnification may be entitled, by bylaw, agreement or otherwise, to extent additional rights are authorized in a corporation is articles of incorporation law further permits a corporation to procure insurance on behalf of its direct officers and agents against any liability incurred by any such individual, even corporation would not otherwise have the power under applicable law to indem the director, officer or agent for such expenses.

The Bylaws of CVB Financial provide that it shall, to the maximum extent permit by the CGCL, have power to indemnify each of its agents against expenigudgments, fines, settlements and other amounts actually and reasonably incurre connection with any proceeding arising by reason of the fact any such person is or an agent of the corporation, and shall have power to advance to each such age expenses incurred in defending any such proceeding to the maximum extent permit by that law.

Directors and Officers Liability Insurance

CVB Financial presently maintains a policy of directors and officers li insurance that provides coverage sufficiently broad to permit indemnification uncertain circumstances for liabilities (including reimbursement for expenses incur arising under the Securities Act of 1933.

II-1

Indemnification Agreements

CVB Financial has entered into an employment agreement with its chief execu officer agreeing, among other things, to indemnify him against liabilities relating his services as a director and officer of CVB Financial and the advancement expenses under certain circumstances. The employment agreement also requires CF Financial to use its reasonable best efforts to purchase and maintain one or no policies of directors—and officers—liability insurance to cover liabilities asserted a or incurred by, its chief executive officer. In addition, CVB Financial has entered indemnification agreements with each of its directors.

Item 21. Exhibits and Financial Statement Schedules

(a) The following is a list of exhibits to this registration statement:

Exhibit

No.	Description
2.1	Agreement and Plan of Reorganization and Merger, dated September 2016, by and between CVB Financial Corp. and Valley Comme Bancorp (included as Appendix A to the proxy statement/prospecontained in Part I of this Registration Statement).
3.1	Articles of Incorporation of the Registrant (incorporated herein reference to Exhibit 3.1 to CVB Financial Corp. s Quarterly Report Form 10-Q filed with the SEC on August 9, 2010).
3.2	Bylaws of the Registrant (incorporated herein by reference to Exhi 3.2(A) and 3.2(B) to CVB Financial Corp. s Annual Report on 10-K filed with the SEC on February 27, 2009).
5.1	Opinion of Manatt, Phelps & Phillips, LLP regarding to the validit the securities being registered.
8.1	Opinion of Manatt, Phelps & Phillips, LLP regarding certain tax mat
10.1	Consulting Agreement dated September 22, 2016 by and betw Citizens Business Bank and Allan W. Stone.
23.1	Consent of KPMG LLP, independent registered public accounting to of the Registrant.
23.2	Consent of Manatt, Phelps & Phillips, LLP (included in Exhibit 5.1).
23.3	Consent of Manatt, Phelps & Phillips, LLP (included in Exhibit 8.1).
24.1	Power of Attorney (included on the signature page to the Registra Statement).
99.1	Consent of Vining Sparks IBG, LP.

99.2 Form of Proxy for Special Meeting of Shareholders of Val Commerce Bancorp.

Item 22. Undertakings

The undersigned registrant hereby undertakes:

- (a) To file, during any period in which offers or sales are being made, a post-effect amendment to this Registration Statement:
- (1) To include any prospectus required by Section 10(a)(3) of the Securities Ac 1933.
- (2) To reflect in the prospectus any facts or events arising after the effective dat the registration statement (or the most recent post-effective amendment thereof) to individually or in the aggregate, represent a fundamental change in the information forth in the registration statement. Notwithstanding the foregoing, any increas decrease in volume of securities offered (if the total dollar value of securities offerwould not exceed that which was registered) and any deviation from the low or lend of the estimated maximum offering range may be reflected in the form prospectus filed with the SEC pursuant to Rule 424(b) if, in the aggregate, the char in volume and price represent no more than a 20 percent change in the maximaggregate offering price set forth in the Calculation of Registration Fee table effective registration statement.

II-2

- (3) To include any material information with respect to the plan of distribution previously disclosed in the registration statement or any material change to sinformation in the registration statement.
- (b) That, for the purpose of determining any liability under the Securities Act of 19 each such post-effective amendment shall be deemed to be a new registra statement relating to the securities offered therein, and the offering of such secur at that time shall be deemed to be the initial bona fide offering thereof.
- (c) To remove from registration by means of a post-effective amendment any of securities being registered that remain unsold at the termination of the offering.
- (d) For purposes of determining any liability under the Securities Act of 1933, of filing of the registrant s annual report pursuant to Section 13(a) or Section 15(d). Securities Exchange Act of 1934 (and, where applicable, each filing of an employenefit plan s annual report pursuant to Section 15(d) of the Exchange Act) to incorporated by reference in the registration statement shall be deemed to be a registration statement relating to the securities offered therein, and the offering of securities at that time shall be deemed to be the initial bona fide offering thereof.
- (e) That prior to any public reoffering of the securities registered hereunder throuse of a prospectus that is a part of this Registration Statement, by any person or p who is deemed to be an underwriter within the meaning of Rule 145(c), the issundertakes that such reoffering prospectus will contain the information called fo the applicable registration form with respect to reofferings by persons who may deemed underwriters, in addition to the information called for by the other item the applicable form.
- (f) That every prospectus (1) that is filed pursuant to paragraph (e) immedia preceding, or (2) that purports to meet the requirements of Section 10(a)(3) of Securities Act of 1933 and is used in connection with an offering of securities subto Rule 415, will be filed as a part of an amendment to the registration statement will not be used until such amendment is effective, and that, for purpose determining any liability under the Securities Act of 1933, each such post-effect amendment shall be deemed to be a new registration statement relating to securities offered therein, and the offering of such securities at that time shall deemed to be the initial bona fide offering thereof.
- (g) Insofar as indemnification for liabilities arising under the Securities Act of 1 may be permitted to directors, officers and controlling persons of the regist pursuant to the provisions described in Item 20 above, or otherwise, the registrant been advised that in the opinion of the SEC such indemnification is against purpolicy as expressed in the Securities Act of 1933 and is, therefore, unenforceable the event that a claim for indemnification against such liabilities (other than payment by the registrant of expenses incurred or paid by a director, office controlling person of the registrant in the successful defense of any action, surproceeding) is asserted by such director, officer or controlling person in connect with the securities being registered, the registrant will, unless in the opinion of

counsel the matter has been settled by controlling precedent, submit to a courappropriate jurisdiction the question of whether such indemnification by it is agapublic policy as expressed in the Securities Act of 1933 and will be governed by final adjudication of such issue.

- (h) To respond to requests for information that is incorporated by reference into prospectus pursuant to Item 4, 10(b), 11, or 13 of this Form, within one business of receipt of such request, and to send the incorporated documents by first-class or other equally prompt means. This includes information contained in docum filed subsequent to the effective date of the registration statement through the dat responding to the request.
- (i) To supply by means of a post-effective amendment all information concernitransaction, and the company being acquired involved therein, that was not the sub of and included in the registration statement when it became effective.

II-3

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, as amended, the Regist has duly caused this Registration Statement on Form S-4 to be signed on its behal the undersigned, thereunto duly authorized, in the City of Ontario, State of Califor on November 10, 2016.

CVB FINANCIAL CORP.

By: /s/ Christopher D. Myer
Christopher D. Myer
President and Cl
Executive Officer

POWER OF ATTORNEY

Each person whose signature appears below has made, constituted and appointed, does hereby make, constitute and appoint, Christopher D. Myers and E. A Nicholson, or either of them, as his or her true and lawful attorney-in-fact, for hir her and in his or her name, place and stead to affix his or her signature as directed officer or both, as the case may be, of the registrant, to any and all amendments post-amendments to this registration statement, and generally to do all things in the names and in their capacities as officers and directors of the registrant to enable registrant to comply with the provisions of the Securities Act of 1933 and requirements of the Securities and Exchange Commission.

Pursuant to the requirements of the Securities Act of 1933, this registration states has been signed by the following persons in the capacities and on the dates indicate

/s/ RAYMOND V. Director and Chairman of the Board November 10, 2 O BRIEN III

Raymond V. O Brien III

/s/ GEORGE A. Director and Vice Chairman November 10, 2

BORBA, JR.

George A. Borba, Jr.

/s/ STEPHEN A. DEL Director November 10, 2

GUERCIO

Stephen A. Del Guercio

/s/ ROBERT M. Director November 10, 2 **JACOBY** Robert M. Jacoby /s/ ANNA KAN November 10, 2 Director Anna Kan November 10, 2 /s/ KRISTINA M. Director **LESLIE** Kristina M. Leslie /s/ CHRISTOPHER D. Director and Chief Executive Officer November 10, 2 **MYERS** President (Principal Executive Christopher D. Myers Officer) /s/ HAL W. OSWALT Director November 10, 2 Hal W. Oswalt /s/ E. ALLEN Executive Vice President and Chief November 10, 2 **NICHOLSON** Financial Officer (Principal Financial and Accounting Officer) E. Allen Nicholson

II-4

EXHIBIT INDEX

Exhibit

No.	Description
2.1	Agreement and Plan of Reorganization and Merger, dated September 2016, by and between CVB Financial Corp and Valley Common Bancorp (included as Appendix A to the proxy statement/prospecontained in Part I of this Registration Statement).
3.1	Articles of Incorporation of the Registrant (incorporated herein reference to Exhibit 3.1 to CVB Financial Corp. s Quarterly Report Form 10-Q filed with the SEC on August 9, 2010).
3.2	Bylaws of the Registrant (incorporated herein by reference to Exhibit 3.2(A) and 3.2(B) to CVB Financial Corp. s Annual Report on 10-K filed with the SEC on February 27, 2009).
5.1	Opinion of Manatt, Phelps & Phillips, LLP regarding to the validit the securities being registered.
8.1	Opinion of Manatt, Phelps & Phillips, LLP regarding certain tax mate
10.1	Consulting Agreement dated September 22, 2016 by and betwee Citizens Business Bank and Allan W. Stone.
23.1	Consent of KPMG LLP, independent registered public accounting to of the Registrant.
23.2	Consent of Manatt, Phelps & Phillips, LLP (included in Exhibit 5.1).
23.3	Consent of Manatt, Phelps & Phillips, LLP (included in Exhibit 8.1).
24.1	Power of Attorney (included on the signature page to the Registra Statement).
99.1	Consent of Vining Sparks IBG, LP.
99.2	Form of Proxy for Special Meeting of Shareholders of Val Commerce Bancorp.

II-5